500-MW DEMONSTRATION OF ADVANCED WALL-FIRED COMBUSTION TECHNIQUES FOR THE REDUCTION OF NITROGEN OXIDE (NO_x) EMISSIONS FROM COAL-FIRED BOILERS

fi

Field Chemical Emissions Monitoring: Overfire Air and Overfire Air/Low NO_x Burner Operation Final Report DCN 93-209-061-01

INNOVATIVE CLEAN COAL TECHNOLOGY (ICCT)

500-MW DEMONSTRATION OF ADVANCED WALL-FIRED COMBUSTION TECHNIQUES FOR THE REDUCTION OF NITROGEN OXIDE (NO_x) EMISSIONS FROM COAL-FIRED BOILERS

Field Chemical Emissions Monitoring: Overfire Air and Overfire Air/Low NO_x Burner Operation Final Report

DOE Contract Number - DE-FC22-90PC89651 SCS Contract Number - C-91-000027

Prepared for:

Southern Company Services, Inc. 800 Shades Creek Parkway Birmingham, Alabama 35209



Prepared by:

Radian Corporation 8501 North Mopac Boulevard Austin, Texas 78759

Cleared by DOE Patent Council on November 22, 1993

CONTENTS

.

Sectio	n	Page
1 1	Introduction Process Operation Sampling and Analysis Protocol Quality Assurance/Quality Control (QA/QC) Data Completeness Data Quality Report Organization	. 1-1 . 1-4 . 1-4 . 1-4 . 1-4 . 1-5
2 :	Site Description Facility Information <i>Flue Gas Treatment Facilities</i> <i>Ash Removal Facilities</i> <i>NO_x Control</i> Sampling Locations	2-1 2-1 2-1 2-4 2-4 2-4 2-4
3	Results Sampling Schedule Data Treatment Coal Ash Streams Water Streams ESP Inlet Gas Stack Gas Emission Factors ESP Performance Speciation of Mercury Size-Fractionated Stack Particulate Matter Other Species Detected	
4	Discussion	4-1 4-1 4-1 4-7 4-9

	4-11
Size-Fractionated Particulate Matter	4-13
Elemental Enrichment in Ashes and Particulate Matter	4-13
5 Data Evaluation	5-1
Process Operation	5-1
Sample Collection	5-4
Analytical Quality Control Results	5-5
OFA Test Analytical QC Data	5-19
Fly Ash	5-19
Coal	5-20
Gas Streams	5-20
Metals	5-21
Anions	5-24
Volatile Organic Compounds	5-24
Semivolatile Organic Compounds	5-25
Aldehydes	5-25
OFA/LNB Test Analytical QC Data	5-26
	5-26
Bottom Ash Samples	5-26
Gas Samples	5-26
Metals	5-28
Anions	5-30
Volatile Organic Compounds	5-31
Semivolatile Organic Compounds	5-32
Aldehydes	5-32
Mercury Speciation (Frontier Geosciences Method)	5-33
Material Balances	5-33
6 Example Calculations	6-1
Stream Flow Rates	6-1
Means and Confidence Intervals for Stream Concentrations	6-2
Unit Energy Emission Factors	6-4
7 Glossary	7-1
-	
Appendix A: Sample Collection, Preparation, and Analysis	A-1
Appendix B: Analytical Data Used in Calculations	B-1
Appendix C: Analytical Data Not Used in Calculations	C-1
Appendix D: Additional Overfire Air Test Results	D-1
Annendix F. Uncertainty Formulas	F-1
	· · · · · · • • • • •

.

Appendix F:	QA/QC Results F-1
Appendix G:	Process Stream Flow Rates and Flue Gas Sampling Data G-1
Appendix H:	Process Data Trend Plots H-1
Appendix I:	Blank Correction Data I-1

.

.

.

.

.

v

LIST OF ILLUSTRATIONS

Figure	e P	age
2-1	Process Flow Diagram and Sampling Locations for Site 16	2-3
3-1	Sampling Schedule for OFA Test	3-2
3-2	Sampling Schedule for OFA/LNB Test	3-3
4-1a	Comparison of Coal Compositions	4-2
4-1b	Comparison of Coal Compositions	4-3
4-2a	Comparison of Emission Factors	4-4
4-2b	Comparison of Emission Factors	4-5
4-2c	Comparison of Emission Factors	4-6

LIST OF TABLES

Table	P	age
1-1	Substances of Interest	1-2
2-1	Unit Summary	2-2
2-2	Process Stream Analyses Performed	2-6
3-1	Coal Composition - OFA Test (mg/kg dry unless noted)	3-6
3-2	Coal Composition - OFA/LNB Test (mg/kg dry unless noted)	3-7
3-3	Ash Stream Compositions - OFA and OFA/LNB Tests (mg/kg dry unless noted)	3-9
3-4	Water Stream Compositions - OFA Test (µg/L)	3-10
3-5	ESP Inlet Gas Composition - OFA Test (μ g/Nm ³ unless noted)	3-12
3-6	ESP Inlet Gas Composition - OFA/LNB Test (µg/Nm ³ unless noted)	3-14
3-7	Stack Gas Composition - OFA Test (µg/Nm ³ unless noted)	3-19
3-8	Stack Gas Composition - OFA/LNB Test (µg/Nm ³ unless noted)	3-21
3-9	Stack Emission Factors - OFA and OFA/LNB Tests (lb/10 ¹² Btu unless noted)	3-24
3-10	ESP Removal Efficiency - OFA and OFA/LNB Tests	3-26
3-11	Mercury Speciation in Stack Gas - OFA/LNB Test (µg/Nm ³ unless noted)	3-29
3-12	Size-Fractionated Stack Particulate Matter - OFA/LNB Test (mg/kg)	3-30

3-13	Other Species Detected
4-1	Comparison of PAH Results 4-8
4-2	Comparison of Mercury Methods - OFA/LNB Test
4-3	Comparison of Chromium Methods - OFA/LNB Test 4-12
4-4	Comparison of Impactor and Multi-Metals Train Results - OFA/LNB Test
4-5	Enrichment Factors - OFA Test 4-16
4-6	Enrichment Factors - OFA/LNB Test 4-17
5-1	Summary of Process Monitoring Data - OFA Test
5-2	Summary of Process Monitoring Data - OFA/LNB Test
5-3	Types of Quality Control Samples 5-7
5-4	Types of Quality Control Data Reported - OFA 5-8
5-5	Types of Quality Control Data Reported - OFA/LNB
5-6	Summary of Precision and Accuracy Estimates - OFA
5-7	Summary of Precision and Accuracy Estimates - OFA/LNB 5-15
5-8	Material Balance Results - OFA and OFA/LNB Tests

1 INTRODUCTION

This report summarizes data gathered by Radian Corporation at a coal-fired power plant, designated Site 16, for a program sponsored by the United States Department of Energy (DOE), Southern Company Services (SCS), and the Electric Power Research Institute (EPRI). The concentrations of selected inorganic and organic substances were measured in the process and discharge streams of the plant operating under two different types of combustion modifications: overfire air (OFA) and a combination of overfire air with low-NO_x burners (OFA/LNB). The Site 16 plant is a participant in the DOEsponsored Clean Coal Technology (CCT) program, and the information contained in this report will allow DOE and EPRI to determine the effects of low-NO_x modifications on plant emissions and discharges. In addition, SCS can use this information to make future decisions about plant modifications and control strategies.

Sampling was performed on an opposed wall-fired boiler burning medium-sulfur bituminous coal. Emissions were controlled by electrostatic precipitators (ESPs). The testing was conducted in two distinct sampling periods, with the OFA test performed in March of 1991 and the OFA/LNB test performed in May of 1993. The specific objectives for each test period were:

- To quantify emissions of target substances from the stack;
- To determine the efficiency of the ESPs for removing the target substances; and
- To determine the fate of target substances in the various plant discharge streams.

Table 1-1 lists the substances of interest to this project.

The Clean Coal Technology program, which DOE began in 1986, demonstrates the commercial readiness and monitors the environmental performance of new, advanced coal utilization technologies. DOE shares the cost on each project, with private sector sponsors providing at least 50% of the funds. Within the CCT program there are currently 46 projects, each of which was selected under one of five rounds of nationwide competition. The Site 16 project, which is a demonstration of advanced combustion techniques to reduce NO_x emissions, was selected under the second round of CCT.

DOE is actively involved in the measurement of potentially hazardous substances identified in Title III of the Clean Air Act Amendments (CAAA) of 1990. The intention of these efforts is to obtain data that will allow a better understanding of the principles

Table 1-1Substances of Interest

Elements	Organic Compounds
Arsenic	Benzene
Barium	Toluene
Beryllium	Formaldehyde
Cadmium	Polycyclic Organic Matter (POM) [*]
Chlorine (as chloride)	
Chromium	
Cobalt	
Copper	
Fluorine (as fluoride)	
Lead	
Manganese	
Mercury	
Molybdenum	
Nickel	
Phosphorus	
Selenium	
Vanadium	

* Also referred to as semivolatile organic compounds. Includes polynuclear aromatic hydrocarbons (PAHs).

and processes involved with the formation, distribution, and fate of toxic substances in power plant systems. Some of the DOE toxics monitoring work is being performed under the CCT program. The first three of the five CCT solicitations were issued before the enactment of the CAAA in 1990, and did not originally include extensive air toxics monitoring activities. However, DOE has expanded several of these projects (including Site 16), through cooperative agreements with private industry and with EPRI, to emphasize measurement of Title III substances. Air toxics monitoring activities were included in the fourth and fifth CCT solicitations. In addition, DOE's Flue Gas Cleanup program issued a solicitation (Comprehensive Assessment of Toxic Emissions from Coal-Fired Power Plants) in 1992 to gather further information on Title III substances at coalfired power plants participating in CCT.

EPRI is cosponsoring the work at Site 16 several reasons. During the Power Plant Integrated Systems Chemical Emissions Studies (PISCES) project (EPRI RP-2933-1), a number of data gaps were identified for certain streams and substances within specific power plant configurations. The work discussed here was done in response to EPRI member utility concerns about the concentrations of trace substances in process streams, the effectiveness of control technologies in reducing emissions of these substances, and the applicability of the results of previous studies discussed in the literature.

The Field Chemical Emissions Monitoring (FCEM) project (EPRI RP-3177-1) sponsored by EPRI was initiated to generate the missing data identified by the PISCES project. Although the Site 16 project was conducted separately from the FCEM project, most of the objectives were the same, and the sampling, analytical, and data handling procedures are consistent with those used in the FCEM project. Reports on several of the plants sampled have already been furnished to the U.S. Environmental Protection Agency (EPA) to use to study emissions from fossil-fuel-fired power plants, as mandated by the Clean Air Act Amendments of 1990.

Radian Corporation conducted the testing and has prepared this report using the following procedures to evaluate the data:

- The type and quantity of quality assurance samples were reviewed to determine the confidence that can be placed in the results; and
- The QA/QC results were compared with data quality objectives to evaluate precision and accuracy.

The results for each substance are presented by individual run and as averaged totals. To quantify the variability of the data, the 95% confidence interval about the mean is also presented. The confidence interval incorporates the combined process, sampling, and analytical variabilities.

Process Operation

The unit operated at nearly full load during each of the sampling periods. Operating parameters were monitored to verify the process stability during sampling. Process operation is discussed in more detail in Section 5. No unusual process upsets were encountered. By all indications, process operation was normal during the sampling.

Sampling and Analysis Protocol

Appendix A describes the sampling and analysis protocol for Site 16. The methods used are comparable to those used by Radian in the FCEM project. The major exception was at the ESP inlet during the OFA/LNB test, where metals, anions, and semivolatile compound trains were equipped with a cyclone precutter ahead of the thimble filter normally used. This prevented the loss of particulate matter from the probe, a problem identified from the OFA test.

Quality Assurance/Quality Control (QA/QC) Data Completeness

The completeness of the quality assurance data was reviewed to judge whether the quality of the measurement data could be evaluated with the available information. In general, the results of the QC checks available for Site 16 indicate that the samples are well characterized. An evaluation of the accuracy, precision, and bias of the data, even if only qualitative, is considered to be an important part of the data evaluation. A full discussion of each of these components of quality can be found in Section 5.

Standard QA/QC checks for this type of sampling program involve the use of: 1) replicate tests, duplicate field samples and lab analyses, and matrix spike and lab control duplicates to determine precision; 2) matrix spikes, surrogate spikes, and laboratory control samples to determine accuracy; and 3) field blanks, trip blanks, method blanks, and reagent blanks to determine if any of the samples were contaminated during collection or analysis. All of these standard QA/QC checks were used on various samples from Site 16. The absence of any of these quality control checks for a given measurement does not necessarily reflect poorly on the quality of the data but does limit the ability to measure the various components of measurement error.

Data Quality

The available QA/QC results were compared to the data quality objectives discussed in Section 5. QA/QC results outside the data quality objectives are noted and discussed, other quality assurance values are evaluated, and the potential effect on data quality is noted. From the detailed information in Section 5, several important issues have been identified that may affect the data.

For the OFA test:

- The high recoveries of copper and barium in a coal performance evaluation sample suggest that copper and barium concentrations in the coal may be biased high.
- The concentrations of molybdenum on the blank filters are significant, compared to the sample results, and may positively bias the results.
- The high spike recoveries and significant blank levels suggest that cadmium concentrations in the fly ash and flue gas samples may be biased high.
- The formaldehyde concentrations measured in the flue gas are highly suspect. Formaldehyde was detected in the lab, trip, and field blanks in concentrations comparable to those found in the samples.

For the OFA/LNB test:

- The PAH concentrations measured in the flue gas were highly variable and should be considered order-of-magnitude estimates. The internal standard and surrogate spike recovery data indicate acceptable analytical performance but suggest sample matrix interference problems.
- The formaldehyde concentrations measured in the stack were near detection limits and were indistinguishable from the lab, trip, reagent, and field blank results.
- High levels of Cr(VI) were found in the KOH reagent blanks, but these were found to be repeatable in three blanks and were subsequently corrected for in the sample results. However, the blank correction exceeded 50% of the result in two of the three samples.
- Toluene was detected in very low but similar concentrations in blank and actual VOST samples. The toluene concentrations measured in the stack may be biased slightly high if not artifacts of blank contamination.
- The variable spike recoveries by GFAAS show that selenium concentrations in the flue gas may show higher-than-expected variability.

Report Organization

Section 2 of this report briefly describes the plant and the sample locations. Section 3 discusses the chemical analyses of the coal, flue gas, and other process streams. Section 4 contains a discussion of the results, focusing primarily on the comparison of the OFA and OFA/LNB tests. Section 5 presents QA/QC and engineering evaluations of the data. Section 6 presents example calculations, and a glossary of terms is provided in Section 7. The appendices contain information on sampling and analytical methods,

Introduction

stream concentrations, sampling data, error propagation equations, and detailed QA/QC data.

.

SITE DESCRIPTION

The FCEM project has a policy of assigning a site code to each plant sampled. The plant covered by this report has been designated Site 16. The test site and sampling locations are described in this section.

Facility Information

The characteristics of the unit are summarized in Table 2-1. The unit tested has a gross generating capacity of approximately 500 MW. The opposed wall-fired, subcritical boiler was designed by Foster Wheeler. A partial, vertical dividing plate within the furnace creates two combustion zones, and very little mixing of the flue gas occurs between the A and B sides.

Figure 2-1 is a process flow diagram of the unit. The plant burns a combination of bituminous coals that have a typical sulfur content of 1.6% and a typical ash content of 10 percent.

Bottom ash is removed from the boiler by an ash sluicing system. Electrostatic precipitators (ESPs) remove fly ash from the flue gases. The flue gas treatment and ash removal facilities are described in greater detail below.

Flue Gas Treatment Facilities

The flue gas exiting each side of the furnace flows into a separate duct, designated the A or B side. Two ESPs, one each for the A and B sides, remove particulate matter from the flue gas. The unit is equipped with a conditioning system capable of injecting SO₃ or NH₃ into the flue gas upstream of the ESPs to improve ESP performance. The conditioning system was not in use during the OFA testing. During the OFA/LNB test, NH₃ was injected at a rate of approximately 25 scfm, which is equivalent to a concentration of about 20 ppmv in the flue gas entering the ESPs. The NH₃ injection was used because of plant concerns about complying with particulate matter emission limits.

Site Description

Table 2-1Unit Summary

Maximum Gross Electrical Output (MW):	500
Particulate Emission Limit (lb/10 ⁶ Btu):	0.24
Particulate Matter Controls:	Cold-Side ESPs
Flue Gas Conditioning:	SO ₃ or NH ₃ ^b
Boiler Type:	Opposed Wall-Fired
Boiler Additives:	None
NO _x Control:	OFA or OFA/LNB
Fuel Type:	Bituminous Coal
Fuel Sulfur Content (% dry):	1.6*
Fuel Ash Content (% dry):	10*
Fuel Heating Value (Btu/lb, dry):	13,700*
Fly Ash Disposal:	Pond
Bottom Ash Disposal:	Pond
Ash Sluice Water Source:	Recycle from Pond
Cooling Water System:	Once Through
Cooling Water Source:	River

* Mean values measured during sampling.

^b No conditioning was used during the OFA test. A concentration of approximately 20 ppmv NH₃ was used during the OFA/LNB test.



Figure 2-1 Process Flow Diagram and Sampling Locations for Site 16

2-3

Ash Removal Facilities

Dry ash collected in the economizer and ESP hoppers is pneumatically transported to a tank where it is mixed with water and sluiced to a settling pond. Bottom ash from the boiler is sluiced to a separate settling pond. The water used for ash sluicing is recycled water from the settling ponds.

NO_x Control

The overfire air ports were installed during a four-week outage in the spring of 1990 by Foster Wheeler Energy Corporation (FWEC). The design includes four overfire air ports on each side of the boiler directly above the top row of burners. Overfire air is diverted from the secondary air ductwork. At full load, approximately 20% of the secondary air is introduced through the overfire air ports.

The low-NO_x burners were installed during a seven-week outage in the spring of 1991. The FWEC burners are of the controlled flow/split flame (CFSF) design. The 24 burners are arranged on opposing walls, with three rows of four burners on each wall. The low-NO_x burners replaced the pre-NSPS Intervane burners previously in service that were in place during the OFA test.

Sampling Locations

Samples were collected at several locations, identified on the process flow diagram, Figure 2-1. Each sampling location is briefly described below:

- During the OFA test, coal composite samples were collected through "clean-out" ports at the bottom of the bunkers that feed each of the six coal mills. For the OFA/LNB test, samples were collected from taps (which weren't present during the OFA test) on each of the six coal feeders directly below the bunkers. Because the samples were collected upstream of the mills in both tests, they were taken before the rejection of pyrites.
- Flue gas entering the ESPs was sampled through four-inch sampling ports on both the A and B inlet ducts. The ducts were angled approximately 20 degrees from horizon-tal, and each duct had 7 ports.
- Stack gas samples were collected from six horizontal ports on the stack at the 250-foot level.
- Bottom ash that had accumulated in the boiler during testing was sampled from the discharge of the sluice pipe during a sluicing event. The sluice water was sampled concurrently.

- Dry ESP fly ash samples were collected from each of the 16 ESP hoppers through ports located near the bottom of each hopper.
- Sluice water supply samples were collected from the recycled pond water intake.

The procedures for collecting, pretreating, and analyzing the samples are discussed in Appendix A. Table 2-2 presents an overview of the types of analyses performed on these streams. Several analyses were not performed during the OFA/LNB test because of the extremely low concentrations found in the OFA test or, in the case of ESP ash, because of concerns about the representativeness of a composite sample of the 16 hopper fractions.

Table 2-2Process Stream Analyses Performed

.

Stream	Metals	Anions	Semivolatile Organic Compounds	Volatile Organic Compounds	Aldehydes
Coal	X,O	X,O			
Bottom Ash	X,O	X,O	x		
Bottom Ash Water	X	X	x		
Sluice Supply Water	X	X	X		
ESP Ash	X	X	X		
ESP Inlet Gas	X,O	X,O	X,O	Х	X
Stack Gas	X,O	X,O	X,O	X,O	X,O

.

X = OFA Test

O = OFA/LNB Test

3

RESULTS

This section summarizes the results of the stream characterization for both the OFA and LNB tests. Sampling, preparation, and analytical methods are summarized in Appendix A. Detailed analytical data can be found in Appendices B and C.

Sampling Schedule

The OFA sampling was conducted in March 1991. The OFA/LNB test was conducted in May 1993. Flue gas samples were collected from the A and B ESP inlet ducts and the stack. Multi-metals trains and Modified Method 5 (MM5, semivolatile organic compound) trains were used to traverse the ducts during each sampling run. Anions, aldehydes, VOST, impactors, Cr(VI), and mercury speciation trains were used to collect samples at single points of average velocity.

Figure 3-1 shows the sample collection schedule for the OFA test; Figure 3-2 shows the schedule for the OFA/LNB test. Three valid runs for each train were completed.

Data Treatment

Several conventions have been developed for treating the test data and developing average concentrations of substances in the various streams.

To determine the total gas concentration for each run, both the solid- and vapor-phase contributions were considered; however, the absence of some detectable concentrations in either (or both) phase(s) required that conventions be developed for dealing with these data. These conventions are summarized below.

For each substance, there are three possible combinations of vapor- and solid-phase concentrations in the flue gas stream. These are:

- Case 1: The concentrations in both the solid and vapor phases are above detection limits.
- Case 2: The concentrations in both the solid and vapor phases are below the detection limits.

Samples:				
Multi-Metals Train	Run 1 (Void)		Run 3	Run 4 Run 5
Anions Train	Run 1 (Void) T		Run 3	Run 5 T Run 5
Aldehyde Train	ξŢ		T B	Ru 4
VOST	T g	Run 2	Run 3	
Semivolatile Train		Run 2	E un 3	
Coal	T Ru	Run 2	Run 3	Run & Run 5
Bottom Ash	Run 1		Run 3	Run 4 Run 5
ESP Ash	Ĩ		Run 3	Run 4 Run 5
Makeup Water	Run 1		Run 3	Run 4 Run 5
Time	0600 1200 1800	0600 1200 1800	0600 1200 1800	0600 1200 1800
Date	3/3/91	3/4/91	3/5/91	3/6/91
Figure 3-1 Sampling Schedule for O	FA Test	 Denotes continuou Denotes single gra 	us or composite sarr ib sample	lple

3-2

Samples:				
Multi-Metals Train	Run 1 (Void)	Run 2	Run 3	Run 4
Anions Train	Run 1 (Void)	T 2 T	Run 3 	Run 4
Aldehyde Train	,	Run 1 (Void)	Run 2	Run 3 Run 4
VOST	Run 1 (Void)	Run 2	Run 3	Run 4
Semivolatile Train	Run 1 (Void)	Run 2	Run 3	Run 4
Cr(VI) Train		Run 2	Run 3	Run 4
Hg Speciation Train	Run 1 (Void)	Run 2	Run 3	Bun 4 ⊥ 4
UW Impactor	(Void)	(Void) Run 2 	(Void) Run 3 H H I	Run 4
Coal	Run 1 (Void)	Run 2	Run 3	Run 4
Bottom Ash	Run 1 (Void)	Run 2	Bun 3	Run 4
L Time	0600 1200 1800	0600 1200 1800	0600 1200 1800	0600 1200 1900
Date	5/18/93	5/19/93	5/20/93	5/21/93
Figure 3-2 Sampling Schedule for OFA	A/LNB Test	Denotes continuor	is or composite sam	lple

Results

Results

Case 3: The concentration in one phase is above the detection limit, and the concentration in the other phase is below the detection limit.

For constituents of interest other than HCl, HF, and mercury, the flue gas stream data from previous studies of coal-fired power plants have shown that most of the material is present in the solid phase, and that only a small fraction is generally found in the vapor phase. Thus, the following conventions were selected for defining the total gas stream concentrations:

• For Case 1, the total concentration is the sum of the concentrations in the vapor and solid phases.

For example, the total arsenic concentration in the stack gas for Run 2 (OFA/LNB test) is calculated as follows:

As in the solid phase = $110 \,\mu g/Nm^3$

As in the vapor phase = $2.0 \,\mu g/Nm^3$

Total As in ESP inlet gas = $112 \mu g/Nm^3$

• For Case 2, the total concentration is considered to be the detection limit in the solid phase.

For example, the total molybdenum concentration in the ESP inlet gas for Run 3 (OFA test) is calculated as follows:

Mo in the solid phase = $ND(290 \mu g/Nm^3)$

Mo in the vapor phase = $ND(22 \mu g/Nm^3)$

Total Mo in the ESP inlet gas = $ND(290 \mu g/Nm^3)$

• For Case 3, the total concentration is considered to be the one above the detection limit, regardless of which phase this represents.

For example, the cobalt concentration in the stack gas for Run 3 (OFA test) is calculated as follows:

Co in the solid phase = $7.6 \,\mu g/\text{Nm}^3$ Co in the vapor phase = $\text{ND}(1.7 \,\mu g/\text{Nm}^3)$ Total Co in the stack gas = $7.6 \,\mu g/\text{Nm}^3$ The above conventions also are in accordance with guidance provided by EPA (*Technical Implementation Document for EPA's Boiler and Industrial Furnace Regulations.* U.S. Environmental Protection Agency, Office of Solid Waste, Washington, D.C., March 1992).

Testing at several sites has indicated that HCl, HF, and mercury are present primarily in the vapor phase (although mercury is sometimes also detected in the solid phase). For Case 2, then, the total concentration of each of these species is considered to be the detection limit in the vapor phase. For Cases 1 and 3 the methodologies are unchanged from those described above.

The following criteria were used when averaging the results of different runs:

- When all values for a given variable were above the detection limit, the mean concentration was calculated as the true arithmetic mean.
- For results that include values both above and below the detection limit, one-half the detection limit was used to calculate the mean. For example:

Analytical Values	Calculation	<u>Mean Value</u>
10, 12, ND(8)	[10+12+(8/2)]/3	8.7

By convention, the calculated mean is not allowed to be smaller than the largest detection limit value. In the following example, using one-half the detection limit would yield a calculated mean of 2.8. This is less than the highest detection level obtained; therefore, the reported mean is ND(4).

Analytical Values	<u>Calculation</u>	<u>Mean Value</u>
5. ND(4), ND(3)	[5+(4/2)+(3/2)]/3 = 2.8	ND(4)

• When all analytical results for a given variable are below the detection limit, the mean is reported as ND(x), where x is the largest detection limit. The bias estimate (used where calculating confidence intervals for other parameters) is one-half of the detection level, and no confidence interval is reported.

Coal

Tables 3-1 and 3-2 show the analytical results for the coal samples collected during the OFA and OFA/LNB tests, respectively. Appendix A describes the analytical methods used for each combination of substance and stream. Measurements of the concentrations reported here were made using what Radian considered to be the best method for each matrix. Typically, the method with the lowest detection limit was chosen, except

Results

Table 3-1Coal Composition - OFA Test (mg/kg dry unless noted)

Substance	Run 1	Run 3	Run 4	Run 5	Mean	95% CI
Date	3/3/91	3/5/91	3/6/91	3/6/91		
Gross Load (MW)	472	477	477	477	476	4
Coal Rate (lb/hr, dry)	330,000	335,000	328,000	326,000	330,000	6,100
HHV (Btu/lb, dry)	13,400	13,900	13,700	13,700	13,700	370
Ash (%, dry)	12	9.1	10	10	10	1.6
Moisture (%)	5.5	4.1	4.3	4.8	4.7	1.0
Sulfur (%, dry)	1.8	1.6	1.6	1.6	1.6	0.2
Major Species						
Aluminum		16,000	14,000	13,000	14,000	3,800
Iroa		15,000	13,000	11,000	13,000	5,000
Sodium		288	266	260	270	37
Titanium		915	694	714	770	300
Target Species						
Arsenic		13	19	19	17	8.0
Barium		210	160	140	170	86
Beryllium		1.7	1.5	1.0@	1.4	0.93
Cadmium		ND(0.10)	0.16@	ND(0.11)	ND(0.11)	
Chloride		310@	500@	400@	410	230
Chromium		26	21	19	22	8.6
Cobalt		8.9	8.3	7.8	8.3	1.4
Copper		56	29	31	38	37
Fluoride		73	38@	100	70	79
Lead		4.8	5.6	4.9	5.1	1.1
Manganese		14	16	22	17	11
Mercury		0.19	0.13	0.15	0.15	0.08
Molybdenum		3.8	5.2	ND(1.4)	3.2	5.7
Nickel		28	27	25	27	3.1
Phosphorus		210	260	270	240	76
Selenium		3.9	3.2	4.2	3.8	1.2
Vanadium		36	41	35	37	8.8

ND = Not detected at the concentration in parentheses.

CI = Confidence Interval.

 \mathcal{Q} = Result is less than five times the detection limit.

Table 3-2Coal Composition - OFA/LNB Test (mg/kg dry unless noted)

Substance	Run 2	Run 3	Run 4	Mean	95% CI
Date	5/19/93	5/20/93	5/21/93		
Gross Load (MW)	467	472	470	470	6
Coal Rate (lb/hr, dry)	313,000	316,000	317,000	315,000	5,200
HHV (Btu/lb, dry)	13,700	13,700	13,900	13,800	200
Ash (%, dry)	9.5	10	8.9	9.5	1.3
Moisture (%)	4.3	4.0	3.0	3.8	1.7
Sulfur (%, dry)	1.8	1.7	1.6	1.7	0.27
Major Species					
Aluminum	12,000	15,000	15,000	14,000	4,590
Iron	14,000	10,000	9,500	11,000	6,200
Sodium	310	390	320	340	110
Titanium	740	980	800	840	310
Target Species					
Arsenic	23	22	24	23	2.3
Barium	94	120	86	99	39
Beryllium	1.9	2.4	2.4	2.2	0.72
Cadmium	ND(2.6)	ND(2.6)	ND(2.6)	ND(2.6)	
Chloride	430@	280@	310@	340	200
Chromium	15	21	15	17	8.7
Cobalt	5.9	6.8	5.5	6.1	1.7
Copper	29	33	40	34	14
Fluoride	110	81	79	91	49
Lead	8.0	8.0	6.0	7.3	2.9
Manganese	14	14	14	14	0.8
Mercury	0.16	0.15	0.11	0.14	0.07
Molybdenum	3.2@	5.0@	ND(3.6)	ND(3.6)	
Nickel	21	19	12	17	12
Phosphorus	170	140	110	140	75
Selenium	3.4	4.0	3.6	3.7	0.8
Vanadium	23	30	25	26	8.7

ND = Not detected at the concentration in parentheses.

CI = Confidence Interval.

Q = Result is less than five times the detection limit.

Results

when the QA/QC results indicated significant problems with precision or bias for a particular technique. For each substance, a mean concentration was calculated, along with the 95% confidence interval about the mean. The confidence interval is the range about the mean wherein the probability is 95% that the true mean lies. For example, according to the three results shown in Table 3-1, it can be said, with 95% certainty, that the true mean cobalt concentration in the coal during the OFA test was between 6.9 and 9.7 mg/kg. Calculation of this confidence interval is discussed in Section 6.

For those substances that could not be quantified, the notation "ND(x)" is used. This term means "not detected at a concentration of x." The detection limit can vary according to sample size, sample preparation, and analytical method. For instance, the detection limit for cadmium in the coal is higher for the OFA/LNB test than for the OFA test, because only INAA was employed for the OFA/LNB test, as opposed to a specific cadmium analysis by GFAAS for the OFA test.

Calculations were performed with unrounded numbers, and the results were rounded for presentation in the table; therefore, slight differences in calculated means and confidence intervals are attributable to round-off errors.

Ash Streams

Table 3-3 shows the mean compositions of the bottom ash and fly ash samples. For the OFA test, bottom ash samples and fly ash samples (taken from the ESP hoppers) were collected and analyzed. For the OFA/LNB test, only the bottom ash samples were analyzed. Fly ash samples were collected from each of the 16 ESP hoppers, but they were archived for possible future analysis because of concern about the representativeness of a composite sample made from equal fractions of each of the 16 hopper samples (as was done for the OFA test). The ESP inlet particulate matter was deemed to be a better measure of the ash collected in the ESP.

In both tests, significant concentrations of mercury were detected in the fly ash. The concentrations are higher than observed for most FCEM project sites. The sorption of mercury onto the fly ash may be influenced by the level of unburned carbon in the ash. The loss on ignition (LOI) of the fly ash was not measured for the OFA test, but the LOI of the ESP inlet particulate matter collected during the OFA/LNB test was 6.2 ± 2.1 percent. This is a relatively high LOI and may be responsible for the higher-than-expected levels of mercury in the fly ash.

Water Streams

Table 3-4 shows the mean compositions of the sluice supply water and the bottom ash sluice water samples collected during the OFA test. These streams were also sampled during the OFA/LNB test, but the samples were archived for possible future analysis.

	Bottom OFA 7	Ash Fest	Bottom OFA/LN	Ash B Test	ESP Col Fly A OFA	llected Ash Test	ESP Partic OFA/L	Inlet culate NB Test
Substance	Mean	95% CI	Mean	95% CI	Mean	95% CI	Mean	95% CI
Flow Rate (lb/hr, dry)	6,770	1,260	7,060	4,370	27,100	5,040	24,300	2,900
Major Species								
Aluminum	140,000	3,500	120,000	29,000	150,000	7,200	130,000	8,300
Iron	92,000	10,000	110,000	19,000	92,000	12,000	93,000	4,700
Sodium	1,800	210	3,600	900	1,900	490	3,400	670
Sulfur	440	900	71	200	2,000	330	4,200	8,800
Titanium	8,600	3,700	6,800	340	8,000	560	6,900	130
Target Species								
Arsenic	52	66	23	78	220	38	310	160
Barium	860	264	920	200	780	110	1,100	24
Beryllium	10	2.9	22	4.7	13	5.1	23	3.0
Cadmium	ND(1.0)	**	ND(1.0)		ND(1.0)		3.6	6.1
Chloride	ND(100)		ND(100)		ND(100)		2,300	1,100
Chromium	110	34	110	15	120	34	150	33
Cobalt	50	9.0	67	34	55	14	51	28
Copper	97	3.6	130	30	160	16	200	45
Fluoride	ND(17)		24	16	56	12	220	230
Lead	20	2.9	14	1.6	75	75	61	12
Manganese	120	51	150	19	140	190	160	44
Мегсигу	ND(0.020)		ND(0.012)		0.23	0.05	0.79	0.35
Molybdenum	ND(200)		24	41	ND(200)		15	22
Nickel	100	7.7	82	3.7	240	620	120	8.2
Phosphorus	1,120	820	540	19	1,400	1,300	810	240
Selenium	ND(5.0)		ND(1.2)	••	17	18	17	1.3
Vanadium	210	11	230	72	240	14	260	67

Table 3-3 Ash Stream Compositions - OFA and OFA LNB Tests (mg/kg dry unless noted)

ND = Not detected at concentration in parentheses. Cl = Confidence Interval.

Results

Table 3-4 Water Stream Compositions - OFA Test $(\mu g/L)$

	Sluice Sup	ply Water	Bottom Ash Sh	uice Water
Substance	Mean	95% CI	Mean	95% CI
Major Species				
Aluminum	680	53	200	220
Iron	790	46	240	220
Sodium	9,300	1,700	6,500	2,000
Sulfate	68,000	7,700	40,000	11,000
Titanium	ND(50)		ND(50)	
Target Species				
Arsenic	16	0	7.2	1.9
Barium	150	3.0	98	17
Beryllium	ND(2.0)		ND(2.0)	
Cadmium	1.5	0.4	ND(1.0)	
Caloride	3,000	50	3,100	540
Chromium	ND(10)		ND(10)	
Cobalt	ND(10)		ND(10)	
Copper	35	1.4	ND(20)	
Fluoride	180	11	130	21
Lead	3.7	5.7	ND(3.0)	
Manganese	90	6.3	52	10
Мегсигу	ND(0.20)		ND(0.20)	
Molybdenum	ND(50)		ND(50)	84
Nickel	ND(20)		ND(20)	
Phosphorus	53	2.9	ND(50)	
Selenium	ND(5.0)		ND(5.0)	
Vanadium	ND(20)		ND(20)	

ND = Not detected at concentration in parentheses. CI = Confidence Interval.

ESP Inlet Gas

Tables 3-5 and 3-6 show the concentrations of the target analytes in the ESP inlet gas for the OFA and OFA/LNB tests, respectively. The A and B ducts were sampled sequentially with a single train for each type of gas sample, producing samples representative of the unit as a whole. The data are presented as solid and vapor compositions, along with the mean concentrations and confidence intervals in the combined phases.

For the OFA test, blank concentrations for both vapor-phase and solid-phase samples were significant for many of the substances when compared with the measured concentrations. Blank corrections were applied, and the details of these corrections can be found in Appendix I. For the OFA/LNB test, blank contributions to the target analyte results were not significant, and blank corrections were not applied.

The analytical results for semivolatile organic compounds were completely different for the two tests. For the OFA test, the samples collected with the MM5 train at the ESP inlet and the stack were analyzed by Method 8270 (GC-MS) for a long list of semivolatile organic compounds, some of which are classified as semivolatile organic matter (POM). During the OFA test, none of the POM compounds were present in the ESP inlet gas (or the stack gas) above detection limits, which were approximately 5-10 μ g/Nm³ for most of the compounds.

For the OFA/LNB test, the MM5 train samples collected at the ESP inlet and the stack were analyzed for a select group of polycyclic aromatic hydrocarbons (PAHs, a subset of POM) by high-resolution GC-MS. The concentrations of several of the compounds measured were above detection limits, but note that these levels are much lower (up to 3 orders of magnitude) than the detection limits available with the 8270 analysis used for the OFA test. The decision to use high-resolution GC-MS was made after the OFA test showed that all of the PAH concentrations were below 8270 analysis detection limits.

There are some anomalous results in the ESP inlet gas stream compositions for chloride, fluoride, mercury, and selenium. Each of these substances can potentially be present in the vapor phase. A problem that may have been encountered during sampling of the high-particulate ESP inlet gas is that, as the flue gas was drawn through the in situ thimble filter, reactions between the vapor and the large quantities of particulate in the thimble may have caused some vapor-phase species to be captured in the particulate fraction of the sampling train. While this would not affect the total concentration measured, it would attribute more of the substance to the particulate phase and less to the vapor phase than was actually present in the duct. Ash alkalinity may be a factor in the case of chloride, fluoride, and selenium, but it is also possible that unburned carbon in the ash may act as a sorbent for these vapor-phase species, including mercury.

For chloride, it appears that some reaction with the fly ash did occur during sampling in both the OFA and OFA/LNB tests. The total amount of chloride in the ESP inlet gas looks reasonable compared to the coal, but the vapor component measured at the ESP inlet is noticeably lower than at the stack. While the particulate/vapor chloride ratio

			and Dhare				Vapor Phase				
						Run 2	Rua 3	Run 4	Run S	Mean	95% CI
Substance	Run 1	Run 2		표 문 문							
	3/3/91	3/4/91	3/5/91	3/6/91	3/6/91						
	1 740 000	1.300.000	1,280,000	1,300,000	1,310,000					1,290,000	34, 100
Gas Flow Kale (use in)	500 For 44	000 000 0	000 000 0	2 060 000	2.070.000					2,040,000	55,900
Gas Flow Rate (Nm ³ /hr)	1,960,000	7,000,000	2,000 000 to							25,600	2,510
Particulate Matter (lb/hr)*			24,400	26,200	29,100					5.650	413
Particulate Matter (mg/Nm ³)*			5,460	5,770	5,720						
-					1						
Major Species			000	000 005	660.000		ND(87)	(£6)QN	ND(90)	670,000	230,000
Aluminum			110,000	000,066	2221222		LIVIN	(61)QN	21@	450,000	190,000
Imp			530,000	380,000	450,000						000 50
			24,000@	18,000@	ND(5,700)		ND(430)	ND(470)	ND(450)		20017
Sodium			000 001	110.000	145.000		4,700,000	4,600,000	4,300,000	4,700,000	480,000
Sulfate			140,000				ND(22)	ND(23)	ND(22)	41,000	14,000
Titanium			41,000	30,000	41,000						
Target Species					2		ND(1.7)	ND(1.9)	ND(1.8)	1,800	1,100
Arsenic			2,200		N/1-		ND(4.3)	ND(4.7)	ND(4.5)	6,300	2,800
Berium			7,500	5,300	Mie						15
Deadline			110	100	120		ND(0.50)	(7C'0)(TN			
Decynnan			6.60	(11) ND(11)	6.1 @		ND(0.43)	ND(0.47)	ND(0.45)	ND(11)	
Cadmium			22.000	21.000	27,000		9,100	12,000	12,000	35,000	10,000
Chloride ⁶			100		000		ND(4.3)	ND(4.7)	ND(4.5)	940	290
Chromium			1,100	640	N64		C OUN	UD CUN	(6.1)UN	370	110
Cobalt			420	320	360				0044	206 1	200
Conner			1,300	1,200	1,300		ND(8.7)	(r-6)(N	(n: c) (n)	11	
Cupres											

OFA Test (µg/Nm³ unless noted)itiom Ċ ç 2-12 FSP Inlet (

Table 3-5 (Continued)

			Solid Phase					Vapor Phase				
Substance	Run 1	Run 2	Run 3	Run 4	Run 5	Run 1	Run 2	Run 3	Run 4	Run 5	Mean	95% CI
Fluoride 4			1,100	1,400	1,400			1,700	2,200	1,700	3,100	1,200
Lead			270	520	530			ND(1.3)	0.86@	2.6@	440	370
Mangancse			980	600	680			ND(4.3)	ND(4.7) ⁶	ND(4.5)	750	500
Mercury *			3.6@	0.050@	5.8			3.0	4.4	4.4	7.1	7.3
Molybdenum			ND(290)*	ND(260)	ND(290)*			ND(22)	ND(23)	ND(22)	ND(290)	1
Nickel			810	540	630			ND(8.7)	ND(9.3)	ND(9.0)	660	340
Phosphorus			9,300	4,000@	4,800@			ND(141)	ND(137)	ND(116)	6,000	7,100
Selenium ^f			310	540	570			ND(1.2)	ND(1.3)	1.7@	470	350
Vanadium			2,000	1,500	1,700			ND(8.7)	ND(9.3)	ND(9.0)	1,800	600
Benzene						1.0@	2.1@	0.67@			1.3	1.8
Toluene						ND(0.52)	ND(0.54)	ND(0.56)			ND(0.56)	;
Formaldehyde ⁴						14@		ND(6.5)*	ND(6.9)		ND(6.8)	:
POM											٩D	

Calculated assuming 80:20 fly ash:bottom ash split

* Blank correction was greater than 50% of uncorrected result.

* Some of the solid-phase chloride measured at this location may have actually been vapor phase chloride captured on the particulate fraction during sampling. See the discussion in the text under the ESP intet results.

⁴ The vapor-phase and total fluoride concentrations may be biased low. See the discussion in the text under the ESP inlet results. • The vapor-phase and total mercury concentrations may be biased low. See the discussion in the text under the ESP inlet results.

Vapor-phase selenium concentrations might be biased low, while particulate-phase and total selenium concentrations may be biased high. See the discussion in the text under the ESP inlet results.

Formaldehyde was detected in blanks at concentrations similar to the samples. Results shown here are blank corrected.

None of the polycyclic organic matter was measured in concentrations above detection limits. The results for the individual compounds can be found in Appendix B.

ND = Not detected at concentration in parentheses.

CI = Confidence Interval.

 \boldsymbol{Q} = Result is less than five times the detection limit.

3-13

Table 3-6 ESP Inlet Gas Composition	a - OFA/LNF	t Test (μg/Nπ	1 ³ unless noted)	_				Results
·		Solid Phase		>	apor Phase			
Substance	Run 2	Run 3	Run 4	Run 2	Run 3	Run 4	Mean	95% CI
Date	5/19/93	5/20/93	5/21/93					
Gas Flow Rate (dscfm)	1,270,000	1,260,000	1,230,000				1,250,000	44,000
Gas Flow Rate (Nm ³ /hr)	2,010,000	1,990,000	1,950,000				1,980,000	70,000
Particulate Matter (lb/hr)	22,900	25,000	24,900				24,300	2,900
Particulate Matter (mg/Nm ³)	5,180	5,710	5,800				5,560	830
Maior Species								
Aluminum	650,000	740,000	760,000	7.4 @	31 @	28 @	720,000	150,000
Iron	490,000	520,000	540,000	6.9 @	36 @	22 @	510,000	61,000
Sodium	19,000	18,000	19,000	18 @	81 @	59 @	19,000	1,400
Sulfate	63,000	13,000	130,000	4,800,000	4,800,000	4,700,000	4,900,000	62,000
Titanium	35,000	40,000	40,000	ND(0.21)	1.1	0.38	38,000	6,200
Target Species								
Arsenic	1,300	2,100	1,900	ND(0.14)	ND(0.14)	0.80	1,800	1,100
Barium	5,500	6,200	6,300	ND(0.11)	0.66 @	0.11	6,000	1,000
Bervllium	110	140	130	ND(0.12)	0.21 @	0.24 @	130	34
Cadmium	6.9 @	35	18 @	0.071 @	0.18	1.3	21	36
Chloride •	10,000	16,000	11,000	14,000	10,000	14,000	25,000	1,800
Chromium	760	950	018	1.3	3.0	3.5	840	250
Cobalt	200 @	340 @	320 @	ND(0.71)	ND(0.72)	ND(0.75)	290	190
Copper	980	1,300	1,100	ND(0.80)	1.1	0.58	1,100	370
Fluoride	680	1,800	1,300	5,200	4,000	4,500	5,800	250
Lead	290	380	360	2.1	2.1 @	5.2 @	340	110

,

		Solid Phase		*	apor Phase			
	•		Din 4	Run 2	Run 3	Run 4	Mean	95% CI
Substance	Kun 2		UL0	0.57	1.0	1.8	890	110
Manganese	940	000	0/0		69	6.8	H	4.9
Mercury	3.2	4.8	3.2	1.5			96	061
Molubenim	ND(54)	110 @	120 @	ND(1.0)	ND(0.97)	(0.1)UN	00	
	640	680	680	0.61 @	1.8 @	2.4 @	660	80
Nickel	4 700	4.400	4,200	ND(180)	ND(180)	ND(190)	4,500	650
Phosphorus	8	001	86	3.5	2.2 @	2.1 @	66	4
Selenum	1 200	1.600	1,500	ND(0.50)	0.47	ND(0.53)	1,500	560
V arracium								
PAHs c			10,0040				ND(0.0048)	1
5-methyl chrysene	ND(0.0013)	(ccn00.0)(IN	ND(0.0040)				ND(0.078)	
7H-dibenzo[c,g]carbazole	(610.0)DN	ND(0.015)	ND(0.078)				0.015	0.039
Acenaphthene	0.0063	0.0048	0.033				1010	0.0045
Acenaphthylene	0.0024	0.0052	0.0058				0100 0	
Anthracene	0.0044	0.0046	0.0058				0.0049	2100'0
	0.023	ND(0.0036)	ND(0.0068)				0.0094	670'0
Benzolajpyrene		0,000	ND(0.032)				ND(0.032)	:
Benzo{b,j&k]fluoranthenes	(connin)(IN						ND(0.041)	1
Benzo[ghi]perylene	ND(0.0027)	(/ £00.0)UN	ND(0.041)				ND(0.015)	
Benz[a]anthracene	0.0011	0.0020	(CI0.0)(1N				0.0045	0.0061
Chrysene	0.0055	0.0017	0.0063				(CIO DIUN	
Dibenzo[a,e]pyrene	ND(0.0023)	ND(0.0015)	ND(0.012)				(210-0)GNI	
Dibenzola, hlpyrene	ND(0.0033)	ND(0.0027)	ND(0.058)				(eco.o)civi	
Dihenzofa, ilpvrene	ND(0.0042)	ND(0.0026)	ND(0.018)				(910.0)(N)	
Ditensia hlacridine	ND(0.011)	ND(0.0027)	ND(0.032)				ND(0-032)	
Ditraufa hlantheara	ND(0.0015)	ND(0.00087)	ND(0.053)				ND(0.053)	;
		ND/0 00301	ND(0.022)				ND(0.022)	;
Dibenz[a,i]acridine	(nom n) riv	(~~~~)/71	//~~					

Table 3-6 (Continued)

...

ued)
Contin
S
Ý
¢,
Table

3-16

		Solid Phase			Vapor Phase			-
Substance	Run 2	Run 3	Run 4	Run 2	Run 3	Run 4	Mean	95% CI
Fluoranthene	0.024	0.0061	0.0085				0.013	0.024
Fluorene	0.015	0.0049	0.0061				0.0087	0.014
Indeno[1,2,3-cd]pyrene	ND(0.0017)	ND(0.0031)	ND(0.021)				ND(0.021)	1
Phenanthrene	0.076	0.022	0.021				0.040	0.078
Pyrene	0.015	0.0049	0.0092				0.0097	0.012

* Some of the solid-phase chloride measured at this location may have actually been vapor-phase chloride captured on the particulate fraction during sampling. See the discussion in the text under the ESP inlet results.

^b Vapor-phase and total selenium concentrations may be biased low. See the discussion in the text under the ESP inlet results.

* PAHs are total particulate + vapor measurements. PAH results are considered order-of-magnitude estimates because of matrix interference problems. See Section 5 for further details.

ND = Not detected at the concentration in parentheses. CI = Confidence Interval.

@ = Result is less than five times the detection limit.
may have been overestimated, it appears that there was at least some chloride actually present in the particulate phase. The total concentrations of chloride are lower at the stack than at the ESP inlet, which suggests removal of particulate-phase chloride across the ESP.

For fluoride, the OFA/LNB results are consistent, with no evidence of artifacts. However, for the OFA test, the ESP inlet vapor-phase and total concentrations appear to be biased low when compared to the coal and stack gas fluoride concentrations, but not because of reaction with the fly ash in the sampling train. The total concentration of fluoride at the ESP inlet for the OFA test is only about 40% of the concentration at the stack, which is clearly inconsistent. The ESP inlet fluoride results obtained during the OFA/LNB test are probably more representative.

For mercury, the OFA test results indicate that the total concentration of mercury measured in the ESP inlet gas, which is only about 60% of that expected based on the coal composition, may be biased low. There is also evidence of vapor-phase mercury being sorbed onto the fly ash during sampling. The concentrations of mercury in the ESP inlet particulate are, on average, more than twice those measured in the ash collected by the ESP, while the vapor-phase mercury concentrations are lower at the ESP inlet than at the stack.

For the OFA/LNB test, the ESP inlet mercury total concentrations agree well with those expected based on the coal mercury concentrations, and the vapor-phase levels are consistent with those measured at the stack. There is no evidence of artifacts, and it appears that the high level of mercury in the ESP inlet particulate, which accounts for roughly half of the total mercury, may reflect the true distribution in this stream. As was previously discussed under the ash stream results, the relatively high levels of unburned carbon in the ash may be sorbing the mercury onto the ash.

For selenium, the OFA test results suggest that both the vapor-phase and total selenium concentrations measured at the ESP inlet are biased low. The total ESP inlet selenium concentration is only about 40% of that expected based on the coal measurements, and the vapor-phase concentrations of selenium are dramatically lower than those measured at the stack. While it is possible that there may be some of the vapor-phase selenium reacting with the fly ash during sampling, there is an indication that significant amounts of vapor-phase selenium may not have been recovered from the ESP inlet sampling train. The multi-metals train used at the ESP inlet location (described in Appendix A) was modified to use an in situ filter for particulate capture, with a Teflon® transfer line transporting the vapor to the impinger train. Because the Teflon® line was not heated to duct temperatures, vapor-phase selenium may have condensed in this line. Even though the transfer line was rinsed during recovery, Radian's experience with selenium deposits indicates that, if selenium had condensed in the transfer line, it could not have been recovered by rinsing with dilute nitric acid. The Method 5 sampling train used at the stack, with its heated glass-lined probe, does not suffer from this same limitation.

Results

For the OFA test, the selenium ESP inlet results are less clear. As with the OFA/LNB test, there is very little vapor-phase selenium measured in the ESP inlet gas. However, considering that the total amount of selenium in the ESP inlet gas is about 170% of that expected based on the coal measurements, it doesn't appear that any selenium was lost during sampling. The lack of vapor-phase selenium in the ESP inlet gas might be attributed to reaction with the fly ash in the sampling train, but there is also an apparent high bias in the total selenium results at this location.

Stack Gas

Tables 3-7 and 3-8 show the concentrations of the target analytes measured in the stack gas during the OFA and OFA/LNB tests, respectively. For the OFA test, blank corrections were applied for many of the target substances. The details of these corrections can be found in Appendix I.

For the OFA/LNB test, blank contributions for most of the target analytes were not significant; therefore, no blank corrections were applied. The exceptions are Cr(VI) and formaldehyde. For Cr(VI), a consistent, significant background concentration was measured in the KOH reagent used in the sampling train. This background contribution was subtracted from the sample results. The details of the blank correction are included in Appendix I. For formaldehyde, blank contributions from field, trip, and reagent blanks were as large as or even greater than the sample results. However, a consistent contamination level could not be defined; therefore, the results were not corrected for the blank results.

Emission Factors

Table 3-9 presents the mean emission factors for target species in the stack gas during both the OFA and OFA/LNB tests. The mass emission rates (lb/hr) for each substance were divided by the heat input (Btu/hr) of the coal.

Mean particulate matter emissions at the stack were $0.21 \text{ lb}/10^6$ Btu for the OFA test and $0.12 \text{ lb}/10^6$ Btu for the OFA/LNB test. Chloride and fluoride emission factors are the highest of the target species, which is expected because the vapor-phase species (HCl and HF) are not effectively removed by the ESP, and because the concentrations of chloride and fluoride in the coal are higher than those of the other target species.

ESP Performance

Table 3-10 shows the removal efficiencies for the target species across the ESP for both the OFA and OFA/LNB tests. Most of the inorganic species are effectively removed by the ESP, except for chloride, fluoride, and mercury. Although selenium shows a removal of 72% for the OFA test, essentially no removal of selenium was measured during the

			Solid Phase					Vapor Phase				
Substance	Run 1	Run 2	Run 3	Run 4	Run 5	Run 1	Run 2	Run 3	Run 4	Run 5	Mean	95% CI
Date	3/3/91	3/4/91	3/5/91	3/6/91	3/6/91							
Gas Flow Rate (dscfm)	1,200,000	1,140,000	1,080,000	1,060,000	1,060,000						1,110,000	75,700
Gas Flow Rate (Nm ³ /hr)	1,900,000	1,800,000	1,710,000	1,680,000	1,680,000						1,750,000	118,000
Particulate Matter (lb/hr)			486	1,290	962						913	1,000
Particulate Matter (mg/Nm ³)			129	349	260						246	275
Major Species												
Aluminum			12,000	27,000	23,000			ND(73)	(<i>11</i>)	ND(68)	21,000	19,000
Iron			000'6	21,000	17,000			ND(15)	ND(15)	ND(14)	16,000	15,000
Sodium			330 @	670 @	580@			ND(370)	ND(380)	ND(340)	530	440
Sulfate			8,200	15,000	14,000			4,100,000	4,000,000	4,100,000	4,100,000	140,000
Titanium			610	1,400	1,300			ND(18)	ND(19)	(11)UN	1,100	1,100
Target Species												
Arsenic			71	:	150			ND(1.5)	ND(1.5)	(1.4)	110	500
Barium			160	320	290			ND(3.7)	ND(3.8)	ND(3.4)	260	210
Beryllium			2.2	5.5	5.1			ND(0.73)	ND(0.44)	ND(0.39)	4.3	4.5
Cadmium			0.34@	-	0.83@			ND(0.37)	ND(0.38)	ND(0.34) ^b	0.59	3.1
Chloride			13	24	57			21000	22000	23000	22000	3200
Chromium			26	53	51			ND(3.7)	ND(3.8)	ND(3.4)	4	37
Cobalt			7.6@	17	15			ND(1.7)	ND(2.2)	ND(1.9)	13	13
Copper			27	59	59			ND(7.3)	(1.1)dN	ND(6.8)	48	46

Table 3-7 Stack Gas Composition - OFA Test $(\mu g/Nm^3 \text{ unless noted})$ · · ·

Results

			Solid Phase					Vapor Phase				
Substance	Run 1	Run 2	Run 3	Run 4	Run 5	Run 1	Run 2	Run 3	Run 4	Run 5	Mean	95% CI
Fluoride			470	380	490			7200	6700	6800	7300	630
Lead			42	1	36			ND(1.1)	ND(1.2)	ND(1.0)*	39	38
Manganese			20	36	30			ND(3.7)	ND(3.8)	ND(3.4)*	56	21
Mercury			0.13@	0.54	0.37			6.6	7.2	7.6	7.5	1.6
Molybdenum			ND(8.7)*	ND(14)*	16@ ه			ND(18)	ND(19)	(11)UN	ND(14)	
Nickel			18	34	33			ND(7.3)	(L.1)(UN	ND(6.8)	28	22
Phosphorus			160@	330@	320 @			ND(96)	(111)QN	ND(120)	270	230
Selenium			7.9	1	34			130	130	120	150	100
Vanadium			50	001	100			ND(7.3)	(<i>t.1</i>)	ND(6.8)	84	73
Benzene						1.3@	3.3	0.65@			1.7	3.4
Toluene						ND(0.58)	ND(0.57)	ND(0.54)			ND(0.58)	;
Formaldehyde ^c						2.7@		ND(4.2)	ND(4.7)		ND(4.7)	;
POM											-QN	
			•		Ē			the formed	in According	_		

- None of the compounds classified as POM were present in concentrations above detection limits. The results for individual compounds can be found in Appendix B.

^b Blank corrections were greater than 50% of uncorrected result.

. Formaldehyde was detected in blanks in concentrations similar to those in the samples. Results shown here are corrected for the blank results.

ND = Not detected at concentration in parentheses.<math>CI = Confidence Interval. @ = Results is less than five times the detection limit.

Results

Continued) 2-7 (Continued)

		Solid Phase			Vapor Pha	e		
Substance	Run 2	Run 3	Run 4	Run 2	Run 3	Run 4	Mean	95% CI
Date	5/19/93	5/20/93	5/21/93					
Gas Flow Rate (dscfm)	1,070,000	1,090,000	1,080,000				1,080,000	19,000
Gas Flow Rate (Nm ³ /hr)	1,690,000	1,720,000	1,710,000				1,710,000	31,000
Particulate Matter (lb/hr)	427	642	502				524	271
Particulate Matter (mg/Nm ³)	114	170	133				139	70
Major Species								
Aluminum	12,000	17,000	11,000	14@	7.5@	20%	13,000	7,500
Iron	6,700	13,000	10,000	ND(4.4)	ND(3.4)	ND(5.0)	11,000	3,700
Sodium	420	730	470	70	15@	23@	570	360
Sulfate	13,000	16,000	17,000					210,000
Titanium	760	1,000	840	030@	ND(0.14)	ND(0.21)	880	340
Target Species								
Arsenic	110	110	150	2.0	1.1	2.1	120	65
Barium	150	190	150	0.11@	ND(0.07)	0.32@	160	67
Beryllium	2.6	4.2	3.4	0.28@	0.10@	ND(0.11)	3.5	1.8
Cadmium	2.9@	4.7@	4.8	ND(0.058)	ND(0.045)	ND(0.066)	4.2	2.7
Chloride	400	230	200	17,000	17,000	19,000	18,000	2,600
Chromium	20	27	23	1.0	0.77	1.1	24	8.6
Chromium (VI).	6.8 ^b	2.6@ ^b	9.3				6.2	8.4
Cobalt	6.4	8.9	7.2	ND(0.62)	ND(0.47)	ND(0.70)	7.5	3.3
Copper	29	43	33	ND(0.69)	ND(0.53)	ND(0.79)	35	18

Table 3-8 Stack Gas Composition - OFA/LNB Test (μ g/Nm³ unless noted)

Results

		Solid Phase			Vapor Phas	Ð		
Substance	Run 2	Run 3	Run 4	Run 2	Run 3	Run 4	Mean	95% CI
Fluoride	98	64	11	5,500	5,700	6,100	5,900	750
Lead	10	15	14	0.31@	ND(0.15)	ND(0.23)	13	5.8
Manganese	22	26	22	0.53	0.14@	0.24@	24	5.5
Mercury	0.29	0.12	0.17	4.4	5.3	6.4	5.5	2.3
Molyb de num	13	15	14	ND(0.83)	ND(0.64)	ND(1.0)	14	1.9
Nickel	16	21	19	0.82@	0.50@	0.64@	19	5.1
Phosphorus	190	220	210	ND(150)	ND(120)	ND(170)	210	33
Selenium	14	16	16	110	120	210	160	140
Vanadium	39	56	47	ND(0.44)	ND(0.34)	ND(0.50)	47	20
Benzene				ND(0.59)	ND(0.52)	0.88@	ND(0.59)	
Toluene				0.72@4	0.79@ ⁴	0.91@	0.81	0.24
Formaldehyde*				2.7°	0.54 *	1.4 °	1.5	2.7
PAHs * 1								
5-methyl chrysene	0.00028	ND(0.00017)	ND(0.00098)				ND(0.00098)	'
7H-dib en zofc,g]carbazolc	0.0051	ND(0.0052)	ND(0.018)				ND(0.018)	1
Accnaphthene	0.018	0.0077	0.0025				0.0094	0.019
Acenap h thylene	0.0041	0.0027	0.0036				0.0035	0.0018
Anthracene	0.0069	0.0014	0.0044				0.0042	0.0069
Benzo(a)pyrene	0.0028	ND(0.0025)	ND(0.0048)	. :			ND(0.0048)	ı
Benzo[b.j&k]fluoranthene	0.0037	0.00059	0.00078				0.0017	0.0044
Benzo[ghi]perylene	0.0036	ND(0.0018)	ND(0.0019)				ND(0.0036)	۰
Benz[a]anthracene	0.023	0.00063	0.00077				0.0081	0.032

Results

_	-
7	-
- 2	1
12	4
- 2	-
÷	-
	-
٠	-
. 6	
- 2	2
7	۱
``	,
`	
·	
`` a	0
0	0
0	
۔ م ر	5
- 0 7	Ş
יי 10 10	505
1.20	
hha 2 0 (
ahla 2 0 (
Pakia 2 0 /	

		Solid Phase			Vapor Phas	ş		
Substance	Run 2	Run 3	Run 4	Run 2	Run 3	Run 4	Mean	95% CI
Chrysene	0.0036	0.0016	0.00091				0.0020	0.0034
Dibenzo[a,e]pyrene	ND(0.0023)	ND(0.00095)	ND(0.0034)				ND(0.0034)	,
Dibenzo[a,h]pyrene	ND(0.0037)	ND(0.00076)	ND(0.0034)				ND(0.0037)	'
Dibenzo[a,i]pyrene	ND(0.0033)	ND(0.0011)	ND(0.0049)				ND(0.0049)	'
Dibenz[a,h]acridine	ND(0.0018)	ND(0.0016)	ND(0.0018)				ND(0.0018)	'
Dibenz[a,h]anthracene	ND(0.0015)	ND(0.0089)	ND(0.0043)				ND(0.0043)	
Dibenz[a,i]acridine	ND(0.0048)	ND(0.0015)	ND(0.0011)				ND(0.0048)	-
Fluoranthene	0.026	0.0045	0.0049	-			0.012	0.030
Fluorene	0.024	0.0047	0.0059				0.011	0.026
Indeno[1,2,3-cd]pyrene	0.0031	0.00074	ND(0.0028)				ND(0.0031)	'
Phenanthrene	0.10	0.025	0.025				0.051	0.11
Pyrene	0.031	0.0036	0.0023				0.012	0.040

* PAHs and Cr(VI) are total particulate matter plus vapor measurements.

^b Blank concentration exceeds 50% of the uncorrected result. Blank contribution was subtracted.

"Blank concentration exceeds 50% of the uncorrected result. Not corrected for the blank results.

^d Tolucne was detected in blank samples at similar levels. Results not corrected for the blank results.

· Formaldehyde was detected in blanks in concentrations similar to those of the samples. Results shown here are not corrected for the blank results.

f PAH results are considered order-of-magnitude estimates because of matrix interference problems. See Section 5 for further details.

ND = Not detected at the concentration in parentheses.

CI = Confidence Interval.

 \boldsymbol{Q} = Result is less than dive times the detection limit.

Table 3-9 Stack Emission Factors - OFA and OFA/LNB Tests (lb/10¹² Btu unless noted)

Substance Combined Mean Combined 95% CI Combined Mean 95% CI Gas Flow Rate (dscfm) 1,110,000 75,700 1,080,000 19,000 Gas Flow Rate (Nm ³ /hr) 1,750,000 118,000 1,710,000 31,000 Coal Flow Rate (lb/hr, dry) 330,000 6,100 315,000 5,200 Heating Value (Btu/lb, dry) 13,700 370 13,800 200 Particulate Matter (lb/l0 ⁶ Btu) 0.21 0.24 0.12 0.061 Target Species		OFA Test		OFA/LNE	Test
Gas Flow Rate (dscfm)1,110,00075,7001,080,00019,000Gas Flow Rate (Nm $^3/hr)$ 1,750,000118,0001,710,00031,000Coal Flow Rate (lb/hr, dry)330,0006,100315,0005,200Heating Value (Btu/lb, dry)13,70037013,800200Particulate Matter (lb/l0 6 Btu)0.210.240.120.061Target Species43011056Barium22018014058Beryllium3.73.83.11.5Cadmium0.502.73.62.4Chloride19,0002,40015,0002,400Chromium3832217.5Chromium (VI)NA5.47.3Cobalt111116.52.9Copper411413015Fluoride6,2005705,100710Lead3334115.1Manganese2518214.9Mickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueaeND(4.0)0.700.21FormaldehydeND(4.0)1.32.4	Substance	Combined Mean	95% CI	Combined Mean	95% CI
Gas Flow Rate (Nm ³ /hr) 1,750,000 118,000 1,710,000 31,000 Coal Flow Rate (lb/hr, dry) 330,000 6,100 315,000 5,200 Heating Value (Btu/lb, dry) 13,700 370 13,800 200 Particulate Matter (lb/10 ⁶ Btu) 0.21 0.24 0.12 0.061 Target Species	Gas Flow Rate (dscfm)	1,110,000	75,700	1,080,000	19,000
Coal Flow Rate (lb/hr, dry) 330,000 6,100 315,000 5,200 Heating Value (Btu/lb, dry) 13,700 370 13,800 200 Particulate Matter (lb/10 ⁴ Btu) 0.21 0.24 0.12 0.061 Target Species	Gas Flow Rate (Nm ³ /hr)	1,750,000	118,000	1,710,000	31,000
Heating Value (Btu/lb, dry) 13,700 370 13,800 200 Particulate Matter (lb/10 ⁶ Btu) 0.21 0.24 0.12 0.061 Target Species	Coal Flow Rate (lb/hr, dry)	330,000	6,100	315,000	5,200
Particulate Matter (lb/10 ⁶ Btu) 0.21 0.24 0.12 0.061 Target Species 94 430 110 56 Barium 220 180 140 58 Beryllium 3.7 3.8 3.1 1.5 Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19	Heating Value (Btu/lb, dry)	13,700	370	13,800	200
Target Species Arsenic 94 430 110 56 Barium 220 180 140 58 Beryllium 3.7 3.8 3.1 1.5 Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19 17 4.5 Phosphorus 230 2	Particulate Matter (lb/10 ⁶ Btu)	0.21	0.24	0.12	0.061
Arsenic 94 430 110 56 Barium 220 180 140 58 Beryllium 3.7 3.8 3.1 1.5 Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19 17 4.5 Phosphorus 230 200 180 31	Target Species			<u></u>	
Barium 220 180 140 58 Beryllium 3.7 3.8 3.1 1.5 Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19 17 4.5 Phosphorus 230 200 180 31 Selenium 128 33 140 120	Arsenic	94	430	110	56
Beryllium 3.7 3.8 3.1 1.5 Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19 17 4.5 Phosphorus 230 200 180 31 Selenium 128 33 140 120 Vanadium 72 64 41 18	Barium	220	180	140	58
Cadmium 0.50 2.7 3.6 2.4 Chloride 19,000 2,400 15,000 2,400 Chromium 38 32 21 7.5 Chromium (VI) NA 5.4 7.3 Cobalt 11 11 6.5 2.9 Copper 41 41 30 15 Fluoride 6,200 570 5,100 710 Lead 33 34 11 5.1 Manganese 25 18 21 4.9 Mercury 6.4 1.1 4.8 2.0 Molybdenum ND(12) 12 1.8 Nickel 24 19 17 4.5 Phosphorus 230 200 180 31 Selenium 128 33 140 120 Vanadium 72 64 41 18 Benzene 1.4 3.0 ND(0.51)	Beryllium	3.7	3.8	3.1	1.5
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Cadmium	0.50	2.7	3.6	2.4
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Chloride	19,000	2,400	15,000	2,400
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Chromium	38	32	21	7.5
Cobalt11116.52.9Copper41413015Fluoride6,2005705,100710Lead3334115.1Manganese2518214.9Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Chromium (VI)	NA		5.4	7.3
Copper41413015Fluoride6,2005705,100710Lead3334115.1Manganese2518214.9Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Cobalt	11	11	6.5	2.9
Fluoride6,2005705,100710Lead3334115.1Manganese2518214.9Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Copper	41	41	30	15
Lead3334115.1Manganese2518214.9Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Fluoride	6,200	570	5,100	710
Manganese2518214.9Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Lead	33	34	11	5.1
Mercury6.41.14.82.0MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Manganese	25	18	21	<u>4.9</u>
MolybdenumND(12)121.8Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Mercury	6.4	1.1	4.8	2.0
Nickel2419174.5Phosphorus23020018031Selenium12833140120Vanadium72644118Benzene1.43.0ND(0.51)TolueneND(0.49)0.700.21FormaldehydeND(4.0)1.32.4	Molybdenum	ND(12)		12	1.8
Phosphorus 230 200 180 31 Selenium 128 33 140 120 Vanadium 72 64 41 18 Benzene 1.4 3.0 ND(0.51) Toluene ND(0.49) 0.70 0.21 Formaldehyde ND(4.0) 1.3 2.4	Nickel	24	19		4.5
Selenium 128 33 140 120 Vanadium 72 64 41 18 Benzene 1.4 3.0 ND(0.51) Toluene ND(0.49) 0.70 0.21 Formaldehyde ND(4.0) 1.3 2.4	Phosphorus	230	200	180	31
Vanadium 72 64 41 18 Benzene 1.4 3.0 ND(0.51) Toluene ND(0.49) 0.70 0.21 Formaldehyde ND(4.0) 1.3 2.4	Selenium	128	33	140	120
Benzene 1.4 3.0 ND(0.51) Toluene ND(0.49) 0.70 0.21 Formaldehyde ND(4.0) 1.3 2.4	Vanadium	72	64	41	18
Toluene ND(0.49) 0.70 0.21 Formaldehyde ND(4.0) 1.3 2.4	Benzene	1.4	3.0	ND(0.51)	
Formaldehyde ND(4.0) 1.3 2.4	Toluene	ND(0.49)		0.70	0.21
	Formaldehyde	ND(4.0)		1.3	2.4

.

Table 3-9 (Continued)

	OFA Test		OFA/LNE	B Test
Substance	Combined Mean	95% CI	Combined Mean	95% CI
PAHs				
5-methyl chrysene	NA		ND(0.0009)	
17H-dibenzo[c,g]carbazole	NA		ND(0.016)	
Acenaphthene	ND(5)		0.0081	0.017
Acenaphthylene	ND(5)		0.0030	0.0016
Anthracene	ND(5)		0.0037	0.0060
Benzo[a]pyrene	ND(5)		ND(0.0041)	
Benzo[b,j&k]fluoranthenes	NA		0.0015	0.0038
Benzo[ghi]perylene	ND(5)	<u>.</u>	ND(0.0031)	
Benz[a]anthracene	ND(5)		0.0070	0.02748
Chrysene	ND(5)		0.0018	0.0030
Dibenzo[a,e]pyrene	NA		ND(0.0030)	
Dibenzo[a,h]pyrene	NA		ND(0.0032)	
Dibenzo[a,i]pyrene	NA		ND(0.0042)	
Dibenz[a,h]acridine	NA		ND(0.0016)	
Dibenz[a,h]anthracene	ND(5)		ND(0.0037)	
Dibenz[a,i]acridine	NA		ND(0.0042)	
Fluoranthene	ND(5)		0.010	0.026
Fluorene	ND(5)		0.0099	0.023
Indeno[1,2,3-cd]pyrene	ND(5)		ND(0.0027)	
Phenanthrene	ND(5)	-	0.044	0.098
Pyrene	ND(5)		0.011	0.035

ND = Not Detected. Value in parentheses is based on the detection limit. CI = Confidence Interval. NA = Not Analyzed.

Results

Table 3-10 ESP Removal Efficiency - OFA and OFA/LNB Tests

	OFA 7	ſest	OFA/LN	B Test
Substance	Removal (%) *	95% CI (%)	Removal (%) *	95% CI (%)
Particulate Matter	96	4	98	1
Target Species	····			
Arsenic	95	25	94	3
Barium	96	2	98	1_
Benzene	0 ^b	190	NA	
Beryllium	97	4	98	1
Cadmium	NC		83	18
Chloride	46	11	39	7
Chromium	96	3	98	1
Cobalt	97	3	98	1
Copper	97	3	98	1
Fluoride	0 •	56	13	9
Formaldehyde	NC		NA	
Lead	92	4	97	1
Manganese	97	2	98	0.4
Mercury	9¢	80	55	17
Molybdenum	NC		86	16
Nickel		3	98	1
Phosphorus	96	3	96	1
Selenium	72	14	0*	130
Toluene	NC		NA	
Vanadium	96	3	97	1
PAHs				
5-methyl chrysene	NA		NC	
7H-dibenzo[c,g]carbazole	NC		NC	
Acenaphthene	NC		45	93
Acenaphthylene	NC		33	48
Anthracene	NC		26	123

.

Table 3-10 (Continued)

	OFA 7	Test	OFA/LN	B Test
Substance	Removal (%) *	95% CI (%)	Removal (%) *	95% CI (%)
Benzo[a]pyrene	NC		> 56	
Benzo[b,j&k]fluoranthenes	NA		NC	
Benzo[ghi]perylene	NC		NC	
Benz[a]anthracene	NC		NC	
Chrysene	NC		61	57
Dibenzo[a,e]pyrene	NA		NC	
Dibenzo[a,h]pyrene	NA		NC	
Dibenzo[a,i]pyrene	NA		NC	
Dibenz[a,h]acridine	NA		NC	
Dibenz[a,h]anthracene	NC		NC	
Dibenz[a,i]acridine	NA		NC	
Fluoranthene	NC		21	170
Fluorene	NC		<u> </u>	220
Indeno[1,2,3-cd]pyrene	NC		NC	
Phenanthrene	NC		0 ^b	210
Pyrene	NC	=-	0 •	370

*Removal efficiencies were calculated based on mass rates, i.e., concentration times flow rate, rather than concentrations alone.

^b Calculated removal was negative but is shown as zero.

^c Based on an ESP inlet concentration that may be biased low. See the discussion in the text under the ESP inlet results.

NC = Not calculated because substance concentration was below the detection limit at the ESP inlet.

NA = Not applicable because substance was not analyzed at the ESP inlet.

CI = Confidence Interval.

Results

OFA/LNB test. However, the confidence interval is very large, indicating considerable uncertainty in this measurement.

The removal efficiencies for volatile organic compounds were measured during the OFA test. Essentially no benzene was removed and the removal efficiencies for toluene and formaldehyde were not calculated because their concentrations in the ESP inlet gas were below detection limits. The removal efficiencies for these substances during the OFA/LNB test could not be determined because they were not sampled in the ESP inlet gas. This decision was made because the concentrations of benzene, toluene, and formaldehyde were low for the OFA test and are not expected to be controlled by the ESP.

The removal efficiencies for the PAHs were measured during the OFA/LNB test. Several of the compounds were below detection limits in the ESP inlet gas; therefore, their removal efficiencies were not calculated. For those compounds for which a removal efficiency could be calculated, the large confidence intervals indicate high uncertainties.

Speciation of Mercury

Stack gas samples were collected with the Frontier Geoscience's mercury speciation train during the OFA/LNB test, and the results are shown in Table 3-11. The analysis only included vapor-phase mercury. The results show that 40% of the stack mercury is present as ionic inorganic mercury, 54% as elemental mercury, and 6% as methyl mercury. These results are discussed and compared to the multi-metals train results in Section 4.

Mercury speciation samples were also collected during the OFA test, but by a substantially different technique that has since been shown to be invalid. These results are presented in Appendix D.

Size-Fractionated Stack Particulate Matter

Table 3-12 shows the results obtained from the chemical analysis of size-fractionated stack particulate matter from the OFA/LNB test. The samples were collected with University of Washington Mark 5 cascade impactors. Individual impactor stages were combined before analysis to produce three target size ranges: $< 3 \mu m$, $3-10 \mu m$, and $> 10 \mu m$. The actual size ranges obtained were $< 4 \mu m$, $4-9 \mu m$, and $> 9 \mu m$. Samples were digested in the microwave and analyzed using inductively coupled plasma mass spectrometry (ICP-MS). The results are reported as mg/kg of the target analytes in each size fraction.

Size-fractionated particulate matter samples were also collected during the OFA test. The samples were analyzed by instrumental neutron activation analysis (INAA), but the

Table 3-11 Mercury Speciation in Stack Gas - OFA/LNB Test (μ g/Nm³ unless noted)

Component*	Run 2 5/19/93	Run 3 5/20/93	Run 4 5/21/93	Mean	95% CI	Percent of Vapor Hg
Ionic Inorganic Hg	2.4	3.3	2.1	2.6	1.6	39%
Elemental Hg	4.1	3.4	3.3	3.6	1.1	54%
Methyl Hg	0.16	0.97	0.20	0.44	1.1	7%
Total Vapor Hg	6.7	7.7	5.6	6.6	2.6	

* Results determined using Frontier Geoscience's mercury speciation train.

CI = Confidence Interval.

Results

Table 3-12

Size-Fractionated Stack Particulate Matter - OFA/LNB Test (mg/kg)

Substance	Run 2 5/19/93	Run 3 5/20/93	Run 4 5/21/93	Mean	95% CI
Gas Volume (Nm ³)	3.74	3.75	3.96	24	
< 4 μ m Fraction					
Collected Mass (mg)	35.4	29.6	27.7		
Antimony	83	87	94	88	13
Arsenic	1,400	1,900	1,700	1,700	580
Barium	2,000	1,900	2,100	2,000	330
Beryllium	24	33	29	29	11
Cadmium	6.4	5.7	8.7	6.9	3.8
Chromium	39 0	490	430	440	130
Cobalt	56	68	58	61	17
Copper	420	400	360	390	82
Lead	320	200	220	250	160
Manganese	200	230	180	200	65
Mercury	2.6	2.8	1.2	2.2	2.2
Molybdenum	240	280	290	270	76
Nickel	170	200	170	180	33
Selenium	100	130	140	120	46
Vanadium	610	770	730	700	200
4-9 µm Fraction			- <u></u>		
Collected Mass (mg)	40.7	33.7	34.4		
Antimony	43	44	42	43	3.1
Arsenic	690	790	740	740	130
Barium	1,800	1,700	1,700	1,700	160
Beryllium	38	41	39	39	3.3
Cadmium	4.2	7.3	3.6	5.0	4.9
Chromium	380	400	370	380	33
Cobalt	98	97	94	96	5.6
Соррег	700	400	400	500	430
Lead	270	200	170	210	130
Manganese	300	250	240	260	81
Mercury	3.2	2.2	1.2	2.2	2.6
Molybdenum	64	62	56	61	11
Nickel	240	220	220	230	22
Selenium	53	36	42	44	21
Vanadium	630	640	640	640	8.9

~

Table 3-12 (Continued)

a b <i>i</i>	Run 2	Run 3	Run 4		
Substance	5/19/93	5/20/93	5/21/93	Меал	95% Cl
> 9 µm Fraction		. <u></u>			
Collected Mass (mg)	164	207	164	40-00	-
Antimony	17	21	18	19	5
Arsenic	310	410	410	380	140
Barium	1,200	1,400	1,300	1,300	240
Beryllium	22	25	28	25	8
Cadmium	3.2	5.1	1.7	3.4	4.3
Chromium	240	230	370	280	190
Cobalt	58	59	71	62	18
Copper	1,700	620	430	900	1,600
Lead	120	120	92	110	35
Manganese	230	280	220	250	75
Mercury	2.4	5.6	1.6	3.2	5.3
Molybdenum	26	29	29	28	4
Nickel	150	140	220	170	110
Selenium	63	100	120	96	77
Vanadium	350	350	430	380	110

CI = Confidence Interval.

,

-

Results

quality of the results was poor because of the relatively low concentrations. These results are presented in Appendix D. (The results from the OFA test led to the modification of the techniques used for the OFA/LNB test.)

Other Species Detected

Other substances not on the target analyte list but listed in Title III of the Clean Air Act Amendments of 1990 were also measured as part of the multi-substance techniques used for the target analytes. Additional Title III substances include antimony and organic compounds available from the VOST, semivolatile compounds, and aldehydes analyses. Table 3-13 shows the concentrations of antimony and organic compounds that were detected in at least one flue gas sample.

		Concentration	us (#g/Nm ³ unles	is noted)"			
Substance	Run 1	Run 2	Run 3	Run 4	Run 5	Mean	95% CI
ESP Inlet Gas - OFA Test							
Acetaldehyde	65		ND(7.8)	ND(8.3)		24	87
Acrolein	12@		ND(9.4)	ND(10)		ND(10)	:
Antimony			ND(590)	ND(540)	ND(590)	ND(590)	:
Bis(2-ethlyhexyl)phthalate ^b	400	55	7.7@			150	530
Bromomethane	0.68@	1.7@	ND(0.56)			0.89	1.8
Dibutylphthalate	270	30	ND(5.0)			100	360
Methylene chloride ^b	1.3@	ND(0.54)	3.4			1.7	4.0
Phenol	ND(5.8)	ND(4.9)	ND(5.0)			ND(5.8)	:
1,1,1-Trichloroethane	0.58@	1.0@	4.5			2.0	5.3
Stack Gas - OFA Test							
Acetaldchyde ^b	4.9@		8.1@	ND(5.6)		ND(5.6)	: [
Acrolein	ND(2.3)		ND(6.2)	ND(6.8)		ND(6.8)	; ;
Antimony			ND(17)	ND(29)	ND(29)	ND(29)	:
Bis(2-ethlyhexyl)phthalatc ^b	11@	ND(5.6)	8.3@			7.3	10
Bromomethane	ND(0.58)	ND(0.57)	ND(0.54)			ND(0.58)	:
Dibutylphthalate ^b	ND(5.3)	ND(5.6)	ND(5.3)			ND(5.6)	:
Methylene chloride ^b	ND(0.58)	2.5@	10			4.4	13
Phenol	ND(5.3)	14@	ND(5.3)			6.4	16
1,1,1-Trichloroethane	ND(0.58)	ND(0.57)	ND(0.54)			ND(0.58)	:
Coal - OFA Test							
Antimony (mg/kg)			160	1.9	1.3	1.4	1.2

Table 3-13 Other Species Detected

3-33

Results

(Continued)
Table 3-13

noted)*
unless
/Nm ³
(a g/
Concentrations

- Substance	Run 1	Run 2	Run 3	Run 4	Run 5	Mean	95% CI
ESP Inlet Gas - OFA/LNB Test							
Antimony		ND(78)	ND(85)	ND(87)		ND(87)	:
Stack Gas - OFA/LNB Test							
Acetaidehyde ^b		5.8	2.4	6.5		4.9	5.5
Antimony		2.4	2.9	2.6		2.6	0.59
Bromomethane		ND(0.59)	1.9@	ND(0.58)		0.83	2.3
Carbon disulfide		ND(0.59)	ND(0.59)	0.61@		ND(0.59)	:
Methylene chloride ^b		6.9	1.1	40		16	52
Coal - OFA/LNB Test							
Antimony (mg/kg)		1.3	1.8	1.3		1.5	0.69

* Total of solid- and vapor-phase concentrations for gas streams.

^b Substances were detected in lab blanks and field blacks. Suspected contaminants.

ND = Not detected at the concentration in parentheses.
 CI = Confidence Interval.
 @ = Result is less than five times the detection limit.

Results

4

DISCUSSION

This section discusses the important results of the testing at Site 16. The results from the OFA and OFA/LNB tests are compared. In addition, special topics such as mercury speciation, chromium speciation, size-fractionation, and elemental enrichment are discussed.

Coal Composition Comparison

Figures 4-1a and 4-1b show the trace element compositions of the coal fired during the OFA and OFA/LNB tests. The error bars represent the 95% confidence intervals about the mean values. The trace element compositions for the two tests are statistically equivalent; for every element, the 95% confidence intervals for the two tests overlap. Cadmium was not detected during either test. The detection limit was higher for the OFA/LNB test, because only INAA was employed, as opposed to a specific Cd analysis by GFAAS for the OFA test. Other coal measurements, such as heating value, moisture, sulfur content, ash content, and major element concentrations are similar for the two tests. Therefore, the coal composition can be considered consistent, and any differences in the emission results between OFA operation and OFA/LNB operation must be attributed to other factors.

Emissions Comparison

Figures 4-2a, 4-2b, and 4-2c show the emission factors for metals, acid gases, and volatile organic compounds for the two tests. The 95% confidence intervals overlap for all of the species shown, indicating that there is not a significant difference between the emissions of the unit under OFA operation or OFA/LNB operation. There is, however, a slight downward trend in the emissions during OFA/LNB operation. For elements primarily associated with the particulate matter, ten (barium, beryllium, chromium, cobalt, copper, lead, manganese, nickel, phosphorus, and vanadium) show lower mean emissions during OFA/LNB operation; only two (arsenic and cadmium) show higher mean emissions during OFA/LNB operation. The total particulate matter emissions were also lower during the OFA/LNB test.

The slight trend toward lower emissions of particulate-phase elements is more a reflection of the ESP performance than it is of the low- NO_x burners. The system



Figure 4-1a Comparison of Coal Compositions

Elemental Concentration in Coal (mg/kg, dry)

4-2

Discussion



4-3



Figure 4-2a Comparison of Emission Factors





Discussion

4-5



Figure 4-2c Comparison of Emission Factors appeared to be more "in tune" during the OFA/LNB test, as shown by the greater overall particulate matter removal efficiency (98% for OFA/LNB vs. 96% for OFA).

Precipitator performance is very sensitive to ash resistivity which, in turn, is sensitive to the unburned carbon content of the fly ash. After the low-NO_x burners were first installed, ESP performance was very poor. However, the burners were tuned by the vendor a short time before Radian conducted the OFA/LNB testing, decreasing the carbon content of the fly ash and improving ESP performance. In addition, NH₃ conditioning during the OFA/LNB test further improved performance. No conditioning was used during the OFA test.

There is some concern that combustion modifications designed to reduce NO_x emissions may also increase the emissions of organic compounds. In response to this, it can be said that the emissions of benzene, toluene, and formaldehyde were very low (either below detection limits or less than five times the detection limits) during both OFA and OFA/LNB operation. Furthermore, the presence of these compounds in laboratory and field blanks makes their measured concentrations in the flue gas highly uncertain.

Comparison of PAH Results

The analysis of PAHs was significantly different for the two tests. For the OFA test, samples were analyzed by GC-MS (Method 8270). For the OFA/LNB test, high-resolution GC-MS was employed for a selected list of PAHs pre-approved by Southern Company Services. The detection limits for high resolution GC-MS were much lower (up to 3 orders of magnitude) than those available through standard GC-MS. As a consequence, the PAH results are difficult to compare.

Table 4-1 lists the PAHs for which results were obtained, along with the range of concentrations measured in the ESP inlet gas and stack gas by each technique. Thirteen of the compounds were measured during both tests. During the OFA test, none of the PAHs were detected; detection limits were about $6 \mu g/Nm^3$ for most of the compounds. For the OFA/LNB test, many of the compounds were detected by high-resolution GC-MS but at concentrations one to four orders of magnitude lower than the detection limits available for the OFA test.

The PAH concentrations measured during the OFA/LNB test show very high variability. As discussed in Section 5, the recoveries of surrogate compounds in actual samples were poor; however, acceptable recoveries of standards spiked into blank samples show that no problem existed with the analytical procedure, and the problem lies with the sample matrix. Possibly, carbonaceous material in the flue gas particulate matter interferes with the analysis; however, high-resolution GC-MS still represents the state of the art for analyzing low levels of PAHs. The PAH results for the OFA/LNB test should be considered order-of-magnitude estimates.

Table 4-1 **Comparison of PAH Results**

PAH Compounds	OFA Test [*] Range of Results (ug/Nm ³)	OFA/LNB Test ^b Range of Results (µg/Nm ³)
Acenaphthene	<6	0.0025 - 0.033
Acenaphthylene	<6	0.0027 - 0.0058
Anthracene	<6	0.0014 - 0.0069
Benz[a]anthracene	<6	0.00063 - 0.023
Benzo[b]fluoranthene	<6	NA
Benzo[k]fluoranthene	<6	NA
Benzo[b,j&k]fluoranthenes	NA	< 0.032
Benzo[ghi]perylene	<6	< 0.041
Benzo[a]pyrene	<6	< 0.0025 - 0.023
Chrysene	<6	0.00091 - 0.0063
Dibenz[a,h]acridine	NA	< 0.032
Dibenz[a,i]acridine	NA	< 0.022
Dibenz[a,h]anthracene	<6	< 0.053
7H-Dibenzo[c,g]carbazole	NA	< 0.078
Dibenzo[a,e]pyrene	NA	< 0.012
Dibenzo[a,h]pyrene	NA	< 0.058
Dibenzo[a,i]pyrene	NA	< 0.018
7,12-Dimethylbenz[a]anthracene	<15	NA
Fluoranthene	<6	0.0044 - 0.026
Fluorene	<6	0.0047 - 0.024
Indeno[1,2,3-cd]pyrene	<6	< 0.021
3-Methylcholanthrene	<6	NA
5-Methyl chrysene	NA	< 0.0048
2-Methylnaphthalene	<6	NA
Naphthalene	<6	NA
Phenanthrene	<6	0.021 - 0.10
Ругепе	<6	0.0023 - 0.031

^{*} Compounds measured by Method 8270 (GC-MS). ^b Compounds measured by high-resolution GC-MS.

NA = Not Analyzed. < = Indicates a detection limit value.

Mercury Speciation

Table 4-2 shows the mercury concentrations measured in the stack gas during the OFA/LNB test using the Frontier Geosciences mercury speciation train and the multimetals train. A different type of mercury speciation train, since shown to be invalid, was used during the OFA test, and those results are discussed in Appendix D.

The Frontier Geosciences train used for the OFA/LNB test is described in detail in Appendix A. The technique was developed by Nicholas Bloom, now with Frontier Geosciences, Inc., but formerly with Brooks Rand, Ltd. The solid sorbent technique uses KCl/soda lime to capture ionic forms of mercury (inorganic Hg^{2+} as well as monomethyl species such as CH_3HgCl), and iodated charcoal to capture elemental mercury. The samples were collected nonisokinetically from a fixed point in the stack. A glass wool plug ahead of the sorbent cartridges prevented particulate matter from entering the sorbents, and this plug was not analyzed; therefore, the results apply to the vapor phase only. Because the particulate loading in the stack gas is relatively low, it is not expected that particulate trapped on the glass wool would have absorbed significant quantities of vapor-phase mercury.

The multi-metals train, also described in Appendix A, uses aqueous impinger solutions to capture vapor-phase metals, including mercury. Samples were collected isokinetically while traversing the stack. Although the multi-metals train was not designed to provide speciation information, it may still give some insight into the mercury species present. Ionic forms of mercury are water-soluble and should be readily captured in the HNO_3/H_2O_2 impingers. Elemental mercury, on the other hand, should pass through the HNO_3/H_2O_2 solutions, because the solubility of elemental mercury in aqueous solutions is very low and the H_2O_2 cannot efficiently oxidize it. The elemental mercury is oxidized and captured in the $H_2SO_4/KMnO_4$ impingers.

The total vapor-phase mercury concentrations measured by the two techniques show good agreement. The mean vapor concentrations are $6.6 \pm 2.6 \,\mu g/\text{Nm}^3$ by the Bloom train and $5.3 \pm 2.5 \,\mu g/\text{Nm}^3$ by the multi-metals train. However, the results for the individual species do not agree as well; the Frontier Geosciences train shows a mean of 39% oxidized mercury, compared with 69% oxidized mercury measured by the multi-metals train.

Blank contamination was not a problem for either type of train, and spike recoveries were within acceptance limits. However, the issue of species conversion during sampling has not been addressed; therefore, although each method can be considered to give reliable results for the total concentration of vapor-phase mercury, less confidence can be placed in the speciation results for either type of train. The possibility of one species converting to another within the sampling equipment or in the sampling media makes it less certain that the species were actually present in the flue gas at the measured levels.

The Frontier Geosciences train is a technique still being developed. Extensive work has been done to improve the capture efficiency of the traps, to increase the analytical

Table 4-2Comparison of Mercury Methods - OFA/LNB Test

Component	Run 2 5/19/93	Run 3 5/20/93	Run 4 5/21/93	Mean	95% CI	Percent of Vapor Hg
Frontier Geosciences I	Ig Speciati	on Train				
Ionic Inorganic Hg	2.4	3.3	2.1	2.6	1.6	39%
Elemental Hg	4.1	3.4	3.3	3.6	1.1	54%
Methyl Hg	0.16	0.97	0.20	0.4	1.1	7%
Total Vapor	6.7	7.7	5.6	6.6	2.6	
Multi-Metals Train						
Ionic Hg*	` 2.7	3.1	5.2	3.7	3.3	69%
Elemental Hg ^b	1.7	2.1	1.2	1.7	1.1	31%
Total Vapor	4.4	5.2	6.4	5.3	2.5	
Solid	0.29	0.12	0.17	0.19	0.22	
Total Vapor + Solid	4.7	5.3	6.6	5.5	2.4	

.

* Mercury collected in the HNO_3/H_2O_2 impingers.

^b Mercury collected in the $H_2SO_4/KMnO_4$ impingers

CI = Confidence Interval.

efficiency, and to minimize the chance for species conversion. However, there are no studies that would conclusively demonstrate the validity of the method for mercury speciation, such as the spiking of specific mercury compounds into the flue gas ahead of the sampling train. Therefore, the method can be considered unproven for mercury speciation.

No reported studies have been done on the ability of the multi-metals train to provide mercury speciation information. The interpretation of the results thus far relies largely on chemical theory. The efficiencies of the HNO_3/H_2O_2 impingers for capturing ionic mercury and for allowing elemental mercury to pass through have not been established. In addition, the extent of species conversion within the train is unknown. Therefore, the method can be considered unproven for mercury speciation.

Chromium(VI)

Table 4-3 shows the concentrations of total chromium and of chromium(VI) measured in the stack gas during OFA/LNB operation. An attempt was made to measure Cr(VI) in the particulate matter during the OFA test by a different technique, and these results are shown in Appendix D. The OFA results are judged to be nonrepresentative because of shortcomings in the sampling method.

The EPA Cr(VI) train, described in Appendix A, was used for the OFA/LNB test. This train does not use a filter; instead, particulate matter is allowed to enter the KOH impinger solution. A portion of this solution is continuously circulated to the tip of the probe. The KOH solution is used to maintain a pH of 8.5 or higher to prevent the reduction of any Cr(VI) species to Cr(III). The analytical result includes both vapor-phase Cr(VI) and soluble particulate-phase Cr(VI) contributions.

The multi-metals train showed that most of the total chromium was in the particulate phase. Total Cr concentrations exhibited relatively low variability $(24 \pm 8.6 \,\mu g/Nm^3)$. However, the measured concentrations of Cr(VI) were highly variable $(6.2 \pm 8.4 \,\mu g/Nm^3)$. It is not known whether the variability in Cr(VI) was due to the process or the sampling and analytical methods.

The KOH impinger solution contributed significant background concentrations of Cr(VI), which were subtracted from the results. The background concentrations measured in three KOH reagent blanks showed very low variability, but the background itself accounted for 48% to 83% of the three uncorrected sample results, contributing considerable uncertainty to the corrected results. The EPA Cr(VI) method calls for a 0.1 N KOH solution, but the technique was not designed to sample gases with the high levels of CO₂ and SO₂ typically found in power plant flue stacks. A 10 N KOH solution was used to guarantee a high pH, but in retrospect this contributed to the high Cr(VI) background. Future measurements made with this technique should focus on determining the minimum required KOH concentration and/or on using higher purity KOH reagents.

Table 4-3Comparison of Chromium Methods - OFA/LNB Test

	Stack	Gas Concent	trations, μg/	Nm ³	
	Run 2 5/19/93	Run 3 5/20/93	Run 4 5/21/93	Mean	95% CI
Multi-Metals Train Results					
Solid Phase Chromium	20	27	23	23	8.6
Vapor Phase Chromium	1.0	0.77	1.1	0.96	0.42
Total Chromium (Solid + Vapor)	21	28	24	24	8.6
Cr(VI) Train Results					·
Chromium (VI) (Solid + Vapor)	6.8	2.6	9.3	6.2	8.4
Chromium (VI) Percent*				26%	

* Mean Cr(VI) result divided by mean total Cr result from the multi-metals train.

CI = Confidence Interval.

Size-Fractionated Particulate Matter

Table 4-4 compares, for the OFA/LNB test, the total (sum of size fractions) concentrations of metals in size-fractionated particulate matter samples to those measured using the multi-metals train. The results are shown on a gas basis in μ g/Nm³.

The mean concentration of total particulate matter collected with the impactor is only 43% of the loading measured with the multi-metals train. The concentrations of the individual metals are, for the most part, also lower in the impactor samples than in the multi-metals train. This is not surprising, because it is difficult to completely recover the particulate matter from the inside surfaces of the impactor. Conversely, the recovery of particulate matter from the multi-metals train is much more complete. Because of the possibility of incomplete recovery from the impactor, the size-fractionated compositions are more meaningful when expressed on a solid-phase basis (mg/kg of particulate matter, as presented in Table 3-12).

Other factors contribute, to a lesser degree, to the difference between the impactor and multi-metals train results. First, the impactor sample was collected at a single point within the stack, while the multi-metals train traversed the stack. Second, since the impactor was inside the stack during sampling, the particulate matter was collected at stack temperature. The multi-metals train, on the other hand, used an out-of-stack filter maintained at 250° F.

Elemental Enrichment in Ashes and Particulate Matter

The relative enrichment factor, RE, is defined here as the ratio of the concentration of an element in the ash or flue gas particulate matter to its equivalent concentration in the coal ash:

RE = $X_s X_A / X_c$

where:

- X_s = the concentration of the element in the ash or particulate matter (mg/kg);
- X_A = the fraction of ash in the coal; and
- X_c = the concentration of the element in the coal (mg/kg).

If an element is uniformly distributed throughout the bottom ash and all size fractions of the fly ash, the enrichment factor for all ash and particulate matter samples would be unity. Factors greater than unity indicate enrichment of the element in a specific solid. This enrichment can occur as the result of vaporization within the boiler and recondensation on the particulate matter. In addition, an element that is undergoing vaporization

Table 4-4

Comparison	of	Impactor	and	Multi-Metals	Train	Results	- (OFA/	'LNB	Test
Combarroow		antipactor.						<u> </u>		2001

Substance	Impactor Total Mean Concentration (µg/Nm ³)	Multi-Metals Mean Concentration (µg/Nm ³)	Ratio Impactor/MM (%)
Particulate Matter (mg/Nm ³)	60	139	43%
Antimony	2.0	2.6	77%
Arsenic	38	121	31%
Barium	94	161	58%
Beryllium	1.8	3.4	52%
Cadmium	0.27	4.2	6%
Chromium	20	23	87%
Cobalt	4.3	7.5	58%
Copper	50	35	142%
Lead	9.2	13	70%
Manganese	15.8	23	67%
Mercury	0.20	0.19	103%
Molybdenum	4.1	14	29%
Nickel	11.2	19	60%
Selenium	5.9	15	38%
Vanadium	29	47	62%

MM = Multi-Metals Train.

and recondensation should be enriched to the greatest extent in the finest particulate fractions, because the surface area per unit mass increases with decreasing particle size.

Tables 4-5 and 4-6 present the elemental enrichments determined from the concentrations measured during the OFA and OFA/LNB tests, respectively. The interpretation of these factors is limited by the potential biases and imprecisions associated with the measured coal and ash compositions. However, three elements -- arsenic, lead, and selenium -- show signs of significant enrichment in the fine particulate matter for both tests, suggesting that these elements are subject to a vaporization/recondensation mechanism. Mercury tends to vaporize almost completely within the boiler but recondensation is limited, as evidenced by the less-than-unity REs for all but the finest (stack) particulate matter.

Table 4-5Enrichment Factors - OFA Test

Substance	Bottom Ash	ESP Ash	ESP Inlet Particulate	Stack Particulate
Arsenic	0.31	1.3	1.8	3.4
Barium	0.52	0.47	0.65	0.66
Beryllium	0.73	0.89	1.4	1.2
Cadmium	NC	NC	NC	NC
Chloride	NC	NC	NC	NC
Chromium	0.52	0.56	0.77	0.87
Cobalt	0,61	0.67	0.78	0.68
Copper	0.26	0.42	0.58	0.54
Fluoride	NC	0.081	NC	NC
Lead	0.41	1.5	1.6	4.6
Manganese	0.72	0.84	0.77	0.74
Mercury	NC	0.16	0.36	0.87
Molybdenum	NC	NC	NC	NC
Nickel	0.38	0.92	0.44	0.46
Phosphorus	0.48	0.60	0.43	0.48
Selenium	NC	0.46	2.3	2.6
Vanadium	0.58	0.66	0.84	1.0

NC = Not calculated because substance was below the detection limit in the coal, ash, or both.

Table 4-6 Enrichment Factors - OFA/LNB Test

Substance	Bottom Ash	ESP Inlet Partic.	Stack Part. Multi-Metals	Stack < 4 µ m	Stack 4-9μm	Stack >9μm	Impactor Wghtd Avg
Arsenic	0.10	1.3	3.7	6.7	3.0	1.5	2.4
Barium	0.88	1.0	1.1	1.9	1.7	1.2	1.4
Beryllium	0.93	0.96	1.0	1.2	1.7	1.1	1.2
Cadmium	NC	NC	NC	NC	NC	NC	NC
Chloride	NC	0.63	0.59	NA	NA	NA	NA
Chromium	0.61	0.84	0.93	2.4	2.1	1.6	1.8
Cobait	1.0	0.79	0.84	0.94	1.5	0.97	1.0
Copper	0.37	0.57	0.71	1.1	1.4	2.5	2.2
Fluoride	0.024	0.23	0.62	NA	NA	NA	NA
Lead	0.18	0.79	1.2	3.2	2.7	1.4	1.8
Manganese	1.0	1.1	1.1	13	1.7	1.6	1.6
Mercury	NC	0.53	1.0	1.5	1.5	2.2	2.0
Molybdenum	NC	NC	NC	NC	NC	NC	NC
Nickel	0.45	0.66	0.75	1.0	1.2	0.92	0.98
Phosphorus	0.36	0.55	1.0	NA	NA	NA	NA
Selenium	NC	0.45	2.9	3.2	1.1	2.5	2.4
Vanadium	0.83	0.95	1.2	2.6	2.3	1.4	1.7

NA = Not Analyzed.

NC = Not calculated because concentration of substance was below the detection limit in the coal, ash, or both.
DATA EVALUATION

Several procedures can be used to evaluate the information developed during a field sampling program. In the case of Site 16, three methods were used to evaluate the quality of the data. First, the process data were examined to determine if the unit operated at normal, steady-state conditions during the sampling periods. Second, the QA/QC protocol for sampling and analytical procedures used at Site 16 (i.e., equipment calibration and leak checks, duplicates, blanks, spikes, standards, etc.) was evaluated. Site 16 QA/QC data were compared with project objectives. Third, material balances were calculated around the unit. Material balances involve the summation and comparison of mass flow rates in several streams, often sampled and analyzed by different methods. Closure within an acceptable range can be used as an indicator of accurate results for streams that contribute significantly to the overall inlet or outlet mass rates.

Process Operation

Process operating data were examined to ensure that operation was stable during the sampling periods. Measurements were available in 5-minute intervals from the plant computerized data acquisition system. Tables 5-1 and 5-2 show the key parameters monitored for the OFA test and the OFA/LNB test, respectively. In addition, process trend plots are included in Appendix H.

The coefficients of variation (CV, standard deviation divided by the mean) were calculated to evaluate the process variability. Steady boiler operation (between 93% and 95% of full load) was maintained during each of the test runs, as indicated by the low CVs for the load, the coal feed rate, and the economizer outlet oxygen levels. Stack opacity and stack CO levels were typically more variable than the other parameters; Cvs for opacity ranged from 13% to 22%, and Cvs for CO ranged from 8% to 60%.

A comparison of the operating parameters between the OFA and OFA/LNB tests showed that only the NO_x levels were substantially different. The stack concentrations of NO_x were, on average, about 50% lower during OFA/LNB operation than during OFA operation.

For one day during each of the two test efforts, operating problems were encountered. On March 3, 1991 (OFA test), two of the burners were feeding coal but were not ignited, and they remained unlit throughout the sampling period. As a result, elevated stack CO levels were observed on this day compared to the other runs. The metals and anions

3/3/									
	/91	3/4	/91	3/8	(/91	3/6	(/91	3/6	(/91
	CV(%)	Mean	CV(%)	Mean	CV(%)	Mean	CV(%)	Mean	CV(%)
472	0.23	477	0.52	477	0.15	477	0.13	477	0.15
349	0.62	354	0.88	349	0.48	343	0.43	342	0.45
247	0.81	266	1.2	265	0.35	271	0.78	267	0.41
2227	0.21	2250	0.42	2228	0.13	2262	0.14	2262	0,15
43	19	41	3.5	54	9.2	9 9	4.1	67	7.2
28.85	0.25	29.28	0.10	29.37	0.14	20.02	0.11	29.01	0.10
23	15	18	13	14	30	18	23	21	15
31	47	5.9	8.3	11	32	13	23	16	23
554	2.8	570	1.6	544	1.6	514	1.3	529	1.4
776	7.5	754	1.5	932	0.93	904	0.88	910	2.3
7.7	1.9	7.2	2.0	6.3	2.7	6.0	2.2	6.1	2.0
4.8	2.4	4.6	1.7	3.5	4.4	3.2	2.6	3.3	2.0
	349 349 247 2347 231 23 233 33 348 43 77 77 77 7.7	349 0.62 247 0.81 227 0.21 43 19 43 19 31 47 31 47 554 2.8 776 7.5 7.7 1.9 7.7 1.9 4.8 2.4	349 0.62 354 247 0.81 266 227 0.21 256 43 19 41 535 0.25 29.28 31 47 59 31 47 59 554 2.8 570 554 2.8 570 554 2.8 570 77 1.9 7.2 4.8 2.4 4.6	349 0.62 354 0.88 247 0.81 266 1.2 227 0.21 2250 0.42 43 19 41 3.5 43 19 41 3.5 43 19 41 3.5 31 47 59.28 0.10 23 15 18 13 31 47 5.9 8.3 34 7.5 754 1.6 77 1.9 7.2 2.0 4.8 2.4 4.6 1.7	349 0.62 354 0.88 349 247 0.81 266 1.2 265 227 0.21 2250 0.42 2258 43 19 41 3.5 54 43 19 41 3.5 54 53 0.25 29.28 0.10 29.37 53 15 18 13 14 31 47 5.9 8.3 11 54 2.8 570 1.6 544 554 2.8 570 1.6 544 57 754 1.5 932 77 1.9 7.2 2.0 6.3 7.7 1.9 7.2 2.0 6.3 4.8 2.4 4.6 1.7 3.5	349 0.62 354 0.88 349 0.48 247 0.81 266 1.2 265 0.35 227 0.21 2250 0.42 2228 0.13 23 19 41 3.5 54 92 43 19 41 3.5 54 92 43 19 41 3.5 54 92 51 19 41 3.5 54 92 53 15 18 13 14 20 31 47 5.9 8.3 11 32 54 2.8 570 1.6 544 1.6 57 754 1.5 932 0.93 77 1.9 7.2 2.0 6.3 2.7 4.8 2.4 4.6 1.7 3.5 4.4	349 0.62 354 0.88 349 0.68 354 0.88 349 0.48 345 346 345 345 346 346 346 345 345 345 345 346 346 346 346 346 346 346 346 346 346 346 346 346 346 346 346 326 346 346 346 346 346 346 346 346 346 346 346 346 346 346 346 34	349 0.62 334 0.88 349 0.48 345 0.43 247 0.81 266 1.2 265 0.35 271 0.78 227 0.21 2260 0.42 2258 0.13 2262 0.14 227 0.21 2250 0.42 2258 0.13 2262 0.14 43 19 41 3.5 54 9.2 66 4.1 885 0.25 29.28 0.10 29.37 0.14 29.02 0.11 885 0.25 29.28 0.10 29.37 0.14 29.02 0.11 885 0.25 29.28 0.16 29.37 0.14 29.02 0.11 885 0.25 29.28 0.14 20 18 23 23 31 47 5.9 8.3 11 32 13 23 23 514 1.5 73 5.3 0.13	349 0.62 354 0.88 349 0.48 349 0.48 349 0.43 542 543 543 543 543 543 543 543 543 543 543 543 543 545 543 543 547 567 547 567 547 567 547 567 547 566 4.1 67 547 566 4.1 67 546 13 14 230 66 4.1 67 5901<

CV = Coefficient of Variation (standard deviation divided by the mean).

2.1

3.2

1.3

3.3

1.2

35

1.9

4.0

2.4

5.0

Duct B

l

Data Evaluation

5-2

•

Table 5-1 Summary of Process Monitoring Data - OFA Test

	Ru	1	Ru	n 2	Rui	13	Ru	n 4
	5/18	-/8	s/13	(93	5/20	/93	5/2	/93
	Mean	CV(%)	Mean	CV(%)	Mean	CV(%)	Mean	CV(%)
Load (MWc)	473	1.6	467	0.84	472	0.81	470	0.77
Coal Flow Rate (1000 lb/hr)	333	3.2	327	1.2	329	1.2	327	1.1
Overfire Air Flow Rate (1000 lb/hr)	568	3.4	743	3.7	734	1.3	713	1.7
Feedwater Flow Rate (1000 lb/hr)	3554	1.7	3411	1.1	3432	1.0	3394	0.95
Main Steam Pressure (psia)	7622	0.78	2336	0.41	2345	0.39	2331	030
Ambient Temperature (° F)	08	3.2	99	6.1	64	3.5	59	8.7
Ambient Pressure (in Hg)	29.35	0.20	29.28	0.062	29.35	0	29.54	0.033
Opacity (%)	24	14	18	22	19	18	17	19
Stack CO (ppmv)	11	37	9.4	60	13	38	15	36
Stack NQ (ppmv)	251	3.1	282	3.2	772	1.4	260	2.6
Stack SO, (ppmv)	596	2.1	912	2.6	954	1.4	970	2.0
Stack O, (%)	5.7	3.4	5.9	3.5	5.4	2.4	5.1	4.1
Economizer Outlet Q. (%):								
Duct A	3.8	4.8	4.4	4.8	3.9	2.7	3.6	6.8
Duct B	3.8	6.1	4.2	9.6	4.2	5.1	3.8	7.2
Secondary Air Heaters:								
A Inlet Gas Temp. (F)	744	0.72	755	0.71	760	0.56	754	0.40
B Inlet Gas Temp. (°F)	730	0.55	743	0.55	747	0.62	742	0.31
A Outlet Gas Temp. (F)	326	0.97	321	1.4	320	1.6	318	2.0
B Outlet Gas Temp. (°F)	311	1.0	300	1.5	304	1.3	667	2.0

Table 5-2 Summary of Process Monitoring Data - OFA/LNB Test CV = Coefficient of Variation (standard deviation divided by the mean).

samples collected on March 3 were not analyzed, and an extra set of samples was collected on March 6. Samples for organic compounds collected on March 3 were analyzed and included in the data set; however, no unusual results were obtained when compared with the results for samples collected during the other runs.

On March 18, 1993 (OFA/LNB test), samples were collected while the unit operated with one burner out of service. An extra day of testing was added to the schedule, and the samples collected on March 18 were not analyzed because of concerns about the non-representativeness of the process on that day.

Sample Collection

Several factors indicate the acceptable collection of gas samples. Key components of the sampling equipment -- pitot tubes, thermocouples, orifice meters, dry gas meters, and sampling nozzles -- were calibrated before use in the field, and those calibrations were checked at the end of sampling. These calibrations are on file at Radian. The methods used to collect samples were comparable to those used at other sites sampled by Radian in the FCEM project. The sampling runs were well documented, and all flue gas samples were collected at rates between 90 and 110% of isokinesis, except for the semivolatile train (86% isokinetic) and the aldehyde train (80% isokinetic) at the ESP inlet on March 3, 1991. Sufficient data were collected using standard sampling and analysis methods to ensure acceptable data completeness and the comparability of the measurements.

Flue gas samples were collected at both the ESP inlet and the stack. It was easier to collect representative samples at the stack, because the flow was more fully developed and the particulate loading was much lower. During the OFA test, difficulty was encountered in obtaining representative particulate loadings at the ESP inlet. The particulate loadings measured at the ESP inlet showed approximately 40-50% of the coal ash partitioning to fly ash. For this type of boiler, a split closer to 80:20 (fly ash to bottom ash) would be expected, and the 80% fly ash figure was later confirmed by another contractor (SRI) and again by Radian during the OFA/LNB test. It is now suspected that particulate matter was lost from the sampling nozzle. An in-stack thimble was used at this location, and because of the inclined orientation of the ESP inlet duct, the nozzle actually faced 20 degrees downward during sampling. Particulate matter captured in the thimble could have fallen out through the nozzle under the force of gravity when the sample flow was halted. For the OFA/LNB test, the sampling probe was fitted with a cyclone precutter ahead of the thimble to prevent the loss of particulate matter.

Because the particulate loadings measured at the ESP inlet during the OFA test were not considered valid, the particulate loading in this stream was calculated from the coal ash content, assuming an 80:20 fly ash to bottom ash split. The gas-based concentrations of particulate phase analytes in this case were obtained by multiplying the assumed particulate loading (g/Nm^3) by the concentration of the analytes in the particulate matter $(\mu g/g)$.

The flue gas flow rates measured at the ESP inlet during both the OFA and OFA/LNB tests were about 15% higher than at the stack, although the O_2 concentration was slightly higher at the stack (which indicates the inleakage of air between the two locations). To check the consistency of the measurements, a combustion calculation was performed using the mean coal composition, the mean coal flow rate, and the mean oxygen concentration at each location to predict a "theoretical" flue gas flow rate. The measured stack gas flow rates agreed with the theoretical flow rates within 5%. However, the measured ESP inlet flow rates were approximately 15% higher than the theoretical flow rates, indicating that the measured flow rates may be biased high at the ESP inlet.

Coal samples are considered to be representative of the coal fired during flue gas sampling. Coal samples for each run were composites of multiple grab samples obtained from every mill. For the OFA test, samples were collected through "clean-out" ports at the bottom of each coal bunker. During the OFA/LNB test, samples were collected from newly installed taps on each of the coal feeders. Although the locations were slightly different, both the clean-out ports and the feeder taps provided samples of the coal immediately before it entered the mills. The coal was sampled before the rejection of pyrites, but this is not expected to affect the results because pyrite rejects are a minute fraction of the total coal feed.

Bottom ash samples are also considered representative. Bottom ash was allowed to accumulate in the bottom of the boiler during each of the test periods, and samples were collected as the ash was sluiced to the pond. Multiple grab samples were taken throughout the sluicing period so that the composite would represent the bottom ash generated during the test.

Freshly-generated ESP ash samples were collected from each of the 16 hoppers. For the OFA test, equal portions of the ash from each hopper were combined into a single composite. It was not known what fraction each hopper contributed to the total ash flow rate; therefore, the results are subject to this limitation. For the OFA/LNB test, the samples from each of the 16 hoppers were kept separate, and these samples were archived for possible analysis in the future.

Analytical Quality Control Results

Generally, the type of quality control information obtained pertains to measurement precision, accuracy (which included precision and bias), and blank effects, determined using various types of replicate, spiked, and blank samples. The specific characteristics evaluated depend on the type of quality control checks performed. For example, blanks may be prepared at different stages in the sampling and analysis process to isolate the source of a blank effect. Similarly, replicate samples may be generated at different stages to isolate and measure the sources of variability. The QA/QC measures commonly used as part of the data evaluation protocol, and the characteristic information obtained, are summarized in Table 5-3. The absence of any of these types of quality control checks from the data does not necessarily reflect poorly on the quality of the data but does limit the ability to estimate the magnitude of the measurement error and, hence, prevents estimating the confidence in the results.

As shown in Table 5-3, different QC checks provide different types of information, particularly pertaining to the sources of inaccuracy, imprecision, and blank effects. Measurement precision and accuracy are typically estimated from QC indicators that cover as much of the total sampling and analytical process as feasible. Precision and accuracy measurements are based primarily on the actual sample matrix. The precision and accuracy estimates obtained experimentally during the test program are compared with the established data quality objectives (DQOs).

These DQOs are not intended to be used as validation criteria but as empirical estimates of the precision and accuracy expected from existing reference measurement methods and that would be considered acceptable. Although analytical precision and accuracy are relatively easy to quantify and control, sampling precision and accuracy are unique to each site and each sample matrix. Data that do not meet these DQOs are not necessarily unacceptable; the intent is to document the precision and accuracy actually obtained, and the objectives serve as benchmarks for comparison. The effects of not meeting the objectives are considered in light of the intended use of the data.

Tables 5-4 and 5-5 present the types of quality control information reported for the OFA and OFA/LNB tests. The results for QC sample analyses can be found in Appendix F. Tables 5-6 and 5-7 present summaries of precision and accuracy estimates obtained during the OFA and OFA/LNB tests, respectively.

Evaluation of the measurement results for the OFA test, conducted in March 1991, are based on QC results previously reported in the "Field Chemical Emission Monitoring Project: Site 16 Report" (1992). Some features of the QC data presentation differ slightly between the 1991 OFA and 1993 OFA/LNB tests. For example, QC data for the 1991 OFA test were presented (as shown in Appendix F.1) in summarized form only, whereas the 1993 OFA/LNB QC data are presented as individual results as well as summaries. The OFA QC data summaries were also broken down by train fraction; the OFA/LNB QC data include all individual fraction results but are summarized in terms of whole trains because no significant performance distinction was evident for the separate fractions. Overall, however, the data sets for the two tests are comparable in terms of both quality control activities performed and performance results.

Comparability is a qualitative parameter expressing the confidence with which one data set can be compared with another. Sampling data should be comparable with other measurement data for similar samples collected under similar conditions. This goal is achieved using standard techniques to collect and analyze representative samples and by

Table 5-3Types of Quality Control Samples

.

QC Activity	Characteristic Measured
Precision	
Replicate samples collected over time under the same conditions	Total variability, including process or temporal, sampling, and analytical, but not bias.
Duplicate field samples collected simultaneously	Sampling plus analytical variability at the actual sample concentrations.
Duplicate analyses of a single sample	Analytical variability at the actual sample concentrations.
Matrix- or media-spiked duplicates	Sampling plus analytical variability at an established concentration.
Laboratory control sample duplicates	Analytical variability in the absence of sample matrix effects.
Surrogate-spiked sample sets	Analytical variability in the sample matrix but at an established concentration.
Accuracy (Including Bias and Precision)	
Matrix-spiked samples	Analyte recovery in the sample matrix, indicating possible matrix interferences and other effects. In a single sample, includes both random error (impreci- sion) and systematic error (bias).
Media-spiked samples	Same as matrix-spiked samples. Used where a matrix- spiked sample is not feasible, such as certain stack sampling methods.
Surrogate-spiked samples	Analyte recovery in the sample matrix, to the extent that the surrogate compounds are chemically similar to the compounds of interest. Primarily used as indicator of analytical efficacy.
Laboratory control samples (LCS)	Analyte recovery in the absence of actual sample matrix effects. Used as an indicator of analytical control.
Standard Reference Material	Analyte recovery in a matrix similar to the actual samples.
Blank Effects	
Field Blank	Total sampling plus analytical blank effect, including sampling equipment and reagents, sample transport and storage, and analytical reagents and equipment.
Trip Blank	Blank effects arising from sample transport and storage. Typically used only for volatile organic compound analyses.
Method Blank	Blank effects inherent in analytical method, including reagents and equipment.
Reagent Blank	Blank effects from reagents used.

		B-	recision				Accura	ley.			_	Blask	
Analysis	Replicate Runs	Dupticate Field Samples	Duplicate Lab Analysis	MSD	UCSD	WS	Surrogate Spike	LCS	Std. Ref. Mat'l	Fjeld Blank	Trip Blank	Metbod Blank	Reagent Blank
Gas Samples					ľ				ſ				
Mctals	`	`		`		`		`	`	`		`	
Anions	`	`		`		`		`			~	•	
Semivolatile Organic Compounds	`	-		`		`	`	`		`		、	
Volatile Organic Compounds	>				-		`	`		`		`	
Aldehydes	`									``	`	•	
Coal Samples						Ī							
Metals	`	`		`		`		`	`			~	
Anions	`	`											
Ultimate/Proximate	•	`							~				
Ash Samples									ſ				
Mctals	•	•		`		`		`				~	
Anions	`	`											

5-8

Table 5-4 Types of Quality Control Data Reported - OFA

		-	recision	·			Accuri	acy	_		-	Blank	
Analysis	Replicate Runs	Duplicate Field Samples	Dupticate Lab Analysis	QSM	LCSD	SW	Surrogate Spike	LCS	Std. Ref. Mat'l	Field Blank	Trip Blank	Method Blank	Reagent Blank
Gas Samples													
Metals (ICP-MS)	>		>		`			`	`		`	`	•
Metals (ICP-AES/AAS)	`				`	`		`	`	`		`	`
Hg Speciation	`							`		`			
Cr (VI)	`			•		`						`	`
Anions	`			•	`	`		>		`		`	
Semivolatile Organic Compounds	`					`	`	`		`		•	
Volatile Organic Compounds	>			>		`	`			`		`	
Aldehydes	`					`		`			`		•
Coal Samples										ſ			
Metals	`	`	`						`			、	
Anions	`	`				•							9
Ultimate/Proximate	`	•											
Ash Samples													
Metals	>	~		>	`	`		`	`			`	
Anions	`	`		>	`	`		`				`	

Table 5-5 Types of Quality Control Data Reported - OFA/LNB

Data Evaluation

5-9

		Objec	ctives	Meas	ured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Ash - Semivolatile Organics	Spiked Samples				
Acenaphthene		35(CV)	50-150	7.8	2
4-Chloro-3-methylphenol		35(CV)	50-150	3.6	82
2-Chlorophenol		35(CV)	50-150	$\tilde{6.1}$	56 8
1.4-Dichlorobenzene		35(CV)	50-150	6.8	38 %
2,4-Dinitrotuluene		35(CV)	50-150	44	88
N-Nitrosodipropylamine		35(CV)	50-150	6.2	33
4-Nitrophenol		35(CV)	50-150	э (22
Pentachlorophenol		33(CV)	50-150	2.2	83
Phenol		33(CV)	50.150	11.2	52
Pyrene 1 2 4-Trichlorobenzene		35(CV)	50-150	3.2	94
Anjons	Spiked Samples				
Chloride (solids)		20	80-120	4.8	94
Chloride (liquid)		20	80-120	•	100
Fluoride (gas)		50	80-120		5 <u>1</u>
Fluoride (solids)		ន៖	80-120	23.2	
Fluoride (liquid)		88	00-120 00 120	C.11	2 1
Phosphate (liquid) Phosphate (solids)		នន	80-120		<u> </u>
Metals (Inlet Cas-Imbineers)	Spiked Samples				
Arcenic		8	75-125	1.3	74
Barium		ନ୍ଦ	75-125	1.1	24
Bervlium		50	75-125	1.0	8
Cadmium		ສ	75-125	7.5	<u>8</u> 8
Chromium		8	12-122	•	26
Cobalt		ສ	75-125	1.1	88
Copper		র	(71-()	- ;	7
Lead		ন	75-125	1.3	4 2
Manganese		28	21-01		7, 8
Mercury		ৰহ	C21-C1 X 1 X	• 5	88
Molybdenum		38	75-125	11	2
NICKEI Selenium		ន	75-125	5.1	2
Vanadium		20	75-125	0	92

.'

Table 5-6 Summary of Precision and Accuracy Estimates - OFA

Table 5-6 (Continued)

Measurement Parameter How Measured % RPD % Rev) % Rev % Rev) % Rev % % Rev % % Rev % % Rev % <th></th> <th></th> <th>Obje</th> <th>ctives</th> <th>Meas</th> <th>sured</th>			Obje	ctives	Meas	sured
Meals (Inlet Gas-Filter) Spiked Samples N	Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Arsenic Arsenic Beylium Arsenic Beylium 20 75-125 2.8 71 Beylium 20 75-125 2.2 36 3 Cadmium 20 75-125 2.2 36 3 Chonkin 20 75-125 2.2 36 3 Cobat 20 75-125 2.2 36 3 Cobat 20 75-125 1.1 92 3 Coper 20 75-125 1.1 92 3 Magance 20 75-125 1.2 93 3 Mercury 20 75-125 1.2 93 3 Nickel 20 75-125 1.2 93 3 Nickel 20 75-125 1.3 93 3 Nickel 20 75-125 1.3 93 3 Nickel 20 75-125 1.4 8 3 Vickel 30 75-125 1.4	Metals (Inlet Gas-Filter)	Spiked Samples				
Barium Barium Barium Barium Barium 20 75-125 6.2 80 Cadvilium Cadout Chromium 20 75-125 6.2 91 Cadout 20 75-125 2.4 83 Cobat 20 75-125 2.4 83 Cobat 20 75-125 2.4 83 Copat 20 75-125 2.4 83 Maganese 20 75-125 2.4 83 Maganese 20 75-125 2.1 93 Molybdenum 20 75-125 2.1 93 Nickel 20 75-125 2.1 93 Selatium 20 75-125 2.1 93 Vaadium 20 75-125 2.1 93 Vickel 20 75-125 2.0 93 Vickel 30 75-125 2.0 94 Selatium 20 75-125 2.0 94	Arsenic		20	75-125	2.8	11
Beryllium 20 75-125 4.0 74 Choantium 20 75-125 2.4 88 Choantium 20 75-125 2.4 88 Cobat 20 75-125 2.4 88 Manganese 20 75-125 2.2 91 Manganese 20 75-125 2.2 310 Nickel 20 75-125 2.1 32 Nickel 20 75-125 2.1 33 Vaadinum 20 75-125 2.0 92 Vickel 50 75-125 2.0 92 Barium 20 75-125 2.0 92 Barium 20 75-125 2.0 92 Barium 20 75-125 0 92	Barium		50	75-125	6.2	8
Cadmium 20 75-125 2.2 91 Cobet 20 75-125 2.4 88 83 Cobet 20 75-125 2.4 88 83 Copet 20 75-125 2.4 88 83 Copet 20 75-125 2.4 88 83 Copet 20 75-125 2.9 310 84 Maganese 20 75-125 2.0 84 83 83 Motybdenum 20 75-125 1.1 82 84 83 Motophenum 20 75-125 1.1 82 84 83 Vatadium 20 75-125 1.1 82 84 83 83 Vatadium 20 75-125 1.1 82 83 83 Vatadium 20 75-125 1.1 82 83 83 Metals Stack Gasy 5piked Samples 20 75-125 <td>Beryllium</td> <td></td> <td>50</td> <td>75-125</td> <td>4.0</td> <td>74</td>	Beryllium		50	75-125	4.0	74
Chromium 20 75-125 2.4 88 Cobalt 20 75-125 3.24 88 Cobalt 20 75-125 3.6 84 Cobalt 20 75-125 3.6 84 Copper 20 75-125 3.0 84 Manganese 20 75-125 2.4 88 Moreury 20 75-125 2.9 310 Moreury 20 75-125 2.9 310 Nickel 20 75-125 2.0 84 Nickel 20 75-125 2.1 82 Nickel 20 75-125 2.1 82 Nickel 20 75-125 2.1 92 Vanadium 20 75-125 2.1 92 Britim 20 75-125 2.0 94 Britim 20 75-125 2.0 94 Cobalt 20 75-125 0 94 </td <td>Cadmium</td> <td></td> <td>50</td> <td>75-125</td> <td>2.2</td> <td>91</td>	Cadmium		50	75-125	2.2	91
Cobalt 20 75-125 2.4 86 Copper 20 75-125 3.6 84 Lead 20 75-125 11 29 84 Marganese 20 75-125 13.6 84 85 Marganese 20 75-125 12.9 84 85 Mercury 20 75-125 12.9 84 85 Nickel 20 75-125 12.9 84 85 Nickel 20 75-125 12.9 87 93 Nickel 20 75-125 12.9 83 93 Selenium 20 75-125 12.0 93 93 Vatadium 20 75-125 12.0 93 93 Metals (Stack Gas) 5piked Samples 20 75-125 10 94 Metals (Stack Gas) 5piked Samples 20 75-125 10 93 Metals (Stack Gas) 5piked Samples 20	Chromium		ଛ	75-125	2.4	83
Copper 20 75-125 3.6 84 Lead 20 75-125 1.1 92 931 Marcury 20 75-125 6.0 84 Mercury 20 75-125 0.0 84 Mercury 20 75-125 1.1 92 Nickel 20 75-125 1.2 87 93 Nickel 20 75-125 1.2 82 93 Nickel 20 75-125 1.2 82 93 Vaadium 20 75-125 1.2 82 93 Vanadium 20 75-125 1.3 93 93 Deryllium 20 75-125 2.0 94 93 Barium 20 75-125 2.0 94 93 Chromium 20 75-125 2.0 94 95 Molybdenum 20 75-125 2.0 95 95 Nolybdenum	Cobalt		8	75-125	2.4	85
Lead 20 75-125 1.1 92 Marganese Morybdenum 20 75-125 1.1 92 Merganese Morybdenum 20 75-125 2.9 310 Morybdenum 20 75-125 2.9 310 Nickel 20 75-125 2.9 310 Nickel 20 75-125 2.9 310 Nickel 20 75-125 2.1 32 310 Nickel 20 75-125 2.1 32 310 Vistel 20 75-125 2.1 33 32 Vandium Spiked Samples 20 75-125 2.0 9 9 Ervinn 20 75-125 2.0 9 9 9 Chromium Cobalt 20 75-125 2.1 9 9 Nolybdenum Margarese 20 75-125 0 9 9 Nolybdenum 77-125 <t< td=""><td>Copper</td><td></td><td>8</td><td>75-125</td><td>3.6</td><td>25</td></t<>	Copper		8	75-125	3.6	25
Manganese 20 75-125 6.0 84 Mercury Mercury 20 75-125 2.9 310 Mercury Nickel 20 75-125 2.9 310 Nickel 20 75-125 2.9 310 32 Nickel 20 75-125 2.9 310 32 Selenium 20 75-125 2.4 83 33 Vanadium 20 75-125 2.4 83 33 Vanadium 20 75-125 2.0 75-125 2.0 93 Barum Barum 20 75-125 2.0 94 92 Ontonium 20 75-125 0 93 92 93 93 Molybdenum 20 75-125 0 94 92 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 93 9	Lead		ຊ	75-125	1.1	8
Mercury Molybdenum 20 75-125 2.9 310 Molybdenum 20 75-125 2.9 79 Nickel 20 75-125 1.2 82 Nickel 20 75-125 1.2 82 Selenium 20 75-125 1.2 82 Vanadium 20 75-125 2.0 94 Barium 20 75-125 2.0 94 Barium 20 75-125 2.0 94 Chromium 20 75-125 1.0 95 Chromium 20 75-125 0 94 Marganese 75-125 0 75-125 0 Marganese 75-125 0 75-125 0 94 Marganese 75-125 0 75-125 0 94 Nicklel 20 75-125 0 94 92 Selenium 20 75-125 0 97 93	Manganese		ន	75-125	6.0	84
Molybdenum 20 75-125 25 79 Nickel 20 75-125 1.2 82 Selenium 20 75-125 1.2 82 Vaadium 20 75-125 1.2 82 Vandium 20 75-125 2.4 83 Vandium 20 75-125 2.4 83 Bartum 20 75-125 2.0 94 Bartum 20 75-125 2.0 94 Chromium 20 75-125 1.0 95 Cobalt 20 75-125 0.0 95 Manganese 75-125 0.0 95 97 Manganese 75-125 0.0 97 97 Nicel 20 75-125 0.0 97 Nicel 20 75-125 0 97 Nicel 20 75-125 0 97 Nicelei 20 75-125 0 97 <td>Mercury</td> <td></td> <td>8</td> <td>75-125</td> <td>2.9</td> <td>310</td>	Mercury		8	75-125	2.9	310
Nickel 20 75-125 1.2 82 Selenium 20 75-125 13 92 Vanadium 20 75-125 13 92 Vanadium 20 75-125 13 92 Vanadium 20 75-125 24 83 Metals (Stack Gas) Spiked Samples 20 75-125 20 94 Barium 20 75-125 0 94 95 95 Chromium 20 75-125 0 95 96 96 Cobalt 20 75-125 0 95 96 96 Copat 20 75-125 0 96 96 96 96 Molydenum 20 75-125 0 75-125 96 96 97 Nickel 20 75-125 0 97 97 96 97 Nickel 20 75-125 0 97 94 94	Molybdenum		8	75-125	R	64
Selenium 20 75-125 13 92 Varadium 20 75-125 2.4 83 Varadium 20 75-125 2.4 83 Metals (Stack Gas) Spiked Samples 20 75-125 2.4 83 Barium 20 75-125 2.0 94 83 Barum 20 75-125 0 94 86 Chromium 20 75-125 0 95 95 Cobalt 20 75-125 0 95 95 Copper 20 75-125 0 95 95 Molydenum 20 75-125 0 95 95 Nickel 20 75-125 0 95 95 Nickel 20 75-125 0 95 95 Vanadium 20 75-125 0 95 95 Selenium 20 75-125 0 95 94	Nickel		ຊ	75-125	1.2	82
Variation ZV 7-1.2/2 ZV 0.3 Metals (Stack Gas) Spiked Samples 20 75-125 2.0 94 Barium 20 75-125 2.0 94 95 95 Baryllium 20 75-125 0 86 95 95 Chromium 20 75-125 1.0 95 95 95 95 Cobalt 20 75-125 0 75-125 0 95 95 Copper 20 75-125 0 75-125 0 95 Molybdenum 20 75-125 0 95 92 Nickel 20 75-125 0 95 92 Vanadium 20 75-125 0 94	Selenium		88	75-125	: 13 13	88
Barium 20 75-125 2.0 94 Beryllium 20 75-125 0 94 Beryllium 20 75-125 0 95 Chromium 20 75-125 1.0 95 Chromium 20 75-125 0 95 Cobalt 20 75-125 0 95 Copper 20 75-125 0 95 Molybdenum 20 75-125 0 91 Nickel 20 75-125 0 92 Vanadium 20 75-125 0 92 Vanadium 20 75-125 0 92	Metals (Stack Gas)	Spiked Samples	07	C7T-C1		60
Beryllium 20 75-125 0 86 Chromium 20 75-125 1.0 95 Chromium 20 75-125 1.0 95 Cobalt 20 75-125 0 95 Copper 20 75-125 0 91 Manganese 20 75-125 0 91 Molybdenum 20 75-125 0 92 Nickel 20 75-125 0 92 Vanadium 20 75-125 0 93 Vanadium 20 75-125 0 94	Barium		20	75-125	2.0	94
Z0 75-125 1.0 95 Chromiun Z0 75-125 1.0 95 Cobalt Z0 75-125 0 91 Copper Z0 75-125 0 91 Manganese Z0 75-125 0 91 Molybdenum Z0 75-125 0 92 Nickel Z0 75-125 0 92 Nickel Z0 75-125 0 92 Vanadium Z0 75-125 0 92 Vanadium 20 75-125 0 94	Beryllium		20	75-125	0	8
Cobalt 20 75-125 2.1 96 Copper 20 75-125 61 92 Copper 20 75-125 0 91 Manganese 20 75-125 0 91 Molybdenum 20 75-125 0 92 Nickel 20 75-125 0 92 Nickel 20 75-125 0 92 Vanadium 20 75-125 0 94	Chromium		20	75-125	1.0	95
Copper 20 75-125 61 92 Manganese 20 75-125 0 91 Manganese 20 75-125 0 95 Molybdenum 20 75-125 0 95 Nickel 20 75-125 63 72 Selenium 20 75-125 63 72 Vanadium 20 75-125 0 94	Cobalt		50	75-125	2.1	8
Manganese 20 75-125 0 91 Manganese 20 75-125 0 95 Molybdenum 20 75-125 0 95 Nickel 20 75-125 63 72 Selenium 20 75-125 63 72 Vanadium 20 75-125 0 94	Copper		50	75-125	61	92
Xolybdenum 20 75-125 0 95 Nickel 20 75-125 4.3 92 Nickel 20 75-125 63 72 Sclenium 20 75-125 63 72 Vanadium 20 75-125 0 94	Manganese		20	75-125	0	91
Nickel 20 75-125 4.3 92 Selenium 20 75-125 63 72 Vanadium 20 75-125 0 94	Molybdenum		50	75-125	0	<u>95</u>
Sclenium 20 75-125 63 72 Vanadium 20 75-125 0 94	Nickel		50	75-125	4.3	22
Vanadium 20 75-125 0 94	Selenium		20	75-125	63	72
	Vanadium		20	75-125	0	94

		Ohje	ctives	Meas	nred
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Metals (Coals)	Spiked Samples				
Arsenic		20	75-125	8.3 2,3	108
Barium		ន	75-125	<u>.</u>	701
Beryllium		88	75-125	12.3	8
		38	75-125	14.6	8
Cobalt		38	75-125	3.9	94
Copper		8	75-125	33	8
Lead		នន	15-125	37 26	010
Manganese		۹ <i>۶</i>	(21-C) 75-125	22	212
Molthdamm		ន	75-125	2.9	95
Nickel		50	75-125	20	88 ;
Selenium		88	75-125 75-125	4.2	910
Vanadium Metals (Flv Ash)	Spiked Samples				
		90	75-125	112	89
Arsenic Rarium		3 8	75-125	7.5	8
Beryllium		50	75-125	0	8
Cadmium		88	75-125	5.3	1 <u>5</u> 1
Chromium		38	75-125	11	88
Conner		ន	75-125	0	8
Lead		92	75-125	21	67
Manganese		88	75-125	2.4	72 50
Mercury		38	75-125	36	2
Nickel		ន	75-125	1.1	8
Selenium Vanadium		នន	75-125 75-125	00	88 88
Metals (Bottom Ash-Sluice H, O)	Spiked Samples				
Rarium		20	75-125	1.0	67
Beryllium		ନ୍ଧ	75-125	0	83
Chromium		28	21-0		\$ 2
Cobalt		38	(21-C) 75,125) c	¥ 3
Copper Manganese		র	75-125	1.1 1.1	8
Molybdenum		8	75-125	1.1	26
Nickel		ន្តទ	75-125 75-125	[]0	¥ 8
Vanadium		2			

Table 5-6 (Continued)

		Obje	tives	Meas	nred
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Metals (Slutce H, O Supply)	Spiked Samples				
Barium		20	75-125	1.0	83
Bervlium		50	75-125	1.1	£ 3
Chromium		ଷ	75-125	1:1	4 2
Cobalt		22	(71-C)	 	19
Copper		28	12-12	7'T	5
Manganese		88	75-125	11	8
Molybdenum Nickel		383	75-125	1.0	88
Vanadium		20	C71-C/	7.7	66
VOST (Inlet Gas)	Surrogates				
1,4-Bromofluorobenzene		35 (CV) 35 (CV)	50-150 50-150	12 (CV) 15 (CV)	103 22
1,2-L)ICIIIOFOCHRAIIC-U14 Toluene-d8		35 (ČV)	50-150	6.0 (CV)	104
VOST (Stack Gas)	Surrogates				
1.4-Bromofluorobenzene		35 (CV)	50-150	14 (CV)	011
1,2-Dichloroethane-d14		33 (CV)	061-06 50-150	0.7 (CV)	t S
l oluene-d8					
Semivolatiles (Gas)	Surrogates				90
2-Fluorobiphenyl		33 (CV)	50-150 50-150	88	38
2-Fluorophenol		20 20 20 20 20 20 20 20 20 20 20 20 20 2	001-00 50-150		2 8
Nitrobenzene-d5		33 (CV)	50-150	27	80
Terphenyl-d14		33 (CV)	50-150	¥.	91 001
2,4,6-Tribromophenol		30 (CV)	OCT-OC	05	107
Semivolatiles (Solids)	Surrogates				
2-Fluorobiphenyl		35 (CV)	50-150	2.5	102 84
2-Fluorophenol		2) 2) 2) 2)	0CT-0C	0.4 V 4	38
Nitrobenzene-d5		2002	50-150	5.1	16
Thenol-do		35 (CV)	50-150	23	94
lerpnenyi-ut4 2 4 6. Tribromonbenol		35 (CV)	50-150	7.4	70
7,4,4 T TIM MIM MIM					

Table 5-6 (Continued)

.

5-13

\sim
Ŋ
ă
Ē
1
5
Ŭ
-
Ģ
5-6
e 5-6 (
ble 5-6 (
able 5-6 (
Table 5-6 (
Table 5-6 (

5-14

٠

		Obje	ctives	Mea	sured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Semivolatiles (Liquid)	Surrogates				
2-Fluorobinhenvl		35 (CV)	50-150	11	67
2-Fluorophenol		35 (CV)	50-150	6.3	65
Nitrobenzene-d5		35 (CV)	50-150	5.8	88
Phenol-d5		35 (CV)	50-150	3.2	8
Terphenyl-d14 2.4.6-Tribromophenol		33 (CV) 33 (CV)	50-150 50-150	11	130 81
Anions (Coal)	Duplicate Samples				
Chloride		20		13.1	
Phosphate		ន្តទ		0 78 0	
Lotar Fridepriorus Antone (Bettom Ach)	Dimlicata Samules	8			
		Ę		₹0	
Cirioride Finoride		38		23.6	
Total Sulfur		20		NC	
Anions (Fly Ash)	Duplicate Samples				
Chloride		20		NC	
Fluoride		50		11.2	
10(al Sultur		3		0.77	

			· Objectives		Measured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Metals in Gas Samples	Precision and Accuracy, MS/MSD				
Arsenic		2	C71-C/	7.1	93-100
Barium		នន	75-125 251-25	4.7	/3-109 00 02
Beryllium		3	c71-c/	21	04-40
Cadmium		ຊະ	75-125		9/-116 07 110
Chromium		38	(21-C) 3CF 3F		9/-110 00 111
Chromium (VI)		38	C21-C1 2C1 2F	0.0	111-06
Cobair		2 2	75-175	0.0	88-101
Cupper Lead		30	75-125	9.2	62-109
Manganèse		ន	75-125	15	82-92
Mercury		50	75-125	6.8	50-132
Molvbdenum		50	75-125	6.0	86-93
Nickel		20	75-125	5.6	80-110
Phosphorus		20	75-125	1.8	79-114
Selenium		50	75-125	15.7	67-142
Vanadium		20	75-125	1.1	89-98
Anions in Gas Samples	Precision and Accuracy, MS/MSD				
Chloride		30 V	80-120	4.8	90-114
Fluoride		28	80-120	4.8	85-102
Sulfate		20	80-120	14.4	74-123
Semivolatiles in Gas Samples	Precision and Accuracy, Surrogate Spike Recoveries				
		36(01)	50 150	νοι	107 019
Biphenyl-d10 Hexachlorobenzene-16		35(CV)	20-1-00 20-150	£88	012-201 66-160 0 125
Perylene-d12		33(UV)	NCT-NC	NC	071-0
Volatiles in Gas Samples	Precision and Accuracy, Surrogate Spike Recoveries				
1.2-Dichloroethane-d4		35(CV)	50-150	13.7(CV)	92-128
4-Bromofluorobenzene		35(CV)	50-150 50 - 550	12.8(CV)	78-117
Toluene-d8		ו ועטוכנ		10.7JC.UL	011-00

Table 5-7 Summary of Precision and Accuracy Estimates - OFA/LNB

	•		Objectives		Measured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Aldehydes in Gas Samples	Precision - NA; Accuracy - Matrix Spike Recovery				
Acetaldehyde Formaldehyde		40	50-150 50-150	A N	88
Mercury Speciation	Precision - NA; Accuracy - Laboratory Control Sample				
Mercury (0) Mercury (II) Methyl Mercury		ଟ୍ଷ	75-125 75-125 75-125	A A A	88 94-97 106
Metals in Ash (ICP-AES)	Precision - Duplicate Samples; Accuracy - Standard Reference Material (NIST1633a)				
Arsenic (GFAAS) Barium Beryllium Cadmium Cadmium Chromium Cobalt Copper		ສ &&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&&	* * * * * * * * * * * * * * * * * * *	191 202 202 202 202 202 202 202 202 202 20	125-130 85-89 85-89 85-89 NA NA 93-101 NA NA NA NA NA NA NA NA NA NA NA NA NA

.

Data Evaluation

Table 5-7 (Continued)

			Objectives		Measured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Metals in Ash (ICP-MS)	Precision and Accuracy, Standard Reference Material				
					00 00
Arsenic		ឧន	21-0		40-00 NA
Barium		38	21-01		24-86
Beryllium		38	21-01	191	51-60
Cadmium		3 2	75,125	80	80-00
Chromium		28	75-125	έ. Έ	68-73
Cobalt		32	75-125	5	94-102
Copper		38	75-125	0.7	88-89
Mananaca		2	75-125	4.5	104-109
Merciry		20	75-125	29.7	133-179
Molvhdenum		20	75-125	12.0	26-98
Nickel		50	75-125	6.8	69-74
Phosphorus		ຊ	12-21-21 221-21	A S	NA 00 100
Selenium		88	75-125 75-125	9.8 3.7	98-108 82-85
	OSM/ SM Accuracy MS/MS/				
METALS IN ASH (ICL-AES/AAS)	I I CONTRA AND DECHIARY MANY MANY				
Arsenic		20	75-125	1.9	104-106
Barium		22	01-0 2,55	0.CI 2.C	54-20 18-82
Beryllium		86	75, 175		94-106
Cadmium		22	75-125	1.1	90-91
Chromium		38	75-125	2.2	16-68
Coosit		20	75-125	1.1	91-92
Lopper		50	75-125	2.1	96-98
Manganese		20	75-125	1.1	16-06
Mercury		2	CZ1-51	0.0	91-91 102-102
Molybdenum		38	21-0	1.U	80-01
Nickel		38	75 175	100	50-20
Phosphorus		38	75-125	4.7	84-88
Seienium Vanadium		20	75-125	1.1	91-92
Anions in Ash	Precision and Accuracy, MS/MSD				
Chloride		20	80-120	13	66-86
Fluoride		20	80-120	46.1	29-88

÷

Table 5-7 (Continued)

,

			Objectives		Measured
Measurement Parameter	How Measured	% RPD	(% Rec)	% RPD	(% Rec)
Metals in Coal (INAA)	Precision - Duplicate Samples; Accuracy - Standard Reference Material (NIST1632a)				
Arcanic		8	75-125	8.2	100
Barium		ন	75-125	5.8	16
Beryllium (ICP-AES)		ន	75-125	10.0	22
(SARM20)		R 2	75-125	10.2	ž S
Chromium		នេះ	75-125	5.2	106
Cobalt		88	75-125	13.3	101
Lead (ICP-AES) (SARM20)		នេះ	75-125	1.1	<u>95</u>
Manganese		ន	C21-C1 2C1 2C	0.61	149
Mercury (SAKM2U) Mercury (CVAAS) (SARM20)		ৰপ্ন	75-125	4.6	104
Molybdenum		នន	75-125	9.2 10.4	108
Nickel		۶ £	75-125	4.0	101
rnospnorus (SAXIM 20) Selenium Vanadium		38	75-125	1.8	95
Anions in Coal	Precision and Accuracy, MS/MSD				
Chloride Fluoride		នន	80-120 80-120	2.1 11.6	95-97 49-91
Ultimate/Proximate Coal	Precision - Duplicate Samples; Accuracy -				
Moisture		ສ	SN	8.2	
Carbon		88	S Z	0.8	
Nitrogen		38	S	0.7	
Sulfur		នន	S S Z	12.9 2.9	
Oxygen		ន្ត	NS	7.7	
Volatiles Fixed Carbon		ৰপ্ন	SN	03	
ННУ		20	SS	0.4	

Table 5-7 (Continued)

reporting analytical results in appropriate units. Data sets can be compared with confidence when the precision and accuracy are known.

Representativeness expresses the degree to which sample data accurately and precisely represent a characteristic of a population, parameter variations at a sampling point, or an environmental condition. The representativeness criterion is based on making certain that sampling locations are properly selected and that a sufficient number of samples are collected. The results of both the OFA and OFA/LNB tests are considered representative and comparable.

Key characteristics of the QC results for the two tests are discussed separately in the following subsections, including a discussion of the overall measurement precision, accuracy, and blank effects.

Precision is a measure of the reproducibility of measurements under a given set of conditions. It is expressed in terms of the distribution, or scatter, of the data, calculated as the standard deviation or coefficient of variation (CV, standard deviation divided by the mean). For duplicates, precision is expressed as the relative percent difference (RPD).

Accuracy is a measure of the degree of conformity of a value generated by a specific procedure with the assumed or accepted true value, and includes both precision and bias. Bias is the persistent positive or negative deviation of the method average value from the assumed or accepted true value. The efficiency of the analytical procedure for a given sample matrix is quantified by the analysis of spiked samples containing target or indicator analytes or other quality assurance measures, as necessary. However, all spikes, unless made to the flowing stream ahead of sampling, produce only estimates of recovery of the analyte through all of the measurement steps occurring after the addition of the spike. A good spike recovery tells little about the true value of the sample before spiking.

Blank effects, including contamination and other artifacts, contribute to low-level measurement bias, which may or may not be significant, depending on the relative quantity of measure found in the investigative samples. A well-known blank may be corrected for, but spurious blank effects generally cannot.

OFA Test Analytical QC Data

The following potential areas of concern were indicated by the OFA test quality control data.

Fly Ash

- A standard fly ash performance evaluation (PE) sample (NIST 1633a) was submitted double blind and analyzed by ICP/AAS following microwave digestion, instrumental neutron activation (INAA), and x-ray fluorescence (XRF). Recoveries for chromium (229%), nickel (202%), and arsenic (148%) were high when the sample was analyzed after digestion in the microwave. Recoveries were high for nickel (140%) when analyzed by INAA, and the recovery of manganese (440%) was high when analyzed by XRF. These results may indicate a high bias for these analytes (when prepared by the respective methods) in flue gas particulate samples. However, neither the XRF nor the INAA analyses were selected as the primary methods for these analytes.
- Recoveries from the fly ash PE sample were low for mercury (65%) after digestion in the microwave. This result may indicate a low bias for mercury in samples prepared by this method.
- The fly ash PE sample was also analyzed after being aspirated into the impinger train. Recoveries indicating a high bias included arsenic (144%) and cadmium (198%). The barium recovery (63%) indicated a low bias after being corrected for the blank results.
- Lead recoveries (average recovery 67%) for spike samples in fly ash indicated low bias.
- Cadmium recoveries (average 150%) in spike samples for fly ash indicated a high bias for this matrix.

Coal

- Recoveries for a standard coal sample (NIST 1632a) indicated a high bias for barium (145%) and selenium (127%) by INAA. The recovery was excessively high for copper (878%).
- The recovery of copper by ICP-AES was low in two spiked samples, and high in one spiked sample. These results indicate higher than expected variability for copper in coal, and the quantitation of the field samples may be suspect.

Gas Streams

• The mean recoveries for selenium (70%) and lead (74%) were slightly below the 75-125% objective in the impinger solution spike samples. Selenium recoveries were also low (72%) in the stack gas probe and nozzle spike samples.

- The mean recovery for arsenic (71%) was slightly low in the ESP inlet gas spike samples.
- Barium and molybdenum are reported in both the field and trip blanks for filters at concentrations greater than five times the detection limits. The filter field sample results may be biased high because of contamination.
- Lead and cadmium are reported in the blanks associated with the impingers in concentrations greater than five times the detection limits. Impinger samples results for these metals near the detection limit may be biased high because of contamination.
- Chloride and sulfate were reported in both the filter trip blank and the field blank samples; however, the blank levels were insignificant compared to the sample concentrations.
- The concentration of formaldehyde was reported above the detection limit in the lab, trip, and field blanks. The field sample data may be biased high because of contamination.

Metals

Precision. The precision of metals analyses was estimated using duplicate spike samples and field duplicate samples. All of the calculated relative percent differences (RPDs) for field duplicate samples met the precision acceptance objective (20%). The spike duplicate RPDs for copper (61%), selenium (63%), and molybdenum (25%) indicate that field sample results for these analytes may be more variable than anticipated.

A majority of the calculated RPDs for metals in coal were acceptable in the field duplicate samples. RPDs for copper (23.9%) and molybdenum (34.9%) were above the 20% objective, but acceptable considering the low concentrations and heterogenous sample material. Six duplicate spiked sample RPDs were reported outside the 20% objective, including copper (33%), lead (32%), and mercury (22%).

RPDs for duplicate fly ash sample analyses were within the precision objective. Duplicate bottom ash samples results met the 20% RPD objective except chromium (160%), manganese (60%), nickel (129%), cadmium (27%), lead (65%), and selenium (65%). These results indicate that the corresponding field sample analyte results for fly ash may also have higher than anticipated variability. No duplicate spike samples were analyzed for bottom ash; however, all but two of the fly ash duplicate spike RPDs were within the acceptance limits.

The field duplicate and duplicate spike sample RPDs for bottom ash sluice water and sluice water supply met the precision acceptance objective.

Accuracy. Accuracy for the metals analyses in stack gas was estimated using laboratory control samples (LCS) and spike samples. The majority of LCS recoveries for metals were within the accuracy acceptance limits (75-125%) and indicate that the analytical systems were in control during gas sample analyses. A majority of recoveries in the spike gas samples (impinger solutions, stack gas probe and nozzle rinses, and inlet gas filters) were also reported within the acceptance limits.

Mercury was recovered high (average recovery 310%) in both spike ESP inlet gas filter samples; however, these spikes were referenced to an anomalously low unspiked sample result (Run 4), and probably do not indicate a positive bias in the particulate-phase mercury results. Selenium recoveries (average 70% and 72%) were slightly low in both impinger solution samples and both spiked stack gas probe and nozzle rinses. Lead (average recovery 74%) was recovered just below the acceptance limit in both spike impinger solutions, and the concentration of arsenic was recovered slightly low (average recovery 71%) in both ESP inlet gas filter spike samples. Concentrations of these analytes may also be biased slightly low in the field sample results for the corresponding matrices.

The accuracy of the measurements of metals in coal was estimated using LCSs, spiked samples, and PE samples. The majority of LCS recoveries for metals were within the accuracy acceptance limits and indicate that the analytical systems were in control during the analysis of coal samples. The recoveries of three analytes were high in the coal PE sample (NIST 1632a). PE sample recoveries for barium (145%), copper (878%), and selenium (127%) indicate a high bias by INAA for these analytes in coal. The concentration of barium was within the acceptance limits in all four spiked samples, selenium recoveries were acceptable in three of four spiked samples, but copper recoveries were acceptable in only one of four spiked samples. The recovery of copper was low in two of four samples and high in one of four spiked samples. These results do not show any definite trends, but the quantitation of copper in coal may be suspect.

The accuracy of measurements of metals in fly ash was estimated using LCSs, spike samples, and PE samples. The majority of LCS recoveries for metals were within the accuracy acceptance limits and indicate that the analytical systems were in control during analysis for fly ash samples. A majority of the spike sample recoveries were also reported as acceptable; however, the recovery of cadmium was high (average recovery 150%) and the recovery of lead was low (average recovery 67%) in both fly ash spike samples. These recoveries indicate that cadmium results may be biased high and the lead result may be biased low in the fly ash field samples. Fly ash PE samples were analyzed after preparation by microwave digestion, INAA, XRF, and after being aspirated into the impinger train. Recoveries were above the acceptance limits for chromium (229%), nickel (202%), and arsenic (148%) when prepared using microwave digestion. The recovery of nickel (140%) was also high when analyzed by INAA. The recovery of manganese was high (440%) when analyzed by XRF. The recoveries of arsenic (144%) and cadmium (198%) were high for the fly ash sample after being aspirated into the impinger train. The nickel and arsenic results indicate that the field sample results may be biased high. Unlike nickel and arsenic, the remaining PE sample

recoveries listed above were high in only one analysis. This does not necessarily indicate an analytical bias for these analytes, but may indicate a non-systematic preparation or analytical problem.

6

PE sample recoveries were low for mercury (65%) after digestion in the microwave, and for barium (63%) after the aspirated fly ash result was corrected for the blank result. These recoveries indicate that the corresponding field sample results may also be biased low.

Blank Effects. One field blank and one trip blank were reported for filter samples. A majority of the target metals were detected in the filter field blank and the filter trip blank. These are trace impurities in the quartz fiber filters. Results above the detection limits were reported for barium (6.75 μ g), chromium (2.13 μ g), manganese (1.28 μ g), molybdenum (29.2 μ g), nickel (2.10 μ g), and lead (0.56 μ g) in the filter field blank. Only barium and molybdenum results were greater than five times the detection limits; all other analytes were less than three times the detection limits. The filter trip blank had results above the detection limits reported for barium (6.27 μ g), chromium (2.06 μ g), manganese (1.17 μ g), molybdenum (29.6 μ g), lead (0.57 μ g), and mercury (0.0424 μ g). Again, barium and molybdenum had results greater than five times the detection limit. Mercury was reported in concentrations greater than four times the detection limit, but the concentrations of all other trip blank analytes were two times the detection limits or less. All of the analytes listed above, except for nickel and mercury, were detected in both the field and trip blanks for filters. The stack results were corrected for the blank values, the details of which are shown in Appendix I. The corrections were small, with the exception of molybdenum.

One laboratory blank was reported for coal. All of the ICP metals and none of the AAS metals were detected in the coal lab blank. Only molybdenum (6.95 mg/kg) was reported above the detection limit (5 mg/kg). The concentration of this analyte may be biased high in the coal field sample results because of low-level laboratory contamination.

One lab blank, two field blanks, and one trip blank were reported for impinger solutions. The concentrations of three analytes were reported above the detection limits in the lab blank, including arsenic (0.005 mg/L), cadmium (0.006 mg/L), and lead (0.0049 mg/L). The cadmium result was six times the detection limit, whereas arsenic and lead were reported at less than two times the detection limits. A majority of the target analytes were reported in the impinger field blanks. Only manganese (0.017 and 0.025 mg/L), molybdenum (0.054 mg/L), and lead (0.0068 and 0.0122 mg/L) were reported in concentrations above the detection limits (0.01, 0.05, and 0.003 mg/L, respectively). A majority of the target analytes were detected in the impinger trip blank. Analytes reported above the detection limits include manganese (0.02 mg/L), molybdenum (0.059 mg/L), cadmium (0.0014 mg/L), and lead (0.015 mg/L). These analytes were reported in concentrations less than two times the detection limits, except for lead, which was five times the detection limit. The results from the lab, field, and trip blanks for impinger solutions indicates that the impinger field sample results may be

biased high for the listed analytes. Lead and cadmium may have significant bias because of the relatively high concentrations reported in these blanks.

One field blank and one trip blank for probe and nozzle rinses were reported. A majority of the target analytes were detected in both blanks. Those analytes reported in concentrations above the detection limits in the field blank include barium $(1.46 \,\mu g)$, arsenic $(0.63 \,\mu g)$, and lead $(0.50 \,\mu g)$. These concentrations are less than two times the detection limits. Mercury (at $0.0302 \,\mu g$) and cadmium (at $0.17 \,\mu g$) are the only analytes reported in concentrations above the detection limits $(0.009 \,\mu g \text{ and } 0.1 \,\mu g$, respectively) in the trip blank. These data indicate that the results may be biased slightly high in the probe and nozzle rinse samples because of contamination.

Anions

Precision. The precision for anions was estimated using field duplicate samples. The RPDs were above the precision acceptance limit (20%) for total phosphorus (28.9%) in the coal, and fluoride (23.6%) in the bottom ash. The variability may be higher than expected for these analytes in field samples.

Accuracy. The accuracy for anions was estimated using LCSs and spiked samples. All of the LCS recoveries and most of the spiked sample recoveries were within the accuracy acceptance limits (80-120%). Only fluoride (77%) and phosphate (76%) in solids were recovered low. These parameters may also be biased low in the field sample results.

Blank Effects. Lab blanks, field blanks, and trip blanks were reported for solids, filters, impingers, and waters. No analytes were reported in any of the lab blanks. Chloride $(4.34 \,\mu g)$ and sulfate $(0.0421 \,\text{mg})$ were reported in concentrations above the detection limit in the filter field blank. Chloride $(3.75 \,\mu g)$ and sulfate $(0.0398 \text{ and } 0.052 \,\text{mg})$ were also reported in concentrations above the detection limit in the filter trip blanks. These analytes may be biased high in the filter field sample results. Concentrations of phosphate $(0.228 \,\text{mg/L})$ and sulfate $(2.04 \,\text{mg/L})$ were reported just below the detection limits $(0.3 \,\text{and } 2.4 \,\text{mg/L})$, respectively) in the impinger field blank. Chloride was detected in the impinger trip blank in a concentration above the detection limit $(0.036 \,\text{mg/L})$. Overall, chloride and sulfate appear to have significant results reported in the filter field and trip blanks.

Volatile Organic Compounds

Precision. Precision is estimated for volatile organic compounds using LCS data. The percent coefficient of variation (CV) for the target compounds is 15.7% for benzene and 2.2% for toluene, which is well within the 35% precision objective. The percent CV for surrogates in gas samples (using mean recoveries) ranges from 6.0% to 13.7%. These results indicate that the precision is acceptable for this analytical method.

Accuracy. Accuracy is estimated for volatile organic compounds by LCS and surrogate recoveries. All of the LCS recoveries reported were within the accuracy acceptance limits. All but one of the surrogate recoveries were within the accuracy acceptance limits. These results indicate that the system was in control during sample analysis.

Blank Effects. Lab blanks and field blanks were analyzed for volatile organic compounds. No target compounds were detected in any of the lab blanks. A total of six field blanks were analyzed, three ESP inlet blanks and three stack blanks. No target analytes were detected in any of the field blanks.

Semivolatile Organic Compounds

Precision. Precision is estimated for semivolatile organic compounds using LCS and duplicate spike sample results. The percent CVs for LCSs ranged from 0 to 5.2%, and all were within the 35% precision objective. The duplicate spike sample precision data were also acceptable.

Accuracy. Accuracy for semivolatile organic compounds is estimated by LCS, spike sample, and surrogate recoveries. All of the LCS recoveries were reported as acceptable. Both spike sample recoveries were reported low for acenaphthene (average recovery 64%), 2,4-dinitrotoluene (average recovery 68%), and 4-nitrophenol (average recovery 63%). The results for these compounds may be biased low in the field sample results.

Blank Effects. Lab blanks and field blanks were analyzed for semivolatile organic compounds. No compounds were reported in concentrations above the detection limit in the lab blanks. Three compounds were reported in the field blanks above the detection limit $(10 \ \mu g)$, including diethylphthalate $(106 \ \mu g)$, dibutylphthalate $(13.8 \ \mu g)$, and bis(2-ethylhexyl)phthalate $(26.3 \ \mu g)$. Concentrations of these compounds may be biased high in the field sample results.

Aldehydes

Precision. No precision data were available for aldehydes.

Accuracy. No accuracy data were available for aldehydes.

Blank Effects. One lab blank, one field blank, and one trip blank were analyzed for aldehydes. Formaldehyde was reported in all three blanks in concentrations approximately two times the detection limits. The lab blank had $1.32 \mu g$ reported, the field blank had $5.4 \mu g$ reported, and the trip blank had $3.06 \mu g$ reported as present. These results indicate that the field sample data for formaldehyde may be biased high because of contamination.

OFA/LNB Test Analytical QC Data

The following potential concerns were identified by the quality control data.

Coal Samples

- The coal standard reference material sample analysis showed a high recovery for mercury (149%) analyzed by INAA, and variable recoveries for fluoride (49-91%). Mercury was also analyzed by CVAAS in the coal reference material, with 100% recovery, but the results of duplicate coal sample analyses showed high variability (37% RPD). Although the INAA and CVAAS QC results for mercury differ, these differences are based on limited data and do not definitively identify which method is superior.
- Duplicate sample results for arsenic in coal by INAA showed variability (58% RPD) outside the 20% objective, although the concentrations were very low.

Bottom Ash Samples

• Recoveries for fluoride in bottom ash samples were variable, measured first at 29 and 71% recovery, and then 81% and 88% in two MS/MSD pairs. Laboratory control sample results for fluoride (71-98% recovery) were outside the 80-120% objective. The results for fluoride in solid matrix laboratory method blanks were about twice the detection limit. For the water blanks, the fluoride results are only slightly above the detection limit. Overall, fluoride results are slightly outside the project accuracy objectives, but may be considered valid, recognizing the limitations in accuracy.

Gas Samples

- Based on internal standard and surrogate spike recovery data, the results for semivolatile organic compounds in gas samples should be considered qualitative or semiquantitative estimates. Relatively heavy loading of high-carbon particulate matter appears to have interfered with accurate quantitation. Laboratory controls that are independent of the sample matrix effects indicate acceptable analytical performance and suggest that the recovery problems are sample matrix related.
- The results for aldehydes were near detection limits and were not distinguishable from blank results, yielding no positive quantitative results for aldehydes.
- High levels of hexavalent chromium were found in the KOH impinger blanks but were found to be repeatable in three blanks and were consequently corrected for in the sample results.

5-26

- Frontier Geosciences reports that the mercury speciation results for the OFA/LNB test should not be compared with the OFA test results because of the experimental nature of the method at the time of the earlier test. Even now, the mercury speciation reports do not include QC results that permit a thorough evaluation of the measurements. Laboratory recovery checks are good, and the field blank results indicate no significant contamination problems (Mercury (II) and methyl mercury were both detected in the field blank at about twice the detection limit). Despite this, no results were given to verify the absence of species conversion during sampling, so the speciation results should be considered with caution.
- The spike recoveries for mercury were low (50-69%) in the permanganate impingers.
- The spike recoveries for selenium in gas samples were variable, ranging from 67% to 142%.
- Recoveries were slightly above the 75-125% objective for arsenic (125-130%) and beryllium (130-137%) in fly ash standard reference material samples analyzed by GFAAS and ICP-AES.
- Recoveries were low for cadmium (51-60%) and cobalt (68-73%) in ash standard reference material samples analyzed by ICP-MS for the size-fractionated particulate matter samples. Recoveries were high (133-179%) and variable (30% RPD) for mercury in the same samples. The recoveries were only slightly low for nickel (69-74%).
- The rinsates from Kapton trip blank samples analyzed for metals by ICP-MS showed concentrations of most metals from 10 to 100 times the reported detection limit. ICP-MS is very sensitive and likely to detect metals in blanks at concentrations much lower than ICP-AES or GFAAS, but the extremely low detection limits reported may reflect how low the method can quantitate elements, but not how low a sample may be distinguished from a blank, or how difficult it may be to obtain sufficiently analyte-free sampling media. For example, relative to the reported detection limits, quite high levels were reported for chromium $(4 \ \mu g/filter, DL=0.01 \ \mu g)$, selenium (2 $\mu g/filter, DL=0.04 \ \mu g)$, and vanadium (2 $\mu g/filter, DL=0.03 \ \mu g)$.
- Filter blank samples for stack gas showed appreciable amounts of barium, chromium, manganese, molybdenum, nickel, copper, mercury, and vanadium. These generally contribute only a small bias to the total sample train measurement, but they are repeatable and may be corrected for.
- Sample results for PAHs showed very high variability. Internal standard and surrogate recoveries indicate significant matrix interferences. The PAH results should be considered order-of-magnitude estimates.
- Field blank results for PAHs were generally very low, except for acenapththene and phenanthrene, which showed 469 and 120 ng/sample, respectively. The high sensitiv-

ity of the HRGC/HRMS method results in the detection of numerous compounds, although mostly at extremely low levels. No significant contamination is suspected.

• Toluene was detected at very low, but similar concentrations in blank and regular VOST samples. The toluene results in the sample may be biased slightly high if not artifacts of blank contamination.

Metals

Precision. The precision of metals analyses was estimated for coal samples using duplicate samples, which include a component of sampling variability. The 20% RPD precision objective was met for all of the target metals, except arsenic (58% RPD). In addition to the INAA analysis, mercury was analyzed by CVAAS and had a duplicate sample precision of 37% RPD, greater than the 15% RPD by INAA. The mercury concentrations measured were low, however, (0.1-0.2 mg/kg) and agreed reasonably well between the methods. These results are presented in Appendix Table F.2-13.

In bottom ash samples, precision was estimated from matrix spike duplicate analyses, all of which were within the precision objective for the target metals. These results are included in Table F.2-7. Duplicate ash sample results also provide a measure of precision, although many of the metals were not detected. In the duplicate samples, precision estimates were slightly higher than the MSD pairs, with RPDs greater than 20% for barium (21%), lead (106%), and phosphorus (24%). Except for lead, these results are typical, compared to previous FCEM data. The lead results show greater variability than expected, but the measured values in the duplicate samples were relatively low (6 and 20 mg/kg).

Precision estimates for ICP-MS analysis of size-fractionated particulate matter were based on replicate analyses of NIST 1633a fly ash reference standard, as shown in Table F.2-10. Only the precision for mercury (29.7% RPD) was outside the 20% objective, but it was acceptable, considering the low concentration in the standard.

Precision estimates for metals in gas samples, based on matrix spike duplicate samples results, were within the 20% objective for all target metals. These results are presented in Table F.2-7.

Another measure of precision for metals analyzed by ICP-AES and AAS techniques is the relative standard deviation (also known as % coefficient of variation, %CV) for laboratory control sample recoveries. Although not indicative of the sample matrix, the percent CV provides a measure of laboratory method repeatability. As shown in Appendix Table F.2-5, the precision is good for all target metals. (Arsenic and selenium results by ICP-AES do not meet the objective, but GFAAS is the primary analytical technique. The ICP-AES results are available because it is a multi-element technique, but it is not recommended for these metals.) **Accuracy.** The accuracy of the metals analyses was estimated for coal samples using NIST 1632a and SARM 20 standard reference coal sample results, which are presented in Table F.2-13. All of the target metals analyzed in the reference sample were within the 75-125% recovery objective, except beryllium by ICP-AES (72%) and mercury by INAA (149%). Mercury was also analyzed by CVAAS and had a recovery of 100%.

The accuracy of metal results from ash samples was estimated from matrix spike (Table F.2-7) recoveries. Matrix spike recoveries were all within the 75-125% objective for target metals.

Analysis of NIST fly ash standard reference material (Table F.2-9) was used to represent metal recoveries in particulate phase gas samples. The matrix of the standard is not identical to that of the samples, especially since stack gas particulate matter samples are digested along with the filters. Recoveries in the ash standard were slightly high for arsenic (125-130%) and beryllium (130-137%). Quartz filters were also digested and subsequently spiked to determine the effects of the filter digestate on recovery. These results, summarized in Table F.2-11, show a tendency to low recoveries for lead (61-67%), selenium (24-76%), molybdenum (40-92%) and mercury (45-46%).

Matrix spikes were used to estimate the accuracy of metal analyses in vapor-phase gas samples. Matrix spike results are presented in Table F.2-7. Recoveries were outside the 75-125% objective for mercury (50-132%) and selenium (67-142%). It has been seen from previous FCEM data that these are difficult elements and such recoveries are not uncommon.

The accuracy of ICP-MS analyses of size-fractionated particulate matter samples was estimated from NIST 1633a fly ash recoveries. Recoveries were rather low for cadmium (51-60%) and high for mercury (133-179%), but the reference concentrations were very small, 1 mg/kg Cd and 0.2 mg/kg Hg. Recoveries were slightly low for cobalt (68-73%) and nickel (69-74%).

Laboratory control sample recoveries for metals (Tables F.2-5 and F.2-6), which demonstrate the analytical accuracy in the absence of sample matrix effects, were all within expected limits.

Blank Effects. Laboratory method blanks, presented in Table F.2-2, representing both solid and liquid sample types, indicated no significant laboratory contamination problems affecting quantitation of target metals. ICP-MS method blanks showed metals detectable above the reported detection limit, but well below amounts measured in samples. Unlike the reports submitted with ICP-AES and AAS measurements, it appears that the reported detection limit for ICP-MS reflects the sensitivity of the method (i.e., the ability to measure extremely low quantities) and not the probability of detecting (or not detecting) an analyte in a blank.

Reagent blank results, presented in Table F.2-3, show relatively high amounts of barium $(27 \mu g)$ and molybdenum $(30 \mu g)$ in the "final filter blanks" analyzed by ICP-MS. Quartz

filter blanks analyzed by ICP-AES and AAS also showed relatively high amounts of molybdenum $(30 \mu g)$ and low amounts of the other target metals. This represents a potential low-level bias for molybdenum in this sample fraction, but barium results were much higher in the samples and not affected by the blank. Significant amounts of hexavalent chromium were detected in the potassium hydroxide impinger solutions, which would bias sample results. Triplicate reagent blanks were analyzed and showed very repeatable results, so reliable blank correction could be performed, as appropriate.

Rinsates from two Kapton blanks were analyzed for metals by ICP-MS. The results of these analyses are shown in Table F.2-4. There were no significantly high levels of any metals. Small amounts of arsenic $(1.5 \ \mu g)$, chromium $(4 \ \mu g)$, selenium $(2 \ \mu g)$, and vanadium $(2 \ \mu g)$ were detected, but should not affect the sample results. In contrast to the "final filter blanks," barium and molybdenum results in the Kapton blank rinsates were quite low.

The field blank results for metals, based on the recovery of assembled sample trains that were leak-checked at the sample location, are summarized in Table F.2-1. These results show small detectable amounts of arsenic $(4 \ \mu g)$, barium $(15 \ \mu g)$, manganese $(7 \ \mu g)$, nickel $(4 \ \mu g)$, and phosphorus $(50 \ \mu g)$ in the ESP inlet gas thimble/precutter fraction. These are about the same magnitude as measured for arsenic, manganese, and nickel in the actual samples and represent a potential low-level bias for these metals in this sample fraction. No significant blank effects were detected in the field blank impinger samples analyzed for metals.

Anions

Precision. Estimates of precision for anion measurements in bottom ash samples were based on matrix spike duplicate recovery data, which are presented in Appendix Table F.2-7. In these samples, the 20% RPD precision objective was met for chloride, but not for fluoride (46% RPD). In bottom ash duplicate samples, only fluoride was detected, and with good agreement between results (3% RPD).

In gas samples, the precision for anions, based on MS/MSD recoveries in impinger solutions, were within the 20% RPD objective.

The precision of anion laboratory control samples results was very good (less than 10% CV).

Accuracy. Estimates of accuracy for anion measurements in bottom ash samples were based on matrix spike recoveries. These results are presented in Table F.2-7. Recoveries of chloride met the 80-120% objective. Two sets of MS/MSD samples were analyzed for fluoride. Recoveries were 29 and 71% in one pair; 81 and 88% in the second pair. While matrix spike data include a component of error (the spiking process) not inherent in unspiked sample measurements, these results suggest that the accuracy of the fluoride measurements does not meet the 80-120% objective.

In gas samples, the accuracy for anions, as determined from matrix spike recovery in impinger solutions, were within the 80-120% objective for chloride and fluoride and slightly outside the objective for sulfate (74-123%).

The accuracy of laboratory control sample results for anions was very good, within 90-110% recovery, except for fluoride. Fluoride recoveries ranged from 71 to 98%, but averaged $90\pm7.4\%$, indicating a slight negative bias.

Blank Effects. The laboratory method blank results associated with solid-phase analyses for fluoride were about twice the detection limit and about the same magnitude as measured in bottom ash samples, contributing a significant potential bias to those analyses, although fluoride contamination was not significant in the field blank impinger. No problems were identified with anion method blanks associated with impinger sample analyses.

The field blank results for anions in impinger solutions showed no problems. On the filter blank, chloride was found at levels similar to those in the samples, although the contribution to the total chloride in the train is small.

Volatile Organic Compounds

Precision. Precision estimates for measurements of volatile organic compounds in gas samples were based on the dispersion of surrogate spike recovery data, expressed as the percent coefficient of variation (%CV). The precision was within the 35% CV objective for each of the three surrogate spike compounds. These results are summarized in Table F.2-19.

Precision was also measured as the relative percent difference for duplicate method spike recoveries. These results, presented in Appendix Table F.2-18, showed very good precision. The RPDs were less than 30% RPD for 37 of 39 spike compounds. For the target compounds, the RPDs were 3.6% for toluene and 6.5% for benzene. The RPDs were relatively high only for vinyl acetate (55% RPD) and chloromethane (53% RPD).

Accuracy. Accuracy estimates for volatile organic analyses of VOST samples, expressed in terms of surrogate spike recoveries (Table F.2-19) and method spike (spiked blank VOST tubes) recoveries (Table F.2-18), were within the recovery objective of 50-150% for all analytes. Method spike recoveries for toluene, although within the objective, were somewhat high, 137-142%. Recoveries for benzene were 89-95%.

Blank Effects. The analytical results for VOST method blanks showed no significant contamination. Methylene chloride and toluene were found just above the 10 ng detection limit, up to 13 ng.

VOST field blanks showed relatively high amounts of acetone (150 ng), methylene chloride (2700 ng), and trichlorofluoromethane (55 ng). The concentration of toluene

(11 ng) was just above the 10 ng detection limit. Similar levels of toluene were detected in a small number of samples and may be biased slightly high if they are not contamination artifacts.

Semivolatile Organic Compounds (PAHs)

Precision. Precision estimates for PAHs in gas samples were based on the dispersion of surrogate spike recovery data. These results are summarized in Table F.2-17. These results show extremely high variability and suggest poor precision in the PAH measurements.

As shown in Table F.2-17, surrogate spike recoveries were acceptable in clean matrix samples, such as the XAD trip spike, lab spike, and lab blank, but very poor in actual project samples. This indicates a matrix interference in the flue gas samples.

Accuracy. Recovery data for the PAHs, like the precision estimates, indicate analytical problems. Surrogate spike recoveries in gas samples ranged from 0-43%, 66-101%, and 102-918% for the three surrogate compounds. In the virgin (not used for sampling) XAD samples, surrogate and analyte spike recoveries were all very good, indicating acceptable laboratory performance.

Internal standard recovery data, shown in Table F.2-15, are excessively low, more so in the ESP inlet samples than in the stack. The samples had high particulate matter levels, and it is presumed that unburned carbon in the particulate matter caused significant interference with the measurements, resulting in the extremely low internal standard recoveries. Because measurement results are factored by the inverse of the internal standard recovery, the effects of bias would tend to be lessened, but the imprecision associated with the standards adds further imprecision to the results.

The PAH measurements should be considered order-of-magnitude estimates.

Blank Effects. Blank results did not show significant problems. Field blank results showed somewhat high levels of acenaphthene (469 ng) and phenanthrene (120 ng). Because of the extreme sensitivity of the method, lesser levels of other semivolatile organic compounds were also detected, but at concentrations that should not be of concern.

Aldehydes

Precision. Duplicate analytical results for aldehydes produced no precision estimates because the analytes were not detected.

Accuracy. Accuracy estimates for aldehydes were based on matrix spike recoveries, which were 90% for formaldehyde and 92% for acetaldehyde. Trip spikes and lab spikes

5-32

were also performed, yielding results of 83% and 89% for formaldehyde, 81% and 100% for acetaldehyde. These are all well within the accuracy objective of 50-150% recovery.

Blank Effects. Aldehyde results in samples were not distinguishable from those in the blank samples, which included field, trip, reagent, and lab blanks, precluding the ability to determine whether aldehydes were present in the gas samples.

Mercury Speciation (Frontier Geosciences Method)

Precision. No precision data were available for mercury speciation results.

Accuracy. Lab spikes were performed to demonstrate instrumental accuracy. Recoveries were good, ranging from 88% to 106%, but these do not address matrix effects or address the sample collection effectiveness, with respect to representative speciation. The method has demonstrably reliable features that make it a useful empirical tool, but species conversion in the sampling train cannot be discounted.

Blank Effects. Field blanks analyzed for mercury speciation showed no blank contamination problems.

Material Balances

Evaluating data consistency can be another overall data quality evaluation tool. Material balances for major elements can be used to verify the internal consistency of stream flow rates. Material balance closures for trace species can be used to indicate whether the samples collected were representative with respect to the trace element concentrations and can help identify analytical biases in one or more types of samples.

The results of material balances around the unit for the OFA and OFA/LNB tests are shown in Table 5-8. Closure is defined as the ratio of outlet to inlet mass rates for a particular substance. A 100% closure indicates perfect agreement. When trace substances are analyzed, a closure of between 70 and 130% has been set as a goal for the project. This range reflects the typical level of uncertainty in the measurements and, therefore, allows one to interpret the inlet and outlet mass flow rates as being equivalent. The 95% confidence intervals about the closures have been calculated using an error propagation analysis, which is discussed in detail in Appendix E.

The material balance calculation included a single inlet stream (coal), and three outlet streams: bottom ash, ESP ash, and stack gas. For the OFA/LNB test, the ESP hopper ash was not analyzed, so its composition was assumed to be equal to that of the ESP inlet particulate matter.

Closures for the major species (except for sodium in the OFA test) met the project goal, which supports the stream flow rates used in the material balances. For the OFA test, 8

Table 5-8Material Balance Results - OFA and OFA/LNB Tests

	OFA	Test	OFA/LN	IB Test
Substance	Out/In, %	95% CI, %	Out/In, %	95% CI, %
Major Species				
Ash	100	^a	100	*
Aluminum	100	24	89	24
Iron	71	17	85	33
Sodium	68	14	102	23
Sulfur	99	11	97	13
Titanium	103	31	81	23
Target Species				
Arsenic	112	36	111	41
Barium	47	16	105	32
Beryllium	87	40	100	24
Cadmium	NC		NC	
Chloride	64	25	113	49
Chromium	53	15	83	32
Cobalt	65	13	89	29
Copper	40	25	55	16
Fluoride	128	120	96	38
Lead	130	90	70	22
Manganese	80	75	110	19
Mercury	71	27	90	31
Molybdenum	NC		NC	
Nickel	78	180	65	40
Phosphorus	57	36	54	22
Selenium	82	30	90	38
Vanadium	65	14	97	24

* Ash balance was forced to close by assuming an 80:20 fly ash to bottom ash ratio.

CI = Confidence Interval.

NC = Not calculated because substance was not detected in the coal.

of the 15 target elements detected in the coal had closures that met the project goal, indicating that, in general, representative samples were obtained. Barium, chlorine, chromium, cobalt, copper, phosphorus, and vanadium all show closures outside the project goal. The fact that none of the 95% confidence intervals for these elements contain 100% indicates that imprecision is not responsible for the poor closures. The low closures may indicate an analytical bias in one or more of the process streams.

For the OFA/LNB test, 12 of the 15 target elements detected in the coal had closures within the desired range of 70 to 130%. The closures for copper, nickel, and phosphorus were below 70%. For copper and phosphorus, this may indicate a bias in one or more of the process streams. For nickel, because the 95% confidence interval contains 100%, imprecision may be partly responsible for the low mean closure.

EXAMPLE CALCULATIONS

This section presents selected examples of the calculations used to develop the results shown in Section 3. Specifically, the calculation of stream flow rates, mean concentration values and confidence intervals, and emission factors are presented.

Stream Flow Rates

Appendix F contains information about the stream flow rates measured at Site 16 during the sampling period. Coal flow rates were obtained directly from plant meters, and flue gas flow rates were measured during sampling. Neither the bottom ash nor the ESP collected ash flow rates could be measured directly. These rates were calculated from other available data.

For the OFA test, the particulate loading measured in the ESP inlet gas is not considered accurate. The measured loadings showed approximately 40-50% of the coal ash appearing in the ESP inlet gas. For this type of boiler, a split closer to 80:20 (fly ash to bottom ash) would be expected. The 80% fly ash figure was later confirmed by another contractor (SRI) and again by Radian during the OFA/LNB test. It is now suspected that during the OFA sampling, ESP inlet particulate matter was lost through the nozzle between sampling and recovery. The sampling probe was modified for the OFA/LNB test to prevent this from occurring.

The bottom ash and fly ash rates were calculated from the coal ash rate, assuming an 80:20 fly ash to bottom ash ratio. For example, for Run 1 on March 3, 1991, the following data were collected:

Coal Flow Rate = 330,000 lb/hr (dry)Coal Ash Content = 12% (dry)

The bottom ash and fly ash flow rates were calculated as:

Fly Ash Rate	= (0.80)(0.12)(330,000 lb/hr) = 31,700 lb/hr
Bottom Ash Rate	= (0.20)(0.12)(330,000 lb/hr) = 7,920 lb/hr
For the OFA/LNB test, the particulate loadings measured at the ESP inlet are considered valid. Therefore, a different approach was taken in calculating the ash flow rates. The ESP collected ash rate was calculated as the difference between the ESP inlet particulate matter rate and the stack particulate matter rate. The bottom ash rate was calculated as the difference between the coal ash rate and the ESP inlet particulate matter rate (adjusted for loss on ignition, LOI). For example, for Run 2 on May 19, 1993, the following data were obtained:

= 313,000 lb/hr (dry)
= 9.47% (dry)
= 427 lb/hr
= 22,920 lb/hr
= 5.19%

The bottom ash and collected fly ash flow rates were calculated as:

Bottom Ash Rate	= (0.0947)(313,000 lb/hr) - (1-0.0519)(22,920 lb/hr) = 7,920 lb/hr
Fly Ash Rate	= 22,920 lb/hr - 427 lb/hr = 22,500 lb/hr

Means and Confidence Intervals for Stream Concentrations

The mean concentrations and 95% confidence intervals (CIs) about the mean were calculated for each target substance in the streams sampled. The means were calculated according to the conventions listed in Section 3. The equations used to calculate the 95% confidence intervals are presented in Appendix E. Example calculations for chromium in the stack gas during the OFA/LNB test follow here; these results were shown in Table 3-8.

The concentration data (in μ g/Nm³) given for chromium in Table 3-8 are:

	<u>Run 2</u>	<u>Run 3</u>	<u>Run 4</u>
Solid Phase	20	27	23
Vapor Phase	1.0	0.77	1.1
Total	21	28	24

The mean is calculated from the individual run totals:

Mean =
$$(21 + 28 + 24)/3$$

= 24

The sample standard deviation of the individual run totals is calculated:

$$S_{p} = \sqrt{\left[\left(21 - 24 \right)^{2} + \left(28 - 24 \right)^{2} + \left(24 - 24 \right)^{2} \right] / 2}$$
(eq. 1)
= 3.5

The standard deviation of the average is calculated according to Equation 6 in Appendix E for N = 3:

$$S_{p} = 3.5/\sqrt{3}$$
 (eq. 2)

= 2.0

The bias error is found by root-sum-squaring the product of the bias error and the sensitivity from each run (see Equation 2 in Appendix E). According to the conventions listed in Section 3, no bias error is assigned to values above detection limits, whereas a bias error of one-half the detection limit is assigned to values below detection limits. The sensitivity of the mean to each run in this case is 1/3.

$$\beta_{\rm r} = \sqrt{\left(1/3 \ge 0\right)^2 + \left(1/3 \ge 0\right)^2 + \left(1/3 \ge 0\right)^2}$$
(eq. 3)
= 0

The total uncertainty in the result is found from Equation 1 in Appendix E:

$$U_{r} = \sqrt{\beta_{r}^{2} + (t \times S_{p})^{2}}$$

$$= \sqrt{0^{2} + (4.303 \times 2.0)^{2}}$$
(eq. 4)

= 8.6

Thus, the result is reported as $24 \pm 8.6 \,\mu g/\text{Nm}^3$.

Unit Energy Emission Factors

In addition to the gas-phase concentrations, unit-energy-based emission factors were developed for each target substance. These values were determined by calculating the mass flow rate of a substance in the flue gas (mean concentration times mean flow rate) and dividing by the mean heat input to the boiler during testing. The mean heat input is the product of the mean coal flow rate and the mean higher heating value (HHV) of the coal.

For example, note the calculation of the emission factor for chromium during the OFA/LNB test. The mean coal flow rate is 315,000 lb/hr on a dry basis. The mean HHV of the coal is 13,800 Btu/lb on a dry basis. Multiplying the coal flow rate by the HHV gives a mean heat input of 4.3×10^{9} Btu/hr. The mean chromium mass flow through the stack (the product of the mean concentration, $24 \,\mu g/\text{Nm}^{3}$, and the mean gas flow rate, 1,710,000 Nm³/hr) is $4.1 \times 10^{7} \,\mu g/\text{hr}$ or 0.090 lb/hr. When the mean mass flow rate is divided by the mean heat input, an emission factor of 21 lb/10¹² Btu is obtained, as shown in Table 3-9.

The 95% confidence intervals for emission factors were calculated according to the equations presented in Appendix E. For each parameter (flue gas flow rate, concentration, coal flow rate, and HHV) the mean, standard deviation, number of points, and bias estimates were used to calculate the combined uncertainty in the mean emission factors.

7

1

GLOSSARY

Btu	British Thermal Unit
CAAA	Clean Air Act Amendments of 1990
CI	Confidence Interval
CV	Coefficient of Variation
CVAAS	Cold Vapor Atomic Absorption Spectrophotometry
DGA	Double Gold Amalgamation
DQO	Data Quality Objective
dscfm	Dry Standard Cubic Feet per Minute (1 atm, 68°F)
ESP	Electrostatic Precipitator
GFAAS	Graphite Furnace Atomic Absorption Spectrophotometry
HGAAS	Hydride Generation Atomic Absorption Spectrophotometry
HHV	Higher Heating Value
IC	Ion Chromatography
ICP-AES	Inductively Coupled Plasma Atomic Emission Spectroscopy
ICP-MS	Inductively Coupled Plasma Mass Spectrometry
INAA	Instrumental Neutron Activation Analysis
ISE	Ion Selective Electrode
LNB	Low NO _x Burners
MS/MSD	Matrix Spike/Matrix Spike Duplicate
MW	Megawatt or Microwave
NC	Not Calculated
ND	Not Detected (below detection limit)
NIST	National Institute of Standards and Technology (formerly National Bureau of Standards, NBS)
Nm ³	Dry Normal Cubic Meter (0°C, 1 atm)
OFA	Overfire Air
РАН	Polynuclear Aromatic Hydrocarbon
РОМ	Polycyclic Organic Matter
QA/QC	Quality Assurance/Quality Control
RPD	Relative Percent Difference

APPENDIX A: SAMPLE COLLECTION, PREPARATION, AND ANALYSIS

This appendix presents the methods used to collect and analyze each type of sample. Difference between the methods used during the OFA and OFA/LNB tests are noted where appropriate. Detailed method tables are included.

Multi-Metals Sampling Trains

Multi-metals samples were collected according to the procedure described in Section 3.1 of 40 CFR, Part 266, Appendix IX, "Methodology for the Determination of Metals Emissions from Hazardous Waste Incineration and Similar Combustion Processes," with modifications as noted here. This method provides for the collection of a flue gas sample at isokinetic conditions while traversing the duct according to EPA Method 1. Particulate matter is collected on a filter (which is also used to determine particulate loading) and the vapor-phase species are absorbed in an impinger train consisting of:

- Two impingers containing 5% $HNO_3/10\%$ H_2O_2 , which are analyzed for all metals of interest; and
- Two impingers containing 4% $KMnO_4/10\% H_2SO_4$, which are analyzed for mercury only.

The multi-metals method specifies that HNO_3/H_2O_2 impinger solutions be evaporated to near dryness prior to analysis. However, due to concern over the possible loss of volatile metals, this procedure was not followed. Instead, the impinger solutions were analyzed as recovered to avoid any loss of volatile metals.

Stack

Samples were collected at the stack according to the method. A Method 5 type train was used, with particulate matter captured on an out-of-stack quartz filter maintained at 250°F during sampling. The glass-lined probe was also maintained at 250°F. After sampling, the glass nozzle and probe liner were rinsed first with acetone, then with 0.1 N nitric acid according to the multi-metals method.

ESP Inlet

The multi-metals method specifies that particulate matter be collected according to the extractive Method 5. However, the high particulate loading at the ESP inlet precluded the use of a Method 5 filter. Instead, particulate matter was collected with an in situ quartz thimble (Method 17) at the ESP inlet. A Teflon[®] transfer line connected the thimble holder to the impinger train.

A single train was used to sample both the A- and B-side ducts. After sampling, the glass nozzle and the thimble holder were rinsed first with acetone, then with 0.1 N nitric acid according to the multi-metals method. The Teflon[®] transfer line was rinsed with 0.1 N nitric acid.

As was discussed in Section 5, it is suspected that particulate was lost through the nozzle during the OFA test. To prevent this from occurring during the OFA/LNB test, a cyclone precutter was added to the train ahead of the thimble. The solids collected in the cyclone were added to those collected in the thimble. The precutter was rinsed with acetone.

Anions Sampling Trains

Anions samples were collected using a Radian procedure designed for collection of HCl, HF, and SO₂. Particulate matter is captured on a filter and the acid gases are absorbed in an impinger train consisting of two impingers containing a solution of 0.013% Na₂CO₃, 0.013% NaHCO₃, and 6% H₂O₂.

Stack

An out-of-stack Method 5 glass filter was used at the stack. The filter and glass-lined probe were maintained at 250°F during sampling. The samples were collected isokinetically at a single point within the stack. After sampling, the glass nozzle and probe liner were rinsed with fresh anions impinger solution.

ESP iniet

An in situ Method 17 glass thimble was used to collect anions samples at the ESP inlet. A Teflon[®] transfer line connected the filter holder to the impinger train. Samples were collected isokinetically at a single point. During sample recovery, the nozzle, thimble holder, and the transfer line were rinsed with fresh anions impinger solution.

For the OFA/LNB test, a cyclone precutter was added to the train ahead of the thimble to prevent the loss of particulate. The precutter was rinsed with fresh anions impinger solution during recovery.

Semivolatiles Sampling Trains (MM5)

Samples were collected according to SW-846 Method 0010, "Modified Method 5 (MM5) Sampling Train for Semivolatile Principal Organic Hazardous Compounds." Particulate matter is collected on a filter and vapor-phase compounds are captured in a chilled XAD resin cartridge. Samples are collected isokinetically while traversing the duct according to EPA Method 1.

Stack

Samples were collected according to the method using an out-of-stack quartz filter maintained at 250°F. The glass-lined probe was also maintained at 250°F. The glass nozzle and probe liner were rinsed with methylene chloride during sample recovery.

ESP Inlet

Particulate collection was the same as for the multi-metals train at the ESP inlet. An in situ quartz thimble was used, and a Teflon[®] transfer line connected the thimble holder to the train holding the XAD cartridge. A single train was used to sample both the A- and B-side ESP inlet ducts. The nozzle, thimble holder, and transfer line were rinsed with methylene chloride during sample recovery.

For the OFA/LNB test, a cyclone precutter was added to the train ahead of the thimble to prevent the loss of particulate. The precutter was rinsed with methylene chloride during recovery.

Volatile Organic Sampling Train (VOST)

Samples for volatile organic compounds were collected according to SW-846 Method 0030 "Volatile Organic Sampling Train (VOST)." Volatile organics are captured by a pair of sorbent resin traps in series and maintained at 20°C. The first trap contains Tenax and the second trap contains Tenax followed by petroleum-based charcoal. Samples are collected at rate of 0.5 liters per minute from a fixed point in the duct. After sampling, the resin traps are sealed and returned to the laboratory for analysis.

For the OFA test, each sampling run included the collection of 1-L, 5-L, and 20-L samples. For the OFA/LNB test, each run included the collection of three 20-L samples.

Aldehyde Sampling Train

Aldehyde samples were collected according to SW-846 Method 0011, "Sampling for Formaldehyde Emissions from Stationary Sources." Aldehydes are absorbed in impingers containing a solution of 2,4-dinitrophenylhydrazine and HCl. A filter is used to prevent particulate matter from entering the impingers. Samples are collected isokinetically from a single point within the duct.

For convenience, the aldehydes impinger trains shared filters with the anions trains, because the particulate is not analyzed for aldehydes. After an anions sample was collected, the anions impinger train was disconnected and the aldehydes impinger train was attached. Therefore, the particulate capture was identical to that previously described for the anions trains. No aldehydes samples were collected at the ESP inlet during the OFA/LNB test.

Chromium (VI) Sampling Train (OFA/LNB Test)

Samples for chromium (VI) were collected at the stack during the OFA/LNB test according to the procedure described in Section 3.2 of 40 CFR, Part 266, Appendix IX, "Determination of Hexavalent Chromium Emissions from Stationary Sources." The method uses a probe/impinger train equipped with a recirculation line to continuously pump the impinger solution to the tip of the probe during sampling. No filter is used; particulate is allowed to enter the impinger train. The impinger solution consists of a KOH solution designed to maintain the pH above 8.5 to prevent reduction of Cr(VI).

The method calls for a 0.1 N KOH solution. However, the method was not designed to contend with the high levels of CO_2 and SO_2 typically found in power plant stack gas. Therefore, the method was modified by using a 10 N KOH solution to guarantee a pH of 8.5 or greater during sampling.

After recovery of the KOH solution, all train components were rinsed with 0.1 N nitric acid, as per the method. The analysis of this rinse is necessary only if total chromium results are desired. Nitric acid rinses were held for possible analysis in the future.

Mercury Speciation Train (OFA/LNB Test)

The Frontier Geosciences mercury speciation train was used to collect samples at the stack during the OFA/LNB test. The solid sorbent technique was developed by Nicholas Bloom, now with Frontier Geosciences, Inc., but formerly with Brooks Rand, Ltd. A quartz-lined probe was inserted into the stack, and flue gas was extracted non-isokinetically from a single point at a rate of 0.5 liters per minute. The flue gas then passed through a series of four solid adsorbent cartridges which were used to trap the various vapor-phase mercury species. The cartridges were maintained at approximately 110°C in a heated jacket outside the duct. The first two cartridges contained KCl-impregnated soda lime which is designed to capture ionic inorganic mercury species (Hg^{2+}) as well as monomethyl mercury species (such as CH_3HgCl). The third and fourth cartridges contained iodated carbon, which is designed to capture elemental mercury. A glass wool plug ahead of the adsorbent cartridges prevented particulate from entering the adsorbents. This plug was not analyzed, because the single-point, non-isokinetic sampling does not provide representative particulate capture. Only vapor-phase species were determined.

The KCl/soda lime traps were dissolved in acetic acid solutions. Ionic and methyl mercury were determined by aqueous-phase ethylation, purging onto a carbotrap, cryogenic GC separation, and detection with cold vapor atomic fluorescence spectrometry (CVAFS). Methyl mercury was determined as methylethyl mercury, while inorganic ionic mercury was determined as diethyl mercury. Elemental mercury on iodated carbon traps was determined by digesting with a mixture of HNO_3/H_2SO_4 and BrCl, reducing with SnCl₂, purging and preconcentrating on gold, and detecting with CVAFS.

Impactor Sampling (OFA/LNB Test)

Size-fractionated particulate samples were collected at the stack using a University of Washington Mark 5 cascade impactor. Samples were collected isokinetically at a fixed point. Particulate on each impactor stage was collected on Kapton (polyamide) substrates, and a quartz fiber filter was used as the final filter. Prior to analysis, adjacent impactor stages were combined to provide three size fractions: approximately $<3 \mu m$, $3-10 \mu m$, and $>10 \mu m$. Quartz final filters were microwave digested and analyzed by ICP-MS. Kapton substrates were rinsed and soaked in dilute nitric acid. The solids and nitric acid solution were then microwave digested and analyzed by ICP-MS.

Process Sample Collection

The details of sample collection at each location are listed below. Differences between the methods used for the OFA and OFA/LNB tests are noted where appropriate.

Coal

For the OFA test, a metal corer was used to collect samples from the clean-out ports at the bottom of each the coal bunkers. For the OFA/LNB test, samples were collected by opening a tap on each of the coal feeders, allowing the coal to fall into a plastic bucket, and subsampling the catch with a plastic scoop. For both tests, multiple grab samples were composited directly into a five-gallon plastic bucket and split with a riffler to provide two one-kilogram samples for analysis.

Bottom Ash and Bottom Ash Water

To obtain representative bottom ash samples, the bottom ash was sluiced prior to each test run and allowed to accumulate in the boiler bottom during the run. At the end of the run, the bottom ash was sluiced again. Wet bottom ash samples were collected from the discharge of the sluice pipe at the ash basin. For the OFA test, a polyethylene dipper was used to collect multiple grab samples. For the OFA/LNB test, a one-liter plastic graduated cylinder was used to traverse the discharge stream during sluicing to obtain one grab sample per minute during sluicing.

For both tests, the multiple grab samples were combined in a large container and the solids were allowed to settle. The bottom ash water was then siphoned off and filtered for collection. The bottom ash water samples collected during the OFA/LNB test were archived for possible analysis in the future.

ESP Ash

A metal sample thief was used to collect samples from the ESP hoppers. For the OFA test, multiple grab samples were collected from each of the 16 ESP hoppers during the gas sampling periods. A composite sample was generated by combining all hopper samples in equal proportions by volume. For the OFA/LNB test, the samples from each of the 16 hoppers were kept separate and archived for possible analysis in the future.

Sluice Supply Water

Grab samples of the water used for ash sluicing were collected at the recycle pond water intake immediately before sluicing. Single grab samples were collected for both the OFA and OFA/LNB tests. The samples collected during the OFA/LNB test were archived for possible analysis in the future.

Detailed Sample Collection/Preparation/Analysis Tables

Table A-1 lists the techniques used to collect, preserve, and handle the samples at Site 16. Analytical methods applied to coal samples are listed in Table A-2. Analytical methods for inorganic species in other samples are listed in Table A-3. The analytical methods applied to organic compounds are listed in Table A-4.

Stream	Collection Method	Fraction Description	Sample Handling and Preservation	Comments
Flue Gas Samples				
ESP Inlet Gas, Stack Gas	Specified in Section 3.1 of 40 CFR, Part 266, Appendix 1X*	Metals Probe and Nozzle Rinse	Acetone portion dried and weighed; nitric acid portion sealed and kept at ambient temperature.	Digested probe and nozzle riases combined with digested filters prior to analysis.
		Metals Filter or Thimble	Desiccated at room temperature to constant weight.	For ESP inlet, a 0.100 g subsampte was digested and analyzed. For the stack, entire filter was digested and analyzed.
		Metals H, Q, /HNQ, Impingers	Scaled and kept at ambient temperature.	Analyzed for all target metals.
		Metals KMnQ4/H4 SQ4 Impingers	Sealed and kept at ambient temperature.	Analyzed for mercury only.
	Radian Method for Anions	Anions Train: Filter Na CQ,/NaHCQ,/H, Q, Impingers Probe and Nozzle Rinse	Scaled and kept at ambient temperature.	Analyzed for chloride, fluoride, and sulfate.
	SW 0030	VOST	Cooled to & C.	Analyzed for volatile organic compounds.
	SW 0010	MMS - Fitter or Thimble MMS - XAD Resin MMS - PNR/Condensate	Cooled to & C. Cooled to & C. Cooled to & C.	Analyzed for semivolatile organic compounds.
	SW 0011	Formaldehyde Impingers	Rinsed with MeCl; cooled to 4 C.	Analyzed for aldchydes.
	Specified in Section 3.2 of 40 CFR, Part 266, Appendix IX* (Method used for OFA/LNB test)	Chromium(VI) Impingers	pH >8.5 with KOH; cooled to & C.	Analyzed for chromium(VI).
Water Samples				
Bottom Ash Water	Decanted & Filtered	Metals	Nitric acid to pH <2; kept in sealed container.	Solids were allowed to settle before decanting the water phase.
		CI, F	Kept in scaled container.	
		Total P	Sulfuric acid to pH <2; kept in a scaled container.	

Table A-1 Sample Collection, Preservation, and Handling Techniques for Site Appendix A: Sample Collection, Preparation, and Analysis

ntinued)
ల్రి
A-1
able

Stream	Collection Method	Fraction Description	Sample Handling and Preservation	Consurents
Recycle Pond Water	Grab	Mctals	Nitric acid to pH <2; kept in a sealed container.	One grab sample during cach test run.
		CI, F	Kept in scaled container.	
		Total P	Sulfuric acid to $pH < 2$: kept in a sealed container.	
Solid Samples				
Coal	Grab/Composite	Ultimate, Proximate Mercury Metals CI, F, P	Placed in sealed container in the field; air dried, ground prior to analysis.	Multiple grab samples collected during the test period were composited directly into buckets, riffled to reduce sample size.
Bottom Ash Solids	Grab/Composite	Metals Cl, F Total P Total S	Placed in sealed container in the field; dried at 105 C and ground prior to analysis.	Multiple grab samples taken during bottom ash sluicing were composited in a large plastic container. Solids were allowed to settle and the water was decanted.
ESP Hopper Ash	Grab/Composite	Metals CI, F Total P Total S	Kept in sealed container, submitted "as collected."	Grab samples taken from individual hoppers. Composite samples analyzed for OFA test. Samples collected during OFA/LNB test were archived.

• 40 CFR, Part 266, Appendix IX, Section 3.1, "Methodology for the Determination of Metals Emissions from Hazardous Waste Incineration and Similar Combustion Processes," 1991.

* 40 CFR, Part 266, Appendix IX, Section 3.2, "Determination of Hexavalent Chromium Emissions from Stationary Sources," 1991.

Table A-2Preparation Procedures and Chemical Analysis MethodsApplied to Coal at Site 16

Component	Method Reference	Coal
Ultimate Analysis of Coal		
Ash	ASTM D3174	X,O
Carbon	ASTM D3178	X,O
Hydrogen	ASTM D3178	X,O
Nitrogen	ASTM D3179	X,O
Sulfur	ASTM D4239	X,O
Heating Value	ASTM D2015	<u>X,O</u>
Proximate Analysis of Coal		
Moisture	ASTM D3173	X,O
Ash	ASTM D3174	X,O
Volatiles	ASTM D3175	X,O
Fixed Carbon	Calculated	X,O
Target Elements by INAA	······	<u></u>
Preparation - None		
Analysis by INAA		
Arsenic	Karr, Chapters 12 and 46	X,O
Barium	Karr, Chapters 12 and 46	X,O
Cadmium	Karr, Chapters 12 and 46	X,O
Chlorine	Karr, Chapters 12 and 46	<u>X,O</u>
Chromium	Karr, Chapters 12 and 46	X,O
Cobalt	Karr, Chapters 12 and 46	<u>X,O</u>
Copper	Karr, Chapters 12 and 46	X,O
Manganese	Karr, Chapters 12 and 46	<u>X,O</u>
Mercury	Karr, Chapters 12 and 46	X,O
Molybdenum	Karr, Chapters 12 and 46	X,O
Nickel	Karr, Chapters 12 and 46	X,O
Selenium	Karr, Chapters 12 and 46	X,O
Vanadium	Karr, Chapters 12 and 46	X,O

Table A-2 (Continued)

Component	Method Reference	Coal
Chlorine and Fluorine in Coal		
Preparation		
Oxygen Bomb Digestion	ASTM D2361/ASTM D3761	X.O
Analysis by Potentiometric Titration	······································	
Chloride	SM 4500	X,O
Analysis by Ion Selective Electrode		····
Fluoride	ASTM D3761	X,O
Total Phosphorus in Coal		
Preparation		· · · · · · · · · · · · · · · · · · ·
Ashing and Acid Digestion	ASTM D2795	X
Spectrophotometric Analysis	······································	
Total Phosphorus	ASTM D2795	<u>X</u>
Be, Pb, P in Coal		
Preparation		
Ashing at 500° C/Acid Digestion	EPA 340.2	<u>X,O</u>
Analysis by ICP-AES		
Beryllium	SW 6010	X,O
Lead	SW 6010	0
Phosphorus	SW 6010	O
Analysis by GFAAS		
Lead	SW 7421	<u> </u>
Arsenic and Selenium in Coal		
Preparation		
Oxygen Bomb Combustion/Acid Digestion	ASTM D3684	X
Analysis by GFAAS	· · · · · · · · · · · · · · · · · · ·	
Arsenic	SW 7060	X
Selenium	SW 7740	X

Table A-2 (Continued)

.

1

Component	Method Reference	Coal
Mercury in Coal		
Preparation		
Double Gold Amalgamation	Karr, Chapter 14	X,O
Analysis by CVAAS		
Mercury	Karr, Chapter 14	X,O
Additional Inorganic Analytes in Coal Preparation - None		
Analysis by INAA	Karr Chapters 12 and 46	X O
Antimony	Karr, Chapters 12 and 46	<u>X,O</u> X,O
Iron	Karr, Chapters 12 and 46	X,O
Sodium	Karr, Chapters 12 and 46	X,O
Titanium	Karr, Chapters 12 and 46	X,O

Karr, C. Jr., (ed)., "Analytical Methods for Coal and Coal Products."

SW is EPA SW-846, "Test Methods for Evaluating Solid Waste".

SM is "Standard Methods for the Examination of Water and Wastewater," 16th Edition. ASTM is American Society for Testing and Materials.

X = Procedure performed on samples from OFA test.

O = Procedure performed on samples from OFA/LNB test.

Component	Method Reference	Ashes	Water Samples	Flue Gas Solids	Metals HNO,/H ₂ O, Impingers	Metals KMnQ4/H ₅ SQ4 Impingers	Anions Impingers
Target Elements by ICP-AES							
Preparation							
Water Digestion	SW 3005		×		X,0		
MW Digestion for Filters	CEM-F			X,0			
MW Digestion for Solids	CEM-FA	Х,О					
Analysis by ICP-AES							
Arsenic	SW 6010	0		0	0		
Barium	SW 6010	X,O	×	X,0	X,0		
Beryllium	SW 6010	X,O	x	X,0	X,0		
Cadmium	SW 6010	X,0	×	X,0	X,0		
Chromium	SW 6010	X,0	x	X,0	X,0		
Cobalt	SW 6010	X,0	×	X,0	X,O		
Copper	SW 6010	Х,О	x	X,0	X,0		
Manganese	SW 6010	X,0	×	X,0	X,0		
Molybdenum	SW 6010	Х,О	x	X,0	X,O		
Nickel	SW 6010	X,0	х	X,0	X,0		
Phosphorus	SW 6010	Х,О		X,0	0		×
Selenium	SW 6010	0		0	0		
Vanadium	SW 6010	X,0	X	Х,О	X,O		

Table A-3

Preparation Procedures and Chemical Analysis Methods for Inorganic Chemical Components in Ashes, Water Samples, and Flue Gas at Site 16

(Continued)	
Table A-3	

	Method	Achee	Water	Flue Gas Solids	Metals HNO3/H, O3 Imningers	Metals KMnQ,/H ₂ SQ, Imvingers	Anions Impingers
CORIDONEUL		CHEL	condumo.		0		-
Target Elements by GFAAS							
Preparation							
Water Digestion	CLP 3/90D-5		x		X,0		
MW Digestion for Filters	CEM-F			Х,О			
MW Digestion for Solids	CEM-FA	Х,О					
Analysis by GFAAS							
Arsenic	SW 7060	X,0		X,0	х,о		
Cadmium	SW 7131	X,0	x	Х,О	X,0		
Chromium	SW 7191				0		
Nickel	EPA 249.2	0	-	0	0		
Lead	SW 7421	Х,О	×	X,0	X,O		
Selenium	SW 7740	0		0	0		
Target Elements by HGAAS							-
Preparation							
MW Digestion for Filters	CEM-F			×			
MW Digestion for Solids	CEM-FA	×					
Analysis by HGAAS							
Arsenic	SW 7061		×				
Selenium	SW 7741	x	×	×	×		

.

1

Comment	Method Reference	Ashes	Water Samples	Flue Gas Solids	Metals HNQ/H, Q, Impingers	Metals KMnQ4/H5O4 Impingers	Anions Impingers
Mercury by CVAAS							
Preparation							
MW Digestion for Filters	CEM-F			X,0			
MW Digestion for Solids	CEM-FA	X,0					
Analysis by CVAAS							
Mercury	SW 7470	X,0	×	X,0	X,O	X,0	
Elements by ICP-MS in							
Size-Fractionated Particulate							
MW Digestion for Filters	CEM-F			0			
MW Digestion for Solids	CEM-FA			0			
Analysis by ICP-MS							
Antimony	SW 6020			0			
Arsenic	SW 6020			0			
Barium	SW 6020			0			
Beryllium	SW 6020		-	0			
Cadmium	SW 6020			0			
Chromium	SW 6020			0			
Cobalt	SW 6020			0			
Copper	SW 6020			0			
Lead	SW 6020			0			
Manganese	SW 6020			0			
Mercury	SW 6020			0			
Molybdenum	SW 6020			0			

Table A-3 (Continued)

\sim
P
e e e
2
0
- 7 \
<u> </u>
E
0
-3 (0
A-3 ((
: A-3 (C
le A-3 (C
ble A-3 (C
able A-3 (C

	Method		Water	Flue Gas	Metals HNQ,/H,Q	Metals KMnO4/H ₂ SO4	Anions
Component	Reference	Ashes	Samples	Solids	Impingers	Impingers	Impingers
Nickel	SW 6020			0			
Selenium	SW 6020			0			
Vanadium	SW 6020			0			
Acid-Forming Anions							
Preparation							
Aqueous Extraction of Solids (CI,F)	Radian	X,0		X,0			
Sodium Hydroxide Fusion (F)	McQuaker and Gurney	X,0					
Chloride by IC	EPA 300.0	X,0	×	Х,О			X,0
Chloride by Pot. Titr.	SM 407C	X,0					
Fluoride by ISE	EPA 340.2	X,0	×	X,0		2	X,0
Total P Spectrophotometry	EPA 365.3		x				
Elements by X-Ray Fluoresce	nce						
Aluminum	ASTM D4326	×					
Calcium	ASTM D4326	×					

Sodium Titanium

Silicon

××

 $\times \times \times \times$

ASTM D4326 ASTM D4326 ASTM D4326 ASTM D4326 ASTM D4326 ASTM D4326 ASTM D4326

> Magnesium Potassium

Iron

Component	Method Reference	Ashes	Water Samples	Flue Gas Solids	Metals HNQ/H,Q Impingers	Metals KMnQ,/H,SQ, Impingers	Anions Impingers
Total Sulfur Analysis							
Sulfur by Leco	ASTM D4239	×					
Additional Inorganic Analytes by ICP-AES							
Preparation							
Water Digestion	SW 3005		х		X,0		
MW Digestion for Filters	CEM-F			X,O			
MW Digestion for Solids	CEM-FA	X,0					
Analysis by ICP-AES							
Aluminum	SW 6010	X,0	x	X,O	X,O		
Antimony	SW 6010	X,O	x	Х,О	X,O		
Iron	SW 6010	X,O	X	X,0	X,O		
Sodium	SW 6010	X,O	х	X,O	X,O		
Titanium	SW 6010	X,0	x	X,0	X,0		
ASTM is American Society for Testing a EPA is EPA Methods for Chemical Anal SW is EPA SW-846, "Test Methods for E SW 6020 is a proposed method for metal CEM-FA CEM Corporation, Matthews, NG CEM-F CEM Corporation, NG CEM CORPORATION, NG CEM F CEM CORPORATION, NG CEM CORPORATION, NG CEM F CEM CORPORATION, NG CEM CORPOR	nd Materials. ysis of Water and Wastes, I valuating Solid Waste", 3rd s analysis by inductively cou VC Procedure for microwave Procedure for microwave m SOW 3/90. microwave m SOW 3/90. attion of Total Fluoride attion of Water and Wastew om OFA/LNB test.	1983. ed. ppled plasma - m e digestion of co digestion of glas igestion of glas re in Soil and Veg ater", 16th Editi	ass spectrometry al fly ash. s or quariz filters ctation Using An	(ICP-MS). 1 Alkali Fusion-Sc	lective Ion Electrode T	cchnique", <u>Analytical Ch</u> c	mistry, Vol. 49,

Table A-3 (Continued)

Table A-4Preparation Procedures and Chemical Analysis Methods Used toMeasure Organic Compounds at Site 16

Component	Method Reference	Flue Gas	Ashes	Water Samples
Volatile Organic Compounds				
Sample Collection				
VOST	SW 00 <u>3</u> 0	X,O		
Analysis by GC-MS				
Benzene	SW 8240	X,O		
Toluene	SW 8240	<u>X,</u> O		
Formaldehyde	<u> </u>			
Sample Collection		<u>.</u>		······································
DNPH Impinger	SW 0011	<u>X,O</u>		<u> </u>
Analysis by HPLC	·			
Formaldehyde	TO5	X,O		
Polycyclic Organic Matter			<u></u>	<u></u>
Sample Collection				
MM5	SW 0010	<u>X,O</u>		
Preparation				
Soxhlet Extraction	SW 3540	X,O	X	X
Analysis by GC-MS				
Semivolatile organics, including PAHs and other POM	SW 8270	X	x	X
Analysis by High Resolution (GC-MS			
PAHs		0		

DNPH is 2,4-Dinitrophenylhydrazine.

SW is EPA SW-846, "Test Methods for Evaluating Solid Waste."

TO5 is "EPA Compendium of Methods for the Determination of Toxic Organic Compounds in Ambient Air," EPA 600/4/84/041.

X = Procedure performed on samples from OFA test.

O = Procedure performed on samples from OFA/LNB test.

APPENDIX B: ANALYTICAL DATA USED IN CALCULATIONS

.

Key to Data Flags

Flag	Description	
@	Concentration is less than five times the detection limit.	
В	Detected in blank.	
Ε	Estimated analyte result exceeds calibration range.	
R	Detected in blank, corrected in sample result.	
<	Less than the detection limit.	
+	Blank level exceeds 50% of uncorrected result.	

30-Oct-93

Table B-1: Site 16 OFA Data USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
1 - Chloronaphthalene	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
1 – Chloronaphthalene	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
1 - Chloronarohthalane	collected fiv ash	GCMS(8270)	6/8n	066.0 >		< 0.990	< 0.970	< 1.00	:
1 - Chloronarohthalane	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
1 - Chlorona ohthalane	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
1 – Chloronaphthalene	Isluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
1 - Chloronaohthaiene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1 – Chloronaohthalene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1 - Nachthylamine	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	066.0 >	066:0 >	
1 - Nachthylamine	bottom ash sluice water	GCMS(8270)	սց/Լ	< 10.0		< 11.0	< 11.0	< 11.0	
1 – Nachthylamine	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
t - Nachthylamine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
1 – Naphthylamine	high dust gas, vepor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503		-	
1 - Naphthylamine	Isluice water supply	GCMS(8270)	ոց/Լ	< 10.0		< 10.0	< 11.0	< 10.0	
1 - Nachthylamine	stack das, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1 – Naphthylamine	stack das, vepor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1 1 – Nichloroathana	high dust gas. 20-L VOST	GCMS	ரு/Nm3	< 0.000525	< 0.000543	< 0.000559			
t t – Dichloroethane	stack gas, 20-L VOST	GCMS	աց/Nm3	< 0.000580	< 0.000569	< 0.000542	I		1
1.1 - Dichtoroethene	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
1 1 – Dichloroethene	stack gas, 20- L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
1 t 1 - Trichloroethane	high dust gas, 20-L VOST	GCMS	EmN/gm	0.000577 @	0.00103 @	0.00447			
1.1.1 – Trichloroethane	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
1.1.2 - Trichloroethane	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
1.1.2 – Trichloroethane	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
1.1.2.2 – Tetrachloroethane	high dust gas, 20 - L VOST	GCMS	EmN/gm	< 0.000525	< 0.000543	< 0.000559			
1.1.2.2 - Tetrachloroethane	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542	-		
1.2 - Dichloropropane	high dust gas, 20-1. VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
1.2 – Dichloropropane	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
1.2 - Diphenythydrazine	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
1.2-Diphenvlhvdrazine	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
1.2 – Diphenvlhvdrazine	collected fly ash	GCMS(8270)	5/6n	< 0.990		< 0.990	0.970 >	< 1.00	
1.2 - Diphenvihvdrazine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			
1.2 - Diphenvlhvdrazine	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
1.2 - Diphenvthydrazine	sluice water supply	GCMS(8270)	1/6n	< 10.0		< 10.0	< 11.0	< 10.0	
1.2-Diphenvihvdrazine	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1.2 - Diphenylhydrazine	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1,2,4 - Trichlorobenzene	bottom ash	GCMS(8270)	0/8	< 0.980		< 0.980	066.0 >	066.0 >	
1,2,4 – Trichlorobenzene	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
1.2.4 – Trichlorobenzene	collected fly ash	GCMS(8270)	0/6n	< 0.990		< 0.990	< 0.970 >	< 1.00	

g
Ĩ
ğ
Ŷ
ĝ

B-4

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
1.2.4 - Trichlorobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
1,2,4 – Trichlorobenzene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
1,2,4 - Trichlorobenzene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	T
1,2,4 - Trichlorobenzene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1,2,4 – Trichlorobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
1 4 - Dichtorobenzene	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
1,4 – Dichlorobenzene	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
1,4 – Dichlorobenzene	collected fly ash	GCMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	< 1.00	
1,4 - Dichlorobenzene	high dust gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
1.4 - Dichlorobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
1,4 – Dichłorobenzene	high dust gas. vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
1.4 - Dichlorobenzene	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
1,4 – Dichlorobenzene	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
1,4 - Dichlorobenzene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		<u> </u>	
1.4 - Dichtorobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 Butanone	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.00525	< 0.00543	< 0.00559			
2 - Butanone	stack gas, 20 - L VOST	GCMS	mg/Nm3	< 0.00560	< 0.00569	< 0.00542			
2 – Chloronaphthalane	bottom ash	GCMS(8270)	B/Bn	< 0.980		< 0.980	< 0.990	< 0.990	
2 - Chloronaphthaiene	bottom ash shrice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
2 - Chloronaphthalene	collected fly ash	GCMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	< 1.00	
2 - Chioronaphthalene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2 - Chioronaphthalane	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
2 - Chloronaphthalene	sluice water supply	GCMS(8270)	VBn	< 10.0		< 10.0	< 11.0	< 10.0	
2 - Chioronaphthalene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 - Chloronaphthalene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 - Methyinaphthalene	bottom ash	GCMS(8270)	6/Bn	0.995 @		< 0.980	< 0.990	< 0.990	
2 - Methyinaphthalene	bottom ash sluice water	GCMS(8270)	1/8n	< 10.0		< 11.0	< 11.0	< 11.0	
2 - Methylnaphthalene	collected fly ash	GCMS(8270)	0/8n	< 0.990		< 0.990	< 0.970	< 1.00	
2 - Methylnaphthelene	high dust gas, solid phase	GCMS(8270)	EmN/Bm	< 0.00997	< 0.00837	< 0.00863			
2 - Methylnaphthalene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			-
2 - Methyinaphthalene	sluice water supply	GCMS(8270)	1/6n	< 10.0		< 10.0	< 11.0	< 10.0	:
2 - Methylnaphthalene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		-	
2 - Methyinaphthalene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 - Methylphenol(o - cresol)	bottom ash	3CMS(8270)	8/6n	< 0.980		< 0.980	< 0.990	< 0.990	
2 – Methylphenol(o – cresol)	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
2 - Methylphenol(o - cresol)	collected fly ash	3CMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	< 1.00	
2 - Methylphenol(o - cresol)	high dust gas, solid phase	3CMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2 - Methylphenol(o - cresol)	high dust gas, vapor phase	3CMS(8270)	mg/Nm3	< 0.00581	< 0.00489	< 0.00503			
2 - Methytphenot(o - cresol)	sluice water supply	3CMS(8270)	Von	< 10.0		< 10.0	< 11.0	< 10.0	

Appendix B: Analytical Data Used in Calculations

2

30-Oct-93

Table B-1: Site 16 OFA Data USED in Calculations

ო

2- Methylphenol(o - cresol) stack gas, solid phase 2- Naphthylamine bottom ash sluice water 2- Unitrophenol bottom ash sluice water 2 - Unintrophenol bottom ash sluice	sse GCMS(8270) usse GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) Phase GCMS(8270) r GCMS(8270) r GCMS(8270) r GCMS(8270) r GCMS(8270) nsse GCMS(8270) nsse GCMS(8270) nsse GCMS(8270) nsse GCMS(8270) nsse GCMS(8270) nsse GCMS(8270) rater GCMS(8270)	mg/Nm3 mg/Nm3 ug/g ug/L mg/Nm3 mg/Nm3 mg/Nm3 ug/L ug/L ug/L ug/L ug/L ug/g ug/g ug/g	 0.00530 0.00530 	< 0.00555	 0.00534 0.00534 0.00534 0.980 	060 0		
P-Methylphenol(o stack gas. solid phase P-Methylphenol(o stack gas. vapor phase P-Naphthylarnine bottom ash P-Dinitrophenol bottom ash	see GCMS(8270) asse GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) phase GCMS(8270) r GCMS(8270) vater GCMS(8270) r phase GCMS(8270) GCMS(8270) vater GCMS(8270) ase GCMS(8270) vater GCMS(8270) ase GCMS(8270) vater GCMS(8270)	mg/km3 mg/km3 ug/g ug/L mg/km3 mg/km3 mg/km3 mg/km3 ug/L ug/L ug/L ug/L ug/L ug/g ug/L	 < 0.00530 < 0.00530 < 0.0800 < 0.0801 < 0.0907 < 0.0930 < 0.00530 < 0.00530 	 < 0.00555 < 0.00555 	 < 0.00534 < 0.00534 < 0.980 	066 U >		
2-Methylphenol(o -cresol)stack gas. vapor phase2-Naphthylarninebottom ash2-Naphthylarninebottom ash2-Naphthylarninebottom ash2-Naphthylarninebottom ash2-Naphthylarninehigh dust gas. solid phase2-Naphthylarninehigh dust gas. vapor phase2-Naphthylarninebottom ash2-Naphthylarninehigh dust gas. solid phase2-Naphthylarninebottom ash2-Naphthylarninebottom ash2-Dinitrophenolbottom ash2-Dinitrophenol<	Iase GCMS(8270) water GCMS(8270) water GCMS(8270) phase GCMS(8270) pr GCMS(8270) pr GCMS(8270) pr GCMS(8270) sse GCMS(8270) ase GCMS(8270) ase GCMS(8270) ase GCMS(8270) uster GCMS(8270) uster GCMS(8270) uster GCMS(8270) uster GCMS(8270) uster GCMS(8270) valer GCMS(8270)	mg/Nm3 ug/2 ug/L ug/L mg/Nm3 mg/Nm3 mg/Nm3 ug/1 ug/1 ug/1 ug/1 ug/1	 < 0.00530 < 0.0980 < 0.980 < 0.0997 < 0.00581 < 0.00530 < 0.00530 < 0.00530 	< 0.00555	< 0.00534 < 0.980	060 V >		
2 - Naphthylamine bottom ash 2 - Naphthylamine bottom ash 2 - Naphthylamine collected fly ash 2 - Naphthylamine bottom ash sluice water 2 - Naphthylamine bigh dust gas, solid phase 2 - Naphthylamine bigh dust gas, solid phase 2 - Naphthylamine bijgh dust gas, solid phase 2 - Naphthylamine bijgh dust gas, solid phase 2 - Naphthylamine bijgh dust gas, solid phase 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol bottom ash sluice wa	GCMS(8270) water GCMS(8270) water GCMS(8270) phase GCMS(8270) prase GCMS(8270) prase GCMS(8270) sse GCMS(8270) ase GCMS(8270) valer GCMS(8270) ase GCMS(8270) valer GCMS(8270) arease GCMS(8270) arease GCMS(8270) arease GCMS(8270)	ug/g ug/L mg/Nm3 mg/Nm3 mg/Nm3 mg/L mg/L ug/L ug/L ug/g ug/g ug/g	 0.980 10.0 10.0 0.0997 0.00581 10.0 10.0530 0.00530 		< 0.980	0 9 9 9 0	100001	
2 - Naphthylamine bottom ash sluice water 2 - Naphthylamine collected fly ash 2 - Naphthylamine collected fly ash 2 - Naphthylamine inigh dust gas, solid phase 2 - Naphthylamine isluice water supply 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol bottom ash sluice <td>vater GCMS(8270) vater GCMS(8270) phase GCMS(8270) r phase GCMS(8270) GCMS(8270) ase GCMS(8270) vater GCMS(8270)</td> <td>ug/L mg/Nm3 mg/Nm3 mg/Nm3 mg/Nm3 mg/L ug/L ug/L ug/L ug/g</td> <td> 10.0 10.0 0.990 0.00937 0.00530 0.00530 </td> <td></td> <td></td> <td></td> <td>C 0.330</td> <td></td>	vater GCMS(8270) vater GCMS(8270) phase GCMS(8270) r phase GCMS(8270) GCMS(8270) ase GCMS(8270) vater GCMS(8270)	ug/L mg/Nm3 mg/Nm3 mg/Nm3 mg/Nm3 mg/L ug/L ug/L ug/L ug/g	 10.0 10.0 0.990 0.00937 0.00530 0.00530 				C 0.330	
2 Naphthylarnine collected fly ash 2 Naphthylarnine high dust gas, solid phase 2 Naphthylarnine high dust gas, vapor phase 2 Naphthylarnine biuice water supply 2 Naphthylarnine bottom ash bluice water 2 - Dinitrophenol bit dust gas, solid phase 2 - Dinitrophenol bottom ash bluice water 2 - Dinitrophenol bottom ash bluice 2 - Dinitrophenol bottom ash bluice 2 - Dinitrophenol bottom a	GCMS(8270) phase GCMS(8270) rr phase GCMS(8270) ase GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) vr phase GCMS(8270) or phase GCMS(8270)	ug/g mg/km3 mg/km3 mg/km3 mg/km3 mg/km3 ug/t ug/t ug/g	 < 0.990 < 0.00997 < 0.00581 < 10.0 < 0.00530 < 0.00530 		< 11.0	< 11.0	< 11.0	
Construction Nigh dust gas, solid phase 2 - Naphthylamine high dust gas, solid phase 2 - Naphthylamine stack gas, solid phase 2 - Naphthylamine stack gas, solid phase 2 - Naphthylamine stack gas, solid phase 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol bottom a	phase GCMS(8270) rr phase GCMS(8270) ase GCMS(8270) valer GCMS(8270) valer GCMS(8270) vale GCMS(8270) vale GCMS(8270) or phase GCMS(8270)	mg/Nm3 mg/Nm3 ug/L mg/Nm3 mg/Nm3 ug/L ug/L ug/L	 < 0.00997 < 0.00581 < 10.0 < 0.00530 < 0.00530 		< 0.990	< 0.970	< 1.00	
2. Naphrtytamine high dust gas. vapor phase 2. Naphrtytamine stuice water supply 2. Vapot phase stack gas. vapor phase 2 Dintrophenol bottom ash sluice water 2 Dintrophenol high dust gas. solid phase 2 Dintrophenol bottom ash sluice water 2 Trichorophenol bottom ash sluice water 2 Trichorophenol bottom ash sluice water 2 Trichorophenol bottom ash sluice w	r phase GCMS(8270) ase GCMS(8270) ase GCMS(8270) ase GCMS(8270) atter GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) r phase GCMS(8270) or phase GCMS(8270)	mg/Nm3 ug/L mg/Nm3 mg/Nm3 ug/L ug/L ug/g	 < 0.00581 < 10.0 < 0.00530 < 0.00530 	< 0.00837	< 0.00863			
2 Naphthylamine Iujce water supply 2 Naphthylamine stack gas, solid phase 2 Naphthylamine stack gas, solid phase 2 Naphthylamine stack gas, solid phase 2 Dinitrophenol bottom ash 2 Dinitrophenol stack gas, solid phase 2 Dinitrophenol bottom ash 2 Dinitrophenol bo	ase GCMS(8270) ase GCMS(8270) Inse GCMS(8270) valer GCMS(8270) vrphase GCMS(8270) vrphase GCMS(8270)	ug/L mg/Nm3 mg/Nm3 ug/g ug/L	 < 10.0 < 0.00530 < 0.00530 	< 0.00488	< 0.00503		4	
2 Naphtiylarnine stack gas, vapor phase 2 - Naphtiylarnine stack gas, vapor phase 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol high dust gas, solid phase 2 - Dinitrophenol stack gas, solid phase 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol stack gas, solid phase 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol bottom ash sluice water 2 - Dinitrophenol bottom ash sluice 2 - Dinitrophe	356 GCMS(8270) IB56 GCMS(8270) valer GCMS(8270) vr phase GCMS(8270) vr phase GCMS(8270)	mg/Nm3 mg/Nm3 ug/g ug/L	< 0.00530		< 10.0	× 11.0	< 10.0	
2 - Trichtorphenol stack gas, vapor phase 2 - Dintrophenol bottom ash sluice water 2 - Dintrophenol high dust gas, solid phase 2 - Dintrophenol high dust gas, solid phase 2 - Dintrophenol stack gas, vapor phase 2 - Dintrophenol stack gas, solid phase 2 - Dintrophenol stack gas, solid phase 2 - Dintrophenol stack gas, solid phase 2 - Dintrophenol bottom ash sluice water 2 - Trichorophenol <t< td=""><td>Insee GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) I phase GCMS(8270) r phase GCMS(8270) r phase GCMS(8270) r phase GCMS(8270)</td><td>mg/Nm3 ug/g ug/L</td><td>< 0.00530</td><td>< 0.00555</td><td>< 0.00534</td><td></td><td></td><td></td></t<>	Insee GCMS(8270) vater GCMS(8270) vater GCMS(8270) vater GCMS(8270) I phase GCMS(8270) r phase GCMS(8270) r phase GCMS(8270) r phase GCMS(8270)	mg/Nm3 ug/g ug/L	< 0.00530	< 0.00555	< 0.00534			
2.4 - Dinitrophenol bottom ash 2.4 - Dinitrophenol bottom ash 2.4 - Dinitrophenol bottom ash 2.4 - Dinitrophenol high dust gas, solid phase 2.4 - Dinitrophenol stack gas, solid phase 2.4 - Dinitrophenol bottom ash 2.4 - Dinitrophenol stack gas, solid phase 2.4 - Dinitrophenol stack gas, solid phase 2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrophenol bottom ash sluice 2.4 - Dinitropheno	acms(8270) vater GCMS(8270) vater GCMS(8270) 0 OMS(8270) 1 phase 0 CMS(8270) or phase GCMS(8270) or phase GCMS(8270)	ug/g. ug/L ug/g		< 0.00555	< 0.00534			
2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrophenol high dust gas, solid phase 2.4 - Dinitrophenol high dust gas, vapor phase 2.4 - Dinitrophenol high dust gas, vapor phase 2.4 - Dinitrophenol high dust gas, vapor phase 2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrotoluene bottom ash sluice water 2.4 - Dinitrotoluene bottom ash sluice water 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol bottom ash sluice water <tr p=""> 2.4.5 - Trichl</tr>	valer GCMS(8270) GCMS(8270) I phase GCMS(8270) br phase GCMS(8270) Sr phase GCMS(8270)	ng/t ug/t	< 4.90		< 4.90	< 4.90	< 5.00	
2.4 - Dinitrophenol collected fly ash 2.4 - Dinitrophenol high dust gas, solid phase 2.4 - Dinitrophenol high dust gas, vapor phase 2.4 - Dinitrophenol atack gas, solid phase 2.4 - Dinitrophenol bottom ash 2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrotoluene bottom ash 2.4 - Dinitrotoluene bigh dust gas, vapor phase 2.4 - Dinitrotoluene stack gas, solid phase 2.4 - Dinitrotoluene bottom ash 2.4.5 - Trichlorophenol bottom ash 2.4.	I phase GCMS(8270) I phase GCMS(8270) or phase GCMS(8270) GCMS(8270)	6/6n	< 52.0		< 53.0	< 54.0	< 56.0	
2.4 - Dinitrophenol high dust gas, solid phase 2.4 - Dinitrophenol high dust gas, vapor phase 2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrophenol bottom ash sluice water 2.4 - Dinitrotoluene bigh dust gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol bot	I phase GCMS(8270) or phase GCMS(8270) GCMS(8270)		< 5.00		< 4.90	< 4.80	< 5.00	
2.4 Dinitrophenol high dust gas, vapor phase 2.4 Dinitrophenol stack gas, vapor phase 2.4 Dinitrophenol bottom ash stuce water 2.4 Dinitrotoluene bottom ash stuce water 2.4 Trichlorophenol bottom ash stuce water 2.4. 5 Trichlorophenol bottom ash stuce 2.4. 5 Trichlorophenol stack gas, vapor phase 2.4. 5 Trichlorophenol stack	or phase GCMS(8270) GCMS(8270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432			
2.4.5 - Trichtrophenol stack gas, solid phase 2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrophene bottom ash sluce water 2.4 - Dinitrotoluene bottom ash sluce water 2.4 - Dinitrotoluene bottom ash sluce water 2.4 - Dinitrotoluene bigh dust gas, vapor phase 2.4 - Dinitrotoluene bigh dust gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Trichlorophenol bottom ash sluice wetler 2.4.5 - Trichlorophenol stack gas, vapor phase 2.4.5 - Trichlorophenol	GCMS(8270)	CmN/gm	< 0.0290	< 0.0244	< 0.0251			:
2.4 Dinitrophenol stack gas, solid phase 2.4 Dinitrophenol stack gas, vapor phase 2.4 Dinitrophene bottom ash sluce water 2.4 Dinitrotoluene high dust gas, vapor phase 2.4 Dinitrotoluene high dust gas, vapor phase 2.4 Dinitrotoluene stack gas, vapor phase 2.4 Trichlorophenol bottom ash sluce water 2.4 Trichlorophenol high dust gas, vapor phase 2.4 Trichlorophenol bottom ash sluce water 2.4 Trichlorophenol high dust gas, vapor phase 2.4 Trichlorophenol stack gas, vapor phase		ng/L	< 52.0		< 52.0	< 53.0	< 52.0	
2.4 - Dinitrophenol stack gas, vapor phase 2.4 - Dinitrotoluene bottom ash sluice water 2.4 - Dinitrotoluene high duti gas, solid phase 2.4 - Dinitrotoluene high duti gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol high dust gas, solid phase 2.4.5 - Trichlorophenol stack gas, vapor phase <tr< td=""><td>ase GCMS(8270)</td><td>mg/Nm3</td><td>< 0.0265</td><td>< 0.0278</td><td>< 0.0267</td><td></td><td>1</td><td></td></tr<>	ase GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267		1	
2.4 Dimitrotoluene bottom ash sluree water 2.4 Dimitrotoluene bottom ash sluree water 2.4 Dimitrotoluene bottom ash sluree water 2.4 Dimitrotoluene high dust gas, solid phase 2.4 Dimitrotoluene high dust gas, solid phase 2.4 Dimitrotoluene high dust gas, vapor phase 2.4 Dimitrotoluene stack gas, vapor phase 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol high dust gas, vapor phase 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol high dust gas, solid phase 2.4.5 Trichlorophenol stack gas, solid phase 2.4.5 Trichlorophenol stack gas, vapor phase 2.4.5 Trichlorophenol	ase GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
2.4 Dimitrotoluene bottom ash sluice water 2.4 Dimitrotoluene high dust gas, solid phase 2.4 Dimitrotoluene high dust gas, vapor phase 2.4 Dimitrotoluene high dust gas, vapor phase 2.4 Dimitrotoluene high dust gas, vapor phase 2.4 Dimitrotoluene stack gas, vapor phase 2.4 Dimitrotoluene stack gas, vapor phase 2.4 Dimitrotoluene stack gas, vapor phase 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol bottom ash sluice weter 2.4.5 Trichlorophenol high dust gas, vapor phase 2.4.5 Trichlorophenol stack gas, solid phase 2.4.5 Trichlorophenol stack gas, solid phase 2.4.5 Trichlorophenol stack gas, solid phase 2.4.5 Trichlorophenol stack gas, vapor phase 2.4.5 Trichlorophenol stack gas, vapor phase 2.4.5 Trichlorophenol stack gas, vapor phase	GCMS(8270)	5/5n	< 0.980		< 0.980	< 0.990	< 0.990	
2.4 - Dinitrotoluene collected fly ash 2.4 - Dinitrotoluene high dust gas, solid phase 2.4 - Dinitrotoluene high dust gas, vapor phase 2.4 - Dinitrotoluene sluice water supply 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4.5 - Trichlorophenol bottom ash sluice weller 2.4.5 - Trichlorophenol bottom ash sluice weller 2.4.5 - Trichlorophenol bottom ash sluice weller 2.4.5 - Trichlorophenol high dust gas, solid phase 2.4.5 - Trichlorophenol bottom ash sluice weller 2.4.5 - Trichlorophenol high dust gas, solid phase 2.4.5 - Trichlorophenol stack gas, vapor phase	vater GCMS(8270)	1/6n	< 10.0		< 11.0	< 11.0	< 11.0	:
2.4 - Dinitrotoluene high dust gas, solid phase 2.4 - Dinitrotoluene high dust gas, vapor phase 2.4 - Dinitrotoluene sluice water supply 2.4 - Dinitrotoluene stack gas, vapor phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, vapor phase	GCMS(8270)	5/6n	< 0.990		< 0.990	< 0.970	< 1.00	
2.4.5. Trichtorobuene high dust gas, vapor phase 2.4. Dinitrotoluene sluice water supply 2.4. Dinitrotoluene stack gas, vapor phase 2.4.5. Trichtorophenol bottom ash sluice water 2.4.5. Trichtorophenol high dust gas, vapor phase 2.4.5. Trichtorophenol bottom ash sluice water 2.4.5. Trichtorophenol high dust gas, vapor phase 2.4.5. Trichtorophenol sluice water supply 2.4.5. Trichtorophenol sluice water supply 2.4.5. Trichtorophenol stack gas, vapor phase 2.4.5. Trichtorophenol stack gas, vapor phase 2.4.5. Trichtorophenol stack gas, vapor phase	Dhase GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2.4 - Dinitrotoluene suitce water supply 2.4 - Dinitrotoluene stack gas, solid phase 2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol bigh dust gas, vapor phase 2.4.5 - Trichlorophenol sluice water supply	or phase GCMS(8270)	mg/Nm3	< 0.00581	< 0.00489	< 0.00503			
2.4 - Dinitrotoluene stack gas, solid phase 2.4 - Dinitrotoluene stack gas, vapor phase 2.4.5 - Trichlorophenol bottom ash sluics weller 2.4.5 - Trichlorophenol high dust gas, solid phase 2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol stack gas, vapor phase 2.4.5 - Trichlorophenol stack gas, vapor phase	GCMS(8270)	ոց/Լ	< 10.0		< 10.0	< 11.0	< 10.0	
2.4.5 Dinitrotoluane stack gas, vapor phase 2.4.5 Trichlorophenol bottom ash sluice water 2.4.5 Trichlorophenol high dust gas, vapor phase 2.4.5 Trichlorophenol high dust gas, vapor phase 2.4.5 Trichlorophenol sluice water supply 2.4.5 Trichlorophenol sluice water supply 2.4.5 Trichlorophenol sluice water supply 2.4.5 Trichlorophenol stack gas, vapor phase 2.4.5 Trichlorophenol stack gas, vapor phase	ase GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			•
2.4.5 - Trichlorophenolbottom ash sluice water2.4.5 - Trichlorophenolbottom ash sluice water2.4.5 - Trichlorophenolcollected fly ash2.4.5 - Trichlorophenolhigh dust gas, solid phase2.4.5 - Trichlorophenolhigh dust gas, vapor phase2.4.5 - Trichlorophenolsluice water supply2.4.5 - Trichlorophenolsluice water supply2.4.5 - Trichlorophenolsluice water supply2.4.5 - Trichlorophenolsluice water supply2.4.5 - Trichlorophenolstack gas, solid phase2.4.5 - Trichlorophenolstack gas, solid phase2.4.5 - Trichlorophenolstack gas, solid phase	1858 GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2.4.5 - Trichlorophenol bottom ash sluice water 2.4.5 - Trichlorophenol collected fly ash 2.4.5 - Trichlorophenol high dust gas, solid phase 2.4.5 - Trichlorophenol high dust gas, vepor phase 2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, solid phase	GCMS(8270)	5/8n	< 0.980	· · · · · · · · · · · · · · · · · · ·	086.0 >	< 0.990	0.990	
2.4.5 - Trichlorophenol collected fly ash 2.4.5 - Trichlorophenol high dust gas, solid phast 2.4.5 - Trichlorophenol high dust gas, vepor phast 2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, vapor phase 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, solid phase	veter GCMS(8270)	ug/L	< 10.0		< 11.0	× 11.0	v # 0	
2.4.5 - Trichlorophenolhigh dust gas, solid phase2.4.5 - Trichlorophenolhigh dust gas, vepor phase2.4.5 - Trichlorophenolsluice water supply2.4.5 - Trichlorophenolstack gas, solid phase2.4.5 - Trichlorophenolstack gas, solid phase	GCMS(8270)	8/6n	< 0.990		< 0.990	< 0.970	× 1.00	
2,4,5 Trichlorophenol high dust gas, vapor phas 2,4,5 Trichlorophenol sluice water supply 2,4,5 Trichlorophenol stack gas, solid phase 2,4,5 Trichlorophenol stack gas, vapor phase 2,4,5 Trichlorophenol stack gas, vapor phase	I phase GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2.4.5 - Trichlorophenol sluice water supply 2.4.5 - Trichlorophenol stack gas, solid phase 2.4.5 - Trichlorophenol stack gas, vapor phase 2.4.6 - Trichlorophenol bottom ash	or phase GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
2,4,5 – Trichtorophenol stack gas, solid phase 2,4,5 – Trichtorophenol stack gas, vapor phase 2,4,6 – Trichtorophenol bottom ash	GCMS(8270)	ug/L	< 10.0	1	< 10.0	< 11.0	< 10.0	
2,4,5 - Trichforophenol stack gas, vapor phase	ase GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2,4,6 - Trichtorophenol bottom ash	ase GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
o a e_Trichloronhanol bottom ash siuice waler	vater GCMS(8270)	ոց/ե	< 10.0		< 11.0	< 11.0	< 11.0	
2 4 6 Trichloronhenol collected fiv ash	GCMS(8270)	8/6n	< 0.990		< 0.990	< 0.970	< 1.00	
2.4.0 Trichtorothanol high dust gas, solid phase	I phase GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
	or ohee GCMS(8270)	ma/Nm3	< 0.00581	< 0.00488	< 0.00503			

Appendix B: Analytical Data Used in Calculations

	•		•
1		ł	
٤	1		Ì
	•	r	•
•	ž		í
1	ŝ	ï	i
k	2		
ľ			

Substance	Stream	Method	NOM	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
				-					
2 ▲ R – Trichloronhanol	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
2.4.6 Trichtorohanol	atack das solid phase	GCMS(8270)	CmN/gm	< 0.00530	< 0.00555	< 0.00534			
		GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
2.4.0 - Intrilotophenoi	bothom ash	GCMS(8270)	ua/a	0.980		< 0.980	< 0.990	< 0.990	
	bottom ash shiring water	GCMS(8270)	uo/L	< 10.0		< 11.0	< 11.0	< 11.0	
3 - Methylcholantherie	collected fly ash	GCMS(8270)	0/07	066.0 >	• 	< 0.990	< 0.970	< 1.00	
3 - Metnyicholanitmetre		00000000000000000000000000000000000000	ma/Nm3	< 0.0097	< 0.00837	< 0.00863			
3 - Methylcholanthrene	nign dust gas, sonid pinase	CUS(B270)	EmN/om	< 0.00581	< 0.00488	< 0.00503			
3 - Methylcholanthrene		(0.00) 0100	1,011	0.01 /		< 10.0	< 11.0	< 10.0	
3 - Methylcholanthrene	sluice water supply	GCM5(92/U)	ug/L	< 0.00530	< 0.00555	< 0.00534	,		
3 - Methylcholanthrene	stack gas, solid priase	001000000	Emilian (Sum	< 0.00530	< 0.00555	< 0.00534			
3 - Methylcholanthrene	stack gas, vapus pnase	GCM3(8270)	cintri Anno 10	< 2.00		< 2.00	< 2.00	< 2.00	
3.3 - Dichlorobenzidine		0.000000000000000000000000000000000000		210		< 21.0	< 22.0	< 22.0	
3,3 - Dichlorobenzidine		COMS(B270)		00 < >		< 2.00	< 1.90	< 2.00	
3,3'-Dichlorobenzidine		0.0000000000000000000000000000000000000		0100	< 0.0167	< 0.0173			
3,3' – Dichlorobenzidine	high dust gas, solid phase	GCMS(82/0)	CEN/Bu	0.0139	0.000	10101		<u>↓</u> — 	
3,3' – Dichlorobenzidine	high dust gas, vapor phase	GCMS(82/0)	mg/Nm3	0110.0 >					
3.3' - Dichtorobenzidine	sluice water supply	GCMS(8270)	ug/L	< 21.0		< 21.0	0.12 >	0.12 >	
3.3' - Dichtorobenzidine	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0106	< 0.0111	< 0.0107			•
a a' - Nichlorohenzidine	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0106	< 0.0111	< 0.0107			
4 - Aminohinhanul	bottom ash	GCMS(8270)	0/0n	< 0.980		< 0.980	< 0.990	< 0.990	
4 - Aminobiohenvi	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
4 - Aminchichenvil	collected fiv ash	GCMS(8270)	6/Bn	< 0.990		066.0 >	< 0.970	< 1.00	
4 Aminobihihanul	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
4 - Aminobiohenut	hinh dust das. vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.0048B	< 0.00503			
4 ~ Aminobinhanvi	stuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
A - Aminchichenul	stack das. solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0 00555	< 0.00534			
4 - Aminobiohenvi	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 - Bromochenvl chervlether	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	066.0 >	
4 - Bromonhenvi phenviether	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	× 11.0	< 11.0	I
4 - Bromochenvi ohenvi ether	collected fly ash	GCMS(8270)	n9/8	066.0 >		< 0.990	< 0.970	< 1.00	
4 - Bromonhanvi nharvi ether	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863	i		
A - Bromonhanvi nhanvi ether	high dust gas. vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
A - Bromonhandi nharvi athar	sinice water subbly	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
4 - Bromonhanvi nhanvi athar	stack das. solid phase	GCMS(8270)	CmN/gm	< 0.00530	< 0.00555	< 0.00534			
4 - Bromoshanvi sharvi athar	stack das, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 - Chlorochanyt prost ather	bottom ash	GCMS(8270)	6/8n	< 0.980		< 0.980	< 0.990	< 0.990	
4 - Chbrochany Inhany ather	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
a - Chloronhanvi phanvi ether	collected fiv ash	GCMS(8270)	5/8n	< 0.990		066.0 >	< 0.970	< 1.00	
4 - Chloronhanvi nhanvi ether	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			

Appendix B: Analytical Data Used in Calculations

4

30-Oct-93

30-Oct-93

Table B-1: Site 16 OFA Data USED in Calculations

Strate 1		Method	Non	Run 1	Bun 2	Bun 3	Run 3D	Run 4	Run 5
4 Chlorophenvl phenvl ether	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
4 - Chlorophanyl phanyl ether	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
4 - Chinonbanyi phanyi ather	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 - Chlorophanyl phanyl ether	stack cas. vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 - Mathvinhanolfo - cresol)	bottom ash	GCMS(8270)	8/6n	< 0.980		< 0.980	066.0 >	< 0.990	
4	bottom ash stuice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
4 - Mathylphenol(n - cresol)	collected fiv ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
4 - Methylphenol(c - cresol)	high dust gas, solid phase	GCMS(8270)	cmN/gm	76600.0 >	< 0.00837	< 0.00863			
4 – Mathvibhenol(c – cresol)	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
4 - Mathvinhanol(o - cresol)	Isluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
4 - Methylphenol(p - cresol)	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 - Methylphenol(p - cresol)	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
4 – Nitrophenol	bottom ash	GCMS(8270)	6/6n	< 4.90		< 4.90	< 4.90	< 5.00	
4 – Nitrophenol	bottorn ash sluice water	GCMS(8270)	ug/t	< 52.0		< 53.0	< 54.0	< 56.0	
4 - Nitrophenof	collected fly ash	GCMS(8270)	0/6n	< 5.00		< 4.90	< 4.80	< 5.00	
4 - Nitrophenol	high dust gas, solid phase	GCMS(B270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432		 	
4 – Nitrophenoł	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0290	< 0.0244	< 0.0251			
4 – Nitronhenol	sluice water suppty	GCMS(8270)	ng/L	< 52.0		< 52.0	< 53.0	< 52.0	
4 Nitrophenol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
4 - Nitrophenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
7,12-Dimethvlbenz(a)anthraœne	bottom ash	GCMS(8270)	6/6n	< 2.50		< 2.40	< 2.50	< 2.50	1
7,12 - Dimethylbenz(a)anthraœne	bottom ash sluice water	GCMS(8270)	ug/L	< 26.0		< 26.0	< 27.0	< 28.0	
7,12-Dimethylbenz(a)anthraœne	collected fly ash	GCMS(8270)	6/6n	< 2.50		. < 2.50	< 2.40	< 2.50	
7,12 - Dimethylbenz(a)anthraœne	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0249	< 0.0209	< 0.0216	-		:
7,12 - Dimethylbenz(a)anthraœne	high dust gas, vapor phase	GCMS(8270)	CmN/gm	< 0.0145	< 0.0122	< 0.0126		:	
7.12 - Dimethylbenz(a) anthracene	sluice water supply	GCMS(8270)	1/6ո	< 26.0		< 26.0	< 26.0	< 26.0	
7.12 - Dimethvibenz(a)anthracene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0132	< 0.0139	< 0.0134			
7.12 - Dimethylbenz(a)anthracene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0132	< 0.0139	< 0.0134			
Acenaphthene	bottom ash	GCMS(8270)	8/6n	< 0.980		< 0.980	< 0.990	066.0 >	
Acenaphthene	bottom ash sluice water	GCMS(8270)	1/6n	< 10.0		< 11.0	< 11.0	< 11.0	
Acenachthene	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
Acenaphthene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Acenaphthene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Acenephthene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Acenaphthene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	-		
Acenephthene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Acenaphthylene	bottom ash	GCMS(8270)	6/6n	< 0.980	ļ	< 0.980	< 0.990	< 0.990	
Acenaphthylene	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
Acenaphthylene	collected fly ash	GCMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	< 1.00	

ŝ

			E 23	Hun	Run 2	Run 3	Run 3D	Run 4	Run 5
u Angela	gli dust gas, solid prase	(0,70) SM20	CUN/BU		< 0.00837	< 0.00863			
nthylene h	gh dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00561	< 0.00488	< 0.00503			
athylene si	uice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
uthylene	ack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
ithytene si	ack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
thyde hyde	gh dust gas, vepor phase	HPLC	mg/Nm3	0.0647		< 0.00782		< 0.00833	
st	ack gas, vapor phase	HPLC	mg/Nm3	0.00489 @		0.00810 @		< 0.00564	
enone	ottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
enone	ottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
enone	llected fly ash	GCMS(8270)	6/6n	066.0 >		066.0 >	< 0.970	< 1.00	
enone hi	gh dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
enone	gh dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
anone sh	rice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
anone st	ack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
anone	ick gas, vapor phase	GCMS(8270)	EmN/gm	< 0.00530	< 0.00555	< 0.00534			
Ĩ	gh dust gas, vapor phase	HPLC	mg/Nm3	0.0118 @		< 0.00944		< 0.0100	,
15	ack gas, vapor phase	HPLC	mg/Nm3	< 0.00232		< 0.00615		< 0.00681	
	ntorn ash	XRF	mg/kg			143321	142157	146126	144750
Ē	ttom ash stuice water	ICP-AES	<i>mg/L</i>			0.248 @	0.283 @	0.261 @	< 0.200
8	8	NAA	ug/g, dry			16477	16591	13681	13083
20	llected fly ash	XRF	mg/kg	,		142898	142580	147238	148349
n.	th dust gas, particulates	ICP-AES	mg/kg			116328 R		115097 R	120181 R
nin nin	th dust gas, solid phase	ICP-AES	cmN/gm			774 H		594 B	657 R
n T	th dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.0869		< 0.0933	< 0.0900
u	ice water supply	ICP-AES	mg/L			0.670 @	0.706 @	0.693 @	0.657 @
n st	ick gas, particulates	ICP-AES	mg/kg			91597 H		77159 R	87035 A
н 81	ick gas, solid phase	ICP-AES	mg/Nm3			11.8 H		26.9 H	226 A
	ck gas, vapor phase	ICP-AES	mg/Nm3			< 0.0733		< 0.0769	< 0.0681
pq	ttom ash	GCMS(8270)	8/6n	< 0.980		0960 >	< 0.990	< 0.990	
9	ttom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
0	lected fly ash	GCMS(8270)	<u>8/8n</u>	< 0.990		066.0 >	< 0.970	< 1.00	
high	h dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
hi	h dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
alu	ice water supply	GCMS(8270)	ոց/Լ	< 10.0		< 10.0	< 11.0	< 10.0	
ste	ck gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		- - -	
ste	ck gas, vapor phase	GCMS(8270)	EmN/Bm	< 0.00530	< 0.00555	< 0.00534			
Pa Pa	tom ash	GCMS(8270)	6/Bn	< 0.980		< 0.980	066.0 >	0.990 ≻	
P P	ttom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
	lected fly ash	3CMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
					- 	000:0 1	1	222.2	0.1 2 1.00

30-Oct-93

Appendix B: Analytical Data Used in Calculations

ø

 \sim

Table B+1: Site 16 OFA Data USED in Calculations

30-Oct-93

Appendix B: Analytical Data Used in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Anthracene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Anthracene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Anthracene	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
Anthracene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Anthracene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Antimony	bottom ash	ICP-AES	mg/kg			< 100	< 100	v 100	< 100
Antimony	bottom ash sluice water	ICP-AES	mg/L			< 0.100	< 0.100	< 0.100	< 0.100
Antimony	coal	NAA	ug/g. dry			0.905	1.62	1.90	1.30
Antimony	collected fly ash	ICP-AES	mg/kg			< 100	< 100	< 100	100
Antimony	high dust gas, particulates	ICP-AES	mg/kg			< 88.1		< 104	< 107
Antimony	high dust gas, solid phase	ICP-AES	mg/Nm3			< 0.586		< 0.538	< 0.587
Antimony	high dust gas, vepor phase	ICP-AES	mg/Nm3			< 0.0435		< 0.0466	< 0.0261
Antimony	sluice water supply	ICP-AES	mg/L			< 0.100	< 0.100	< 0.100	< 0.100
Antimony	stack gas, particulates	ICP-AES	mg/kg			< 135		< 83.0	< 113
Antimony	stack gas, solid phase	IC P ~ AES	mg/Nm3			< 0.0174		< 0.0290	< 0.0295
Antimony	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.0200		< 0.0164	< 0.0194
Arsenic	bottom ash	GFAAS	mg/kg			61.3@	87.0	30.0 @	45.0
Arsenic	bottom ash stuice water	HGAAS	mg/L			0.00680 @	< 0.0250	0.00810 @	0.00670 @
Arsanic	coal	NAA	ug/g, dry	-		13.3	24.0	18.9	18.8
Arsenic	collected fly ash	GFAAS	m <u>g/kg</u>			225	224	226	199
Arsenic	high dust gas, particulates	GFAAS	mg/kg			324 H		250 R	342 R
Arsenic	high dust gas, solid phase	GFAAS	mg/Nm3			2.15 R		1.29 R	1.87 R
Arsenic	high dust gas, vapor phase	GFAAS	CmN/gm			< 0.00174		< 0.00187	< 0.00180
Arsenic	stuice water supply	HGAAS	mg/L			0.0160 @	< 0.0250	0.0160 @	0.0160
Arsenic	stack gas, particulates	GFAS	mg/kg			553 R			581 B
Arsenic	stack gas, solid phase	GFAS	mg/Nm3			0.0713 R		-	0.151 B
Arsenic	slack ges, vapor phase	GFAS	mg/Nm3			< 0.00147		< 0.00154	< 0.00136
Ash	coal	Ultimate	%dry	11.5		9.05	8.67	10.3	10.01
Barium	bottom ash	ICP-AES	mg/kg			893	871	948	743
Barium	bottom ash sluice water	ICP-AES	mg/L			0.0940	0.0950	0.106	0.0940
Barlum	coat	NAA	ug/g, dry			206	154	158	139
Berium	collected fly ash	ICP-AES	mg/kg			827	775	756	747
Barium	high dust gas, particulates	ICP-AES	mg/kg			1126 R		1030 R	1092 R
Barium	high dust gas, solid phase	ICP-AES	mg/Nm3			7.50 R		5.32 R	5.97 H
Barium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.00435		< 0.00466	< 0.00450
Barium	sluice water supply	ICP-AES	mg/L			0.150	0.151	0.148	0.148
Barium	stack gas, particulates	ICP-AES	mg/kg			1267 R		925 R	1102 B
Barium	stack gas, solid phase	ICP-AES	mg/Nm3			0.163 R		0.323 R	0.287 A
Bariem	stack das vebor phase	ICP-AES	mg/Nm3			< 0.00366	 	< 0.00384	< 0.00341

B-9

Calculations
Ĩ,
USED
ata
õ
•
Ľ,
3
₽
Site
7
ĊÓ.
9
ab.
Ĥ

30-Oct-93

Substance	Stream	Method	NOM	Run 1	Run 2	Run 3	Hun 3D	Hunt	
		01100		0.00105.@	0.0006.00	0.000671.@			
euzene	high dust gas, zu - L VOSI	C MO	Curvian Curvia	0.00128.@	0.0030	0.000650 @			
6/12010	stack gas, zu - L vost	00100		0 080		< 0.980	060.0 >	066.0 >	
enzidine	bottom asn	GCM3(0210)	R/Rn	< 10.0		< 11.0	< 11.0	< 11.0	
enzidine	DOTIOR ASIA SILICE WARE	00110100000				066.0 >	0/6/0 >	< 1.00	
enzidine	collected ny asn	00000000000000000000000000000000000000		< 0 00907	< 0.00837	< 0.00863			
enzidine			C m N D m	< 0.00581	< 0.00488	< 0.00503			
enzidine	nign dusi gas, vapor priase	GCMC(8270)		< 10.0		< 10.0	< 11.0	< 10.0	
enzidine		COMPLEX (0)		< 0.00530	< 0.00555	< 0.00534			
8/12/dime	stack gas, solid pitase	0.0000000000000000000000000000000000000	E antiour	< 0.00530	< 0.00555	< 0.00534			
enzidine	stack gas, vapor pnase	GCM3(8270)	CHINA/BIN		20000	0.080	0.990	0.990	
erzo(a)anthracene	boltom asn	00100 (0100)	R/Rm			< 11.0	< 11.0	< 11.0	
en zo (a) anthracene	boltom ash sluice water	CCMS(85/0)	- navr	0.01			0.070	100	
enzo(a)anthracene	collected fly ash	GCMS(BZ/U)	<u>8/6n</u>			000.0 4	2	8	
enzo(a)anthracene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			
enzo(a)anthracene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
enzo(a)anthracene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
enzo(a)anthracene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
anthracene	stack das, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
anzo(a)ovrene	bottom ash	GCMS(8270)	ng/g	< 0.980		< 0.980	< 0.990	< 0.990	
anyo (a) avrene	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		A 11.0	< 11.0	- · · · · · · · · · · · · · · · · · · ·	
	collected fiv ash	GCMS(8270)	8/6n	< 0.990		066.0 >	< 0.970	< 1.00	
	htoh dust cas. solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
	hich dust gas vanor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			1
anzo(a)pyrana ozo(a)purana	shire water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
	etack das solid ohase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	stack das vanor nhasa	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			1
on so (h) financiant han a	bottom ash	GCMS(8270)	8/Bn	< 0.980		< 0.980	< 0.990	< 0.990	į
anzo(h)finoranthana	bottom ash sluice water	GCMS(B270)	ng/L	< 10.0		< 11.0	< 11.0	11.0	
anzo(h)thintanthana	collected flv ash	GCMS(8270)	6/8n	< 0.990		066.0 >	< 0.970	1.00	
(h)fliorenthene	high dust das solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
or to (b) fill or souther to	high dust cas vanor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
or to (b) (i. or an them a	etuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	{
(h)(incentione	stack das solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	eteck res vandr nhase	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
	bottom ash	GCMS(8270)	0/01	< 0.980		< 0.980	< 0.990	< 0.990	
	hottom ash shrice water	GCMS(8270)	uo/L	< 10.0		< 11.0	< 11.0	< 11.0	
utor(a, interviewa	collected fiv ash	GCMS(8270)	8/6n	066.0 >		< 0.990	< 0.970	< 1.00	
	binh dust ass solid nhase	GCMS(8270)	ma/Nm3	< 0.00997	< 0.00837	< 0.00863	-		
euso(ĝ'u'u)perviene				102000		~ 0.00503			

Appendix B: Analytical Data Used in Calculations

∞

30-Oct-93

Table B-1: Site 16 OFA Data USED in Calculations

თ

Substance	Stream	Method	NON	Run 1	Run 2	Run a	Bun 3D		
Benzo(g.h.i)perylene	shuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0		
Benzo(g.h.i)perylene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	> - -	/	
Benzo(g.h.i)perylene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Ben zo (k) fluoranthene	bottom ash	GCMS(8270)	6/6n	< 0.980		0.980 >	066 0 >	0000	
Benzo(k)fluoranthene	bottom ash stuice water	GCMS(8270)	1/ 6 /T	< 10.0		< 11.0	< 11.0		
Benzo (k) fluoranthene	collected fly ash	GCMS(8270)	ng/g	< 0.990		066.0 >	< 0.070 <		
Benzo(k)fluoranthene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0 00837	< 0.00863	0/6-0 /		
Benzo(k)fluoranthene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Benzo(k)fluoranthene	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0			
Benzo(k)fluoranthene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	,	2 2 2	
Benzo(k)fluoranthene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Beryllium	bottom ash	ICP-AES	mg/kg			8.80 @	R 70 @	a 01	
Beryllium	bottom ash stuice water	ICP-AES	mg/L			< 0.00200		000000	0.11
Beryflium	coat	ICP-AES	mg/kg, dry			1 73			0.00200.0 ×
Beryllium	collected fly ash	ICP-AES	mg/ka						1.01 (6)
Beryllium	high dust gas, particulates	ICP-AES	mg/kg			165 81	2	1.0	11.2
Beryllium	high dust gas, solid phase	ICP-AES	CmN/pm					LOS	H 6.02
Beryllium	high dust gas, vapor phase	ICP-AES	mg/Nm3		· · · · · · · · · · · · · · · · · · ·	0.000		0.102 H	0.115 H
Beryllium	sluice water supply	ICP-AES	mg/L			000000		620000.0 >	006000.0 >
Beryllium	stack gas, particulates	ICP-AES	mg/kg	+ 		16 8 8			
Berytlium	stack gas, solid phase	ICP-AES	mg/Nm3	+		0.00217 8	*		H / BL
Beryllium	stack gas, vapor phase	ICP-AES	mg/Nm3		, ,	< 0.000233			H HIGONO
bis(2 - Chloroethyl)ether	bottom ash	GCMS(8270)	6/6n	0.980		< 0.980		 0.00041 0.000 	680000 >
bis(2 Chloroethyl)ether	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		 11 0 		066.0 >	
bis(2 - Chloroethyl)ether	collected fly ash	GCMS(8270)	8/8n	0.990 >				0.11 2	
bis(2 - Chloroethyt)ether	high dust gas, solid phase	GCMS(8270)	CmN/Bm	< 0.0097	< 0.00837	0.00883			
bis(2 - Chloroethyl)ether	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	•	,	
bis(2 Chloroethyl)ether	sluice water supply	GCMS(8270)	1/60	< 10.0		< 10.0	× 110		
bis(2 - Chloroethyl)ether	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		21 22 21 21 21 21 21 21 21 21 21 21 21 2	
bis(2 - Chloroethyl)ether	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
bis(2 - Ethylhexyl)phthalate	bottom ash	GCMS(8270)	8/8n	< 0.980		< 0.980	0.990	0000	
bis(2 - Ethylhexyl)phthalate	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	0110	
bis(2 - Ethylhexyl)phthalate	collected fly ash	GCMS(8270)	6/Bn	< 0.990		066.0 >	< 0.970		
bis(2 - Ethylhexyl)phthalate	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	0.0109	< 0.00863	 		- - - - - -
bis(2 - Ethylhexyl)phthalate	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	0.396	0.0438	0.00767 @	<u> </u>		
bis(2 - Ethylhexyl)phthatate	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	100	
bis(2 - Ethylhexyl)phthalate	stack gas, solid phase	GCMS(8270)	mg/Nm3	0.0109 @	< 0.00555	< 0.00534	 		
bis(2 - Ethylhexyl)phthalate	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	0.00832 @			
Bromotorm	high dust gas, 20-1. VOST	GCMS	EmN/gm	< 0.000525	< 0.000543	< 0.000559		, 	

· .

Appendix B: Analytical Data Used in Calculations

9
ĊĎ.
Ť
ŭ
0
1
2

Substance	Stream	Method	MOU	Run 1	Aun 2	Run 3	Run 3D	Run 4	Run S
Bromoform	stack das 20-1 VOST	GCMS	ma/Nm3	< 0.000580	< 0.000569	< 0.000542			
Bromomethane	high dust gas 20-1 VOST	GCMS	mo/Nm3	0 000682 @	0.00168.@	0.000559			
Bromomethane	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Butylbenzylphthalate	bottom ash	GCMS(8270)	5/6n	< 0.980		0.980 >	0.990	066.0 >	
Butylbenzyiphthalate	bottom ash sluice water	GCMS(8270)	лgл	< 10.0		< 11.0	< 11.0	< 11.0	
Butyłbenzylphthalate	collected fly ash	GCMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	4 1.00	
Butytbenzylphthalate	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Butytbenzylphthalate	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Butytbenzylphthalate	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Butytbenzylphthalate	stack gas, solid phase	GCMS(8270)	EmN/Bm	< 0.00530	< 0.00555	< 0.00534			70.0 × 0
Butytbenzytphthalate	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Cadmium	bottom ash	GFAAS	mg/kg		,	< 1.00	< 1.00	< 1.00	< 1.00
Cadmium	bottom ash sluice water	GFAAS	mg/L			< 0.00100	< 0.00100	< 0.00100	0.00200
Cadmium	coat	GFAAS	mg/kg. dry			< 0.104	< 0.104	0.157 @	< 0.105
Cadmium	collected fly ash	GFAAS	mg/kg			< 1.00	< 1.00	< 1.00	< 1.00
Cadmium	high dust gas, particulates	GFAAS	mg/kg			0.994 @R		< 2.04 R	1.11 @F
Cadmium	high dust gas, solid phase	GFAAS	mg/Nm3			0.00662 @R	1	< 0.0105 R	0.00608 @F
Cadmium	high dust gas, vapor phase	GFAAS	CmN/gm			< 0.000435		< 0.000466	< 0.000450
Cadmium	sluice water supply	GFAAS	mg/L			0.00140 @	0.00170 @	0.00140 @	0.00170
Cadmium	stack gas, particulates	GFAAS	mg/kg			2.65 @			3.20 (9)
Cadmium	stack gas, solid phase	GFAS	mg/Nm3			0.000342 @			0.000833 @
Cadmium	stack gas, vapor phase	GFAAS	EmN/Bm			< 0.000366		< 0.000384	< 0.000341
Catcium	bottom ash	XRF	mg/kg			7433	7433	7504	5861
Calcium	bottom ash stuice water	ICP-AES	mg/L			16.0	16.1	17.0	16.1
Calcium	coal	NAA	ug/g. dry			2879	1763	2408	1296
Calcium	coltected fly ash	XRF	mg/kg			11650	9863	6361	4646
Calcium	high dust gas, particulates	ICP-AES	mg/kg			9761 R		9898 R	4041 @F
Celcium	high dust gas, solid phase	ICP-AES	mg/Nm3			65.0 A		51.1 B	22.1 @H
Calcium	high dust gas, vapor phase	ICP~AES	mg/Nm3			< 0.435	+ 	< 0.466	< 0.450
Calcium	sluice water supply	ICP-AES	_\6m			21.1	21.4	20.6	20.8
Calcium	stack gas, particulates	ICP-AES	mg/kg			9398 R		5905 R	6471 B
Calcium	stack gas, solid phase	ICP-AES	mg/Nm3			1.21 R		2.06 R	1.68 H
Calcium	stack gas, vapor phase	ICP-AES	CmN/gm			< 0,366		< 0.384	< 0.341
Carbon	coal	Ultimate	%dry	75.0		77.6	78.0	76.5	76.6
Carbon disulfide	high dust gas. 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Carbon disuffide	stack gas, 20-1 VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Carbon tetrachloride	high dust gas. 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Carbon tetrachloride	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Chloride	bottom ash	Pot. Titr.	mg/kg			< 100	167 @	< 100	< 100

Appendix B: Analytical Data Used in Calculations

10

٢

∽.
ന
1
ະ
0
ī
2
3

Ξ

	Stream	Method	NON	Bun 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Chioride	bottom ash sluice water	0	mg/L			2.92	2.89	3.35	3.07
Chloride	coal	Pot. Titr.	mg/kg. dry			314 @	460 @	501 @	404 @
Chloride	collected fly ash	Pot. Titr.	mg/kg			< 100	< 100	< 100	100
Chloride	high dust gas, solid phase	5	EmN/Bm			22.5 R		20.7 B	27.5 R
	high dust das vapor phase	0	mg/Nm3			9.11		11.7	11.6
Chlorida	sluice water supply	0	mg/L)	3.01	2.96	2.99	2.97
Chloride	stack cas, solid phase	ç	mg/Nm3			0.0126 R	ĺ	0.0242 R	0.0572 R
Chloride	stack gas, vapor phase	2	EmN/Bm			20.9		21.7	23.4
Chlorobenzene	high dust gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Chlorobenzene	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Chloroethane	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Chloroethane	stack gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Chloroform	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Chloroform	stack gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Chloromethane	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Chloromethane	stack gas, 20-1, VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Chromium	bottom ash	ICP-AES	mg/kg			6.86	100	116	125
Chromium	bottom ash sluice water	ICP-AES	mg/L			< 0.0100	< 0.0100	< 0.0100	< 0.0100
Chromium	coal	NAA	ug/g. dry			25.6	20.0	20.9	18.8
Chromium	collected fly ash	ICP-AES	mg/kg			903	131	105	
Chromium	high dust gas, particulates	ICP-AES	mg/kg			161 R		162 R	169 B
Chromium	high dust gas, solid phase	ICP-AES	EmN/Bm			1.07 R		0.837 B	0.925 R
Chomin	high dust gas, vapor phase	IC P-AES	EmN/gm			< 0.00435		< 0.00466	< 0.00450
Chromium	stuice water supply	ICP - AES	mg/L			< 0.0100	< 0.0100	< 0.0100	< 0.0100
Chromium	stack gas, particulates	ICP-AES	mg/kg			205 R		153 A	196 R
Chromium	stack gas, solid phase	ICP-AES	mg/Nm3			0.0264 B		0.0534 R	0.0511 8
Chromium	stack gas, vapor phase	ICP-AES	CmN/Bm			≤ 0.00366		< 0.00384	< 0.00341
Chrysene	bottom ash	GCMS(8270)	6/6n	< 0.980		0.980	0660	0660 >	
Chrysene	bottom ash sluice water	GCMS(8270)	1/8ո	< 10.0		< 11.0	<pre>< 11.0</pre>	< 11.0	
Chrysene	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
Chrysene	high dust gas, solid phase	GCMS(8270)	CmN/Bm	< 0.00997	< 0.00837	< 0.00863		•	
Chrysene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Chrysene	stuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0 <	< 10.0	
Chrysene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Chrvsene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
cis-1.3-Dichloropropene	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
cis - 1,3 - Dichtoropropene	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Cobaft	bottom ash	ICP-AES	mg/kg			52.7	49.9 @	51.7	46.0 @
Cobatt	bottom ash sluice water	ICP-AES	mg/L			< 0.0100	< 0.0100	< 0.0100	< 0.0100

Appendix B: Analytical Data Used in Calculations

Substance	Stream	Method	NON	Run 1	Run 2	Hun 3	Run 3D	Bun 4	Hun 5
Cobalt	coal	NAA	ug/g. dry			8.93	6.84	8.26	1.82
Cobalt	collected fly ash	ICP-AES	mg/kg			60.6	57.3	49.8 @	54.0
Cobalt	high dust gas, particulates	ICP-AES	mg/kg			62.3		62.8	999
Cobalt	high dust gas, solid phase	ICP-AES	mg/Nm3			0.415		0.324	0.364
Cobalt	high dust gas vapor phase	ICP-AES	mg/Nm3			< 0.00248		< 0.00262	< 0.00189
		IC P - AFS	ma/L			< 0.0100	< 0.0100	< 0.0100	< 0.0100
CODBIG	stuck ase nationales	CP-AFS	ma/ka			59.0 @		48.9	59.5
Cobalt	SIBCK Has, Painvulaios		C million			0.00761.@		0.0171	0.0155
Cobalt	stack gas, solid phase	ICF-AES				< 0.00167		< 0.0020	< 0.00194
Cobalt	stack gas, vapor pnase		cilla/Ran			e au	000	0.7 6 (0)	0.70
Copper	bottom ash	ICP-AES	mg/kg			97.C6	84.J (2000 C / R	0000
Copper	bottom ash sluice water	ICP-AES	mg/L			< 0.0200	< 0.0200	< 0.0200	< 0.0200
Copper	coal	ICP-AES	mg/kg. dry			55.5	31.6	29.2	30.6
Connet	collected fly ash	ICP-AES	mg/kg			161	146	158	170
Conner	high dust gas, particulates	ICP-AES	ву/вш			196 R		223 R	233 <u>F</u>
Cupper	high dust das solid phase	ICP-AES	ma/Nm3			1.31 R		1.15 R	1.27 B
	high duet das vanor phase	IC P-AES	ma/Nm3		 	< 0.00869		< 0.0033	< 0.00900
Copper			Por la			0.0340 @	0.0330 @	0.0350 @	0.0350 @
Copper	sluice water supply		- TAIL			0 000		170.0	906 B
Copper	stack gas, particulates	ICP AES	Bw/6m			1 507 C			
Copper	stack gas, solid phase	ICP-AES	mg/Nm3			H 0/200		H FRCD.D	1 6900.0
Copper	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.00733		< 0.00768	< 0.00681
Dibenzofuran	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
Dibenzofu(an	bottom ash sluice water	GCMS(8270)	1/6n	< 10.0		< 11.0	< 11.0	× 11.0	
Dihenzohutan	collected fly ash	GCMS(8270)	B/Bn	< 0.990		066.0 >	< 0.970	< 1.00	
	high dust gas solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			-7
	hich dust das vanor phase	GCMS(8270)	ma/Nm3	< 0.00581	< 0.00488	< 0.00503			
	alline water enough	GCMS(8270)	110/1	< 10.0		< 10.0	< 11.0	< 10.0	
	stark rae enlid these	GCMS(8270)	ma/Nm3	< 0.00530	< 0 00555	< 0.00534		- 	· · · · · · · · · · · · · · · · · · ·
		GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
Diharta Hanihrena	hottom ash	GCMS(8270)	ua/a	< 0.980		< 0.980	< 0.990	< 0.990	1
Diborata blanthracena	hottom ash strice water	GCMS(8270)	ua/L	< 10.0		< 11.0	< 11.0	< 11.0	
Dihangla hisothracana	collected fiv ash	GCMS(8270)	8/8n	066.0 >		< 0.990	< 0.970	< 1.00	
Dihana(a hianthracana	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Discrete blanthranena	hich dust das vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Diserv(a h)anthracene	elvice water subbly	GCMS(8270)	na/r	< 10.0		< 10.0	< 11.0	< 10.0	
Disconta Hanthreesia	ctack das solid nhasa	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
Dihanya hanthraana	stack das vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	hotton ash	GCMS(8270)	0/07	086.0 >		< 0.980	< 0.990	< 0.990	
	bottom ash aluica water	GCMS(8270)		< 10.0		< 11.0	< 11.0	< 11.0	
Dibenz(a, i)acrioirte	DOILUIII dail ature mere	02(8270)		066.0 >		066.0 >	< 0.970	< 1.00	-
Dibenz(a,))acridine	COLLECTED ILY ANI	הייין הייין	1					F = = + + +	

30-Oct-93

Appendix B: Analytical Data Used in Calculations

₽

30-Oct-93

Table B-1: Site 16 OFA Data USED in Calculations

.

₽

				L UNH		C MUN	hun Ju		
henzta ibacridine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			
thenzie Nacridine	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
benz(a i)acridine	stuice water supply	GCMS(8270)	ոց/Լ	< 10.0		< 10.0	< 11.0	< 10.0	
bany(a Nacridina	stack das solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
bont(u.))aurunu bont(a i)acridina	stack das vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
<u>Versiantshinalata</u>	hottod ash	GCMS(8270)	na/a	< 0.980 B		< 0.980 B	066.0 >	< 0.990	
	bottom ash shrice water	GCMS(8270)	ua/L	20.1 @		20.4 @	31.5 @	77.2	
Dutypricialate	collected fly ash	GCMS(8270)	0/0n	066.0 >		< 0.990	< 0.970	< 1.00	
	high duct are solid where	GCMS(R270)	mo/Nm3	< 0.0097	< 0.00837	< 0.00863			
butyipninalate	high dust das, sond pridad	GCMS(8270)	EmN/pm	0.266	0.0303	< 0.00503			
	lehine water sumpty	GCMS(8270)	ua/L	< 10.0		< 10.0	< 11.0	< 10.0	
	stark dae solid phase	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
	stack das vanor nhase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
uryipminated mothulatitedate	hottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
	hottom ash shrice water	GCMS(8270)	ug/L	< 10.0		< 11.0	× 11.0	< 11.0	
mothurbithelate	collected fiv ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
math.unhthelate	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
methytohthalate	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
mathylinhthalata	stuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	:
methylohthalate	stack das, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
methylohthalate	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
ohanvlamine	bottom ash	GCMS(8270)	5/Bn	< 0.980		< 0.980	< 0.990	066 .0 >	1
tion famine	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
nhanvlamine	collected fly ash	GCMS(8270)	6/6n	< 0.990 ×		< 0.990	< 0.970	< 1.00	
ohenvlamine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
ohenvlamine	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
ohenvlamine	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
ohenvlamine	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			:
bhenvlamine	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
hvi benzene	high dust gas, 20-1 VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
hvi benzene	stack gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			-
ad carbon	coal	Proximate	%dry	53.9		55.3	55.6	54.4	54.7
inventhane	bottom ash	GCMS(8270)	ß/6n	< 0.980		< 0.980	< 0.990	< 0.990	
instanthene	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
locanthene	collected fly ash	GCMS(8270)	0/8	066:0 >		< 0.990	< 0.970	< 1.00	
oranthana	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
uoranthene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
locanthene	stuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
uoranthene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			

Appendix B: Analytical Data Used in Calculations
Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D
			-				
Fluoranthene	stack ges, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	
Fluorene	bottom ash	GCMS(8270)	6/61	< 0.980	-	< 0.980	< 0.990
Fluorene	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0
Fluorene	collected fly ash	GCMS(8270)	ug/g	< 0.990		< 0.990	< 0.970
Fluorene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863	
Fluorene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	
Fluorene	sluice water suppty	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0
Fluorene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	
Fluorene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	
Fluoride	bottom ash	ISE	mg/kg			< 17.0	< 17.0
Fluoride	bottom ash stuice water	ISE	mg/L			0.123 @	0.122 @
Fluoride	coal	ISE	mg/kg. dry			72.9	9.96
Fluoride	collected fly ash	ISE	mg/kg			51.6	61.1
Fluoride	high dust gas, solid phase	ISE	mg/Nm3			1.06 R	
F luoride	high dust gas, vapor phase	ISE	CmN/Bm			1.68 R	
F luoride	stuice water supply	ISE	mg/L			0.179@	0.173 @
Fluoride	stack gas, solid phase	ISE	mg/Nm3			0.466 R	
Fluoride	stack gas, vapor phase	ISE	mg/Nm3			7.15 R	
Formaldehyde	high dust gas, vapor phase	HPLC	mg/Nm3	0.0135 @R		< 0.00647 R	
Formaldehyde	stack gas, vapor phase	HPLC	mg/Nm3	0.00274 @R		< 0.00422	
Heating value	CO2)	Proximate	Btu/lb, dry	13385		13948	13992
Hexachlorobenzene	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990
Hexachlorobenzene	bottom ash stuice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0
Hexachlorobenzene	collected fly ash	GCMS(8270)	100/0	< 0.990		< 0.990	0/6.0 >
Haxachlorobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863	
Hexachlorobenzene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	
Hexachlorobenzene	stuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0
Hexachlorobenzene	stack gas. solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	
Hexachtorobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	
Hexachlorobutadiene	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990
Hexa chlorobutadiene	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0
Hexa chlorobutadiene	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970

Site 16 OFA Data USED in Calculations Table B-1:

4

Run 5

Run 4

Appendix B: Analytical Data Used in Calculations

•

t

< 11.0 < 1.00

< 0.990

0.135 @

0.139 @

37.5 @

96.8

55.4

< 17.0

< 17.0

ī.

< 10.0

< 11.0

1.39 R 1.65 A 0.170 @ 0.486 A

1.41 H 2.23 H

0 175 @ 0 379 A 6.74 H

6.81 R

< 0.00689 < 0.00467

13740 1

13733 < 0.990

ł

× 1.00

< 10.0

< 11.0

1

< 11.0

< 1.00

< 0.990 < 0.00863 < 0.00503 < 10.0

< 11.0

< 10.0

< 0.00488

< 10.0

< 0.00837

< 0.00997

< 0.00581

mg/Nm3 mg/Nm3

high dust gas, vapor phase high dust gas, solid phase

Hexachlorobutadiene Hexachlorobutadiene Hexachlorobutadiene Hexachlorobutadiene Hexachlorobutadiene Hexachlorobutadiene

8/Bn

GCMS(8270) GCMS(8270) GCMS(8270) < 0.00534

< 0.00555 < 0.00555

< 0.00530

mg/Nm3 mg/Nm3 6/6n

GCMS(8270)

GCMS(8270)

stack gas, vapor phase stack gas, solid phase sluice water supply

bottom ash

Hexachlorocyclopentadiene. Hexachlorocyclopentadiene

GCMS(8270) GCMS(8270)

1/Bn

< 0.00530

< 0 990 < 11.0

< 11.0

< 11.0

< 0.980 < 0.00534

> < 0.980 < 10.0

> > ng/L

bottom ash sluice water

< 0.990

< 0.990

61.2

101

30-Oct-93

30--Oct--93

Table B-1: Site 16 OFA Data USED in Calculations

15

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Bun 4	Run 5
				000 0		000 0 /	< 0 070		
Hexachlorocyclopentadiene	collected fly ash	GCMS(8270)	6/6n	0.88.0 >			0.00	8	
Hexachlorocyclopentadiene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			
Hexachlorocyclopentadiene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Hexachlorocyclopentadiene	sluice water suppty	GCMS(8270)	ug/t	< 10.0		< 10.0	< 11.0	< 10.0	
tionaction of the second of th	stack cas solid ohasa	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
uevechiorocyclopentaciene uevechiorocyclopentaciene	stack cas. vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	hortom ash	GCMS(8270)	p/on	< 0.980		< 0.980	< 0.990	< 0.990	
	bottom ash shrice water	GCMS(8270)	UQ/L	< 10.0		< 11.0	< 11.0	< 11.0	
	collected fly ash	GCMS(8270)	na/a	066.0 >		< 0.990	< 0.970	< 1.00	
	thick duet day solid phase	GCMS(8270)	ma/Nm3	< 0.0097	< 0.00837	< 0.00863			
Hexacnioroenane	hind dust gas, sond prose	GCMS(8270)	mg/m3	< 0.00581	< 0.00488	< 0.00503			
	chice water supply	GCMS(8270)	ua/L	< 10.0		< 10.0	< 11.0	< 10.0	
	stark ne enlid nhee	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
	stack dae vanor chase	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
		Ultimate	%drv	4.72		5.04	4.95	4.85	4.96
	hottom ach	GCMS(8270)	ua/a	< 0.980		< 0.980	< 0.990	< 0.990	
	bottom ash shrica water	GCMS(8270)	ua/L	< 10.0		< 11.0	< 11.0	< 11.0	
	collected flv ash	GCMS(8270)	0/00	066.0 >		< 0.990	< 0.970	< 1.00	
	bich duet nee exid nhase	GCMS(8270)	ma/Nm3	< 0.00997	< 0.00837	< 0.00863			
indeno(),2,3 - cujpyrene		CCMS(8270)	C.m.V.om	< 0.00581	< 0.00488	< 0.00503			
tndeno(1,2,3 - cd)pyrene	nign dust yas, vaput prase	COMO(0210)		< 10.0		< 10.0	< 11.0	< 10.0	
Indeno(1,2.3 - cd)pyrene	Sturce water suppry	COMP (0220)	2 muluus	< 0.00530	< 0.00555	< 0.00534			
Indeno(1,2,3 - cd)pyrene	stack gas, solid phase	0CM3(0210)	Curvie and	V 0.00530	 0.00555 0.00555 	< 0.00534			
Indeno(1,2,3 - cd)pyrene	stack gas, vapor pnase	(CLASICAL)	curv/fur	00000.0 /		93864	96172	95333	87499
Iron	bottom ash	AHF ICD AFE	Ru/Bui			0.293	776.0	0.296	0.140 @
Iron	DOROTH ASH SINICE WARE					14510	0407	12AGO	11356
Iron	coat	NAA	have any			20020			
Iron	collected fly ash	XRF	mg/kg			31,231	16716	00000	60000
lon	high dust gas, particulates	ICP-AES	mg/kg	!		79782 H		73463 H	82430 H
lian	high dust gas, solid phase	ICP-AES	mg/Nm3			531 R		379 H	451 H
	high dust gas, vapor phase	ICP-AES	mg/Nm3		 	< 0.0174		< 0.0187	0 0206 @
	shice water supply	ICP-AES	mg/L			0.805	0.850	0.788	0.768
	eteck das particulates	ICP-AES	mg/kg			69896 R		60304 R	64137 R
	stack nas solid nhase	ICP-AES	ma/Nm3			9.02 H		21.1 R	16.7 B
1101	etack das vanor phase	ICP-AES	mg/Nm3			< 0.0147		< 0.0154	< 0.0136
	bottom ash	GCMS(8270)	na/a	< 0.980		< 0.980	< 0.990	< 0.990	
150priorone	Hottom seh slitice water	GCMS(8270)	ua/L	< 10.0		< 11.0	< 11.0	< 11.0	
	contactact for ash	GCMS(8270)	na/a	066.0 >		066.0 >	< 0.970	< 1.00	
	Lich durt me exlid sheep	GCMS(8270)	mo/Nm3	7990.0 >	< 0.00837	< 0.00863			
Isophorone	nigir urat yas, avin prises	00100000000	C. Min	< 0.00581	< 0.004.88	< 0.00503			
lso phorone	high dust gas, vapor pnase	GUMSIOZIU	CUNN/BUI	1 00001 2	1 777777	1 >>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>		L	[

33	
-0ct-1	
g	

sopharone

sophorane

sopharone

Lead Lead Lead Lead Lead Lead

16

Substance	Stream	Method	NON	Run 1	Run 2	P UNH		* 1171	NUN U
		101000000		0.01		10.01	011 /		
	sluice water supply	GCM3(02/0)		0.00640	V D DOEEK	LCSOD O			
	stack gas, solid phase	GCM2185/U)	Smn/gm	nccon n >		10000			
	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	CCC00'0 >	+Reonin >			
	bottom ash	GFAAS	mg/kg			21.5	21.4	20.7	6
	bottom ash sluice water	GFAAS	mg/L			< 0.00300	< 0.00300	0.00310 @	< 0.0030
	coal	GFAAS	mg/kg. dry			4.80	4:16	5.64	Š,
	collected fiv ash	GFAS	mg/kg			40.2	7.8.7	0.06	95
	hich dust cas particulates	GFAAS	mg/kg			40.0 R		101 R	96.1
	hich duet cas solid phase	GFAS	mg/Nm3	 		0.266 R		0.523 R	0.526
	hinh dust das vapor phase	GFAS	CmN/gm			< 0.00130		0.000864 @	0.00260
	skitce water supply	GFAS	mg/L			< 0.00300	0.00690 @	0.00340 @	0.00610 (
	stack das. Darticulates	GFAS	mg/kg			326 R			139
	stack das solid phase	GFAAS	mg/Nm3			0.0420 R			0.0360
	stack das. vabor phase	GFAS	mg/Nm3			< 0.00110		< 0.00115	< 0.00103
	hottom ash	ICP-AES	ma/kg			140	142	101	9
	hottom ash stuice water	ICP-AES	mg/L		1	0.0500	0.0510	0.0570	0.050(
	coal	NAA	ug/g, dry			13.6	16.7	15.6	22
	coffected flv ash	ICP-AES	ma/ka			230	130	93.0	96.1
	high dust gas particutates	ICP-AES	mg/kg			147 R		116 R	124
	high dust gas, solid phase	ICP-AES	EmN/Bm			0.9B0 R		0.601 R	0.679
	i high dust das. vapor phase	ICP-AES	EmN/Bm			< 0.00435		< 0.00466	< 0.00450
	stuice water supply	ICP-AES	mg/L			0.0920	0.0930	0.0870	0.090
	stack das, Darticulates	ICP-AES	mg/kg			152 R		104 R	116
	stack das, solid phase	ICP-AES	EmN/Bm			0.0196 B		0.0363 H	0.0301
	stack cas. vepor phase	ICP-AES	EmN/Bm			< 0.00366		< 0.00384	< 0.0034
	bottom ash	DGA/CVAAS	6/6n			< 0.0200		< 0.0200	< 0.0200
	bottom ash sluice water	CVAS	mg/L			< 0.000200	< 0.000200	< 0.000200	< 0.00020
	coal	DGA/CVAAS	ug/g, dry			0.168	0.448	0.125	0.14
	collected fly ash	DGA/CVAAS	0/0		+	0.240	0.270	0.210	0.25(
	high dust gas, particulates	CVAAS	mg/kg			0.536 @ H		0.00961 @	1.06
	high dust gas, solid phase	CVAS	mg/Nm3			0.00357 @R		0.0000496	0.00562
	high dust gas, vapor phase	CVAAS	mg/Nm3			0.00297		0.00437	0.0044
	Isluice water supply	CVAAS	mg/L			< 0.000200	< 0.000200	< 0.000200	< 0.00020(
	stack gas, particulates	CVAAS	mg/kg			0.984 @R		1.53 R	1.41
	stack das, solid phase	CVAAS	mg/Nm3			0.000127 @R		0.000536 B	0.000366
	stack gas, vapor phase	CVAAS	mg/Nm3			0.00663		0.00718	0.00757
thoride	high dust gas, 20-L VOST	GCMS	mg/Nm3	0.00131 @	< 0.000543	0.00341			
thioride	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	0.00250 @	0.0103			
	leng	Ultimate	*	5.51	-	4.12	3.94	4.27	4.82

Manganese

Manganese Manganese Manganese

Manganese

Manganese

Lead

Lead Lead

Manganese

Manganese

Manganese Manganese

Mercury

Mercury

Mercury Mercuty Mercury

Mercury

Mercury

Manganese

Mercury Methylene chloride Methylane chloride

Moisture

Mercury Mercury

Mercury

Table B-1: Site 16 OFA Data USED in Calculations

17

Substance	Stream	Method	NON	Run 1	- Bun 2	Run 3	Run 3D	Run 4	0F
Molybdenum	bottom ash	IC P-AES	mg/kg	 	-	< 200	006 >	000	
Molybdenum	bottom ash sluice water	ICP-AES	mg/L			< 0.0500	00500	2 200	002. >
Molybdenum	coal	NAA	up opri				1	nnen n >	< 0.0500
Molybdenum	collected fiv ash	ICP-AFS				200 ····	14.G	5.19	
Molvbdentim	birth duet ras contourister						200	× 200	< 200
Molthdamm			By/Bu			< 43.2 H		< 51.0 R	< 52.5 R
	nigh dust gas, solid phase	ICP-AES	CmN/Bm			< 0.287 R		< 0.263 R	< 0.287 R
Molybdenum	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.0217		< 0.033	× 0.025
Molybdenum	sluice water supply	IC P ~ AES	mg/L			< 0.0500	< 0.0500		
Molybdenum	stack gas, particulates	ICP-AES	mg/ka			61 S 13	200		
Molybdenum	stack gas, solid phase	ICP-AES	CmN/om	· · · · · · · · · · · · · · · · · · ·					HO 97.60
Molybdenum	stack gas, vapor phase	ICP-AES	mg/Nm3					H CHION >	0.0155 @R
m.pXylene	high dust gas. 20-L VOST	GCMS	mo/Nm3	< 0.00555	- 0.000517			< 0.0192	< 0.0170
m.p-Xvlane	stack rae 20~1 VOST	GCMS	E	0.000500		Second >		+	
Nachthatana		CHOLON CHOC	Clistick	noconnin	R90000 >	249000.0 >			
	DOUD BSN	6CMS(82/0)	<u>6/6n</u>	< 0.980		< 0.980	< 0.990	< 0.990	
Naphthalene	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
Naphthalene	collected fly ash	GCMS(8270)	0/6n	< 0.990		< 0.990	020 >		
Naphthalene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0 00863		8	1
Naphthalene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Naphthalene	sluice water supply	GCMS(8270)	ug/L	< 10.0		0.01 >			
Naphthalene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0 00555	< 0.00534	2		
Naphthalene	stack gas, vapor phase	GCMS(8270)	CmN/Bm	< 0.00530	< 0.00555	< 0.00534		• • • • • • • • • • • • • • • • • • • •	
Nickel	bottom ash	ICP-AES	mg/ka			EU		() e ee	
Nickel	bottom ash sluice water	ICP-AES	ma/L					90.9 (0)	00 6 86 0
Nickel	coat	NAA	ua/a. drv			7 70		> 0.200	< 0.0200
Nicket	collected fiv ash	ICP-AES				21.1	5.52	27.0	25.3
Nickel	high dust gas particulates	IC P - AFS	make					0.1.66	100
Nickel	high dust gas, solid phase	ICP-AFS	mo/Nm3					105 R	116 R
Nickel	high dust gas, vapor phase	ICP-AES	mo/Nm3					H 44 C	0.634 H
Nickel	sluice water supply	ICP-AES	ma/l			60000 V		< 0.00933	< 0.00900
Nickel	stack gas, particulates	ICP-AES	mg/ka			1.18.1	< 0.0200	0.0250 @	< 0.0200
Nickel	stack gas, solid phase	ICP-AFS	ma/Nm3					H 0.05	H CZL
Nickej	stack das vapor phase	IC P - AFS	mo/nm3					0.0337 H	0.0325 R
Vitrobenzene	bottom ash	GCMS(8270)	na/a	0.080		< 0.00/33		< 0.00768	< 0.00681
Vitrobenzene	bottom ash sluice water	GCMS(8270)	na/L	< 10.0		0.00	066-0 2	< 0.990	
Vitrobenzene	collected fly ash	GCMS(R270)	uala				2 1 2	21.0	
Vitrobenzene	high dust gas, solid phase	GCMS(8270)	ma/Nm3		10000	088.0 >	< 0.9/0	< 1.00	
ditrobenzene	high dust gas, vapor phase	GCMS(8270)	ma/Nm3	< 0.00581	 0.00488 	 V.VUGUU V.00503 			
ditrobenzene	sluice water suboly	GCMS(R270)	1/01	001 1		0000 ·			
ditrobenzene	stack cas. solid phase	GCMS(8270)	CmN/ow	00000	10.00555	10.01	0.11	< 10.0	
			1 MILLIAM	200000	erenn'n y	< 0.00334			

6
t
ŏ
ģ
e

Table B-1: Site 16 OFA Data USED in Calculations

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Nitrobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Nitroaen	coal	Uttimate	۶dry	1.59		1.52	1.52	1.53	1.45
N - Nitrosodimethylamine	bottom ash	GCMS(8270)	6/Bn	< 0.980		< 0.980	< 0.990	< 0.990	
N - Nitrosodimethylemine	bottom ash sluice water	GCMS(8270)	ոց/Լ	< 10.0		< 11.0	< 11.0	< 11.0	
N - Nitrosodimathylamine	collected fiv ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
N Nitrosodimethylamine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
N - Nitroscotimethylamine	high dust gas vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
N - Nitrosodimethylemine	sluice water sucolv	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
N - Nitrosodimethula mine	stack cas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
N-Nitrosodimethylamine N-Nitrosodimethylamine	stack das, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	bottom ash	GCMS(8270)	ug/a	< 0.980		< 0.980	< 0.990	< 0.990	
N - Nitrosodinhanulamina	hottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
N - Nitrosodinhaoulamine	collected fiv ash	GCMS(8270)	8/8n	< 0.990		< 0.990	< 0.970	< 1.00	
N - Nitrosodinhanvlamina	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
N - Nitrosodiohenvlamine	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
N - Nitrosodiohenvlemine	siuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	× 10.0	
N - Nitrosodinhanulamina	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	-		
N - Nitrosodiohenvlamine	stack gas. vepor phese	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Ovvdan	coal	Ultimate	%dry	5.41		5.17	5.41	5.20	5.34
o - Xviana	high dust gas. 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
o - Xviana	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
o - Dimethylaminazobenzene	bottom ash	GCMS(8270)	6/Bn	< 0.980		< 0.980	066.0 >	066.0 >	
o Dimethylamimazobenzene	bottom ash stuice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
p – Dimethylaminoszobenzene	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	× 1.00	
o - Dimethylaminoazobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
o - Dimethvlamiroszobenzene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			1
n - Dimethylaminoszobenzene	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	× 10 0	
n - Dimethylaminoszobenzene	stack gas, solid phase	GCMS(8270)	EmN/gm	< 0.00530	< 0.00555	< 0.00534			,
n Dimethvlaminoazobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pentachloronitrobenzene	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	066.0 >	
Pentachloronitrobenzene	bottom ash sluice water	GCMS(8270)	1/Bn	< 10.0		< 11.0	< 11.0	< 11.0	
Pentachloronitrobenzene	collected fiy ash	GCMS(8270)	8/8n	066.0 >		< 0.990	< 0.970	< 1.00	-
Pentachloronitrobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Pentachloronitrobenzene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Pentachloronitrobenzene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Pentachionitrobenzene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pentachtoronitrobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pentachtorophenol	bottom ash	GCMS(8270)	6/6n	< 4.90		< 4.90	< 4.90	< 5.00	
Pantachibrophenol	bottom ash sluice water	GCMS(8270)	ng/L	< 52.0	-	< 53.0	< 54.0	< 56.0	

Appendix B: Analytical Data Used in Calculations

19

i

Table B--1: Site 16 OFA Data USED in Calculations

30-Oct-93

Substance	Stream	Mothod	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Pentachtorophanol	collected fly ash	GCMS(8270)	6/6n	< 5.00		< 4.90	< 4.80	< 5.00	
Pentachlorophenol	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432			
Pentachtorophenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0290	< 0.0244	< 0.0251			
Pentachtorophenol	stuice water supply	GCMS(8270)	ug/L	< 52.0		< 52.0	< 53.0	< 52.0	
Pentachlorophenol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
Pentachtorophenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267	-		
Phenanthrene	bottom ash	GCMS(8270)	6/Bn	< 0.980		< 0.980	< 0.990	066:0 >	
Phenanthrene	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
Phenanthrene	collected fly ash	GCMS(8270)	6/6n	< 0.990		< 0.990	< 0.970	< 1.00	
Phenanthrene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Phenanthrene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Phenanthrene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Phenanthrene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Phenanthrene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
henol	bottom ash	GCMS(8270)	6/8n	< 0.980		< 0.960	< 0.990	< 0.990	
"herol	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
herol	collected fly ash	GCMS(8270)	6/8n	066.0 >	: ! 	066.0 >	070.0 >	< 1.00	
herol	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
henol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
henol	sluice water supply	GCMS(8270)	ug/t.	< 10.0		< 10.0	< 11.0	< 10.0	
henol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
ohenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	0.0138 @	< 0.00534			
Phosphate as P	bottom ash stuice water	Spectrophot.	mg/L			< 0.0500	< 0.0500	< 0.0500	< 0.0500
^o hosphate as P	stuice water suppty	Spectrophot.	mg/L			0.0535 @	0.0535 @	0.0519 @	0.0542 @
phosphorus	bottom ash	XRF	mg/kg	j		1266	1309	1353	742
hosphorus	coat	ASTM D2795	mg/kg, dry_			209	189	256	267
hosphorus	collected fly ash	XRF	mg/kg			1920	1833	1527	873
hosphorus	high dust gas, particulates	ICP-AES	mg/kg		1	1395		@ 692	872 @
hosphorus	high dust gas, solid phase	ICP-AES	mg/Nm3			9.28		3.97 @	4.77 @
hosphorus	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.141		< 0.137	< 0.116
hosphorus	stack gas, particulates	ICP-AES	mg/kg			1256 @		936 @	1217 @
hosphorus	stack gas, solid phase	ICP-AES	mg/Nm3			0.162 @		0.327 @	0.317 @
hosphorus	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.0963		< 0.111	< 0.120
yrene	bottom ash	GCMS(8270)	8/Bn	< 0.980		< 0.980	< 0.990	066.0 >	
yrene	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
yrene	collected fly ash	GCMS(8270)	6/8n	< 0.990		< 0.990	< 0.970	1.00	
yrene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863		-	
yrene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Viene	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	

69
1
U
0
ຮ

Table B-1: Site 16 OFA Data USED in Calculations

Cubetance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
	stack das solid ohase	GCMS(8270)	CmN/gm	< 0.00530	< 0.00555	< 0.00534			
-yrana		GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pyrene	Stack yas, variet press	HGAS	ma/ka			00.5 ×	< 5.00	< 5.00	× 5.00
Selenium						< 0.00500	< 0.0250	< 0.00500	< 0.00500
Selenium	Dottom asn siuice water		and day			3.88	3.84	3.24	4.21
Selenium	coat	NAA	Vin 6/Rn			25.0	12.8 @	6.11	14.9 @
Selenium	collected fly ash	HGAS	Bx/Bu			17.0		TUT	104
Selenium	high dust gas, particulates	HOAS	mg/kg					0.530	0.566
Selenium	high dust gas, solid phase	HGAS	mg/Nm3						0.00175.@
Selenium	high dust gas, vepor phase	HGAAS	mg/Nm3			< 0.00124		< 0.00101	
Salanium	stuice water supply	HGAAS	mg/L			< 0.00500	< 0.00500	00000 >	00000.0 >
Colonium	Istack gas, particulates	HGAAS	mg/kg			61.0			000
	stack cas. solid phase	HGAAS	mg/Nm3			0.00787			6660.0
	ctack ras vanor phase	HGAS	mg/Nm3		+	0.129		0.128	0.116
Seignum		XAF	ma/ka			1706	1410	1855	1855
Sodium		AFC	mo/l			5.52	5.55	7.05	6.79
Sodium	DOTTOM BSR SILICE WALE					268	358	266	260
Sodium	COB					2077	2003	1780	1706
Sodium	collected fly ash	XHr	By/B			3627 @R		3407 @R	< 1050 R
Sodium	high dust gas, particulates	ICP AES	mg/Kg		+-			47 6 @ B	5 74 B
Sodium	high dust gas, solid phase	ICP-AES	mg/Nm3			24.1 G			0 150
Sodium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.430		001.0	
Sodium	sluice water supply	ICP-AES	mg/L			- 10 - 6	9.06	8.64	
	stack das, particulates	ICP-AES	mg/kg			2523 @R		1920 @H	100 112 110 110 110 110 110 110 110 110
	stark nes solid bhase	ICP-AES	mg/Nm3			0.325 @R		0.670 @H	0.583 @H
Sogium		ICP-AES	ma/Nm3	,	:	< 0.366		< 0.384	< 0.341
Sodium	bidto gas, tapor press	GCMS	ma/Nm3	< 0.000525	< 0.000543	< 0.000559			
Styrene		Shoo	mo/Nm3	< 0.000580	< 0.000569	< 0.000542			
Styrene	stack gas, zu- L vost					35.1	35.6	43.5	42.4
Sultate	DONOM ASN SWICE WALET	2				117 H		110 R	145 R
Sulfate	high dust gas, solid phase	2 9	cuiv/Rui			4695		4598	4336
Sulfate	high dust gas, vapor phase	2	cinv/Bit			64 B	6.43	69.8	70.5
Sullate	sluice water supply	2	mg/r			0.10.0		15.0 8	14.3 B
Sullate	stack gas, solid phase	2	EmN/Bm					2005	ADER
Suttate	stack gas, vapor phase	2	mg/Nm3					VEDO O	
	bottom ash	Leco	×			00000 >	00000 >	5000.0	0,0034
Cuthur Cuthur	COR	Ultimate	%dry	1.78		1.58	41	1.59	.00
	collected fly set	Leco	*			0.197	0.224	0.197	0.220
Sullur		GOMS	ma/Nm3	< 0.000525	< 0.000543	< 0.000559			
Tetrachioroethere		ecus.	ma/Nm3	< 0.000580	< 0.000569	< 0.000542			1
Tetrachloroethene	Stack gas, zu-L VOS	XBF	ma/ka			8153	7554	7434	10312
Titanium						< 0.0500	< 0.0500	< 0.0500	< 0.0500
Titanium	bottom ash sivice water	ILL-160	7/6/11						

Appendix B: Analytical Data Used in Calculations

ŝ
D.
1
77
×
Ý.
1
Q
~

Table B--1: Site 16 OFA Data USED in Calculations

Substance	Stream	Method			HUN Z	Run J		+ UNH	
								-	
Titanium	coat	NAA	ug/g, dry			915	1099	694	714
Titanhum	collected fly ash	XRF	mg/kg			7794	8153	7854	8213
Tranium	high dust gas, particulates	ICP-AES	mg/kg			7128 R		6927 R	7474 R
Tiening	high dust gas, solid phase	ICP-AES	mg/Nm3			47.4 R	<u> </u>	35.B R	40.9 R
Thank	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.0217		< 0.0233	< 0.0225
Thanim	sluice water supply	ICP-AES	mg/L		_	< 0.0500	< 0.0500	< 0.0500	< 0.0500
Thaning	stack das, particulates	ICP-AES	mg/kg			4706		3908	4960
Tianing	stack gas, solid phase	ICP-AES	mg/Nm3			0.607		1.37	1.29
Tianium	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.0183		< 0.0192	< 0.0170
Toluane	high dust gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Toluene	stack gas, 20-L VOST	GCMS	CmN/gm	< 0.000580	< 0.000569	< 0.000542			
trans-13-Dichloropropene	high dust gas, 20-L VOST	GCMS	EmN/gm	< 0.000525	< 0.000543	< 0.000559			
trans-13-Dichloropropene	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
Trichloroethene	high dust gas, 20 - L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Trichloroethene	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			,
Vanadium	bottom ash	ICP-AES	mg/kg			213	211	213	205
Vanadium	bottom ash sluice water	ICP-AES	mg/L			< 0.0200	< 0.0200	< 0.0200	< 0.0200
Vanadium	coal	NAA	ug/g, dry			35.5	39.6	41.3	34.8
Vanadium	collected fly ash	ICP-AES	mg/kg			243	242	239	250
Vacadium	high dust gas, particulates	ICP-AES	mg/kg			303 R		298 R	315 H
Vanadium	high dust gas, solid phase	ICP-AES	mg/Nm3			2.02 R		1.54 R	1 72 A
Vanadium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.00869		< 0.00933	< 0.0000
Vanadium	sluice water supply	ICP-AES	∏/B⊞			< 0.0200	< 0.0200	< 0.0200	< 0.0200
Vanadium	stack gas, particulates	ICP-AES	mg/kg			390 R		295 R	383 R
Vanadium	stack gas, solid phase	ICP-AES	mg/Nm3			0.0504 R		0.103 R	0.0996 P
Vanadium	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.00733		< 0.00768	< 0.00681
Vinvl acetate	high dust gas, 20-L VOST	GCMS	EmN/Bm	< 0.00525	< 0.00543	< 0.00559			
Vinvl acetate	stack gas, 20 - L VOST	GCMS	mg/Nm3	< 0.00580	< 0.00569	< 0.00542			:
Vinvl chloride	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
Vinyl chloride	stack gas, 20-L VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542		1	
Volatiles	coal	Proximate	%dry	34.7		35.6	35.8	35.3	35.3

8	
1	
8	
Ō	
1	
N -	

Site 16 OFA/LNB Data USED in Calculations Table B-2:

÷*

		Mathod	I NON	Run 2	Run 2D	Run 3	Run 4
Substance							
+ 1 - Dicklorathene	Stack das. VOST	GCMS	ug/Nm3	< 0.568		< 0.588	< 0.578
1 1 - Dichlorosthene	Stack das. VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
1,1	Stack nas VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
1,1,1-1HCHIOLOGNIAIO	Stack nes VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
1, 1, 2 - Truthoroeniario	Stack das. VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
	Stack das VOST	GCMS	ua/Nm3	< 0.588		< 0.588	< 0.578
1,2 - Didillorougilizario	Stack das VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
1,2- Dichloropioparia	Stack das. VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
	Stack das. VOST	GCMS	ug/Nm3	< 2.94		< 2.94	< 2.89
5 - Dutantono 5 - mathyl chrysane	ESP inlet das	HRGCMS	ug/Nm3	< 0.00127		< 0.000345	< 0.00485
5 mathul chrusone	Stack das	HRGCMS	ug/Nm3	0.000280		< 0.000168	< 0.000978
7H_dihenzofo alcarbazole	ESP inlet das	HRGCMS	ug/Nm3	< 0.0186		< 0.0147	< 0.0781
7H	Stack das	HRGCMS	ug/Nm3	< 0.00511		< 0.00520	< 0.0180
Acarenthane	ESP iniet das	HRGCMS	ug/Nm3	0.00628		0.00485	0.0331
Aconsolutions	Stack das	HRGCMS	ug/Nm3	0.0178		0.00772	0.00254
Acenentitutere	ESP inlet cas	HRGCMS	ug/Nm3	0.00241		0.00518	0.00582
Accurately to the second	Stack ras	HRGCMS	ug/Nm3	0.00414		0.00268	0.00364
Acetaldahury iono Acetaldahuda	Stack das, vabor phase	HPLC	ug/Nm3	5.84 +		2.40 +	6.54 +
Aliminim	Bottom ash	ICP-AES	mg/kg, dry	108216	131791	122000	131000
Aliminum	Coal	NAA	mg/kg, dry	11965	11229	15393	14873
	ESP inter das particulate	ICP-AES	mg/kg	125126	122244	129353	131737
Automic	ESP intet das. solid phase	ICP-AES	ug/Nm3	647443	632533	737965	762468
Aliminim	ESP inlet das. vapor phase	ICP-AES	ug/Nm3	7.42 @		31.0 @	27.7 @
Aliminim	Stack das. particulate	ICP-AES	mg/kg	100789		98132	84195
Alimhum	Stack gas, solid phase	ICP-AES	ug/Nm3	11543		16616	11207
Aluminum	Stack gas, vapor phase	ICP-AES	ug/Nm3	14.2 @		7.51 @	19.9 @
Anthracana	ESP inlet gas	HRGCMS	ug/Nm3	0.00444		0.00459	0.00581
Anthracana	Stack das	HRGCMS	ug/Nm3	0.00691		0.00136	0.00442
Antimony	Bottom ash	ICP-AES	mg/kg, dry	< 18.9	< 19.0	22.6 @	< 18:3
Antimony	Coal	NAA	mg/kg, dry	1.33	1.40	1.78	1.28
Antimony	ESP inlet gas, particulate	ICP-AES	mg/kg	< 15.1	< 15.0	< 14.9	< 15.0
Antimony	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 78.3	< 77.8	< 85.1	< 86.6
Antimony	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 5.04		< 5.08	< 5.28
Antimony	Stack gas, particulate	ICP-AES	mg/kg	20.9 @		16.9	19.3

63
1
5
0
1
ଷ

	Ctreent	Method	NON	Run 2	Run 2D	Run 3	Run 4
Suusialice							
Antimonu	Stack das solid phase	ICP-AES	ug/Nm3	2.39 @		2.86	2.57
Antimony	Stack des vapor phase	ICP-AES	ug/Nm3	< 4.35		< 3.35	< 4.97
Antimory	Stack das <4 um	ICP-MS	ug/Nm3	0.785		0.690	0.656
	Stack cas 4-9 um	ICP-MS	ug/Nm3	0.472		0.396	0.363
Antimony	Stack das. >9 um	ICP-MS	ug/Nm3	0.735		1.14	0.766
Antimony	Stack particulate, <4 um	ICP-MS	mg/kg	83.0		87.4	93.6
Antimo.ov	Stack particulate, 4-9 um	ICP-MS	mg/kg	43.5		44.1	41.7
Antimony	Stack particulate, >9 um	ICP-MS	mg/kg	16.7		20.6	18.5
Arsenic	Bottom ash	GFAAS	mg/kg, dry	5.01	4.15 @	59.8	5.65
Arsenic	Coat	NAA	mg/kg. dry	23.4	42.7	22.1	23.9
Arsenic	ESP inlet gas, particulate	GFAAS	mg/kg	245	214	372	324
Arsenic	ESP inlet gas, solid phase	GFAAS	ug/Nm3	1269	1110	2123	1877
Atsenic	ESP inlet gas, vapor phase	GFAAS	ug/Nm3	< 0.137		< 0.138	0,802
Arsenic	Stack gas, particulate	GFAAS	mg/kg	935		624	1134
Arsanic	Stack gas, solid phase	GFAAS	ug/Nm3	107		106	151
Arsanic	Stack gas, vapor phase	GFAAS	ug/Nm3	2.03		1.06	2.13
Aranic	Stack gas, <4 um	ICP-MS	ug/Nm3	13.1		14.4	12.2
Areanio	Stack das, 4-9 um	ICP-MS	ug/Nm3	7.52		7.13	6.46
Arsanic	Stack gas, >9 um	ICP-MS	ug/Nm3	13.5		22.7	16.9
Arsenic	Stack particulate, <4 um	ICP-MS	mg/kg	1387		1826	1739
Areanic	Stack particulate, 4-9 um	ICP-MS	mg/kg	692		794	744
Arsanic	Stack particulate, >9 um	ICP-MS	mg/kg	308		410	407
Ach	Coal	Ultimate	%, dry	9.47	9.20	9.97	8.93
Barium	Bottom ash	ICP-AES	mg/kg, dry	831	1026	947	988
Barium	Coal	NAA	mg/kg, dry	94.5	100	117	86.0
Barium	ESP inlet gas, particulate	ICP-AES	mg/kg	1070	1042	1085	1088
Berium	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	5535	5392	6188	6296
Rarium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.111		0.663 @	0.109
Barium	Stack gas, particulate	ICP-AES	mg/kg	1265		1135	1099
Barium	Stack gas, solid phase	ICP-AES	ug/Nm3	145		192	146
Barium	Stack gas, vapor phase	ICP-AES	ug/Nm3	0.112 @		< 0.0740	0.323 @
Barium	Stack gas, <4 um	ICP-MS	ug/Nm3	19.4		14.8	15.0
Barium	Stack gas, 4–9 um	ICP-MS	ug/Nm3	19.6		15.2	14.8
Barium	Stack gas, >9 um	ICP-MS	ug/Nm3	52.4		76.4	54.9

g
3
8
Ţ
ଷ୍ଣ

B-26

ო

Substance	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
Barium	Stack particulate, <4 um	ICP-MS	mg/kg	2047		1876	2137
Barium	Stack particulate, 4-9 um	ICP-MS	mg/kg	1806		1692	1700
Barium	Stack perticulate, >9 um	ICPMS	mg/kg	1192		1381	1324
Benzene	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.521	0.877 @
Benzo[a] pyrene	ESP inlet gas	HRGCMS	ug/Nm3	0.0229		< 0.00357	< 0.00678
Benzola]pyrene	Stack gas	HRGCMS	ug/Nm3	0.00283		< 0.00253	< 0.00477
Benzo(b,j&k)fluoranthenes	ESP inlet gas	HRGCMS	ug/Nm3	< 0.00848		0.000699	< 0.0316
Benzo(b,j&k)fluoranthenes	Stack gas	HRGCMS	ug/Nm3	0.00373		0.000588	0.000782
Benzo[ghi]perylene	ESP inlet gas	HRGCMS	cmN/gu	< 0.00272		< 0.00371	< 0.0407
Benzolghijperylene	Stack gas	HRGCMS	ug/Nm3	< 0.00361		< 0.00184	< 0.00193
<u>Benz</u> [a]anthracene	ESP inlet gas	HRGCMS	ug/Nm3	0.00114		0.00197	< 0.0149
Benz[a]anthracene	Stack gas	HRGCMS	emN/gu	0.0228		0.000633	0.000770
Beryllium	Bottom ash	ICP-AES	mg/kg, dry	19.7	20.8	22.9	23.1
Beryllium	Coal	ICP-AES	mg/kg, dry	1.90	2.10	2.40	2.40
Beryllium	ESP inlet gas, particulate	ICP-AES	mg/kg	21.3	18.0	23.7	22.8
Beryllium	ESP intet gas, solid phase	ICP-AES	ug/Nm3	110	93.3	135	132
Beryllium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.116		0.213 @	0.238 @
Beryllium	Stack gas, particulate	ICP-AES	mg/kg	23.0		24.8	25.5
Beryllium	Stack gas, solid phase	ICP-AES	ug/Nm3	2.64		4.21	3.39
Beryllium	Stack gas, vapor phase	ICP-AES	ug/Nm3	0.278 @		0.101 @	< 0.114
Berylilum	Stack gas, <4 um	ICP-MS	ug/Nm3	0.227		0.259	0.204
Beryllum	Stack gas, 4-9 um	ICP-MS	ug/Nm3	0.417		0.368	0.341
Beryllium	Stack gas, >9 um	ICP-MS	ug/Nm3	0.958		1.38	1.18
Beryllium	Stack particulate, <4 um	ICP-MS	mg/kg	24.0		32.7	29.1
Beryllium	Stack particulate, 4-9 um	ICP-MS	mg/kg	38.3	_	41.0	39.2
Beryllium	Stack particulate, >9 um	ICP-MS	mg/kg	21.8		24.9	28.4
Bromolotm	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Bromomethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		1 89 @	< 0.578
Cadmium	Bottom ash	GFAAS	mg/kg, dry	< 0.998	< 1.00	< 0.922	< 0.963
Cadmium	Coal	NAA	mg/kg, dry	< 2.61	< 2.62	< 2.61	< 2.58
Cadmium	ESP inlet gas, particulate	GFAAS	mg/kg	1.34 @		6.22	3.18@
Cadmium	ESP inlet gas, solid phase	GFAAS	ug/Nm3	6.94 @		35.5	18.4 @
Cadmium	ESP inlet gas, vapor phase	GFAAS	ug/Nm3	0.0715 @		0.179	1.31
Cadmium	Stack gas, particulate	GFAAS	mg/kg	25.2 @		27.9 @	36.3

Substance	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
Cadmium	Stack gas, solid phase	GFAAS	ug/Nm3	2.89 @		473@	4.83
Cadmium	Stack gas, vapor phase	GFAAS	ug/Nm3	< 0.0581		< 0.0447	< 0.0663
Cadmium	Stack gas, <4 um	ICP-MS	ug/Nm3	0.0604		0.0454	0.0607
Cadmium	Stack gas, 4~9 um	ICP-MS	ug/Nm3	0.0454		0.0657	0.0316
Cadmium	Stack gas, ≥9 um	ICP-MS	ug/Nm3	0.142		0.284	0.0703
Cadmium	Stack particulate, <4 um	ICP-MS	mg/kg	6.38		5.75	8.67
Cadmium	Stack particulate, 4-9 um	ICP-MS	mg/kg	4,18		7.31	3.64
Cadmium	Stack particulate, >9 um	ICP-MS	mg/kg	3.24		5.14	1.70
Calcium	Bottom ash	ICP-AES	mg/kg, dry	4008 @	5573	4560 @	5130
Calcium	Coal	NAA	mg/kg, dry	1812	1509	1698	6287
Calcium	ESP intet gas, particutate	ICP-AES	mg/kg	5570	5541	5602	5619
Celcium	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	28822	28671	31960	32520
Calcium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 31.5		31.0	27.0
Calcium	Stack gas, particulate	ICP-AES	mg/kg	6778		6118	6201
Calcium	Stack gas, solid phase	ICP-AES	ug/Nm3	776		1036	825
Celcium	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 27.2		< 21.0	< 31.1
Carbon	Coal	Ultimate	%, dry	5.77	77.3	76.7	77.6
Carbon Disulfide	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	0.611@
Carbon Tetrachloride	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Chloride	Bottom ash	Pot. Titr.	mg/kg, dry	< 100.0	< 99.8	< 98.2	8.66 >
Chloride	Coal	Pot. Titr.	mg/kg, dry	432 @	346 @	276 @	311@
Chloride	ESP intet gas, particulate	Q	mg/kg	2011		2777	1970
Chloride	ESP intet gas, solid phase	Q	ug/Nm3	10380		15863	11190
Chloride	ESP inlet gas, vapor phase	<u>ں</u>	ug/Nm3	14439		10354	13967
Chloride	Stack gas, perticulate	Q	mg/kg	3455		1350	1502
Chloride	Stack gas, solid phase	Q	ug/Nm3	399		229	200
Chloride	Stack gas, vapor phase	<u>0</u>	ug/Nm3	16950		17009	18910
Chlorobenzene	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Chloroethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Chloroform	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Chloromethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Chromium	Bottom ash	ICP AES	mg/kg, dry	102	109	113	113
Chromium	Coel	NAA	mg/kg, dry	14.7	16.2	21.0	15.2
Chromium	ESP inlet gas, particulate	ICP-AES	mg/kg	146	128	166	141

Table B-2: Site 16 OFA/LNB Data USED in Calculations

29-Oct-93

Gubatanca	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
Chromium	ESP inlet das, solid phase	ICP-AES	ug/Nm3	757	664	948	814
Chromium	ESP intet cas. vepor phase	GFAAS	ug/Nm3	1.26		3.01	3.54
Chromitum	Stack case perficulate	ICP-AES	ma/ka	170		157	173
Chromitim	Stack das. solid phase	ICP-AES	ug/Nm3	19.5		26.6	23.0
Chromium	Stack das, vapor phase	GFAAS	ug/Nm3	0.962		0.768	1.06
Chromium	Stack cas. <4 um	ICP-MS	ug/Nm3	3.72		3.89	3.01
Chromium	Stack das. 4-9 um	ICP-MS	ug/Nm3	4.14		3.57	3.22
Chromium	Stack das, >9 um	ICP-MS	ug/Nm3	10.6		12.6	15.2
Chromium	Stack particulate, <4 um	ICP-MS	mg/kg	393		493	429
Chromium	Stack particulate, 4-9 um	ICP-MS	mg/kg	381		397	371
Chromium	Stack particulate, >9 um	ICP-MS	mg/kg	242		228	367
Chromium (VI)	Stack das	BIF Cr6	ug/Nm3	6.77 R+		2.61 @R+	9.32 R
Chrvana	ESP injet das	HECMS	ug/Nm3	0.00546		0.00169	0.00627
Chrvana	Stack das	HRGCMS	ug/Nm3	0.00357		0.00162	D.000905
cis-13-Oichioropropene	Stack das, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Cohalt	Bottom ash	ICP-AES	mg/kg, dry	61.6 @	57.4 @	57.0 @	82.8 @
Cobat	Coal	NAA	mg/kg, dry	5.95	6.27	6.84	5.45
Cobalt	ESP inlet gas, particulate	ICP-AES	mg/kg	37.8 @	42.6 @	59.1 @	55.8 @
Cohalt	ESP inlet gas, solid phase	ICP-AES	emN/gu	196 @	220 @	337 @	323 @
Cohalt	ESP inlet das. vapor phase	ICP-AES	ug/Nm3	< 0.715		< 0.719	< 0.747
Cohalt	Stack das, particulate	ICP-AES	mg/kg	55.5		52.8	53.9
Cobalt	Stack das, solid phase	ICP-AES	em//Bn	6.36		8.93	7.18
Cohalt	Stack das, vapor phase	ICP-AES	ug/Nm3	< 0.617		< 0.475	< 0.704
Cobalt	Stack das, <4 um	ICP-MS	6mN/Bu	0.527		0.539	0.407
Cobalt	Stack das, 49 um	ICP-MS	cm//Bn	1.06		0.875	0.815
Cobalt	Stack das, >9 um	ICP-MS	ug/Nm3	2.53		3.25	2.93
Cobalt	Stack particulate, <4 um	ICP-MS	mg/kg	55.8		68.3	58.1
Cobalt	Stack particulate, 4–9 um	ICP-MS	mg/kg	97.9		97.4	93.8
Cobalt	Stack particulate, >9 um	ICP-MS	mg/kg	57.6		58.8	70.8
Copper	Bottom ash	ICP-AES	mg/kg, dry	120	128	132	144
Conner	Coal	NAA	mg/kg, dry	28.8	34.1	32.5	40.0
Copper	ESP inlet gas, particulate	ICP-AES	mg/kg	189	193	223	196
Copper	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	976	1001	1272	1132
Copper	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.799		1.12	0.585

ŝ

Copper Stack gas, periculate CP – AES ug/km3 252 Copper Stack gas, polid phase CP – AES ug/km3 268 Copper Stack gas, a fold phase CP – AES ug/km3 268 Copper Stack gas, a fold phase CP – AES ug/km3 268 Copper Stack gas, 4 – 9 um CP – MS ug/km3 7.56 Copper Stack gas, 4 – 9 um CP – MS ug/km3 7.26 Copper Stack gas, 4 – 9 um CP – MS ug/km3 7.26 Copper Stack gas, 4 – 9 um CP – MS ug/km3 7.26 Copper Stack gas HROCMS ug/km3 6.00032 Dibenzol(a)[Privene ESP inite gas HROCMS ug/km3 < 0.0023 Dibenzol(a)[Privene ESP inite gas HROCMS ug/km3 < 0.0032 Dibenzol(a)[Privene ESP inite gas HROCMS ug/km3 < 0.0033 Dibenzol(a)[Privene ESP inite gas HROCMS ug/km3 < 0.0033 Dibenzol(a)[Pri	Substance	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
Copper Stack gas, and phase ICP-AES ug/Mm3 252 Copper Stack gas, and phase ICP-AES ug/Mm3 26.0 Copper Stack gas, and phase ICP-AES ug/Mm3 2.6.0 Copper Stack gas, 4-9 um ICP-MS ug/Mm3 7.6.5 Copper Stack gas, 4-9 um ICP-MS ug/Mm3 7.04 Copper Stack gas IROCMS ug/Mm3 < 0.00239								
Copper Stack gas, solid phase ICP – AES ug/Mm3 28.9 Copper Stack gas, 4-9 um ICP – AES ug/Mm3 3.9 Copper Stack gas, 4-9 um ICP – MS ug/Mm3 5.6 Copper Stack gas, 4-9 um ICP – MS ug/Mm3 7.6 Copper Stack gas, 9-9 um ICP – MS ug/Mm3 7.6 Copper Stack gas, 9-9 um ICP – MS ug/Mm3 7.6 Copper Stack gas, 9-9 um ICP – MS ug/Mm3 7.6 Copper Stack gas Pum ICP – MS ug/Mm3 7.6 Copper Stack gas HRGCMS ug/Mm3 5.000231 Dibenzole, IPyrene ESP indet gas HRGCMS ug/Mm3 5.00136 Dibenzole, IPyrene	Copper	Stack gas, particulate	ICP-AES	mg/kg	252		253	252
Corper Stack gas, vapor phase CP – AES ug/Mm3 < 0689 Copper Stack gas, 4 um CP – MS ug/Mm3 7.65 Copper Stack gas, 4 um CP – MS ug/Mm3 7.65 Copper Stack gas, 4 um CP – MS ug/Mm3 7.65 Copper Stack particulate, 4 - 9 um CP – MS mg/kg 7.64 Copper Stack particulate, 4 - 9 um CP – MS mg/kg 7.65 Copper Stack particulate, 4 - 9 um CP – MS mg/kg 7.64 Copper Stack gas HR3CMS ug/Mm3 <0.00231	Copper	Stack gas, solid phase	ICP-AES	ug/Nm3	28.9		42.8	33.5
Copper Stack ges, 4-4 um ICP - MS Ug/Mm3 3.98 Copper Stack ges, 4-9 um ICP - MS Ug/Mm3 7.65 Copper Stack parten.let, 4 um ICP - MS Ug/Mm3 7.65 Copper Stack parten.let, 4 um ICP - MS Ug/Mm3 7.65 Copper Stack parten.let, 4 um ICP - MS mg/kg 7.64 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Copper Stack particulate, 4 um ICP - MS mg/kg 7.04 Dibenzole, Dipenzel, Dipenzene ESP inlet gas HRGCMS Ug/Mm3 <0.00331	Capper	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 0.689		< 0.531	< 0.787
Copper Saek gas, 4 – 9 um ICP – MS ug/Mm3 7/55 Copper Stack gan – 9 um ICP – MS ug/Mm3 7/55 Copper Stack gan loudiate, < 4 um	Conner	Stack oes. <4 um	ICP-MS	ug/Nm3	3.98		3.15	2.49
CopperStack gas, > 9 umCP - MSwg/Mm372.6CopperStack particulate, < 4 um	Copper	Stack das, 4-9 um	ICP-MS	ug/Nm3	7.65		3.64	3.51
Copper CopperStack particulate, <1 um benzole(CP – MS)mg/kg421CopperStack particulate, <1 e um Stack particulate, <1 e um	Conner	Stack das. >9 um	ICP-MS	ug/Nm3	72.6		34.1	17.7
Copper Stack particulate, 4 9 um CP-MS mg/kg 704 Copper Stack particulate, 5 um ICP-MS mg/kg 1654 Copper Stack particulate, 5 um ICP-MS ug/hm3 < 0.00232	Conner	Stack particulate. <4 um	ICP-MS	mg/kg	421		399	356
Opper Sack particulate, > 9 um RCP-MS mg/kg 1654 Copper ESP inlet gas HRGCMS ug/hm3 < 0.00231	Conner	Stack particulate, 4-9 um	ICP-MS	mg/kg	704		405	404
DistrictESP inlet gasHFIGCMSug/Nm3< 0.00232Dismrzola, elpyreneStack gasHRGCMSug/Nm3< 0.00231	Conner	Stack particulate. >9 um	ICP-MS	mg/kg	1654		616	427
Diberzole, al pyreme Stack gas HRGCMS ug/hm3 < 0.00231 Diberzole, hlpyreme ESP intet gas HRGCMS ug/hm3 < 0.00239	Obenzola el ovrene	ESP inlet gas	HRGCMS	ug/Nm3	< 0.00232		< 0.00146	< 0.0116
Diberzola, IhPyrene ESP inlat gas HRGCMS ug/hm3 < 0.00329 Diberzola, IhPyrene Stack gas HRGCMS ug/hm3 < 0.00329	Dibenzofa.e) Dvrene	Stack gas	HRGCMS	ug/Nm3	< 0.00231		< 0.000953	< 0.00342
Dibenzo(a.h)[pyreneStack gasHRGCMSug/Nm3< 0.00368Dibenzo(a.i)[pyreneESP inlet gasHRGCMSug/Nm3< 0.00324	Dibenzofa.h]bvrene	ESP inlet gas	HRGCMS	ug/Nm3	< 0.00329		< 0.00269	< 0.0581
Dibenzola.iJpyreneESP inlet gasHRGCMSug/Nm3< 0.00424Dibenzola.iJpyreneStack gasHRGCMSug/Nm3< 0.00153	Dihenzofa hinvrene	Stack das	HRGCMS	ug/Nm3	< 0.00368		< 0.000765	< 0.00344
Dibenzolesi.Macrolesi.Esck gaaMRGCMSug/Mm3< 0.00331Dibenzolesi.EsP inlet gaaHRGCMSug/Mm3< 0.00163	Dibenzola, ilovrene	ESP intet gas	HRGCMS	ug/Nm3	< 0.00424		< 0.00255	< 0.0181
Dibenz(a,h)acridineESP inlet gasHRGCMSug/Mm3< 0.0165Dibenz(a,h)acridineStack gasHRGCMSug/Nm3< 0.00163	Dihanzofa ilovrana	Stack das	HRGCMS	ug/Nm3	< 0.00331		< 0.00110	< 0.00491
Dibenz[a,h]actridineStack gasHRGCMSug/Nm3< 0.00163Dibenz[a,h]anthraceneESP inlet gasHRGCMSug/Nm3< 0.00163	Dibanzia hlacridine	ESP inter gas	HRGCMS	ug/Nm3	< 0.0106		< 0.00271	< 0.0320
Dibenz(a,h)anthraceneESP inlet gasHRGCMSug/Nm3< 0.00153Dibenz(a,h)anthraceneStack gasHRGCMSug/Nm3< 0.00148	Dibenz fa. hlacridine	Stack gas	HRGCMS	ug/Nm3	< 0.00183		< 0.00157	< 0.00180
Dibenz (a. h) antificationStack gasHRGCMSug/Mm3< 0.00148Dibenz (a. h) antificationESP inlet gasHRGCMSug/Mm3< 0.00148	Dibenzia blanthracene	ESP inlet ges	HRGCMS	ug/Nm3	< 0.00153		< 0.000868	< 0.0527
Dibenz[a,]acridineESP intet gasHRGCMSug/Mm3< 0.00798Dibenz[a,]acridineStack gasHRGCMSug/Mm3< 0.00482	Olbenzia hlanthracane	Stack gas	HRGCMS	ug/Nm3	< 0.00148		< 0.000893	< 0.00433
Dibenz[a.]]acridineStack gasHRGCMSug/Mm3< 0.00482Dibenz[a.]]acridineStack gas, VOSTGCMSug/Nm3< 0.588	Dihanzla ilacridina	ESP inter gas	HRGCMS	ug/Nm3	< 0.00798		< 0.00296	< 0.0218
Ethyl BenzeneStack gas, VOSTGCMSug/Nm3< 0.588Fixed CarbonCoal%, dry58.258.2FluorantheneCoalProximate%, dry58.2FluorantheneESP inlet gasHRGCMSug/Nm30.0259FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gasHRGCMSug/Nm30.0236FluorantheneStack gas, particutateISEmg/kg, dry114FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, solid phaseISEug/Nm35239FluorideESP inlet gas, vapor phaseISEug/Nm35239FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm35239FluorideESP inlet gas, particutateISEug/Nm35239FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm35239FluorideESP inlet gas, particutateISEug/Nm35239FluorideESP inlet gas, particutateISEug/Nm35239	Olbenzfe ilacridine	Stack das	HRGCMS	ug/Nm3	< 0.00482		< 0.00155	< 0.00114
Fixed CarbonCoalProximate%, dry58.2*FluorantheneESP inlet gasHRGCMSug/Mm30.0240FluorantheneESP inlet gasHRGCMSug/Mm30.0259FluorantheneStack gasHRGCMSug/Mm30.0259FluoreneStack gasHRGCMSug/Mm30.0149FluoreneStack gasHRGCMSug/Mm30.0149FluoreneStack gasHRGCMSug/Mm30.0149FluoreneStack gasIRGCMSug/Mm30.0236FluorideEsP inlet gas, particutateISEmg/kg, dry30.6 (g)FluorideESP inlet gas, particutateISEug/Mm36.82FluorideESP inlet gas, vapor phaseISEug/Mm35239FluorideESP inlet gas, vapor phaseISEug/Mm35239FluorideESP inlet gas, vapor phaseISEug/Mm35239FluorideESP inlet gas, particutateISEug/Mm35239FluorideESP inlet gas, particutateISEug/Mm35239	Filvy Benzene	Stack gas, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
FluorantheneESP inlet gasHRGCMSug/Nm30.0240FluorantheneStack gasHROCMSug/Nm30.0149FluorantheneStack gasHRGCMSug/Nm30.0149FluoreneStack gasHRGCMSug/Nm30.0149FluoreneStack gasHRGCMSug/Nm30.0149FluoreneStack gasHRGCMSug/Nm30.0236FluorideBottom ashISEmg/kg. dry30.6 (g)FluorideCoalISEmg/kg. dry114FluorideESP inlet gas, particulateISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm35239FluorideESP inlet gas, vapor phaseISEug/Nm35239FluorideESP inlet gas, vapor phaseISEug/Nm35239FluorideESP inlet gas, particulateISEug/Nm35239FluorideESP inlet gas, particulateISEug/Nm35239FluorideESP inlet gas, particulateISEug/Nm35239FluorideESP inlet gas, particulateISEug/Nm35239FluorideISEISEISEmg/kg687	Fixed Carbon	Coal	Proximate	%, dry	58.2	58.1	57.7	58.9
FluorantheneStack gasHR3CMSug/Nm30.0259FluorantheneESP inlet gasHR3CMSug/Nm30.0149FluoreneESP inlet gasHR3CMSug/Nm30.0136FluoreneStack gasHR3CMSug/Nm30.0236FluorideBottom ashISEmg/kg. dry30.6 @FluorideCoalISEmg/kg. dry114FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm3682FluorideESP inlet gas, particutateISEug/Nm35239FluorideISEmg/kgISEug/Nm35239	Fluoranthana	ESP inlet gas	HRGCMS	ug/Nm3	0.0240		0.00608	0.00852
FluoreneESP inlet gasHRGCMSug/Nm30.0149FluoreneStack gasHRGCMSug/Nm30.0236FluorideBottom ashISEmg/kg. dry30.6 (g)FluorideCoal15Emg/kg. dry11481FluorideESP inlet gas, particutateISEmg/kg132FluorideESP inlet gas, solid phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, vapor phaseISEug/Nm3682FluorideESP inlet gas, particulateISEug/Nm3682FluorideESP inlet gas, particulateISEug/Nm3682	Fluoranthene	Stack gas	HRGCMS	ug/Nm3	0.0259		0.00447	0.00493
FluoreneStack gasHRGCMSug/Nm30.0236FluorideBottom ashISEmg/kg. dry30.6 @29FluorideCoalISEmg/kg. dry11481FluorideESP inlet gas, particutateISEmg/kg13281FluorideESP inlet gas, solid phaseISEug/Nm368281FluorideESP inlet gas, vapor phaseISEug/Nm35239847FluorideStack gas, particutateISEug/Nm3847	Fluorene	ESP inlet gas	HRGCMS	ug/Nm3	0.0149		0.00492	0.00609
FluorideBottom ashISEmg/kg, dry30.6 @29.FluorideCoal11481FluorideESP inlet gas, particulateISEmg/kg, dry11481FluorideESP inlet gas, particulateISEmg/kg13281FluorideESP inlet gas, solid phaseISEug/Nm3682823FluorideESP inlet gas, vapor phaseISEug/Nm35239847FluorideStack gas, particulateISEmg/kg847	Fluorene	Stack gas	HRGCMS	ug/Nm3	0.0236		0.00474	0.00587
Fluoride Coal ISE mg/kg. dry 114 81 Fluoride ESP inlet gas, particulate ISE mg/kg 132 81 Fluoride ESP inlet gas, solid phase ISE ug/Nm3 682 82 Fluoride ESP inlet gas, vapor phase ISE ug/Nm3 5239 847 Fluoride Stack gas, particulate ISE ug/Nm3 847 847	Fluoride	Bottom ash	ISE	mg/kg, dry	30.6 @	29.9 @	17.4 @	22.6 @
Fluoride ESP inlet gas, particulate ISE mg/kg 132 Fluoride ESP inlet gas, solid phase ISE ug/Nm3 682 Fluoride ESP inlet gas, vapor phase ISE ug/Nm3 682 Fluoride ESP inlet gas, vapor phase ISE ug/Nm3 5239 Fluoride ISE mg/kg 847	Fluoride	Coal	ISE	mg/kg, dry	114	81.9	80.9	78.6
Fluoride ESP inlet gas, solid phase ISE ug/Nm3 682 Fluoride ESP inlet gas, vapor phase ISE ug/Nm3 5239 Fluoride ISE ug/Nm3 5239 847 Fluoride ISE mg/kg 847	Fhioride	ESP inlet gas, particulate	ISE	mg/kg	132		314	225
Fluoride ESP inlet gas, vepor phase ISE ug/Nm3 5239 Fluoride Stack gas, particulate ISE mg/kg 847	Fluorida	ESP intet gas, solid phase	ISE	ug/Nm3	682		1793	1276
Fluoride Stack gas, particulate ISE mg/kg 847	Fluoride	ESP inlet gas, vapor phase	ISE	ug/Nm3	5239		3961	4466
	Fluoride	Stack gas, particulate	ISE	mg/kg	847		379	576
Elinorida Stack das. solid phase ISE ug/Nm3 97.8	Fluorida	Stack das, solid phase	ISE	ug/Nm3	97.8		64.4	76.8

Table B-2: Site 16 OFA/LNB Data USED in Calculations

29-Oct-93

g

83
Ĩ
ő
6
N

Table B-2: Site 16 OFA/LNB Data USED in Calculations

				C C		0 0	0 4
Substance	Stream	Method	MON	Kun z			
Fluoride	Stack gas, vapor phase	ISE	ug/Nm3	5520		5704	6123
Formaldehvde	Stack gas, vapor phase	HPLC	ug/Nm3	2.72 +		0.544 +	1.37 +
HHV	Coal	Proximete	Btu/Ib, dry	13728	13779	13730	13889
Hudronen	Coal	Ultimate	%, dry	4.93	4.97	4.65	4.94
Indenoit 23-collovrene	ESP inter das	HRGCMS	ug/Nm3	< 0.00172		< 0.00314	< 0.0212
Indeno[1 2 3-cd]pyrene	Stack das	HRGCMS	ug/Nm3	< 0.00311		0.000738	< 0.00284
	Bottom ash	ICP-AES	mg/kg, dry	102204	113682	112000	117000
Iron	Coal	NAA	mg/kg, dry	14100	10369	10145	9522
lon	ESP intet gas, particulate	ICP-AES	mg/kg	94551	98196	90746	92914
lion	ESP intet cas, solid phase	ICP-AES	ug/Nm3	489237	508100	517711	537771
	ESP intet gas, vapor phase	ICP-AES	ug/Nm3	6.87 @		35.5 @	22.2 @
lton	Stack gas, particulate	ICP-AES	mg/kg	85013		74112	76884
Iton	Stack gas, solid phase	ICP-AES	ug/Nm3	9737		12549	10234
lion	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 4.35		< 3.35	< 4.97
tead	Bottom ash	GFAAS	mg/kg, dry	13.7	13.7	15.0	14.1
laad	Coal	ICP-AES	mg/kg, dry	8.00	7.00	8.00	6.00
l aad	ESP inlet gas, particulate	GFAAS	mg/kg	56.1		66.1	61.5
Lead	ESP intet gas, sofid phase	GFAAS	ug/Nm3	290		377	356
	ESP intet gas, vapor phase	GFAAS	ug/Nm3	2.10		2.14 @	5.18 @
	Stack gas, particulate	GFAAS	mg/kg	91.1		90.7	102
	Stack gas, solid phase	GFAAS	ug/Nm3	10.4		15.4	13.6
laad	Stack gas, vapor phase	GFAAS	ug/Nm3	0.308 @		< 0.154	< 0.228
lead	Stack gas, <4 um	ICP-MS	ug/Nm3	3.02		1.61	1.54
Lead	Stack gas, 4–9 um	ICP-MS	ug/Nm3	2.94		1.81	1.44
Lead	Stack gas, >9 um	ICP-MS	ug/Nm3	5.13		6.35	3.80
Lead	Stack particulate, <4 um	ICP-MS	mg/kg	320		204	220
Lead	Stack particulate, 4–9 um	ICP-MS	mg/kg	271		201	165
Lead	Stack particulate, >9 um	ICP-MS	mg/kg	117		115	91.6
Mandanese	Bottom ash	ICP-AES	mg/kg, dry	160	171	148	146
Manganese	Coat	NAA	mg/kg, dry	14.4	13.4	13.9	14.5
Mandanese	ESP inlet gas, particulate	ICP-AES	mg/kg	181	185	149	151
Mangenese	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	935	959	851	872
Manganese	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	0.572		1.04	1.81
Mandanese	Stack gas, particulate	ICP-AES	mg/kg	194		155	166

~

Manganese Manganese Manganese Manganese Manganese						7 1171
Manganese Manganese Manganese Manganese Manganese						
Manganese Manganese Manganese Manganese Mandanese	Stack gas, solid phase	ICP-AES	ug/Nm3	22.2		26.2
Menganese Manganese Manganese Manganese	Stack gas, vapor phase	ICP-AES	ug/Nm3	0.528		0.135@
Manganese Manganese Manganese	Stack gas, <4 um	ICP-MS	ug/Nm3	1.90	-	1.81
Manganese Manganese	Stack gas, 4-9 um	ICP-MS	ug/Nm3	3.27		2.23
Manganese	Stack gas, >9 um	ICP-MS	ug/Nm3	10.3		15.4
	Stack particulate, <4 um	ICP-MS	mg/kg	201		230
Mandanese	Stack particulate, 4–9 um	ICP-MS	mg/kg	301		248
Mancanese	Stack particulate, >9 um	ICP-MS	mg/kg	234		279
Mercurv	Bottom ash	CVAAS	mg/kg, dry	< 0.0120	< 0.0121	< 0.0120
Mercurv	Coal	DGA/CVAAS	mg/kg, dry	0.160	0.110	0.150
Mercurv	ESP inlet gas, particulate	CVAAS	mg/kg	0.626	0.782	0.836
Marcurv	ESP inlet gas, solid phase	CVAS	ug/Nm3	3.24	4.04	4.77
Marcurv	ESP inlet gas, vapor phase	CVAS	ug/Nm3	5.18		6.90
Marcurv	Stack das, particulate	CVAS	mg/kg	2.52		0.710
Marcurv	Stack das. solid phase	CVAS	ug/Nm3	0.289		0.120
Mercurv	Stack gas, vapor phase	CVAS	ug/Nm3	4.41		5.26
Mercurv	Stack gas, <4 um	ICP-MS	ug/Nm3	0.0246		0.0219
Marcurv	Stack gas, 4-9 um	ICP-MS	ug/Nm3	0.0350		0.0200
Mercurv	Stack gas, >9 um	ICP-MS	ug/Nm3	0.107		0.310
Mercury	Stack particulate, <4 um	ICP-MS	mg/kg	2.60		2.77
Mercurv	Stack particulate, 4-9 um	ICP-MS	mg/kg	3.22		2.23
Mercury	Stack particulate, >9 um	ICP-MS	mg/kg	2.43		5.60
Mercury (0)	Stack gas, vapor phase	Bloom	ug/Nm3	4.07		3.44
Mercury (II)	Stack gas, vapor phase	Bloom	ug/Nm3	2.40		3.30
Methyl Mercury	Stack gas, vapor phase	Bloom	ug/Nm3	0.161		0.965
Methylene Chloride	Stack gas, VOST	GCMS	ug/Nm3	6.93 @B		1.07 @B
Moisture	Coal	Ultimate	ሄ	4.32	4.69	4.04
Molvbdenum	Bottom ash	ICP-AES	mg/kg, dry	< 12.9	< 13.0	27.4 @
Molvbdenum	Coal	NAA	mg/kg, dry	3.19	3.34	5.02
Molvbdenum	ESP inlet gas, particulate	ICP-AES	mg/kg	< 10.5	< 10.4	18.9 @
Molybdenum	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 54.3	< 53.9	108 @
Molybdenum	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.967		< 0.973
Molybdenum	Stack gas, particulate	ICP-AES	mg/kg	115		87.0
Molvbdenum	Stack gas, solid phase	ICP-AES	ug/Nm3	13.2		14.7

¥

0.00834

6.37

0.0660 1.19 1.16 1.59 3.28

0.110 0.898

< 0.0120

177 241 221

5.20

6.81 1.31 0.174

Site 16 OFA/LNB Data USED in Calculations Table B–2:

29-Oct-93

ω

0.240 @

1.24 2.09 9.18

22.0

Run 4

2.11 0.201 39.7 B

21.4 @ 124 @ < 1.01

106 14.1

38.7 @

< 3.61

3.02

Substance	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
Molubdanim	Stack das vapor phase	ICP-AES	ug/Nm3	< 0.835		< 0.642	< 0.953
	Stack nas <4 um	ICP-MS	ua/Nm3	2.23		2.22	2.06
Methodenum	Stack das 4–9 um	ICP-MS	ug/Nm3	0.700		0.560	0.484
	Stack das >9 um	ICP-MS	ug/Nm3	1.15		1.60	1.20
	Stack particulate. <4 um	ICP-MS	mg/kg	236		281	294
	Stack perticulate, 4-9 um	ICP-MS	mg/kg	64.4		62.3	55.7
Motvbdenum	Stack particulate, >9 um	ICP-MS	mg/kg	26.2		28.9	29.0
	Stack gas, VOST	GCMS	cmN/gu	< 0.588		< 0.588	< 0.571
Nickel	Bottom ash	GFAS	mg/kg, dry	80.1	85.2	81.7	83.0
Nickel	Coal	NAA	mg/kg, dry	20.6	18.8	18.9	117
Nickal	ESP inlet gas, particulate	GFAAS	mg/kg	123	106	118	117
Nickel	ESP inlet gas, solid phase	GFAS	ug/Nm3	637	550	676	676
Nickel	ESP inlet gas, vapor phase	GFAAS	ug/Nm3	0.609 @		1.81 @	2.41 @
Nickal	Stack das, particulate	GFAAS	mg/kg	144		124	140
Nickel	Stack das solid phase	GFAAS	ug/Nm3	16.5		20.9	18.7
	Stack das. vapor phase	GFAS	ug/Nm3	0.816 @		0.503 @	0.642 @
Nickel	Stack ass. <4 um	ICP-MS	ug/Nm3	1.61		1.54	1.22
Nickel	Stack cas. 4-9 um	ICP-MS	ug/Nm3	2.55		1.96	1.92
Nickal	Stack das, >9 um	ICP-MS	ug/Nm3	6.38		7.60	8.95
Nickal	Stack particulate, <4 um	ICP-MS	mg/kg	170		195	174
	Stack particulate. 4–9 um	ICP-MS	mg/kg	235		219	221
i Nickel	Stack particulate, >9 um	ICP-MS	mg/kg	145		138	216
Nitrogen	Coal	Ultimate	%, dry	1.50	1.51	1.61	1.64
Oxvoan	Coal	Ultimate	%, dry	5.01	5.41	5.36	5.29
o-Xviene	Stack gas, VOST	GCMS	ug/Nm3.	< 0.588		< 0.588	< 0.578
Percent moisture	Bottom ash	Gravimetric	8	0.200	0.600	0	0
Phenenthrene	ESP injet gas	HROCMS	ug/Nm3	0.0763		0.0224	0.0214
Phenanthrene	Stack gas	HRGCMS	ug/Nm3	0.104		0.0247	0.0253
Phosphorus	Bottom ash	ICP-AES	mg/kg, dry	527 @	674 @	542 @	535 @
Phosphorus	Coal	ICP-AES	mg/kg, dry	170	140	140	110
Phosphorus	ESP inlet gas, perticulate	ICP-AES	mg/kg	917	769	222	731
Phosphorus	ESP intet gas, solid phase	ICP-AES	ug/Nm3	4746	3977	4433	4228
Phosphorus	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 177		< 178	< 185
Phosphorus	Stack gas, particulate	ICP-AES	mg/kg	1683		1289	1598

ი

				0.00			Bun 4
Substance		nolitow	B				
Phoenhorite	Stack das solid phase	ICP-AES	ug/Nm3	193		218	213
Dhoshorus	Stack das, vapor phase	ICP-AES	ug/Nm3	< 152		< 117	< 174
Dyrana	ESP inlet das	HRGCMS	ug/Nm3	0.0148		0.00493	0.00924
	Stack das	HRGCMS	ug/Nm3	0.0309		0.00361	0.00231
Calonium	Bottom ash	GFAAS	mg/kg, dry	< 1.16	< 1.17	< 1.07	< 1.12
Selantium	Coat	NAA	mg/kg, dry	3.36	3.53	4.00	3.60
Selentium	ESP inlet das. particulate	GFAAS	mg/kg	17.4		17.9	16.9
Salantim	ESP inlet das, solid phase	GFAAS	ug/Nm3	89.8		102	97.6
Selantium	ESP inlet das, vapor phase	GFAAS	ug/Nm3	3.53		2.20 @	2.12 @
Selenium	Stack gas, particulate	GFAAS	mg/kg	123		96.5	119
Calorium	Stack cas solid phase	GFAS	ug/Nm3	14.1		16.3	15.9
Salantim	Stack das vapor phase	GFAAS	ug/Nm3	108		124	213
Selentium	Stack cas. <4 um	ICP-MS	ug/Nm3	0.963		1.05	0.946
Selection	Stack cas. 4-9 um	ICP-MS	ug/Nm3	0.580		0.326	0.369
Celevium	Stack cas: >9 um	ICP-MS	ug/Nm3	2.76		5.53	5.14
Seatim	Stack particulate. <4 um	ICP-MS	mg/kg	102		133	135
Selantita	Stack particulate, 4-9 um	ICP-MS	mg/kg	53.3		36.3	42.5
Salanium	Stack particulate, >9 um	ICP-MS	mg/kg	62.7		5	124
Sodium	Bottom ash	ICP-AES	mg/kg. dry	3397	3481	3320	3980
Sodium	Coat	NAA	mg/kg, dry	310	- 304	386	317
	ESP inlet das, particulate	ICP-AES	mg/kg	3713	2956	3194	3323
	ESP inter das. solid phase	ICP-AES	ug/Nm3	19214	15295	18222	19235
	ESP inlet cas, vapor phase	ICP-AES	ug/Nm3	17.7 @		80.7 @	59.2 @
Sodium	Stack gas, particulate	ICP-AES	mg/kg	3681		4291	3504
Sodium	Stack cas, solid phase	ICP-AES	ug/Nm3	422		727	466
Sodium	Stack das, vapor phase	ICP-AES	ug/Nm3	70.4		14.7 @	22.8 @
Stutene	Stack das. VOST	GCMS	ng/Nm3	< 0.588		< 0.588	< 0.578
Sulfate	ESP inlet gas, particulate	<u>0</u>	mg/kg	12222		2191	23465
Sulfate	ESP injet das, solid phase	Q	ug/Nm3	63085		12520	133269
Sulfata	ESP inlet gas, vapor phase	<u>2</u>	ug/Nm3	4829957		4840293	4714625
Suifate	Stack gas, particulate	Q	mg/kg	113068		93141	130542
Sulfate	Stack gas, solid phase	<u>0</u>	ug/Nm3	13064		15823	17396
Sulfate	Stack gas, vepor phase	Q	ug/Nm3	4022447		3986056	4146332
Sulfur	Bottom ash	Leco	%, dry	< 0.00501	0.0145@	0.0164 @	< 0.00500

Table B-2: Site 16 OFA/LNB Data USED in Calculations

29-Oct-93

S. the second	Stream	Method	NON	Run 2	Run 2D	Run 3	Run 4
Sulfur	Coal	Ultimate	%, dry	1.82	1.60	1.67	1.61
Tatrachloroathana	Stack das, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Titadium	Bottom ash	ICP-AES	mg/kg, dry	6670	6974	6794	6942
Titonium	Coal	NAA	mg/kg, dry	739	724	978	802
Ttantum	ESP inlet das, particulate	ICP-AES	mg/kg	6821	6774	6925	6866
Thomas	ESP inter cas. solid phase	ICP-AES	ug/Nm3	35296	35049	39509	39741
Ttanium	ESP inlet das. vepor phase	ICP-AES	ug/Nm3	< 0.210		1.09	0.382
Titanium	Stack das, particulate	ICP-AES	mg/kg	6632		6056	6327
Titaninm	Stack das, solid phase	ICP-AES	ug/Nm3	760		1025	842
	Stack das. vapor phase	ICP AES	emN/gu	0.305 @		< 0.140	< 0.207
Toluene	Stack das. VOST	GCMS	ug/Nm3	0.719 @		0.789 @B	0.911@B
trans-13-Dichloropropene	Stack das, VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Trichloroathana	Stack das. VOST	GCMS	emN/Bn	< 0.588		< 0.588	< 0.578
Venedium	Bottom ash	ICP-AES	mg/kg, dry	200	229	224	258
Venedium	Coal	NAA	mg/kg. dry	23.0	22.6	29.8	25.1
Vanactium	ESP intet gas, particulate	ICP-AES	mg/kg	233	234	287	264
Vanadium	ESP intet das, solid phase	ICP-AES	ug/Nm3	1206	1213	1635	1531
Vanadium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.504		0.466	< 0.528
Vanadium	Stack gas, particulate	ICP-AES	mg/kg	345		328	355
Vanadium	Stack gas, solid phase	ICP-AES	ug/Nm3	39.5		55.6	47.3
Vanadium	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 0.435		< 0.335	< 0.497
Vanadium	Stack gas, <4 um	ICP-MS	ug/Nm3	5.80		6.04	5.10
Vanadium	Stack gas, 4-9 um	ICP-MS	ug/Nm3	6.83		5.76	5.55
Vanadium	Stack gas, >9 um	ICP-MS	ug/Nm3	15.2		19.5	2.21
Vanadium	Stack particulate, <4 um	ICP-MS	mg/kg	614		765	729
Vanadium	Stack particulate, 4-9 um	ICP-MS	mg/kg	634		641	638
Vanadium	Stack particulate, >9 um	ICP-MS	mg/kg	347		352	428
Vinul Acetate	Stack das, VOST	GCMS	ug/Nm3	< 2.94	-	< 2.94	< 2.89
Word Chlorida	Stack das. VOST	GCMS	ug/Nm3	< 0.588		< 0.588	< 0.578
Volatiles	Coal	Proximate	%, dry	32.3	32.7	32.3	32.1

Ŧ

APPENDIX C: ANALYTICAL DATA NOT USED IN CALCULATIONS

σ.
1
•
0
æ
C 3
-
•
~
\sim

C-2

Table C-1: Site 16 OFA Data NOT USED in Calculations

-

e
6
1
5
Õ
Ĩ.
0
ģ

Stream		Method	MOU	Run 1	Run 2	Run 3	Hun 3U	
	I VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105		
hick due one 5-1	L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226		
etack one 1-1 VO	0.8T	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110	· · · · · · · · · · · · · · · · · · ·	
etect case 5-1 VO	OST	GCMS	mg/Nm3	< 0 00231	< 0.00227	< 0.00218		
high duct age 1-1	I VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105	· · · · · · · · · · · · · · · · · · ·	
	1 VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226		
stark nas 1-1 VC	ost	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110		
	0sT	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218		
Fire Australe A	I VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105		
	LaON -	SMOO	ma/Nm3	< 0.00202	< 0.00211	0.00770 @		
		SMOO	ma/nm3	< 0.0115	< 0.0114	< 0.0110		
e stack gas, 1-L VU	100			~ 0.0031	< 0.00227	< 0.00218		
ie stack gas 5-t VC	051	GCMS		0.0001	0.100	< 0.0105		
high dust gas, 1-1	-L <u>VOST</u>	GCMS	SmN/gm					
high dust gas. 5-	-L VOST	GCMS	mg/Nm3	< 0.00202	<pre>< 0.00211</pre>			
start dag 1-1 VC	ost	GCMS	mg/Nm3	< 0.0115	< 0.0114	× 0.0110		
	OST OST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218		
		GCMS	ma/Nm3	< 0.00887	< 0.0139	< 0.0105		
		SCMS	ma/Nm3	< 0.00202	< 0.00211	< 0 00226		:
sthane high dust gas, 3-	-1.4031			< 0.0115	< 0.0114	< 0.0110		:
stack gas, 1 - L V	091			< 0.00231	< 0.00227	< 0 00218		
stack gas, 5 - L V(051	GCMS		080		< 0.980	066.0 >	066.0 >
e boltom ash		GCMS(BZ(U)	<u></u>		+ 	< 11.0	< 11.0	× 11 0
e bottom ash sluice	e water	GCMS(82/0)				000 0	0 0 0 0	< 1.00
collected fly ash		GCMS(8270)	6/6n	0.66.0 >		DEE N V		
high dust gas, 1-	-L VOST	GCMS	mg/Nm3	< 0.00687	80100 ×	C010.0 V		
high dust das. 20	I-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0 000559		
high dust das 5-	-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226		
high dust oas sol	hid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863		
	nor phase	GCMS(8270)	EmN/gm	< 0.00581	< 0.00488	< 0.00503		:
shire water silon		GCMS(8270)	ng/L	< 10.0		< 10.0	× 11.0	× 10.0
	nosT	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110	i	:
	VOST	GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542		
	LaC.	SMC	ma/Nm3	< 0.00231	< 0.00227	< 0.00218		
		COMS(8270)	ma/Nm3	< 0.00530	< 0 00555	< 0.00534		
Be stack gas, solid p	Diasa	CUS(8270)	mo/Nm3	< 0.00530	< 0.00555	< 0.00534		
16 Stack gas, vapur	Pilesu		mn/nm3	< 0.00887	< 0.0139	< 0.0105		
			mo/Nm3	< 0.000525	< 0.000543	< 0.000559		
high dust gas, 20	1-L VU31		Sund/Nm3	< 0.00202	< 0.00211	< 0.00226		
high dust gas, 5-		0.00 0.00 0.00		< 0.0115	< 0.0114	< 0.0110		
stack gas, 1-L V	/081	N NO		000500	0.00569	< 0.00542		
International Sharts	VOST	SNDG			00000		T	

Table C--1: Site 16 OFA Data NOT USED in Calculations

N

Modelhane stack gas. 5-1 VOST GC propropane high dust gas. 5-1 VOST GC propropane high dust gas. 5-1 VOST GC propropane stack gas. 1-1 VOST GC propropane stack gas. 1-1 VOST GC propropane stack gas. 1-1 VOST GC propropane stack gas. 5-1 VOST GC propropane bottom ash slute water GC propropane bottom ash slute water GC prachbrobenzene slute water GC probenzene bottom ash slute water GC	MS MS MS MS MS MS MS MS MS MS MS MS MS M	6mN/gr 6mN/gr	1 0 0 0					
harre stack gas, 5 - L VOST GC opare high dust gas, 1 - L VOST GC opare stack gas, 1 - L VOST GC opare stack gas, 1 - L VOST GC opare stack gas, 1 - L VOST GC blorobenzene bottom ash sluce water GC blorobenzene bigh dust gas, solid phase GC blorobenzene bigh dust gas, solid phase GC brorobenzene bigh dust gas, solid phase <t< td=""><td>MS MS MS MS MS MS MS MS MS MS MS MS MS M</td><td>EmN/g</td><td></td><td>< 0.00227</td><td>< 0.00218</td><td></td><td></td><td></td></t<>	MS MS MS MS MS MS MS MS MS MS MS MS MS M	EmN/g		< 0.00227	< 0.00218			
opane high dust gas. 5-L VOST GC opane stack gas. 5-L VOST GC opane stack gas. 5-L VOST GC opane stack gas. 5-L VOST GC biotobenzene bottom ash bottom ash GC biotobenzene bottom ash sluice water GC biotobenzene bottom ash sluice water GC biotobenzene high dust gas, solid phase GC biotobenzene biotom Stack gas, solid phase GC biotobenzene high dust gas, vapor phase GC GC biotobenzene biotom ash sluice water GC GC biotobenzene biotom ash sluice water GC GC biotobenzene biotom ash sluice water GC GC bigh dust gas, solid phase GC GC GC bigh dust gas, solid phase GC GC GC rizzene bigh dust gas, i - L VOST GC GC rizzene bottom ash sluice water GC GC rizene	MS MS MS MS MS MS MS MS MS MS MS MS MS M	CmN/gr						
Ppane high dust gas. 5 - L VOST GC Ppane stack gas. 1 - L VOST GC Pottomarbne bottom ash GC Pottobenzene bottom ash stuice water GC Pottobenzene high dust gas. solid phase GC Pottobenzene stuice water supply GC Pottobenzene stuck gas. solid phase GC Pottobenzene stuck gas. solid phase GC Pottobenzene stack gas. solid phase GC Pottobenzene bottom ash stuce water GC Pottom ash stuce water GC GC Pottom ash stuce water GC GC Pottom ash stuce water GC GC Pottom ash stuce water	MS MS MS MS MS MS MS MS MS MS MS MS MS M	Nm3	< 0.00887	8010.0 >	C010 0 >			
pare stack gas. 5-1 VOST GC pere stack gas. 5-1 VOST GC borobenzene bortom ash sluice water GC borobenzene sluice water supply GC borobenzene skuice water supply GC borobenzene skuice water supply GC borobenzene skuice water supply GC nzene bortom ash sluice water GC nzene stack gas. vapor phase GC nzene bortom ash sluice water GC nzene bortom ash sluice water GC nzene stack gas. solid phase GC nzene bortom ash sluice water GC	MS MS MS(8270) MS(827	CHANGE IN COMPANY	< 0.00202	< 0.00211	< 0.00226			
pare stack gas, 5-1, VOST GC pare bottom ash stuice water GC borobenzene bigh dust gas, solid phase GC borobenzene stuice water supply GC borobenzene stuck gas, vapor phase GC borobenzene stuck gas, vapor phase GC botobenzene stuck gas, vapor phase GC bottom ash stuck gas, solid phase GC bottom ash bottom ash L nzene bottom ash L <tr< td=""><td>MS MS(8270)</td><td>CmN/gn</td><td>< 0.0115</td><td>< 0.0114</td><td>< 0.0110</td><td></td><td>!</td><td></td></tr<>	MS MS(8270)	CmN/gn	< 0.0115	< 0.0114	< 0.0110		!	
parte bottom ash sluice water GG biobenzene bigh dust gas, vapor phase GC biobenzene sluice water supply GC biobenzene stack gas, vapor phase GC biotom ash sluice water supply GC bottom ash bottom ash Sluice water GC zene bottom ash supor phase GC zene bottom ash supor phase GC zene <td< td=""><td>MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270)</td><td>Nm3</td><td>< 0.00231</td><td>< 0.00227</td><td>< 0.00218</td><td></td><td></td><td></td></td<>	MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270) MS(8270)	Nm3	< 0.00231	< 0.00227	< 0.00218			
Otopenaene Dottom ash sluice water orobenzene bottom ash sluice water orobenzene high dust gas, solid phase orobenzene high dust gas, vapor phase orobenzene sluice water supply orobenzene sluice water supply orobenzene sluice water supply orobenzene stack gas, vapor phase orobenzene stack gas, solid phase collected fly ash stack gas, solid phase collected fly dust gas, upply GC zene stack gas, solid phase stack gas, solid phase GC zene stack gas, solid phase stack gas, solid phase GC zene stack gas, solid phas	MS(8270) MS(8270) MS(8270) MS(8270) MS(8270)	0/0	< 0.980		< 0.980	< 0.990	< 0.990	
Olicobencene Douclim sail subre recol orobencene high dust gas, vapor phase GC orobencene bigh dust gas, vapor phase GC orobencene sluice water supply GC orobencene sluice water supply GC orobencene sluice water supply GC orobencene stack gas, vapor phase GC orobencene stack gas, vapor phase GC orobencene stack gas, vapor phase GC zene bottom ash sluice water GC zene high dust gas, 1-L VOST GC zene stack gas, solid phase	MS(8270) MS(8270) MS(8270) MS(8270)	1/0	< 10.0		< 11.0	< 11 0	< 11.0	
Olicitude valer supply Solid phase GC orobenzene high dust gas, solid phase GC orobenzene sluice water supply GC orobenzene stack gas, vapor phase GC zene bottom ash sluice water GC zene high dust gas, 1-L VOST GC zene stack gas, solid phase	MS(8270) T MS(8270) T MS(8270) T	0/0	066.0 >	+	066.0 >	0.970 >	< 1.00	
Orobenzene Inign dost gas, surup phase Oco orobenzene high dust gas, vapor phase GC orobenzene sluice water supply GC zene bottom ash sluice water GC zene high dust gas, solid phase GC zene stack gas, f-L VOST G	MS(8270)	(Mm.)	< 0.00997	< 0.00837	< 0.00863			
onobenzene probenzene probenzene probenzene stack gas, vapor phase stack gas, vapor phase stack gas, vapor phase bottom ash sluice water zene bottom ash sluice water bottom ash sluice water bottom ash sluice water collected fly ash collected fly ash dust gas, 1-L VOST prob stack gas, 20-L VOST cone bligh dust gas, 20-L VOST prob stack gas, solid phase stack gas, solid phase stack gas, solid phase prob stack gas, 5-L VOST prop phigh dust gas, 1-L VOST prob stack gas, 5-L VOST prop phigh dust gas, 1-L VOST prop phigh dust gas, 1-L VOST prop phigh dust gas, 1-L VOST prop phigh dust gas, 1-L VOST prop prop phase phigh dust gas, 1-L VOST prop phigh dust gas, 1-L VOST phigh phigh dust phigh phigh dust phigh dust phigh phigh dust phigh dust phigh dust phigh dust phigh dust phigh dust	MS(8270)	Mm3	< 0.00581	< 0.00488	< 0.00503			
anobenzene aurocenter vareur uterunt 000 zene stack gas, vapor phase 600 zene bottom ash 600 zene high dust gas, zo-L VOST 600 zene high dust gas, vapor phase 600 zene stack gas, solid phase 600 stack gas, 5-L VOST 600 zene stack gas, 5-L VOST 600 zene stack gas, 5-L VOST 600 stack gas, 5-L VOST 600 zene stack gas, 5-L VOST 600 stack gas, 5-L VOST 60			< 10.0		< 10.0	< 11.0	< 10.0	
Otopentation Stack gas. vapor phase GC Zene bottom ash stuice water GC Zene bottom ash stuice water GC Zene bottom ash stuice water GC Zene bottom ash bottom ash GC Zene bottom ash bottom GC Zene high dust gas. vapor phase GC Zene high dust gas. solid phase GC Zene high dust gas. vapor phase GC Zene stuice water supply GC Zene stack gas. solid phase GC Zene stack gas. J - L VOST GC Stack gas. J		Nm3	< 0.00530	< 0.00555	< 0.00534			
Zene bottom ash sluice water GC Zene high dust gas. 5-L VOST GC Zene stack gas. 1-L VOST GC Zene stack gas. 1-L VOST GC Zene stack gas. stack gas. GC Zene stack gas. stack gas. GC Zene stack gas. 5-L VOST GC Zene stack gas. 1-L VOST GC Zene stack gas. 5-L VOST GC <tr< td=""><td>MS(8270)</td><td>ng/Nm3</td><td>< 0.00530</td><td>< 0.00555</td><td>< 0.00534</td><td></td><td></td><td></td></tr<>	MS(8270)	ng/Nm3	< 0.00530	< 0.00555	< 0.00534			
Zene boltom ash sluice weler GC zene collected fly ash 1 UOST 60 zene collected fly ash 1 - UOST 60 zene collected fly ash 5 - L VOST 60 zene high dust gas, zolle phase 60 zene high dust gas, solid phase 60 zene sluice water supply 60 zene sluice water supply 60 zene stack gas, solid phase 60 stack gas, 5 - L VOST 60	MS(8270)	D/0	< 0.980		0.980 >	066.0 >	< 0.990	
Zene Doulding ash Duce water Sec Zene high dust gas, 1-1 VOST GC Zene high dust gas, 1-1 VOST GC Zene high dust gas, 1-1 VOST GC Zene high dust gas, solid phase GC Zene high dust gas, solid phase GC Zene stuce water supply GC Zene stuck gas, solid phase GC Zene stuck gas, solid phase GC Zene stack gas, 1-1 VOST GC Zene stack gas, 5-1 VOST GC	MS(8270)	10	< 10.0		< 11.0	< 11.0	< 11.0	
zene conected my asn conected m	10200	20	0.000		066.0 >	< 0.970	< 1.00	
Zene nigh dust gas, 1 - L VOST GC Zene high dust gas, 20 - L VOST GC Zene high dust gas, solid phase GC Zene high dust gas, solid phase GC Zene stock gas, 1 - L VOST GC Zene stock gas, solid phase GC Zene stock gas, 1 - L VOST GC Zene stock gas, 5 - L VOST GC			C 0 00887	< 0.0139	< 0.0105		1	
Zence high dust gas. 20 - L VOST GC Zence high dust gas. solid phase GC Zence high dust gas. vapor phase GC Zence stuice water supply GC Zence stuice water supply GC Zence stack gas. 1 - L VOST GC Zence stack gas. 5 - L VOST GC Stack gas. 5 - L VOST GC GC Stack gas. 5 - L VOST GC GC	2		000055	0.0001	< 0.00559			
Zene high dust gas, 5-L VOS1 GC Zene high dust gas, solid phase GC Zene sluice water supply GC Zene sluice water supply GC Zene stack gas, 1-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene high dust gas, 1-L VOST GC stack gas, 5-L VOST GC GC Zene stack gas, 5-L VOST GC zene stack gas, 5-L VOST GC stack gas, 5-L VOST GC GC stack gas, 5-L VOST GC GC	2			0.00011	< 0.0026	i	•	
zene high dust gas, solid phase GC zene bigh dust gas, vapor phase GC zene sluice water supply GC zene stack gas, 1 - L VOST GC zene stack gas, solid phase GC stack gas, solid phase GC zene stack gas, 1 - L VOST GC zene stack gas, 1 - L VOST GC stack gas, 1 - L VOST GC GC zene stack gas, 1 - L VOST GC stack gas, 1 - L VOST GC GC stack gas, 5 - L VOST GC GC	2		202000			•		
zene high dust gas, vapor phase GC Zene succe water supply GC Zene stack gas, 1-L VOST GC Zene stack gas, solid phase GC Rene stack gas, solid phase GC Zene stack gas, solid phase GC Rene stack gas, solid phase GC Sene stack gas, solid phase GC Rene stack gas, solid phase GC Rene stack gas, solid phase GC Sene stack gas, solid phase GC Sene stack gas, solid phase GC stack gas, solid phase Solid phase GC stack gas, solid phase Solid phase GC stack gas, solid up to to gas, 5-L VOST GC stack gas, 5-L VOST GC stack gas, 5-L VOST GC	MS(8270)	6mN/gr	16600.0 >		50000 n V			
Zene sluice water supply GC Zene stack gas, 1-t VOST GC Zene stack gas, 20-t VOST GC Zene stack gas, 5-t VOST GC Zene high dust gas, 1-t VOST GC Zene stack gas, 5-t VOST GC Zene high dust gas, 1-t VOST GC Zene stack gas, 5-t VOST GC Zene stack gas, 5-t VOST GC Zene stack gas, 1-t VOST GC Stack gas, 5-t VOST GC GC Zene stack gas, 1-t VOST GC stack gas, 5-t VOST GC GC	MS(8270)	10/Nm3	< 0.00581	< 0.00488	20200.0 >			
Zene stack gas, 1-L VOST GC Zene stack gas, 20-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, solid phase GC Zene high dust gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Stack gas, 5-L VOST GC SC Zene stack gas, 5-L VOST GC Stack gas, 1-L VOST GC SC Stack gas, 1-L VOST GC SC Stack gas, 5-L VOST GC SC Stack gas, 5-L VOST SC SC<	MS(8270)	10/F	< 10.0		10.0	< 11.0	< 10.0	
Zene stack gas, 20-L VOST GC zene stack gas, 5-L VOST GC zene stack gas, solid phase GC zene stack gas, solid phase GC zene stack gas, solid phase GC zene high dust gas, 1-L VOST GC zene stack gas, 1-L VOST GC stack gas, 5-L VOST GC GC	MS	6mN/Bu	< 0.0115	< 0.0114	< 0.0110	+		
cone stack gas, 5 - L VOST GO zene stack gas, solid phase GC zene stack gas, vapor phase GC zene high dust gas, 1 - L VOST GC zene stack gas, 5 - L VOST GC zene high dust gas, 1 - L VOST GC zene stack gas, 5 - L VOST GC zene stack gas, 5 - L VOST GC zene stack gas, 5 - L VOST GC stack gas, 5 - L VOST GC GC stack gas, 5 - L VOST GC GC stack gas, 5 - L VOST GC GC bigh dust gas, 5 - L VOST GC GC	WS SW	ng/Nm3	< 0.000580	< 0.000569	< 0.000542		i	
Zone stack gas, solid phase GC zene stack gas, vapor phase GC zene high dust gas, 1 - L VOST GC zene high dust gas, 1 - L VOST GC zene stack gas, 5 - L VOST GC zene stack gas, 1 - L VOST GC zene stack gas, 1 - L VOST GC zene stack gas, 1 - L VOST GC stack gas, 5 - L VOST GC GC stack gas, 1 - L VOST GC GC bigh dust gas, 1 - L VOST GC GC	MS -	CmN/gu	< 0.00231	< 0.00227	< 0 00218			
Zane stack gas, vapor phase GC Zene high dust gas, 1-L VOST GC Zene high dust gas, 1-L VOST GC Zene stack gas, 5-L VOST GC Zene stack gas, 1-L VOST GC Zene stack gas, 1-L VOST GC Inigh dust gas, 1-L VOST GC GC stack gas, 5-L VOST GC GC stack gas, 1-L VOST GC GC high dust gas, 1-L VOST GC GC stack gas, 5-L VOST GC GC bigh dust gas, 1-L VOST GC GC	MS(8270) r	CmN/gu	< 0.00530	< 0.00555	< 0.00534			
Zene high dust gas, 1-L VOST GC zene high dust gas, 5-L VOST GC zene stack gas, 1-L VOST GC zene stack gas, 5-L VOST GC zene stack gas, 1-L VOST GC high dust gas, 5-L VOST GC GC stack gas, 5-L VOST GC GC stack gas, 5-L VOST GC GC high dust gas, 5-L VOST GC GC stack gas, 5-L VOST GC GC high dust gas, 5-L VOST GC GC stack gas, 5-L VOST GC GC	MS(8270)	ng/Nm3	< 0.00530	< 0.00555	< 0.00534			
Ingh dust gas, 5-L VOST GC zene tiack gas, 1-LVOST GC zene stack gas, 5-L VOST GC zene stack gas, 1-LVOST GC ingh dust gas, 1-LVOST GC GC high dust gas, 1-LVOST GC GC stack gas, 5-LVOST GC GC stack gas, 5-LVOST GC GC	MS	CmN/gr	< 0.00887	< 0.0139	< 0.0105			
zene stack gas. 1 - L VOST GO zene stack gas. 5 - L VOST GO stack gas. 5 - L VOST GO high dust gas. 1 - L VOST GO stack gas. 1 - L VOST GO stack gas. 1 - L VOST GO stack gas. 5 - L VOST GO stack gas. 5 - L VOST GO	L SM	Pm3	< 0.00202	< 0.00211	< 0.00226			
zene stack gas. 5 - 1 VOST GC ziack gas. 5 - 1 VOST GC high dust gas. 5 - 1 VOST GC stack gas. 1 - 1 VOST GC stack gas. 5 - 1 VOST GC stack gas. 5 - 1 VOST GC	MS MS	EmN/Bu	< 0.0115	< 0.0114	< 0.0110			
Inight dust gas, 1-L VOST GC high dust gas, 1-L VOST GC stack gas, 1-L VOST GC stack gas, 5-L VOST GC	SM	CmV/pu	< 0.00231	< 0.00227	< 0.00218			
high dust gas, 5-1 VOST GC stack gas, 1-1, VOST GC stack gas, 5-1, VOST GC bottom ash	SN	no/Nm3	< 0.0887	< 0.139	< 0.105			
nign uusi gas, J-t, VOST GC stack gas, 1-L VOST GC stack gas, 5-L VOST GC bottom ash		EmV/pd	< 0.0202	< 0.0211	< 0.0226			
stack gas, 5-L VOST 60 bottom ash		emu)ou	< 0.115	< 0.114	< 0 110			
tack gas, 5-L VUSI GG bottom ash GG			11000	× 0.027	< 0.0218			
bottom ash GC	0		1020.0					
	MS(8270) 1	6/6	20.900				0000	
bottom ash sluice water GC	MS(8270) L	jø∕L	< 10.0		11.0	0.11 ×		1
collected fly ash GC	MS(8270) 4	6/6	0.090		< 0.990	< 0.970	< 1.00	
high dust gas, solid phase [GC	MS(8270)	EmN/Bu	26600 0 >	< 0.00837	< 0.00863			

93
Ť
ö
ī
30

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
			+						
2 - Chloronhenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503		 	
2 - Chloronhanol	Iskiice water supply	GCMS(8270)	ng/t	< 10.0		< 10.0	× 11.0	< 10.0	
2 - Chloronhanol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0 00555	< 0.00534			
	stack das vanor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	hich dust ras 1-1 VOST	GCMS	ma/Nm3	< 0.0887	< 0.139	< 0.105			
	high dust cas 20-1 VOST	GCMS	mg/Nm3	< 0 00525	< 0.00543	< 0.00559			
	high dust das 5-1 VOST	GCMS	ma/Nm3	< 0.0202	< 0.0211	< 0.0226			:
	stark ras 1-1 VOST	GCMS	ma/Nm3	< 0.115	< 0.114	< 0.110			
	stack nas 20-1 VOST	GCMS	ma/Nm3	< 0.00580	< 0.00569	< 0.00542			
	stark nes 5-1 VOST	GCMS	ma/Nm3	< 0.0231	< 0.0227	< 0.0218			
	bottom ash	GCMS(8270)	0/67	<pre></pre>		< 4.90	4.90	< 5.00	
	bottom ash sluice water	GCMS(8270)	ng/L	< 52.0		< 53.0	< 54.0	< 56.0	
2 - Mirroaniine	collected fiv ash	GCMS(8270)	B/Bn	< 5.00		< 4.90	< 4.80	< 5.00	•
2 – Nitroanilne	high dust gas, solid phase	GCMS(8270)	EmN/Bm	< 0.0498	< 0.0419	< 0.0432			
2_Nitroanijira	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0290	< 0.0244	< 0.0251			
2 - Nitroaniine	Isluice water supply	GCMS(8270)	ug/L	< 52.0		< 52.0	< 53.0	< 52.0	
	stack das. solid phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			·
	stack das. vapor phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
2 – Nitronhamol	bottom ash	GCMS(8270)	b/bn	< 0.980 >		0.980	066.0 >	< 0.990	:
2 Mitroshand	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0 	
	collected fly ash	GCMS(8270)	6/60	066.0 >		066.0 >	< 0.970	< 1.00	
	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2 - Nitronhanol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
2 – Nitronhenol	isluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	1
2	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		- ··· -	
2 – Nitrophenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0 00534			
2 - Ptcoline	bottom ash	GCMS(8270)	6/6n	< 0.980 <		< 0.980 ≥	< 0.990	< 0.990	
2 - Picoline	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	v 11 0	< 11.0	
2 - Picoline	collected fly ash	GCMS(8270)	5/61	< 0.990		066.0 ≥	0.970	< 1.00	 - -
2 - Picoline	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
2 - Picoline	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00408	< 0.00503			
2 - Picoline	sluice water supply	GCMS(8270)	1/61	< 10.0		< 10.0	< 11.0	< 10.0	
2 - Picoline	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 Picoline	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2 3 4 6 ~ Tetrachbrophenol	bottom ash	GCMS(8270)	<u>6/6n</u>	< 2.00		< 2.00	< 2.00	< 2.00	
2 3.4.6 - Tetrachlorophenol	bottom ash sluice water	GCMS(8270)	ug/L	< 21.0		< 21.0	< 22.0	< 22.0	
2.3.4.6 - Tetrachloro pheriol	collected fly ash	GCMS(8270)	6/6n	< 2 00		< 2.00	< 1.90	< 2.00	
2 3,4,6 - Tetrachlorophenol	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0199	< 0.0167	< 0.0173			
2.3,4,6 - Tetrachtorophenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0116	≤ 0.00976	< 0.0101	- -	 - - - -	: : :

Appendix C: Analytical Data Not Used in Calculations

ო

C-4

Table C-1: Site 16 OFA Data NOT USED in Calculations

-

Substance	Stream	Method	Non	Ben 1	Run 2	Run 3	Run 3D	Hun 4	gun 5
2,3,4,6 - Tetrachlorophenol	sluice water supply	GCMS(8270)		< 21.0		< 21.0	< 21.0	< 21.0	
2.3.4.6 - Tetrachlorophenol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0106	< 0.0111	< 0.0107			
2.3,4,6 - Tetrachbrophenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0106	< 0.0111	< 0.0107			
2.4 - Dichlorophenol	bottom ash	GCMS(8270)	0/6 0	< 0.980		< 0.980	066:0 >	< 0.990	
2,4 - Dichlorophenol	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	 11.0 	
2.4 - Dichlorophenol	collected fly ash	GCMS(8270)	ug/g	066.0 >		< 0.990	< 0.970	< 1 00	
2.4 - Dichlorophenol	high dust gas, solid phase	GCMS(8270)	Բ ա <mark>/</mark> /Խա3	< 0.0097	< 0.00837	< 0.00863			
2.4 - Dichlorophenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
2 4 - Dichtorophenol	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
2,4 - Dichlorophenol	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	<pre>< 0 00534</pre>	+ 		
2.4 - Dichtorophenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	-		•
2.4 - Dimethylphenol	bottom ash	GCMS(8270)	6/6n	< 0.980	, r ;	< 0.980	066.0 >	< 0.990	
2.4 - Dimethylphenol	bottom ash sluice water	GCMS(8270)	V6n	< 10.0		< 11.0	< 11.0	< 11.0	
2.4 - Dimethylphenol	collected fly ash	GCMS(8270)	6/6n	066.0 >		066.0 >	< 0.970	4 1 00	r •
2.4 - Dimethylphenol	high dust gas, solid phase	GCMS(8270)	CmN/Bm	< 0.0097	< 0.00837	< 0.00863			
2.4 - Dimethylphenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503		· · -	
2.4 - Dimethylphenol	sluice water supply	GCMS(8270)	<u>ug/L</u>	< 10.0		< 10.0	< 11.0	< 10.0	
2.4 - Dimethylphenol	stack gas, solid phase	GCMS(B270)	mg/Nm3	< 0.00530	< 0 00555	< 0.00534			
2.4 - Dimethylphenol	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		-	
2.6 - Dichlorophenol	bottom ash	GCMS(8270)	6/8n	< 0.980		< 0.980 >	< 0.990	066.0 >	
2.6 - Dichlorophenol	bottom ash sluice water	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
2.6 - Dichtorophenol	collected fly ash	GCMS(8270)	5/6n	066.0 >		< 0.990	< 0.970	< 1.00	
2.6 - Dichlorophenol	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			· · · · · · · · · · · · · · · · · · ·
2.6 - Dichlorophenol	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	<u>.</u>	. ,	•
2.6 - Dichtorophenol	stuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
2,6 - Dichlorophenol	stack gas, solid phase	GCMS(8270)	m9/Nm3	< 0.00530	< 0.00555	< 0.00534		• -	•
2,6 – Dichtorophenol	stack gas, vapor phase	GCMS(8270)	EmN/8m	< 0.00530	< 0.00555	< 0 00534			
2,6 - Dinitrotoluene	boltom ash	GCMS(8270)	6/ðn	< 0.980		< 0.980	< 0.990	066.0 >	
2.6 - Dinitrotoluene	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
2,6 - Dinitrotoluene	collected fly ash	GCMS(8270)	ng/g	< 0.990		066.0 >	< 0.970	< 1.00	
2,6 - Dinitrotoluene	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863		 	! ! !
2,6 - Dinitrotoluene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
2.6 - Dinitratoluene	sluice water supply	GCMS(8270)	<u>1/6</u>	< 10.0		< 10.0	< 11.0	< 10.0	
2.6 - Dinitrololuene	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
2,6 - Dinitratokene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			-
3 - Nitroanijine	bottom ash	GCMS(8270)	<u>5/6n</u>	4.90		< 4.90	< 4.90	< 5.00	! !
3 – Nitroaniline	bottom ash sluice water	GCMS(8270)	ug/L	< 52.0		< 53.0	< 54.0	< 56.0	
3 – Nitroaniline	collected fly ash	GCMS(8270)	<u> </u>	< 5.00		< 4.90	< 4.80	< 5.00	,
3 - Nitroaniline	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432		 	

6
1
ğ
Ò.
1
8

Table C-1: Site 16 OFA Data NOT USED in Calculations

Nitroaniline Nitroaniline Nitroaniline Nitroaniline Chloro – 3 – methylphenol Chloro – 3 – methylphenol		GCMS(8270)							r
Nitroaniline Nitroaniline Nitroaniline Nitroaniline Chloro - 3 - methylphenol Chloro - 3 - methylphenol Chloro - 3 - methylphenol	inh dust nas vapor phase 1		mg/Nm3	< 0.0290	< 0.0244	< 0.0251			
- Nitroanitine 11 - Nitroanitine 11 - Chloro - 3 - methylphenol 1 - Chloro - 3 - methylphenol 1 - Chloro - 3 - methylphenol 0	luica water supply	GCMS(8270)	ng/L	< 52.0		< 52 0	< 53.0	< 52.0	
- Chloro - 3 - methylphenol - 5 - Chloro - 3 - methylphenol - 5 - Chloro - 3 - methylphenol - 5 - Chloro - 3 - methylphenol - 5	tack das, solid phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0270	< 0.0267			:
- Chloro - 3 - methylphenol - b - Chloro - 3 - methylphenol - b - Chloro - 3 - methylphenot - c	tack das. vapor phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
- Chloro - 3 - methylphenol - Chloro - 3 - methylphenol - Chloro - 3 - methylphenot	ottom ash	GCMS(8270)	5/6n	< 0.980		< 0.980	< 0.990	< 0.990	
- Chloro - 3 - methylphenot	ottom sch sture water	GCMS(8270)	ua/L	< 10.0		< 11.0	< 11.0	< 11.0	;
	ollacted fly ash	GCMS(8270)	6/6n	066.0 >		066.0 >	< 0.970	< 1.00	
Chieren 3. mathematical history	inth duet rase solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
- Chloro - 3 - meiny prieros	Hundras gas, sond pruss	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Chloro - 3 - methylinhanol	Auice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Chloro – 3 – methylinhenol	tack das. solid phase	GCMS(8270)	EmN/gm	< 0 00530	< 0.00555	< 0.00534			
Chloro - 3 - mathulpherol	tack das vanor ohase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	indh dust das 1-t VOST	GCMS	mg/Nm3	< 0.0887	< 0.139	< 0.105	 		:
	inch dust cas 20-1 VOST	GCMS	mg/Nm3	< 0.00525	< 0.00543	< 0.00559			
	The first day 5-1 VOST	GCMS	ma/Nm3	< 0.0202	< 0.0211	< 0.0226			
		GCMS	mg/Nm3	< 0.115	< 0.114	< 0.110		 - - - - - - - -	
		GCMS	mo/Nm3	< 0.00580	< 0.00569	< 0.00542			
			E mN/om	< 0.0231	< 0.0227	< 0.0218			
	LACK GAS, J L VOI	GCMS(8270)	uolo -	< 4.90		< 4.90	< 4 90	< 5.00	
		COMERCE OF		< 52.0		< 53.0	< 54.0	< 56.0	
	OTOM ASI SIUCE WALL	OCMS(8270)		5.00		< 4.90	<pre>4.80</pre>	< 5.00	
				0.0498	< 0.0419	< 0.0432			
- Nitroaniira	IQU DUST Gas, soird priase				- 0 0 24A	× 0.051	· · ·		
-Ntroaniline	igh dust gas, vapor phase	GCMS(82/0)				1 50 0	< 510 < 510	< 52 D	
- Nitroaniline	luice water supply	GCMS(82/0)		0.00.0			2		
- Nitroanitine	tack gas, solid phase	GCMS(8270)	mg/Nm3	0.0265	0.02/0	1020.0 2		1	-
- Nitroaniline	tack gas, vapor phase	GCMS(8270)	mg/Nm3		2/20 0 ×				
6 - Dinitro - 2 - methylphenol b	ottom ash	GCMS(8270)	n0/0	→ 7 1 10	•				
6 - Dinitro - 2 - methylphenol b	ottom ash sluice water	GCMS(8270)	ug/L	< 52.0		< 23.0	× 34.0	20.0	
6 - Dinitro - 2 - methylphenol	ollected fly ash	GCMS(8270)	<u></u>	< 5.00		× 4.90	< 4.80	00.6 >	
6 - Dinitro - 2 - methylphenol	igh dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432		:	
6 - Dinitro - 2 - methviphenol	ingh dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.0290	< 0.0244	< 0.0251			
6 – Dinitro – 2 – methviohenol sl	luice water supply	GCMS(8270)	ng/L	< 52.0	1	< 52.0	< 53.0		
6 - Dinitro - 2 - methvlohenol	tack das. solid phase	GCMS(8270)	mg/Nm3	< 0.0265	< 0.0278	< 0.0267			
6 - Dinitro - 7 - methylohenol	tack das. vapor phase	GCMS(8270)	CmN/gm	< 0.0265	< 0.0278	< 0.0267		 	
catona	lich dust das. 1-L VOST	GCMS	CmN/gm	< 0.0887	< 0.139	< 0.105		, 1	
	inch dust das 20-1 VOST	GCMS	mg/Nm3	< 0.00525	< 0.00543	< 0.00559			
	Nich dust gas. 5-L VOST	GCMS	mg/Nm3	< 0.0202	< 0.0211	< 0.0226			
	tack das 1-L VOST	GCMS	mg/Nm3	< 0.115	< 0.114	< 0.110			
	dack das 20-1 VOST	GCMS	ma/Nm3	< 0.00580	< 0.00569	< 0.00542			

Appendix C: Analytical Data Not Used in Calculations

ŝ

Table C-1: Site 16 OFA Data NOT USED in Calculations

ø

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 30	Run 4	Bun 5
									1 1 1
Acetone	STACK gas, 5-L VUSI	GCMS	5mN/gm			< 0.0218			:
Aluminum	bottom ash	ICP-AES	mg/kg		-	129000	128000	126000	112000
Aluminum	bottom ash	NAA	6/6n					133440	
Aluminum	coat	ICP-AES	mg/kg, dry			9856	8120	11073	7628
Aluminum	collected fiv ash	ICP - AES	ma/ka			108000	101000	07100	01000
	collected fly ash	NAA	0/0/					118440	
	high dust gas, particutates	NAA			+	 		10 64064	
Aluminum	istack das particulates	NAA	0/001	 					I
Antimony	bottom seh	NAA		· 1					
Antimony	coat		ma/ko drv			- 10 T	101		
Antimony	collected fiv ash	NAA	- A B C			/-			
Antimony	high dust gas, particulates	NAA			:				
Antimony	stack nas narticulates	NAA					•		
Arsenic	hottom ash	NAA	B/An				- -	H CZ-0	:
Arsanic	coal	GFAAS	motho dry			3 F	C 10		
Areanic	cottacted fivesh	NAA	Lo Rula			2			0.4°
								0.1	
Arsenic	nigh dust gas, particulates	NAA	<u></u>					142 H	,
Arsenic	stack gas, particulates	NAA	6/6n					149	
Barium	bottom ash	NAA	6/6n					612	1
Barium	bottom ash	XRF	mg/kg				1164	1164	
Barium	coal	ICP-AES	mg/kg, dry			102	89.2	166	106
Barium	collected fly ash	NAA	<u> </u>					1449	
Barium	collected fly ash	XRF	mg/kg			1075		1343	1
Barium	high dust gas, particulates	NAA	6/6n					595 R	
Barium	stack gas, particulates	NAA	5/6n				 ; ,	497 B	
Benzaldehyde	high dust gas, vapor phase	HPLC	mg/Nm3	0.0313 @		< 0.0151		< 0.0161	
Benzaldehyde	stack gas, vapor phase	HPLC	mg/Nm3	< 0.00371		< 0.00984		< 0.0109	
Benzene	high dust gas, 1-L VOST	GCMS	EmN/gm	< 0.00887	< 0.0139	< 0.0105			1
Benzene	high dust gas, 5-1. VOST	GCMS	EmN/gm	< 0.00202	< 0.00211	< 0.00226		÷ —	*** =
Benzene	stack gas, 1-L VOST	GCMS	EmN/gm	< 0.0115	< 0.0114	< 0.0110		+	
Benzene	stack gas, 5-1, VOST	GCMS	mg/Nm3	< 0.00231	0.00862 @	< 0.00218		+ 	*
Benzoic acid	bottom ash	GCMS(8270)	6/67	< 4.90		< 4.90	< 4.90	< 5.00	,
Benzoic acid	bottom ash stuice water	GCMS(8270)	ոց/Լ	< 52.0		< 53.0	< 54.0	< 56.0	
Benzoic acid	collected fly ash	GCMS(8270)	6/6n	< 5.00		< 4.90	< 4.80	< 5.00	
Benzoic acid	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0498	< 0.0419	< 0.0432		- 	
Benzoic acid	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	0.106 @	0.0400	< 0.0251			
Benzoic acid	sluice water supply	GCMS(8270)	ng/L	< 52.0		< 52.0	< 53.0	< 52.0	
Benzoic acid	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0 0265	< 0.0278	< 0.0267	+ ; ;	 	
Benzoic acid	stack gas, vapor phase	GCMS(8270)	EmN/gm	0.105 @	0.0983 @	0.168			

83
C,
0
ł
8

Table C-1: Site 16 OFA Data NOT USED in Calculations

	Stream	Method	MON	Run 1	Run 2	Run 3	Aun 3D	Run 4	Run 5
bottorn ash		GCMS(8270)	6/6n	< 0.980	-	< 0.980	< 0.990	0.990	
bottom ash stuice	e waler	GCMS(8270)	ug/L	< 10.0		< 11.0	< 11.0	< 11.0	
collected fly ash	:	GCMS(8270)	6/8n	066.0 >		066 0 >	< 0.970	 1.00 1.00 	
high dust gas, sol	id phase	GCMS(8270)	m9/Nm3	< 0.00997	< 0.00837	< 0.00863			
high dust gas, vap	or phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0 00503		•	
sluice water supp	<u>v</u>	GCMS(8270)		< 10.0		< 10.0	< 11.0	< 10.0	
stack gas, solid pl	nase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
stack gas, vapor p	hase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
bottom ash		GCMS(8270)	6/6n	< 0.980		< 0.980	066 0 >	066.0 >	7
bottom ash sluice	vater	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
collected fly ash		GCMS(8270)	6/6n	< 0.990		066.0 >	< 0.970	< 1.00	:
high dust gas, solid	d phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863	+ ; ; ;	+ !	
high dust gas. vap	or phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	 		
sluice water supply		GCMS(8270)	<u>ug/L</u>	< 10.0		< 10.0	< 11.0	< 10.0	
stack gas, solid phi	856	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
stack gas, vapor ph	a 5e	GCMS(8270)	mg/Nm3	< 0.00530	< 0.005555	< 0.00534		3	
bottom ash	1	GCMS(8270)	<u>0/6</u>	< 0.980	,	< 0.980	066 0 >	066.0 >	
bottom ash sluice wa	Iter	GCMS(8270)	u p /L	< 10.0	•••• •	< 11.0	< 11.0 <	< 11.0	t
coltected fly ash		GCMS(8270)	5,6,7	< 0.990	 	< 0.990	0/6/0 >	< 1.00	
high dust gas, solid p	hase	GCMS(8270)	EmN/8m	< 0.00997	< 0.00837	< 0.00863			5
high dust gas, vapor	phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503	 		
shrice water supply		GCMS(8270)	10/L	< 10.0		< 10.0	< 11 0 <	< 10.0	
stack gas, solid pha		GCMS(8270)	CmN/bm3	< 0.00530	< 0.00555	< 0.00534		<u>.</u>	
stack gas, vapor ph	356	GCMS(8270)	EmN/Bm	< 0.00530	< 0.00555	< 0.00534		, ;	
bottom ash sluice	vater	ICP-AES	mg/L	1		< 0.600	< 0.600	< 0 600	< 0 600
high dust gas, vap	or phase	ICP-AES	mg/Nm3			0.538 @		0.600	0.624 @
stuice water supply		ICP-AES	mg/L			< 0.600	< 0.600	< 0.600	< 0.600
stack gas, vapor pi	nase	ICP-AES	mg/Nm3			0.837 @		0 805 @	0 883 @
bottom ash	<u>-</u> : :	NAA	6/6n				 	< 3.93	
coal	!	NAA	ug/g, dry	 i		3.52	4.92	1.79	5.16
collected fly ash		NAA	B/5 n					6.58	!
high dust gas, parti	culates	NAA	<u>10,01</u>					34.0	
stack gas, particula	tes	NAA	6/6n					3.90 R	
high dust gas, 1-L	VOST (GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
high dust gas, 20 -	L VOST	GCMS	CmN/gm	< 0.000525	< 0.000543	< 0.000559			
high dust gas, 5-L	VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
stack gas, 1-L VO	IST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
stack gas, 20-L V	/0ST (GCMS	mg/Nm3	< 0.000580	< 0.000569	< 0.000542			
stack gas, 5-1 V(DST (GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			

Appendix C: Analytical Data Not Used in Calculations

~

Table C-1: Site 16 OFA Data NOT USED in Calculations

8

Substance	Stream	Method	MON	Bun -	Run 2	Bun 3	Run 30	Run 4	Run 5
			;						
Bromolorm	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0 00887	< 0.0139	< 0.0105			
Bromoform	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Bromotorm	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110	- +		
Bromotorm	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218	· :		
Bromomethane	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105		,	
Bromomethane	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Biomomethane	stack gas, 1 - L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Brothomethane	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			
Cadmium	bottom ash	ICP-AES	mg/kg			< 5.00	< 5 00	< 5.00	< 5 00
Cadmium	bottom ash	NAA	6/6n			i		< 100	
Cadmium	bottom ash sluice water	ICP-AES	mg/L			< 0.00500	< 0.00500	< 0.00500	< 0.00500
Cadmium	coal	NAA	ug/g, dry			< 2.49	A 3.28	2.88	< 2.59
Cadmium	collected fly ash	ICP-AES	mg/kg	 !		< 5.00	< 5.00	< 5.00	< 5 00
Cadmium	collected fly ash	NAA	6/6n	 1 1 1		!		< 22.9	
Cadmium	high dust gas, particulates	ICP-AES	mg/kg			< 4.41 R		< 5.10 H	< 5.25 B
Cadmium	high dust gas, particulates	NAA	6/6n		•			< 25 4	
Cadmium	high dust gas, solid phase	ICP-AES	mg/Nm3			< 0.0293 B		< 0.0263 H	< 0.0287 H
Cadmium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.00217	 !	< 0.00233	< 0.00225
Cadmium	sluice water supply	ICP-AES	mg/L			< 0.00500	< 0.00500	< 0.00500	< 0 00200
Cadmium	stack gas, particulates	ICP-AES	mg/kg			< 6.75		< 4.15	< 5.67
Cadmium	stack gas, particulates	NAA	6/6n		 		;	4 30	
Cadmium	stack gas, solid phase	ICP-AES	EmN/6m			1 28000.0 >		< 0.00145	< 0.00147
Cadmium	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.000834		_ < 0.00192	< 0 00170
Calcium	bottom ash	ICP-AES	mg/kg		+ !	7330	7260	5980	3700 @
Calcium	bottom ash	NAA	ng/g	 : : ; ;			 :	< 10000	
Calcium	coal	ICP-AES	mg/kg. dry			<u>6</u> 44	1011	1254	961
Calcium	collected fly ash	ICP-AES	mg/kg			7520	7650	4460 @	4570 @
Catcium	collected fly ash	NAA	- <u>6/6</u> -		 ,	i ;		< 12653	
Całcium	high dust gas, particulates	NAA	6/ðn					< 13589 R	
Calcium	stack gas, particulates	NAA						2171 B	
Carbon disulfide	high dust gas, 1-t. VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
Carbon disulfide	high dust gas, 5-1 VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Carbon disulfide	stack gas 1-1 VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Carbon disuffide	stack gas, 5 - t vOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218	• • •		
Carbon tetrachtoride	high dust gas, 1-L VOST	GCMS	CmN/gm	< 0.00887	< 0.0139	< 0.0105			
Carbon tetrachloride	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Carbon tetrachloride	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Carbon tetrachforide	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0 00227	< 0.00218			
Cerium	bottorn ash	NAA	<u></u>			- 4		94.4	

Appendix C: Analytical Data Not Used in Calculations

6
-T
Ö
ł
g

Table C-1: Site 16 OFA Data NOT USED in Calculations

6

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Cerium	coal	NAA	ug/g. dry			28.9	23.3	21.4	21.9
Cerium	collected fly ash	NAA	6/6n					161	
Cerium	high dust gas, particulates	NAA	6/6n	, , ,		;		82.0 R	
Cerium	stack gas, particulates	NAA	6/6n					47.B.H	
Cesium	bottom ash	NAA	6/6n					6.51	
Cesium	coal	NAA	ug/g, dry			1.30	1.01	1.07	0.943
Cesium	collected fly ash	NAA	6/6n					12.5	
Cesium	high dust gas, particulates	NAA	6/6n				· · ·	3.73 R	
Desium	stack gas, particulates	NAA	6/6n		 		+ :	2.76 R	
Chtoride	collected fly ash	<u>0</u>	mg/kg			9.16			!
Chlorine	bottom ash	NAA	6/6n					< 411	
Chlorine	coat	NAA	ug/g. dry			< 95.6	365	574	402
Chlorine	collected fly ash	NAA	6/6n					< 324	· · · · · · · · · · · · · · · · · · ·
Chtorine	high dust gas, particulates	NAA	6/6n					1957 RI]
Chforine	stack gas, particulates	NAA	0/0n		· · · · · · · · · · · · · · · · · · ·			< 97.2	
Chlorobenzene	high dust gas, 1-L VOST	GCMS	CmN/gm	< 0.00887	< 0 0139	< 0.0105			
Chlorobenzene	high dust gas, 5-L VOST	GCMS	CmN/pm	< 0.00202	< 0.00211	< 0.00226			
Chlorobenzene	stack pas. 1-1 VOST	GCMS	ma/Nm3	< 0.0115	< 0.014	< 0110			
horobenzene	stack das 5-t VOST	GCMS	ma/Nm3	- 0 0031	0.0027	81000			
Chloroathana	high duct ras 1-1 VOST	SUNS		202000					
							J		
nioroetnane	ngn dust gas, 5-L VUST	GCMS	EUN/Gm	< 0.00202	< 0.00211	< 0.00226		:	-
hloroethane	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
hioroethane	stack gas, 5-1 VOST	GCMS	EmN/Bm	<pre>< 0.00231</pre>	< 0.00227	< 0 00218			
hlorolorm	high dust gas, 1-1 VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
thoroform	high dust gas, 5-1 VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			 - -
hloroform	stack gas, 1-L VOST	GCMS	EmN/gm	< 0.0115	< 0.0114	< 0.0110		 - 	· · ·
thoroform	stack gas, 5-L VOST	GCMS	CmN/gm	< 0.00231	< 0.00227	< 0.00218			
thoromethane	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0 00887	< 0.0139	< 0.0105			
chloromethane	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226		:	
hloromethane	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
hloromethane	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00238			
hromium	bottom ash	NAA	ng/g				•	9.0	 - - - -
homium	coal	ICP-AES	mg/kg drv	• • • • • • • • • • • • • • • • • • •	·	23.2	319	48.1	10.5
homium	collected fly ash	NAA	6/6n		:	1 1 1 1		136	
hromium	high dust gas, particulates	NAA	p/bn				-	A A P	
homium	stack das particulates	NAA					1		
s 1 2 - Dichtoroathana	high duet dae 1 -1 VOST	GMS	L mN)om	< 0.00087	01100	10.010E			
		onco							
		OF CO			240000 V	ACCOULD >	-		
Is-1.2 - UIChioroethene	I high dust gas, 5-L VOSI	GCMS		< 0.00202	< 0.00211	< 0.002261	•		

Appendix C: Analytical Data Not Used in Calculations

C-10

Table C-1: Site 16 OFA Data NOT USED in Calculations

₽

Run 5						}	•	• • •	3.96 @			- · · · · · · · · · · · · · · · · · · ·		< 53.9																								· · ·	
Run 4						!	;	29.9	6.41	619	27.8 R	21.0 R	< 438	57.2	< 364	< 558	< 112			. <u> </u>	, ,	 .		< 0 990	< 11.0	< 1 00			< 10.0			< 0.990	< 11.0	< 1.00			< 10.0		
Run 3D									4.62 @	, ,				80.8										066.0 >	< 11.0	< 0.970			< 11.0			066.0 >	< 11.0	< 0.970			< 11.0		
Run 3	< 0.0110	< 0.000542	< 0.00218	< 0.0105	< 0.00226	< 0.0110	< 0.00218	-	6.51					80.0				< 0.0105	< 0 000559	< 0.00226	< 0.0110	< 0.000542	< 0.00218	< 0.980	< 11.0	< 0.990	< 0.00863	< 0 00503	× 10 0	< 0.00534	< 0.00534	< 0.980	< 11.0	0.990	< 0.00863	< 0.00503	< 10.0	< 0.00534	< 0.00534
Run 2	< 0.0114	< 0.000569	< 0.00227	< 0.0139	< 0.00211	< 0.0114	< 0.00227											< 0.0139	< 0.000543	< 0.00211	< 0.0114	< 0.000569	< 0 00227		 -		< 0.00837	< 0.00488		< 0.00555	< 0.00555				< 0 00837	< 0.00488		< 0.00555	< 0.00555
Run 1	< 0.0115	< 0.000580	< 0.00231	< 0 00887	< 0.00202	< 0.0115	< 0.00231											< 0.00887	< 0.000525	< 0.00202	< 0.0115	< 0.000580	< 0.00231	< 0.980	< 10.0	< 0.990	× 0 00997	< 0.00581	< 10.0	< 0.00530	< 0 00530	< 0.980	< 10.0	< 0.990	< 0.00917	< 0.00581	< 10.0	< 0.00530	< 0.00530
NON	mg/Nm3	EmN/gm	mg/Nm3	mg/Nm3	mg/Nm3	mg/Nm3	mg/Nm3	6/6n	mg/kg. dry	6/6n		6/6n	8/8n	ug/g, dry	5/Bn	6/ 5 л	6/6n	CmN/6m	mg/Nm3	mg/Nm3	mg/Nm3	EmN/6m	mg/Nm3	6/6n	<u>1/6</u>	5/6n	mg/Nm3	mg/Nm3	т. т. Т/бл	mg/Nm3	mg/Nm3	5/6n	<u>ug/t</u>	. <u>6,6n</u>	EmN/6m	CmN/gm	ug/L	mg/Nm3	mg/Nm3
Method	GCMS	GCMS	GCMS	GCMS	GCMS	GCMS	GCMS	NAA	ICP-AES	NAA	NAA	NAA	NAA	NAA	NAA	NAA	NAA	GCMS	GCMS	GCMS	GCMS	GCMS	GCMS	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)	GCMS(8270)
Stream	stack gas, 1-L VOST	stack gas, 20 - L VOST	stack gas, 5-L VOST	high dust gas. 1-L VOST	high dust gas, 5-L VOST	stack gas, 1 – L VOST	stack gas, 5-L VOST	bottom ash	coal	collected fly ash	high dust gas, particulates	stack gas, particulates	bottom ash	coal	collected fly ash	high dust gas, particulates	stack gas, particulates	high dust gas, 1-L VOST	high dust gas, 20-L VOST	high dust gas, 5- t VOST	stack gas. 1 - L VOST	stack gas, 20 - L VOST	stack gas, 5- t. VOST	bottom ash	bottom ash stuice water	collected fly ash	high dust gas, solid phase	high dust gas, vapor phase	stuice water supply	stack gas, solid phase	stack gas, vapor phase	bottom ash	bottom ash sluice water	collected fly ash	high dust gas, solid phase	high dust gas, vapor phase	sluice water supply	stack gas, solid phase	stack gas, vapor phase
Substance	s - 1,2 - Dichloroethene	s - 1,2 - Dichloroethene	s - 1,2 - Dichkoroethene	s – 1, 3 – Dichloropropene	s - 1.3 - Dichloropropene	s-1,3 - Dichloropropene	s - 1,3 - Dichloropropene	obalt	obalt	obalt	obalt	obalt	opper	opper	opper	opper	pper	bromochloromethane	bromochloromethane	bromochloromethane	bromochloromethane	bromochloromethane	bromochloromethane	ethylphthalate	ethylphihalate	ethylphthatate	ethylphthalate	ethytphthalate	ethytphthalate	ethylphthalate	ethylphthatate	methylphenethylamine	methytphenethylamine	methylphenethylamine	methylphenethytamine	methylphenethylamine	methylphenethylamine	methylphenethylamine	methylphenethylamine

30-Oct-93	F	able C-1:	Site 16 OFA D	ata NOT USE	D in Calculatic	Suc			
Substance	Stream	Method	Mon	Run 1	Run 2	Run 3	Run 3D	Bun 4	Run 5
Di - A - Actual Helete	to the set	GCMC/8070V	v /on	080 0 1			000 0		
Di-n-octvinhthalate	bottom ash strice water	GCMS(8270)	2,01	100			011 /		
Di-n-octylohthalate	collected fly ash	GCMS(8270)	B/Bn	066.0 >		066.0 >	0.970 >	- 100 - 100	
Di – n – octylphthalate	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0997	< 0.00837	< 0.00863			
Di - n - octylphthatate	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Di-n-octylphthatate	sluice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	· · · · · · · · · ·
Di – n – octylphthalate	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534] ; ;
Di - n - octylphthalate	stack gas. vapor phase	GCMS(8270)	mg/Nm3	< 0 00530	< 0.00555	< 0.00534			
Ethyl benzene	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105	•	 '	
Ethyl benzene	high dust gas, 5- L VOST	GCMS	mg/Nm3	< 0 00202	< 0.00211	< 0.00226			:
Ethyl benzene	stack gas, 1 ~ L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Ethył benzene	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218	 	: ! !	
Ethyl methanesultonate	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
Ethyl methanesultonate	bottom ash stuce water	GCMS(8270)	ug/L	< 10.0		11.0	< 11.0 11.0	< 11.0	:
Ethyl methanesultonate	collected fly ash	GCMS(8270)	5/6n	066.0 >		< 0 990	< 0.970	× 1.00	: ;
Ethyl methanesultonate	high dust gas, solid phase	GCMS(8270)	CmN/gm	< 0.00997	< 0.00837	< 0.00863			
Ethyl methanesulfonate	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0 00581	< 0.00488	< 0.00503			
Ethyl methanesultonate	skuice water supply	GCMS(8270)	ng/L	< 10.0		< 10.0	< 11.0	< 10.0	
Ethyl methanesulfonate	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		i 	
Ethyl methanesultonate	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	- - - -		
Europium	bottom ash	NAA	0/6n				-	- 	
Europium	coal	NAA	ug/g. dry			0.537	0 349	76E.0	0.408
Europium	collected fly ash	NAA	6/6n	 			 	1.95	
Europium	high dust gas, particulates	NAA	6/6n					0.668 R	· · · ·
Europium	stack gas, particulates	NAA	6/60					0.393 R	<u></u>
Gold	coal	NAA	ug/g. dry			< 0.00104	< 0.00104	< 0.00104	< 0.00105
Hafnium	bottom ash	NAA	5/57			- - -		5.04	
Hafnium	coal	NAA	λip -β/ðn			1.20	0.775	0.788	0.728
Hafnium	collected fly ash	NAA	b/6 n	, , ,	 	 		8.71	
Hafnium	high dust gas, particulates	NAA						2.57 B	
Hathium	stack gas, particulates	NAA	6/6n					1.54 B	
lodine	bottom ash	NAA	6/6n					< 10,6	
todine	coal	NAA	ug/g. dry			< 2.26	< 2.18	1.77	< 1.78
lodine	collected fly ash	NAA	6/6 n					< 9.36	 1
lodine	high dust gas, particulates	NAA	6/6 n	 				< 14.4	
lodine	stack ges, particulates	NAA	ng/g					< 2.63	
lron	bottom ash	ICP-AES	mg/kg			90500	90600	91600	82600
lton	bottom ash	NAA	6/Bn					62080	
Iron	coal	ICP-AES	mg/kg. dry			8605	5705	8211	6304

Appendix C: Analytical Data Not Used in Calculations

Ξ

₽

Table C-1: Site 16 OFA Data NOT USED in Calculations

30-Oct-93

Substance	Strea	Method	WON	ftun t	Run 2	Run 3	Run 3D	₩ Un H	Run 5
							-		
	collected fly ash	ICP-AES	mg/kg			19600	17600	68100	69900
	collected fly ash	NAA	6/6n			- +		90531	
	high duet and modified ates	NAA	110/0					36211 H	
(OU					 			26496 R	
ron	stack gas, particulates		Alfan		· ·	:	+ - - -	105	
antharum	bottom ash	NAA	5/6n			· · · · · · · · · · · · · · · · · · ·			
antharum	coal	NAA	Vib. 6/2			10.8	13.9	-1 <u>-1</u>	
	collected fly ash	NAA	ng/g					90.1	
	List that are restricted	NAA	ua/a		•			51.1	
anthanum	Ligh ous day bar baine				- - -		-	17.4 B	
anthamum	stack gas, particulates		- RiAn			•		0.693	
utetium	bottom ash	NAA	<u>6/6n</u>				0.452	0.150	0 115
utetium	coal	NAA	10.01V		·				
uteticum	collected fly ash	NAA	- <u>6/6</u> -					71	
	high dust gas, particulates	NAA	n9/9	 , 1	,		ì	0 592 H	
	tack das particulates	NAA	6/6n		-	 	;	0.301 8	
		ICP-AES	ma/ka			4500 @	4480 @	4130 @	
Magnesium		NAA						7395	
Magnesium					 	4885	4945	4885	4583
Magnesium	Dollom asn		Ru Ru	 	! 	2 R2 @	287 @	3.02 @	2.86 @
Hagnesium	bottorn ash sluice water					0.995		504	@ 7 21
Magnesium	coal	ICP-AES	mg/kg. dry	***					
Magnesium	coal	NAA	1 10, dry		: : :	0/9		+DC	n (
Magnesium	collected fiy ash	ICP-AES	mg/kg			3670 @	2940 @	23000 @	
Madrasiittii	collected fly ash	NAA	6/6n					6591	
	collected fiv ash	XRF	mg/kg			5488	5246	5126	4583
	high dust gas particulates	ICP-AES	mg/kg			4744 B		4745 @R	3359 @ F
Magreenuit	biob duct nas narticulates	NAA	0/00				· · · · · · · · · · · · · · · · · · ·	2827 R	
		IC P AFS	ma/Nm3	-		31.6 R		24.5 @R	18.4 @R
Magnesium		ICD AFS	mo/Nm3		 	< 0.435		< 0.466	< 0 450
Magnesium	high dust yas, vaput pilasa					3.56.6	3610	3.44 @	3.46 @
Magnesium	sluice water supply	CP-AES				a Cooot		2015/610	14516
Magnesium	stack gas, particulates	ICP-AES	mg/kg			1 A A A A A A A A A A A A A A A A A A A			
Magnesium	stack gas, particulates	NAA		•	:	;		H 2401	
Machashim	stack gas, solid phase	ICP-AES	mg/Nm3			0.473 @H	1	1.02 @H	0 838 0
	stack das. vapor phase	IC P-AES	mg/Nm3	:		< 0.366		< 0.384	< 0.341
	hottom ach	NAA	na/a					124	
Mangarese		X DE				792	792	792	720
Manganese					•	661	4 61	1012	13.8
Manganese	coal	(ICP - AES	And Ballon			2		2	
Manganese	collected fly ash	NAA	6/67					201	
Manganese	collected fly ash	XAF	mg/kg			864		N21	26
Mandarese	high dust gas, particulates	NAA	B/Bn					46.1 H	
Mannangsa	stack gas, particulates	NAA	6/6n					40.6 H	

	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
2003/01/20					' 	0.0450	< 0.0450	< 0.0450	< 0.0450
	bottom ash	CVANS	mg/kg		• • •			< 0.701	
	bottom ash	NAA	<u>8/8n</u>			OSIC C	- 0.046A	< 0.0470	< 0.0473
	Icoal	CVAS	mg/kg. dry			0.175	< 0.228	0.237	< 0.179
	coal	NAA	<u>vip. 6/gu</u>				626.0	0.248	0.248
	contected fly ash	CVAS	mg/kg			402.0		116	
Mercury	collected fly ash	NAA	.6/Bn						1
Merculty	high duet ros nationates	NAA	ő/Bn						
Mercury		NAA	04/9					AU. 04	
Mercuty	Stack gas, particularies	GCMS(8270)	ua/a	< 0.980	- +	0.980	0660 >		
Methyl methanesulfonale	bottom asn	COMPANY OF	- Non	< 10.0		A 11.0	× 11.0	0.11 v	
Methyl methanesulfonate	bottom ash sluice water	COMPLEX O		066.0 >		066.0 >	< 0.970	00 V	
Methyl methanesulfonate	collected fty asn		EmN/om	< 0.0097	< 0.00837	< 0.00863			
Methyl methanesultonate	high dust gas, solid phase	10 100 100 101 01		< 0.0581	< 0.00488	< 0.00503	:		*****
Methyl methanesulfonate	high dust gas, vapor phase	GCMS(82/U)				< 10.0	< 11.0	. < 10.0	
Methyl methanesultonate	sluice water supply	GCMS(82/U)		0.00530	 0.00555 	< 0.00534			1
Methyl methanesulionale	stack gas, solid phase	GCMS(8270)		00000	0.00555	< 0.00534			i
Muttin methodenificiate	stack das, vapor phase	GCMS(8270)	mg/Nm3			3010 0			
	high dust new 1-1 VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139			:	
Methylene chloride	TSON 1-2 and the Long to the L	GCMS	mg/Nm3	< 0.00202	< 0.00211	0.00589 @			
Methylene chloride		SMCO	ma/Nm3	< 0.0115	< 0.0114	0.208			
Methylene chloride	stack gas, 7-1, VOS	SMCG	ma/Nm3	< 0.00231	0.125	0.0392			
Methylene chloride	stack gas, 3-L VUSI		0/01					- 12.4	
Molybdenum	bottom ash		- Alto day	F I I		< 5.21	< 5.21	6.08 @	< 2 5
Molvbdenum	coal		in By Bu	-				< 12.5	;
	collected fly ash	NAA	6/67				·	< 58 4 B	
Mothdamm	high dust gas, particulates	NAA	<u>19/9</u>		1	I		10.7 R	1
	stack das. particulates	NAA	n9/9						
	hink dust das 1-L VOST	GCMS	mg/Nm3	< 0.00887	ecto.0 >				
		GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
m.pXylene		GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
m.p - Xylene		SNUS	ma/Nm3	< 0.00231	< 0.00227	< 0.00218			-
m.p-Xylene	stack gas, 2 - L VOSI								
Neodymium	boltom ash					21.2	15.1	14.6	13
Neodymium		NAA		-				131	:
Neodymium	collected fly ash	NAA	<u> </u>					43.6	
Neodymium	high dust gas, particulates	NAA	6/6n					20.3 F	
	stack gas, particulates	NAA	<u> 6/6</u>			-		191	
Nickel	bottom ash	NAA	<u>5/6</u> n				37.9	62.8	18
	coal	ICP-AES	mg/kg, dry					<pre>159</pre>	
	collected fly ash	NAA	6/6n					52.3 F	
	high dust das, particulates	NAA	6/6 n						, _ ↓
			I	-				126	

Appendix C: Analytical Data Not Used in Calculations

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Aun 4	Run 5
N - Nitroso - di - n - butylamine	bottom ash	GCMS(8270)	6/ <u>6</u> n	< 0.980		< 0.980	066.0 >	066.0 >	
N - Nitroso - dí - n - hutMamine	bottom ash sluice water	GCMS(8270)	1/0 n	< 10.0		<pre>< 11.0</pre>	< 11.0	< 11.0 11.0	
N - Nitroso - di - n - butvlarnine	collected fiv ash	GCMS(8270)	6/Bn	< 0.990		< 0.990	0/6.0 >	< 1.00	
N - Nhroeo - di - n - but da mine	high dust gas, solid phase	GCMS(8270)	EmN/gm	< 0.00997	< 0 00837	< 0.00863			
N - Mitcoo - di - n - hutdamine	high dust gas vapor phase	GCMS(8270)	CmN/pm	< 0.00581	< 0.00488	< 0.00503			
	strice water supply	GCMS(8270)		< 10.0		< 10.0	< 11.0	< 10.0 <	
	stack gas solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
N - Mitcool - di - n - but da mine	stack das vanor phase	GCMS(8270)	EmN/gm	< 0.00530	< 0.00555	< 0.00534			
		GCMS(8270)	ua/a	< 0.980		< 0.980	066.0 >	066.0 >	
N - Mitcoodioropyanina	hottom ash shrice water	GCMS(8270)	J/Bn	< 10.0		< 110	< 11.0	< 11.0	
N Nitrosodintonvisnine	collected fly ash	GCMS(8270)	6/6n	066.0 >		< 0.990	< 0.970	< 1.00	- - -
	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.0097	< 0.00837	< 0.00863			
	high dust gas vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
		GCMS(8270)	uo/L	< 10.0		< 10.0	< 11.0	< 10.0	
	and the solid these	GCMS(8270)	ma/Nm3	< 0.00530	< 0 00555	< 0.00534			
		GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
		GCMS(8270)	na/a	< 0.980		< 0.980	066.0 >	< 0.990	,
N - Nitrosopiperialne		00100 (02.0)		1001 >		< 11.0	< 11.0	< 11.0	:
N - Nitrosopiperidine	bottom ash stuice water	CCMS(0210)				- 066 0 >	- 0.970 >	< 1.00	
N - Nitrosopiperidine	collected ny ash	10 20 20 10	- <u>6/8</u> 7						
N - Nitrosopiperidine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	26600.0 >	100001 >	< 0.00803			
N – Nitrosopiperidine	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			:
N - Nitrosopiperidine	sluice water supply	GCMS(8270)	<u>ug/L</u>	< 10.0		< 10.0	< 11.0	< 10.0	
N - Nitrosopiperidine	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534		:	i
N - Nitrosoniberidine	stack das, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
o - Xvlane	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
o	high dust gas, 5-L VOST	GCMS	EmN/gm	< 0.00202	< 0.00211	< 0.00226			
oXvlana	stack das. 1 - L VOST	GCMS	EmN/Bm	< 0.0115	< 0.0114	< 0.0110			
o - Xviene	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			
o - Chloroaniline	bottom ash	GCMS(8270)	₿/đn	< 0.980		< 0.980	< 0.990	< 0.990	
n - Chloroanilina	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0	 	< 11.0	< 11.0	< 11.0	
P	collected fly ash	GCMS(8270)	6/ 5 n	< 0.990		< 0.990	< 0.970	< 1.00	
p. Chloroantine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
P - Chloroaniine	hidh dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
	shrice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
	etack nas solid Dhase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
	stark das vanor ohase	GCMS(8270)	ma/Nm3	< 0.00530	< 0.00555	< 0.00534			
	bottom ach	GCMS(8270)	no/a	< 0.980		< 0.980	< 0.990	066.0 >	
Pentachioropenzene		COMS(8270)		< 10.0		< 11.0	< 11.0	< 11.0	
Pentachlorobenzene	DOGOTI ASII SIUGE WANT	00000000000000000000000000000000000000			-		01970		
Pentachlorobenzene	collected fly ash	GCM2(02/0)	<u>6/8</u>	1 1 2 2 1		2002			

_...

Appendix C: Analytical Data Not Used in Calculations

ട്ട	
1	
ರ	
Ó	
Ĩ.	
0	
ന	

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
					1				
Pentachlorobenzene	high dust gas, solid phase	GCMS(8270)	mg/Nm3 .	< 0.00997	< 0.00837	< 0.00863			
Pentachlorobenzene	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Pentachlorobenzene	stuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Pentachlorobenzene	stack gas. solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pentachlorobenzene	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534	-	• :	-
Phenacetin	bottom ash	GCMS(8270)	6/6n	< 0.980		< 0.980	< 0.990	< 0.990	
Phenacetin	bottom ash stuice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	< 11.0	
Phenacetin	collected fly ash	GCMS(8270)	B/6n	066.0 >	1	066.0 >	< 0.970	× 1.00	
Phenacetin	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863			
Phenacetin	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0.00503			
Phenacetin	stuice water supply	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	
Phenacetin	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Phenacetin	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Phosphate	collected fly ash	Spectrophot.	mg/kg			< 160			
Phosphate	high dust gas, solid phase	Spectrophot.	mg/Nm3			0.862 @		0.914	1.14
Phosphate	stack gas, solid phase	Spectrophot.	mg/Nm3			< 0.0375		< 0.0402	< 0.0573
Phosphorus	bottom ash	ICP-AES	mg/kg			< 300	< 300	< 300	< 300
Phosphorus	coal	ICP-AES	mg/kg. dry			< 31.3	< 31.2	< 31.3	< 31.5
Phosphorus	coat	ICP-AES	mg/kg. dry			30.2 @	38.0	@ 06 F	10.7
Phosphorus	collected fly ash	ICP-AES	mg/kg			765 @	764 @	< 300	< 300
Potassium	bottom ash	ICP-AES	mg/kg			18400	18230	19100	19100
Potassium	bottom ash	NAA	6/6n					< 31691	
Potassium	bottom ash	XAF	mg/kg			19508	19342	11539	19425
Potassium	bottom ash sluice water	ICP-AES	mg/L			3.00	< 3.00	3.02 @	< 3.00
Potassium	coal	ICP-AES	mg/kg, dry			1241 @	1040 @	1964	1261 @
Potassium	coal	NAA	ug/9. dry			2463	2400	1833	2046
Połassium	collected fly ash	ICP-AES	0%) B		-	16200	15500	17300	17200
Polassium	collected fly ash	NAA	6/ðn					28764	
Potassium	collected fly ash	XRF	<u>64/6</u> m			20671	20505	22829	21667
Potassium	high dust gas, particulates	ICP-AES	mg/kg			17544 R		18920 H	19531 R
Potassium	high dust gas. particulates	NAA	6/Bn					< 42562	
Potassium	high dust gas, solid phase	ICP-AES	CmN/gm			117 A	-	97.7 R	107 B
Potassium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 1.30		< 1.40	< 0.568
otassium	sluice water supply	IC P - AES	mg/L			3.54 @	3.39 @	353@	3.27 @
otassium	stack gas, particulates	ICP-AES	mg/kg			14773 @R		12441 @R	14808 @R
botassium	stack gas, particulates	NAA	6/6n					< 6173	
otassium	stack gas, solid phase	ICP-AES	mg/Nm3			1.91 @H		4.35 @R	3.85 @R
otassium	stack gas, vapor phase	ICP-AES	mg/Nm3			< 0.500		< 0.661	< 1.02
ronamide	bottom ash	GCMS(8270)	B/6л	< 0.980		< 0.980	< 0.990	066.0 >	

Appendix C: Analytical Data Not Used in Calculations

15

C-16

16

Table C-1: Site 16 OFA Data NOT USED in Calculations

30--Oct--93

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Pronamide	bottom ash sluice water	GCMS(8270)	ng/L	< 10.0		< 11.0	< 11.0	<pre>< 11.0</pre>	
Pronamide	collected fly ash	GCMS(8270)	6/67	066.0 >	• 	066 0 >	< 0.970	< 1.00	
Pronamide	high dust gas, solid phase	GCMS(B270)	mg/Nm3	10:00997	< 0.00837	< 0.00863			
Pronamide	high dust gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00581	< 0.00488	< 0 00503			
Pronamide	slutce water suppfy	GCMS(8270)	ug/L	< 10.0		< 10.0	< 11.0	< 10.0	I
Pronamide	stack gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			:
Ргопатіde	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0.00530	< 0.00555	< 0.00534			
Pyridine	bottom ash	GCMS(8270)	6/6n	0.980 >		< 0.980	< 0.990	066.0 >	
Pyridine	bottom ash sluice water	GCMS(8270)	ոց/Լ	< 10.0		< 11.0	< 11.0	< 11.0	
Pyridine	collected fly ash	GCMS(8270)	6/6n	066.0 >		066.0 >	< 0.970	< 1.00	
Pyridine	high dust gas, solid phase	GCMS(8270)	mg/Nm3	< 0.00997	< 0.00837	< 0.00863		 :	-
Pyridine	high dust gas, vapor phase	GCMS(8270)	CmN/gm	< 0.00581	< 0.00488	< 0.00503			
Pyridine	sluice water supply	GCMS(8270)	ng/L	< 10.0		< 10.01 >	< 11.0	< 10.0	
Pyridine	stack gas, solid phase	GCMS(8270)	CmN/Bm	< 0.00530	< 0.00555	< 0.00534			
Pyridine	stack gas, vapor phase	GCMS(8270)	mg/Nm3	< 0 00530	< 0.00555	< 0.00534			
Aubidium	bottom ash	NAA	6/6n					85.6	
Rubidium	coal	NAA	V10.9/84			17.71	13.2	14.3	12
Rubidium	collected fly ash	NAA	6/6n					137	1
Rubidium	high dust gas, particulates	NAA	6/8n					48.5 R	:
Rubidium	stack gas, particulates	NAA	6/6n					31.7 R	•
Samarium	bottom ash	NAA	<u>6/6</u> n					11.8	:
Samarium	coat	NAA	<u>ν. dry</u>	 		174	2.22	1 83	
Samarium	collected fly ash	NAA	6/67					13.2	
Samarium	high dust gas, particulates	NAA	ug/9					8.13 H	
Samarium	stack gas, particulates	NAA	6/6n				-	4.44 B	
Scandium	bottom ash	NAA	ng/9			· · · · · · · · · · · · · · · · · · ·	.	22.4	
Scandium	coal	NAA	ug/g, dry			5.70	4 45	5.12	4 30
Scandium	collected fly ash	NAA	6/6n				1	41.7	1
Scandium	high dust gas, particulates	NAA	6/6n					17 6 R	
Scandium	stack gas, particulates	NAA	6/6n					10.2 R	
Selenium	bottom ash	NAA						< 2.39	
Selenium	coal	HGAAS	mg/kg. dry	 		1.56 @	1.46 @	1.57 @	2.00 @
Selenium	collected fly ash	NAA	8/8n					16.4	r
Selenium	high dust gas, particulates	NAA	6/6n					29.4	
Selenium	stack gas, particulates	NAA	<u> </u>					41.7	
Silicon	bottom ash	ICP-AES	mg/kg			190000	189000	192000	195000
Silicon	bottom ash	XAF	mg/kg			242599	238158	243113	239888
Silicon	bottom ash sluice water	ICP-AES	mg/L			3.57 @	3.65 @	3.73 @	3.46 @
Siticon	collected fly ash	ICP-AES	mg/kg		 	149200	153000	209000	202200
Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 30	Run 4	Run 5
Cilicon .		YRE	molka	1		00000	300300	-30510	11111
2000	conected iny and		Ru/Am			066007	1 000007	100012	240114
Silicon	high dust gas, particulates	ICL-AES	ву/вш			2/42/2 H		259754 R	279978 R
Silicon	high dust gas, solid phase	ICP-AES	mg/Nm3			1825 R		1341 B	1532 R
Silicon	high dust gas. vapor phase	ICP-AES	EmN/6m			< 0.435		< 0.466	< 0.450
Silicon	sluice water supply	ICP-AES	<u>mg/L</u>			3.91 @	4.01@	3.85 @	387@
Silicon	stack gas, particulates	ICP-AES	mg/kg			370983 R			333978 H
Silicon	stack gas, solid phase	ICP-AES	mg/Nm3			47.9 H			86 9 R
Silicon	stack gas, vapor phase	ICP-AES	mg/Nm3			0.794 @	; ; ;	0.457 @	0.705@
Silver	bottom ash	ICP-AES	mg/kg			< 40.0	< 40.0	< 40.0	< 40.0
Silver	bottom ash	NAA	n9/g					< 3.82	
Silver	bottom ash stuice water	ICP-AES	mg/L			< 0.0100	< 0.0100	< 0.0100	< 0.0100
Silver	coat	ICPAES	mg/kg, dry			< 1.04	< 1.04	< 1.04	< 1.05
Silver	coal	NAA	ug/g. dry			< 0.935	< 0.732	< 0.845	< 0.797
Silver	collected fly ash	ICP-AES	mg/kg			< 40.0	< 40.0	< 40.0	< 40.0
Silver	collected fiv ash	NAA	D/Bn	-				< 7.97	
Silver	high dust gas, particulates	ICP-AES	mg/kg			< 8.81	- -	< 10.4	< 10.7
Silver	high dust gas, particulates	NAA	0/01		· · ·			2.49	
Silver	high dust gas, solid phase	ICP-AES	mg/Nm3		-	< 0.0586		< 0.0538	< 0.0587
Silver	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.00435		< 0.00466	< 0.00450
	stuice water supply	ICP-AES	mg/L			0.0110 @	< 0.0100	< 0.0100	< 0.0100
Silver	stack das perticulates	ICP-AES	ma/ka	+ + -		< 13.5		< 8.30	
Silver	stack das, particulates	NAA	ng/a					910 >	
2 \var	stack cas. solid phase	ICP-AES	ma/Nm3			< 0.00174			~ 0.00205
Silver	stack das, vapor phase	ICP-AES	EmN/em			< 0.00167		< 0.00384	116200.0
	bottom ash	ICP-AFS	mo/ka			2370.@	2580.60	WUU W	
	thattom ash	NAA	ua/a				9	2464	
Sodium.	coal	ICP-AFS	ma/ka drv			260.@	223 @	() 9UL	955
	collected fiv ash	ICP-AES	ma/ka			2320 @	2352 @	0000	
Sodium	collected fiv ash	NAA	ng/a			×	»	2475	
Sodium	high dust gas, particulates	NAA	0/60					< 51694 B	:
Sodium	stack das, particulates	NAA	b/6n					386.81	
Strontium	bottom ash	ICP-AES	mg/kg			1000	956	840	582
Strontium	bottom ash	NAA	6/ 6 n					454	
Strontium	bottom ash	XRF	mg/kg			3721	3467	3636 {	3636
Strontium	bottom ash stuice water	ICP-AES	mg/L			0.168	0.170	0.200	0.182
Strontium	coal	ICP-AES	mg/kg, dry			98.7	96.7	99.2	80.8
Strontium	coal	NAA	ug/g, dry			1.68	68.4	85.8	116
Stontium	collected fly ash	CP-AES	mg/kg			915	858	624	625
steatium	collected fly ash	NAA	0/81	 : :				1339	

Appendix C: Analytical Data Not Used in Calculations

17

ന
=

30-Oct-93

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Bun 4	Aun 5
Strontium	collected fly ash	XRF	mg/kg			3636	3805	2960	2875
Strontium	high dust gas, particulates	ICP-AES	mg/kg			1135 R		893 R	1000 R
Strontium	high dust gas, particulates	NAA	6/6n					355 R.	
Strontium	high dust gas, solid phase	ICP-AES	mg/Nm3			7.55 R	1	4.61 R	547 H
Strontium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.00130		< 0.000785	< 0.00135
Strontium	sluice water supply	ICP-AES	mg/L			0.266	0.268	2.66	0.267
Strontium	stack gas, particulates	ICP-AES	mg/kg			1081 B		756 R	862 A
Strontium	stack gas, particulates	NAA	5/6n					256 R	
Strontium	stack gas, solid phase	ICP-AES	mg/Nm3			0.140 R		0.264 R	0.224 B
Strontium	stack gas, vapor phase	ICP-AES	EmN/6m			< 0.00110		< 0.000661	< 0 00102
Styrene	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0 0139	< 0.0105			
Styrene	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Styrene	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Styrene	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218		-	
Suflate	collected fly ash	0	6√Pm			4650			
Sulfur	bottom ash	XRF	mg/kg		 i i	120	120	120	120
Sultur	collected fiy ash	XRF	mg/kg			841	360	120	120
Tantalum	bottom ash	NAA	6/6n					1.54	
Tantalum	coat	NAA	ug/g. dry			0.436	0.233	0.185	0.256
Tantakum	collected fly ash	NAA	6/ 6 n				4	2.79	:
Tantalum	high dust gas, particulates	NAA	6/6n				+ 1 1	1,05	
Tantalum	stack gas, particulates	NAA	6/Bn					0.685	·····
Terbitum	bottom ash	NAA	5/ <u>8</u> n		+	:		1.23	
Terbium	coal	NAA	ug/g. dry			0.381	0.282	0.263	0.287
Terbium	collected fly ash	NAA	- 6/6n	 - - -				2.56	
Terbium	high dust gas, particulates	NAA	6/6n					1.24 R	
Terbium	stack gas, particulates	NAA	6/6 n			:		0.618 R	
Tetrachloroethene	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
Tetrachloroethene	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Tetrachloroethene	stack gas. 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Tetrachloroethene	stack gas, 5-1 VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			
Thallium	bottom ash	ICP-AES	mg/kg			< 100	< 100	< 100	< 100
Thalkium	bottom ash sluice water	ICP-AES	mg/L			< 0.100	< 0.100	< 0.100	< 0.100
Thallium	coal	ICP-AES	mg/kg. dry			< 10.4	< 10.4	< 10.4	< 10.5
Thallium	collected fly ash	ICP-AES	mg/kg			< 100 <	< 100	< 100	< 100
Thallium	high dust gas, particulates	ICP-AES	mg/kg			< 88.1 R		< 104 B	< 107 R
Thallium	high dust gas, solid phase	IC P-AES	mg/Nm3			< 0.586 A	, 	< 0.538 R	< 0.587 R
Thallium	high dust gas, vapor phase	ICP-AES	mg/Nm3			< 0.0435		< 0.0466	< 0.0450
Thallium	sluice water supply	ICP-AES	mg/L			< 0.100	< 0.100	< 0.100	< 0.100

Appendix C: Analytical Data Not Used in Calculations

Site 16 OFA Data NOT USED in Calculations
Table C-1:

<u>~</u>	
~~	
σ	
0	
ĩ	
2	
₩ #	

~

	mealle	DOUISM				5 IIDH	TC UNH	- 474	
						115		0.58 /	< 113
<u>[hallium</u>	stack gas, particulates		By/Bu		-+	12100		0000 0 ~	2000
Thallium	stack gas, solid phase		CIIIN BIII			00000		Faco O	
Thallium	stack gas, vapor phase	ICP-AES	EmN/Qm			DOCU.U A			
Thorium	bottom ash	NAA	6/ <u>6</u> n					# 0	
Thorium	coal	NAA	ug/g, dry			4.43	<u>9.5</u>	3.42	RO.5
Thorism	collected fly ash	NAA	6/Bn					28.1	
Thorium	high dust gas, particulates	NAA	6/6n			-		13.9 H	
Thorium	stack gas, particulates	NAA	6/65					7.93 H	
tin.	bottom ash	NAA	6/6n			-		< 10.0	
	coal	NAA	ug/g. dry			< 15.6	< 15.6	< 15.7	15.8
	collected fly ash	MAA	6/6n					< 10.01 ×	
	high dust cas particulates	NAA	B/Bn					< 10.0	
	start de harticulates	NAA	0/80					< 10.0	
	bottom ach	HCP AES	ma/ka			7220	7210	7130	7120
		NAA	110/0			1	1	6179	
Itanum	DONOTI #31				-	617	469	601	429
Titanium	coal						0733	0073	
Titaníum	collected fly ash	ICP-AES	m0/kg			0000	0000		0.250
Titanium	collected fly ash	NAA	8/6n		-+			6609	
Titanium	high dust gas, particulates	NAA	6/6n					3924	
Titanium	stack gas, particulates	NAA	6/6n					1751 R	
Tolyane	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
Toluene	hich dust gas. 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226		1	
	stack cas. 1-L VOST	GCMS	EmN/Bm	< 0.0115	< 0.0114	< 0.0110	,		
	stack das. 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			
	high dust cas. 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105	i		; ; ;
	high duet one 20-1 VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0.000559			
		GCMS	ma/Nm3	< 0.00202	< 0.00211	< 0.00226			
		SW09	EmV/om	< 0.0115	< 0.0114	< 0.0110			
		or we		0.000580	< 0.000569	< 0.000542		;	
Irans - 1,2 - Dichloroethene	Stack gas, zu-L VOSI			100001	× 0.0027	< 0.0021B			•
trans-1,2-Uichloroethene	Stack gas, 3-L VO3		C-Now	0.00687	< 0.0139	< 0.0105			
Irans - 1,3 - Lichloropropene		CE CO	Carly Carl	0.000	11200 0 2	< 0.0026			
trans-1,3 ~ Dichloropropene	Thigh dust gas, 3-L VOS	CM00		~ 0.0115	< 0.014	< 0.0110			
trans - 1,3 - Dichloropropene	stack gas, 1-L VUSI	CWOD				0.0010			
trans-1,3-Dichloropropene	stack gas, 5-L VOST	GCMS	Em0/0m3	< 0.00231	< 0.00227	4 0.002 10			
Trichloroethene	high dust gas, 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
Trichloroethene	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Trichlaroethene	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Trichloroathana	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	< 0.00227	< 0.00218			
Tricklorofluoromathane	high dirst das 1-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			

Appendix C: Analytical Data Not Used in Calculations

19

20

30-Oct-93

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	MON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
Trichlorofluoromethane	high dust gas, 20-L VOST	GCMS	mg/Nm3	< 0.000525	< 0.000543	< 0 000559	+- 		
Trichlorofluoromethane	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Trichloroftuoromethane	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
	stack gas, 20-1 VOST	GCMS	mg/Nm3	< 0.000580	0.000910	< 0.000542		:	
Trichlorofluoromethane	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.00231	0.00295 @	< 0.00218			
Tunasten	bottom ash	NAA	6/6n					< 5.00	
Tunosten	coal	NAA	ug/g, dry			< 10.4	< 10.4	< 10.4	< 10.5
Tunasten	collected fly ash	NAA	6/6n					< 1.50	
Tuncstan	hinh dust gas, particulates	NAA	b/bn					< 1.50	
Tungsien	stack gas, particulates	NAA	6/6n					< 1.50	
Uranium	bottom ash	NAA	6/ 5 n					8.21	
Uranium	coat	NAA	ug/g. dry			1.57	1.72	1.84	1.48
Uranium	collected fly ash	NAA	6/6n					9.49	
Uranium	high dust gas, particulates	NAA	6/6n					5.15	ĺ
lltanitum	stack das, particulates	NAA	5/6n					4.27 H	
Vanadium	bottom ash	NAA	6/8n					223	
Vanadium	coal	IC P-AES	mg/kg, dry		· · · ·	25.2	214	28.5	208
Vanadium	collected fly ash	NAA	6/8n		· · ·			257	
Vanadium	high dust gas, particulates	NAA	6/6n					152	
Vanadium	stack gas, particulates	NAA	6/Bn					110 R	
Vinvi acetate	high dust gas. 1-L VOST	GCMS	mg/Nm3	< 0.0887	< 0.139	< 0.105			
Vinvl acetate	high dust gas, 5-L VOSF	GCMS	mg/Nm3	< 0.0202	< 0 0211	< 0.0226			
Vinvl acetate	stack gas, 1-L VOST	GCMS	mg/Nm3	< 0.115	< 0 114	< 0.110			
Vinvl acetate	stack gas, 5-L VOST	GCMS	mg/Nm3	< 0.0231	< 0.0227	< 0.0218			
Vinvl chloride	high dust gas. t-L VOST	GCMS	mg/Nm3	< 0.00887	< 0.0139	< 0.0105			
Vinvl chtoride	high dust gas, 5-L VOST	GCMS	mg/Nm3	< 0.00202	< 0.00211	< 0.00226			
Vinvl chtoride	stack gas, 1-1, VOST	GCMS	mg/Nm3	< 0.0115	< 0.0114	< 0.0110			
Vinvl chtoride	stack gas, 5-L VOST	GCMS	EmN/gm	< 0.00231	< 0.00227	< 0.00218			
Ytterbium	bottom ash	NAA	6/6n					< 1.02	
Ytterbium	coal	NAA	ug/g. dry			< 0.457	0.479	1.34	0.541
Ytterbium	collected fly ash	NAA	6/6n					5.82	-
Ytterbium	high dust gas, particulates	NAA	5/Bn					3.70 H	
Ytterbium	stack gas, particulates	NAA	6/6n					1.85 R	-
Zinc	bottom ash	ICP-AES	mg/kg			32.6 @	29.9 @	28.3 @	24.3 @
Zinc	bottom ash	NAA	6/8n					37.7	
Zinc	bottom ash sluice water	ICP-AES	mg/L			< 0.0200	< 0.0200	< 0.0200	0.0230 @
Zinc	coal	ICP-AES	mg/kg, dry			15.4	9.75 @	14.7	14.9
Zinc	coal	NAA	ug/g, dry			23 2	15.7	19.0	13.8
Zinc	collected fly ash	ICP-AES	mg/kg			90.5 @	86.1 @	86.1 @	970@

Appendix C: Analytical Data Not Used in Calculations

8
Ĩ
ğ
9
ģ

Table C-1: Site 16 OFA Data NOT USED in Calculations

Substance	Stream	Method	NON	Run 1	Run 2	Run 3	Run 3D	Run 4	Run 5
		NAA	0/01					107	
UC	CORECIED ILY ASI		AR						
	high dust gas, particulates	ICP-AES	mg/kg		1	131 R		H /61	1381
	high dust gas particulates	NAA	0/b0					50.4 B	
		ICD_AFC	mo/Nm3			0.871 R		0.705 R	0.754 F
20						0 431		0.371	1.43
nc	high dust gas, vapor prase		CIIINUBILI				00000	0 0110 0	0.0270.6
UC	sluice water supply	ICP-AES	mg/L			0.0430 @	0.0300 @		
	stack das narticulates	ICP-AES	mg/kg			237 R		197 R	268 1
	track das Darticulates	NAA	na/a					57.2 H	
<u> </u>	ator and abao	ICP-AES	mo/Nm3			0.0305 R		0.0687 R	0.0698 F
10	static gas, soild prides		E my/om			0.696		0.243	< 0.00681
2	Stack gas, vapor prices							189	
rconium	bottom ash	NAM	A.B.n						
rconium	coal	NAA	ug/g. dry			76.0	< 50.3	200	00
	collected fly ash	NAA	B/Bn		-			< 376	
	high dust gas, particulates	NAA	5/6n					175 R	
	stack as nationates	NAA	ua/a					< 116	

Appendix C: Analytical Data Not Used in Calculations

3

Substance	Stream	Method	MON	Run 2	Run 2D
1.2-Dichloroethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588	
1,3-Dichlorobenzene	Stack gas, VOST	GCMS	ng/Nm3	< 0.588	
2-Hexanone	Stack gas, VOST	GCMS	ug/Nm3	< 2.94	
4-Methyl-2-Pentanone	Stack gas, VOST	GCMS	ug/Nm3	< 2.94	
Acetone	Stack gas, VOST	GCMS	ug/Nm3	3.13 @	
Arsenic	Bottom ash	ICP-AES	mg/kg, dry	< 159	< 160
Arsenic	ESP inlet gas, particulate	ICP-AES	mg/kg	85.8 @	83.6 @
Arsenic	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	444	432 @
Arsenic	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 4.83	
Arsenic	Stack gas, particulate	ICP - AES	mg/kg	508	
Arsenic	Stack gas, solid phase	ICP-AES	ug/Nm3	58.2	
Arsenic	Stack gas, vapor phase	ICP-AES	rug/Nm3	< 4.17	
Boron	ESP inlet gas, vapor phase	ICP - AES	ug/Nm3	628	
Boron	Stack gas, vapor phase	ICP-AES	ug/Nm3	733	
Bromine	Coat	NAA	mg/kg, dry	5.73	5.59
Bromodichioromethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588	
Cadmium	Bottom ash	ICP-AES	mg/kg, dry	< 2.90	4.03 @
Cadmium	ESP inlet gas, particulate	ICP-AES	mg/kg	< 2.93	< 2.91
Cadmium	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 15.1	< 15.0
Cadmium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	0.382 @	
Cadmium	Stack gas, particulate	ICP-AES	mg/kg	12.8	
Cadmium	Stack gas, solid phase	ICP-AES	ug/Nm3	1,47	
Cadmium	Stack gas, vapor phase	ICP-AES	emN/gu	< 0.308	
Cerium	Coat	NAA	mg/kg, dry	171	15.1
Cesium	Coal	NAA	mg/kg, dry	1.37	1.1
Chłorine	Coat	NAA	mg/kg, dry	473	448
Chromium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 0.525	
Chromium	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 0.454	
Dibromochloromethane	Stack gas, VOST	GCMS	ug/Nm3	< 0.588	
Europium	Coal	NAA	mg/kg, dry	0.328	0.315
Hafnium	Coal	NAA	mg/kg, dry	0.699	0.681
lodine	Coal	NAA	mg/kg, dry	0.233	0.214
Lanthanum	Coal	NAA	mg/kg, dry	9.74	9.71
Lead	Bottom ash	ICP-AES	mg/kg, dry	6.11 @	20.0

164 @

197 1124 < 4.87 452 76.5

ł

< 147

947 @

< 5.06 592 78.8

< 4.76

< 3.21

842

708

820

467 4.22

ł

< 0.578 2.98 @

3.28 @

< 0.588

< 2.89 < 16.8 0.176

4.21 @ 24.0 @ 15.8 15.8 2.68

2.49 < 0.352

< 0.237

18.7

4.95

< 2.89</pre>
< 2.89</pre>
< 15.89</p>

< 2.94</pre>< 2.94</pre>

< 0.578

< 0.578

< 0.588 < 0.588

Run 4 ļ

> ę Run

Table C-2:

16.1 1.03 320

23.3

1.86@ 0.725 @

1.51 @

417

< 0.349

< 0.588

< 0.578

0.348 0.728 0.170

0.490 1.09 0.147 12.4

8.46 3.85 @

9.84 @

Table C-2: Site 16 OFA/LNB Data NOT USED in Calculations

N

Substance	Stream	Method	MON	Run 2	Run 2D	Run 3	Run 4
	ESP inlet das, particulate	ICP-AES	mg/kg	61.7 @	68.6 @	57.4 @	63.0 @
	ESP inlet das, solid phase	ICP-AES	emN/gu	319@	355 @	328 @	364 @
	ESP inlet cas. vapor phase	ICP-AES	ng/Nm3	< 5.67		< 5.71	< 5.94
	Stack cas, particulate	ICP-AES	mg/kg	154		120	145
	Stack das, solid phase	ICP-AES	ug/Nm3	17.71		20.3	19.4
	Stack das, vapor phase	ICP-AES	ug/Nm3	< 4.90		< 3.77	< 5.59
		NAA	mg/kg, dry	0.148	0.146	0.138	0.138
Magnasium	Bottom ash	ICP-AES	mg/kg, dry	3788	5724	4230	4960
Manaatium	Coal	NAA	mg/kg, dry	538	484	623	685
Machaelim	ESP inlet cas, particulate	ICP-AES	mg/kg	4601	4349	4687	4950
	ESP inlet das, solid phase	ICP-AES	ug/Nm3	23809	22502	26737	28650
Macnesium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 4.83		< 4.87	2.30
Macnesium	Stack das, particulate	ICP-AES	mg/kg	4703		4393	3832
Manasim	Stack das, solid phase	ICP-AES	ug/Nm3	539		744	510
Marnasium	Stack gas, vapor phase	ICP ~ AES	ug/Nm3	< 4.17		< 3.21	< 4.76
	Coal	NAA	mg/kg, dry	0.100	0.0861	0,171	0.116
	Coal	NAA	mg/kg, dry	1.1	11.6	18.2	16.8
Nickel	Bottom ash	ICP-AES	mg/kg, dry	88.9	88.5	82.8	84.1
	ESP inlet das. particulate	ICP-AES	mg/kg	113	96.6	104	106
Nickel	ESP inlet gas, solid phase	ICP - AES	ug/Nm3	585	200	296	612
Nickal	ESP inlet das, vapor phase	ICP - AES	ug/Nm3	< 2.08		< 2.09	1.40
Nickal	Stack gas, particulate	ICP-AES	mg/kg	108		96.3	103
Nickel	Stack gas, solid phase	ICP AES	EmN/gu	12.3		16.3	13.7
Nickel	Stack gas, vapor phase	ICP-AES	ug/Nm3	1.80		< 1.38	< 2.05
Potassium	Bottom ash	ICP - AES	mg/kg, dry	21242	25352	24100	26400
Potassium	Coal	NAA	mg/kg, dry	823	860	1650	739
Potassium	ESP inlet gas, particulate	ICP-AES	mg/kg	19980	20541	21990	21058
Potassium	ESP inter das, solid phase	ICP - AES	ug/Nm3	103382	106286	125454	121879
Potassium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	5 77.8		< 78.3	< 81.3
Potassium	Stack gas, particulate	ICP - AES	mg/kg	20713		21556	18629
Potassium	Stack gas, solid phase	ICP - AES	emN/gu	2372		3650	2480
Potassium	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 67.1		< 51.7	< 76.6
Rubidium	Coal	NAA	mg/kg, dry	21.8	16.8	25.1	15.0
Samarium	Coal	NAA	mg/kg, dry	1.74	1.78	2.22	1.68

Appendix C: Analytical Data Not Used in Calculations

30-Oct-93

Table C-2: Site 16 OFA/LNB Data NOT USED in Calculations

Substance	Stream	Method	NON	Run 2	Run 2D	Run 3	Run 4
Scandium	Coal	NAA	mg/kg, dry	3.58	3.45	4.94	3.54
Selenium	Bottom ash	ICP-AES	mg/kg, dry	< 43.9	< 44.1	44.7 @	< 42.4
Selenium	ESP inlet gas, particulate	ICP-AES	mg/kg	< 178	< 176	< 175	< 176
Selentum	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 919	< 913	666 >	< 1017
Selenium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	8.83		< 8.89	< 9.23
Selenium	Stack gas, particulate	ICP-AES	mg/kg	160 @		89.3 @	68.8 @
Selenium	Stack gas, solid phase	ICP-AES	ug/Nm3	18.4 @		15.1@	9.16 @
Selenium	Stack gas, vapor phase	ICP-AES	ug/Nm3	126		122	224
Silicon	Bottom ash	ICP-AES	mg/kg, dry	211423	217304	220000	215000
Silicon	ESP inlet gas, particulate	ICP-AES	mg/kg	297679	241483	270647	247505
Silicon	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	1540288	1249511	1544049	1432516
Silicon	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	609		428	441 @
Silicon	Stack gas, vapor phase	ICP-AES	ug/Nm3	354		402	541
Silver	Bottom ash	ICP-AES	mg/kg, dry	< 1.79	< 1.80	< 1.66	< 1.73
Silver	Coat	NAA	mg/kg. dry	0.201	0,180	0.227	0.262
Silver	ESP intet gas, particulate	ICP-AES	mg/kg	1.82	< 1.80	< 1.79	< 1.80
Silver	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 9.40	< 9.33	< 10.2	< 10.4
Silver	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 1.03		0.796	< 1.08
Silver	Stack gas, particulate	ICP-AES	mg/kg	2.27 @		2.65	2.26 @
Silver	Stack gas, solid phase	ICP_AES	ug/Nm3	0.260 @		0.448	0.300 @
Silver	Stack gas, vapor phase	ICP-AES	ug/Nm3	< 0.889		< 0.684	< 1.01
Strontium	Bottom ash	ICP-AES	mg/kg, dry	680	980	778	822
Strontium	Coal	NAA	mg/kg, dry	189	107	72.6	97.2
Strontium	ESP inlet gas, particulate	ICP-AES	mg/kg	941	921	935	945
Strontium	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	4871	4765	5336	5470
Strontium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	0.109@		0.536	0.282 @
Strontium	Stack gas, particulate	ICP-AES	mg/kg	1052		951	910
Strontium	Stack gas, solid phase	ICP-AES	ug/Nm3	120		161	121
Strontium	Stack gas, vapor phase	ICP-AES	ug/Nm3	0.223		0.0978 @	0.108 @
Tantalum	Coal	NAA	mg/kg, dry	0.171	0.167	0.214	0.189
Terbium	Coal	NAA	mg/kg, dry	0.187	0.189	0.284	0.160
Thallium	Bottom ash	ICP - AES	mg/kg, dry	€. 99.8 V	< 69.1	< 63.6	< 66.5
Thallium	ESP inlet gas, particulate	ICP-AES	mg/kg	< 20.2	< 20.0	< 19.9	< 20.0
Thallium	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	< 104	< 104	< 114	< 116

ო

30-Oct-93

C-26

4

	Stream	Method	NOM	Run 2	Run 2D	Run 3	Run 4
Thallium	ESP inlet gas, vapor phase	ICP-AES	ug/Nm3	< 3.57		10.9 @	- 74
Thallium	Stack gas, particulate	ICP-AES	mg/kg	< 5.84		< 411	
Thallium	Stack gas, solid phase	ICP - AES	ug/Nm3	< 0.669		C 0 605	
Thallium	Stack gas, vapor phase	ICP-AES	ug/Nm3	4.75 @		0.09	5 00 S
Thorium	Coal	NAA	mg/kg, dry	2.30	2.43	3.75	2 45
Tin	Coal	NAA	mg/kg, dry	< 10.5	< 10.5	< 10.4	2 U 2
trans-1,2-Dichloroethene	Stack gas, VOST	GCMS	ug/Nm3	< 0.588	4 	< 0.588	< D 578
Trichtorofluoromethane	Stack gas, VOST	GCMS	ug/Nm3	4.96		5.31 @	0.00
Tungsten	Coal	NAA	mg/kg, dry	< 5.23	< 5.25	< 5.21	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
Uranium	Coal	NAA	mg/kg, dry	1.94	1.63	1.87	
Ytterbium	Coal	NAA	mg/kg, dry	0.304	0.327	0.326	0.563
Zinc	Bottom ash	ICP-AES	ma/ka. drv	51.1@	516.0	780.0	
Zinc	Coal	NAA	ma/ka. dry	12.5	16.5		
Zinc	ESP inlet gas, particulate	ICP-AES	mg/kg	102	66.2	115	4 H
Zinc	ESP inlet gas, solid phase	ICP-AES	ug/Nm3	527	513	658	589
Zinc	ESP intet gas, vapor phase	ICP-AES	ug/Nm3	5.91		16.5	
Zinc	Stack gas, particulate	ICP-AES	mg/kg	245		228	303
Zinc	Stack gas, solid phase	ICP-AES	ug/Nm3	28.0		38.6	42.9
Zinc	Stack gas, vapor phase	ICP-AES	ug/Nm3	0.929 @	!	0.216 @	< 0.311
Zirconium	Coal	NAA	mg/kg, dry	40.3	0.68	49.0	66.7

Appendix C: Analytical Data Not Used in Calculations

¥

APPENDIX D: ADDITIONAL OVERFIRE AIR TEST RESULTS

This appendix discusses additional results from the OFA test. Specifically, the arsenic, chromium, and mercury speciation results, and the analytical results for size-fractionated particulate matter samples are presented and discussed. These results are not considered valid for the following reasons:

- Arsenic speciation: The spike results showed that As(III) was oxidized to As(V) during sample recovery, preservation, or analysis. This procedure was not used for the OFA/LNB test.
- Chromium speciation: The spike results showed that Cr(VI) was reduced by the flue gas during sample collection, probably to Cr(III). This procedure was not used for the OFA/LNB test; instead, the EPA Cr(VI) method was selected.
- Mercury speciation: The procedure used for the OFA test has since been shown to be invalid. The technique has been changed significantly, and an improved method was used for the OFA/LNB test.
- Size-fractionated particulate matter samples: INAA has been shown to give poor results for this type of sample; therefore, ICP-MS was selected as the technique to analyze the samples collected during the OFA/LNB test.

Each of these topics is discussed in detail below.

Arsenic Speciation

Special stack gas samples were collected during the OFA test to determine the oxidation state of arsenic present in the stack gas and to verify the sampling and analytical methods applied. Studies by EPRI have shown that As(III) and As(V) concentrations can be determined in aqueous samples by adjusting the sample pH during analysis by hydride generation atomic absorption spectrophotometry (HGAAS). This analytical approach was applied to the impinger and extraction solutions for vapor- and particulate-phase samples collected for this test.

Sampling and Analytical Method

Stack gas samples were collected isokinetically at a single point in the stack representative of the overall gas velocity obtained from previous velocity profiles. Three EPA Method 5 samples were collected simultaneously from separate ports to provide triplicate samples for each sampling period. Teflon®-coated fiber filters were used to collect the particulate phase, and an impinger solution of nitric acid and hydrogen peroxide was used to collect vapor-phase arsenic.

Two sets of triplicate samples were collected to determine the concentration of As(III) and As(V) in the stack gas. Two additional sets of triplicate samples were collected using filters previously spiked with two different levels of As(III) and As(V). The purpose of analyzing these samples was to determine the recovery of As(III) and (V) after exposure to flue gas. One set was spiked with aqueous As(III) and (V) solutions to provide $3 \mu g$ of arsenic at each oxidation state. A second set was spiked with $15 \mu g$ each. Additional filters were spiked at the same levels but not exposed to flue gas. These filters were analyzed to determine spike recoveries without interference from the sample matrix.

The solids present in the sampling probe and nozzle were recovered by rinsing with a solution of 0.1 N sodium hydroxide (reducing-agent free). The filter was recovered from the hot box apparatus and placed in a covered Petri dish and sealed with Teflon[®] tape. Impinger solutions were transferred to polyethylene bottles. After the sample was recovered, the probe and nozzle rinse samples and impinger solutions were refrigerated at 4°C; the filters were placed in a freezer at approximately -20°C. Samples were shipped to the laboratory on ice by overnight express and refrigerated on arrival.

On the analysis date, the filters were combined with the corresponding probe and nozzle rinse samples and sonicated to solubilize the arsenic in the filtered solids. The leachates were analyzed by HGAAS. As(III) was analyzed directly at an adjusted pH of 6. Total arsenic was determined from the same sample with the pH adjusted to less than one. The concentration of As(V) was determined by subtracting the As(III) values from the total As value.

Arsenic Results and Discussion

Table D-1 shows the arsenic test results for the unspiked samples. Total arsenic averaged $108 \mu g/Nm^3$ and corresponded well with the $112 \mu g/Nm^3$ result obtained by analyzing the multi-metals train samples collected on the two preceding days. However, the test results for As(III) and As(V) are considered suspect because of the poor spike recovery obtained in the analysis of the spiked blank filters. The analytical results from these control samples are presented in Table D-2. The average spike recovery for the As(III) spikes on blank filters was less than 3 percent. The recovery of As(V) on the same filters exceeded 175%, demonstrating the oxidation of As(III) to As(V). Total

Run No.	Collection Date	(111)64		IUUAL AS
1-1	3/7/91	3.25	131	134
1-2	3/7/91	3.14	114	117
1-3	3/7/91	2.63	1.79	100
4-1	3/8/91	3.26	107	111
4-2	3/8/91	2.71	96.7	99.4
4-3	3/8/91	2.52	81.4	83.9
Average		2.92	105	108
5% CI		0.35	18	18

Table D-1 Speciation of Arsenic in Stack Particulates - OFA Test The conversion of As(III) to As(V) is evident in both sets of control samples.

As(V).

Appendix D: Additional Overfire Air Test Results

	Arsen	ic(111)	Arser	ic(V)*	Total	Arsenic
Sample No.	Spike Amount (#g)	Spike Recovery (%)	Spike Amount (#g)	Spike Recovery (%)	Spike Amount (4g)	Spike Recovery (%)
l " o Sniked Filt	ers ^b					
1	6	4.7	3	156	6	80.0
	,	43	m	163	6	83.3
7 4		43	e	156	9	80.0
0 4		ND(2.0)	6	186	9	93.8
+ 4		ND(2.0)	6	192	6	96.8
0	، ار		6	197	9	99.3
9	c	10.2/11	, , ,	361	9	88.9
Mean	e	2.1		C/T	>	
15 " ø Sniked Fi	lters ^b					
1	15	3.7	15	196	30	9.99
- c	1 1 2	30	15	186	30	95.1
v r	C Y	ND/UA)	15	180	30	90.7
• •	2 2	ND(0.4)	15	195	30	98.0
t 4	5	ND(04)	15	189	30	94.9
	51	ND(0.4)	15	194	30	97.6
		()	1	100	30	0.96

^b Arsenic species spiked onto blank filters. No flue gas sample was collected. • Arsenic (V) determined indirectly by subtracting As(III) from total As.

ND = Not detected. Value in parentheses is based on the detection limit.

Appendix D: Additional Overfire Air Test Results

Table D-2

arsenic recoveries averaged 88.9% and 96.0% for the $3 \mu g$ and $15 \mu g$ spikes, respectively. The results for the spiked sample filters are not shown because the spike levels were much too low, compared with the native arsenic concentration in the stack gas, rendering them meaningless.

Because nitric acid/peroxide impinger solutions were not amenable to analysis by HGAAS or speciation, they were analyzed for total arsenic by graphite furnace atomic absorption spectroscopy. Arsenic was not detected in these impinger solutions at a limit of detection of $1.2 \,\mu g/Nm^3$.

A problem experienced with the sample sparger in the hydride generation unit during the analysis of As(III) may have contributed to the low As(III) results obtained during this test. However, a review of the QC check samples run during analysis tends to discredit the significance of this equipment problem, and Radian assumes that the poor recovery of As(III) is the result of oxidation to the more stable As(V) valence state. The high recovery of As(V) also supports this assumption.

Greater attention to sample handling, preservation, and preparation techniques will be necessary to maintain the oxidation state of the samples up to the time of analysis. An EPRI study of arsenic speciation in natural waters and sediments suggests that samples should be quick-frozen in liquid nitrogen and maintained at -80°C until analysis. Future sampling efforts should implement this preservation technique.

Chromium Speciation

Stack gas samples were collected during the OFA test to determine chromium concentrations in the +6 oxidation state. A modified version of California's Air Resources Board (CARB) Method 425 was used at this site. This method has been undergoing field application studies for detecting and quantifying Cr(VI) emissions in flue gas at several coal-fired generating stations. The method validation test results at this site are presented in this section.

Sampling and Analytical Methods

Three individual sampling trains were used to simultaneously collect triplicate sample sets from three sampling points of average flue gas velocity. Samples were collected isokinetically with Teflon®-coated filters used to separate the particulate material from the vapor phase. Unspiked filters and filters spiked with Cr(VI) were used to determine the efficacy of the sampling and analytical procedure. Instead of sodium hydroxide impingers being used to collect vapor-phase Cr(VI), impingers containing nitric acid/hydrogen peroxide solution were used to collect total vapor-phase chromium. Previous experience with impinger collection techniques for Cr(VI) in the presence of SO_2 have been unsuccessful.

Five sample sets were collected over a two-day period. Two sample sets (one each day) were collected on unspiked filters for determining the baseline Cr(VI) concentration in the particulate phase. The other sample sets were collected on filters pre-spiked with potassium dichromate solution to a level of 7.6 micrograms of Cr(VI). Additional filters were spiked at $1.25 \ \mu$ g, $6.25 \ \mu$ g, and $25 \ \mu$ g of Cr(VI) for spike recovery tests and sample degradation studies.

Particulate material was recovered from the sampling trains by rinsing the probe and nozzle with a solution of 0.1 N sodium hydroxide (reducing agent free). The particulate filters were recovered and immediately placed into the probe and nozzle rinse samples. The total volume was then adjusted to 100 mL and quantified gravimetrically. The samples were well mixed for approximately 30 minutes to extract Cr(VI) from the particulate material. Filtered aliquots of the extracts were then reacted with diphenylcarbazide reagent to produce a colored Cr(VI) complex suitable for spectrophotometric analysis.

Blank and pre-spiked filters not subjected to flue gas were extracted and analyzed alongside the samples to determine extraction efficiency and background corrections. To determine potential holding times for the samples, 18 pre-spiked filters of each spiking level (1.25, 6.25, and $25 \mu g$) were collected for a time-related degradation study. Six filters of each Cr(VI) level were analyzed after one day, one week, and one month of storage under ambient conditions.

Chromium (VI) Results and Discussion

Table D-3 shows the results of Cr(VI) analyses for the spiked and unspiked filters used to sample the flue gas particulate material. Each series of three runs represents three samples obtained simultaneously using three stack sampling trains. The Series 1 runs provide an estimate of the Cr(VI) particulate material on the first day of sampling. The filters used contained no spike. The mean (\pm 95% confidence interval) particulate matter Cr(VI) concentration was $5.4 \pm 1.9 \,\mu g/Nm^3$. The Series 4 runs found the mean flue gas particulate matter Cr(VI) concentration to be $3.0 \pm 4.5 \,\mu g/Nm^3$. The mean particulate-phase total chromium concentration in the stack gas, as measured by the multi-metals train, was $44 \pm 37 \,\mu g/Nm^3$. These results show that about 10% of the particulate-phase total chromium was attributed Cr(VI).

Table D-3 also shows the detailed results for the spiked filters exposed to flue gas. The Series 2 results were obtained when the particulate filter was spiked with the medium spike before sampling on March 7. The Series 3 and Series 5 filters were obtained in the same way, except the sampling was performed on March 8. Because the daily flue gas particulate matter Cr(VI) values differ, the daily values were used to calculate expected filter Cr(VI) weights for spiked filters on the corresponding days. Comparison of the expected quantity of Cr(VI) on the filter with the measured Cr(VI) by the candidate method shows that, in every case, the measured Cr(VI) was less than the expected value. The mean recovery was 30 ± 38 percent.

	Spike Recovery (%)				16 -29 -62	41 89 92	41 38 46	30 38
	Cr(VI) Concentration 1 (#g/Nm ²)	5.2 4.7 6.2	3.2 NA 2.7	4.4 1.8				
Filter	Total (μg)				15.8 16.0 16.0	12.2 12.3 12.2	11.5 12.1 11.5	
Expected on	Flue Gas (#g)				8.2 8.4 8.4	4.6 4.7 4.6	3.9 4.5 3.9	
Cr(V])	Spike (#g)				7.6 7.6 7.6	7.6 7.6 7.6	7.6 7.6 7.6	
I	Cr(V1) Measured on Filter (#g)	8.1 7.5 9.6	5.0 NA 4.3		9.4 6.2 3.7	7.7 11.5 11.6	7.0 7.4 7.4	
aration	Flue Gas Sampled (Nm ³⁾	1.57 1.58 1.55	1.56 1.60 1.57		1.52 1.56 1.57	1.55 1.59 1.56	1.30 1.53 1.30	
Filter Prepa	Cr(VI) Spike "	None None None	None None Nonc		Mcdium Mcdium Medium	Mcdium Mcdium Medium	Medium Medium Medium	
	Collection Date	3/7/91 3/7/91 3/7/91	3/7/91 3/7/91 3/7/91		3/7/91 3/7/91 3/7/91	3/8/91 3/8/91 3/8/91	3/8/91 3/8/91 3/8/91	
	Run No.	1-1 1-2 1-3	4-1 4-2 4-2	Mean 95% CI	2-1 2-2 2-3	3-1 3-2 3-3	5-1 5-2 5-3	Mean 95% CI

Table D-3 Cr(VI) Measurement in Stack Particulate - OFA Test

Appendix D: Additional Overfire Air Test Results

D-7

NA = Not Analyzed.

* Spikes were made to filter before sampling.

Table D-4 shows the results of the sample degradation study. Cr(VI) was spiked onto the blank filters at low, medium, and high levels. The results indicate the variability in the spiking and analysis at each of the three spiking levels. The results also show that there is no significant difference in the amount of Cr(VI) on the filters after one day after one month of storage. Filters spiked at the same time as the medium-level spiked filters in this test were used to conduct flue gas testing at Site 16. The results in Table D-4 show that the best estimate of the amount of Cr(VI) on these medium-spiked filters is 7.6 μ g.

Chromium (VI) (as potassium dichromate) spiked onto clean, Teflon[®]-coated filters, can easily be extracted and analyzed with close to 100% recovery at a level comparable to the amount of Cr(VI) detected in the particulate matter collected on unspiked filters from the flue gas at this site. However, when spiked filters are exposed to a large quantity of flue gas, the recoveries of these medium-level spikes are highly variable. The high degree of variability and the poor spike recoveries associated with pre-spiked sample results have been the primary concern when evaluating the data from this and other test sites. The low recoveries suggest that the Cr(VI) collected on the filter is being reduced to Cr(III) during sampling.

Mercury Speciation

Samples of stack gas were collected during the OFA test to characterize the forms of mercury being emitted from a coal-fired utility boiler and to validate the sampling and analytical methods applied. Elemental mercury (Hg^0) , inorganic ionic mercury (Hg^{2+}) , and methyl mercury (compounds such as CH_3HgCl) were the targeted species. The analysis was conducted by Brooks Rand, Ltd. in Seattle, Washington. This technique, developed by EPRI contractor Nicholas Bloom (formerly with Brooks Rand), was in the early stages of development when the OFA test was conducted.

Sampling and Analytical Methods

Four stack gas samples were collected non-isokinetically from a single port with a special impinger train. A Teflon[®] tube was inserted into the stack port and directed towards the gas stream for sample collection. The sampling train was designed for in-stack filtration of particulates from the gas stream before bubbling the gas through a solution of 1.0 N potassium chloride (KCl) and on through a series of iodated-charcoal sorbent tubes. The purpose of the KCl impinger solution is to collect water-soluble forms of mercury such as CH₂Hg and Hg²⁺, and the charcoal sorbent is used to collect elemental mercury.

Three Teflon[®] filters and one ashed, quartz-fiber filter were used in an effort to compare the effects of the filtering media on sample recovery and analysis. Three Teflon[®] impingers, each with approximately 100 mL of 1.0 N KCl solution, were placed in an ice bath during sample collection, and three iodated-charcoal sorbent tubes followed in

	Cr(VI) Measured on Filte	r (μg)
Target Cr(VI) Level *	One Day	One Week	One Month
Low (1.25 µg)	0.91	0.94	1.18
	0.89	0.89	1.14
	0.84	0.85	0.87
	0.91	0.75	1.21
	0.91	0.89	1.18
	0.84	0.99	1.18
	0.88 ± 0.04^{b}	0.89 ± 0.09	1.13 ± 0.13
Medium $(6.25 \mu g)$	7.91	7.69	7.11
	7.84	7.83	8.10
	7.42	8.01	7.07
	7.42		7.68
	7.69		7.98
	7.34		7.41
	7.60 ± 0.25	7.84 ± 0.40	7.56 <u>+</u> 0.46
High $(25 \mu g)$	27.4	28.6	29.1
	26.1	27.1	29.3
	27.7	28.4	28.0
	27.9	29.2	29.5
	27.4	28.8	28.2
	28.2	28.6	29.4
	27.4 ± 0.8	28.4 ± 0.8	28.9 ± 0.7

Table D-4Cr(VI) Spiked Filter Degradation Results - OFA Test

* Cr(VI) spiked onto a blank filter. No flue gas sample was collected.

^b Mean ± 95% CI.

series to complete each sampling train. All connecting pieces of the sampling train were made of Teflon[®].

Impinger samples were combined in Teflon[®] bottles and refrigerated immediately after collection. Filters were placed in separate sealed containers and also refrigerated. Charcoal sorbent tubes were left connected with the ends capped and refrigerated after collection. All samples were packed on ice and sent by next day air to the laboratory. At the laboratory, the impinger samples were preserved with the addition of 10 mL/L of HCl and the filters were transferred to a freezer.

Aqueous (impinger) samples were first analyzed for ionic mercury by the direct reduction of a small aliquot with SnCl₂, purging onto gold, and analysis by cold vapor atomic fluorescence spectroscopy (CVAFS). The remaining sample was then divided into aliquots for methyl mercury extraction and total mercury analysis.

For the methyl mercury extractions, one sample was extracted in triplicate and the others were performed in duplicate. Approximately 40-mL aliquots were accurately weighed into Teflon® bottles, the volume was brought to exactly 60 mL with deionized water, and the samples were extracted by shaking overnight with 40 mL of methylene chloride. The methylene chloride layer was then separated, and 50 mL of deionized water was added. The methylene chloride was removed by boiling off at 60°C and then sparging with nitrogen. The methyl mercury, thus transferred to pure water, was analyzed by aqueousphase ethylation, cryogenic gas chromatography, and CVAFS detection.

The remaining aliquots of impinger samples were cold oxidized with BrCl and analyzed for total mercury by the techniques described for ionic mercury analysis.

The filters were pre-digested in 5 mL of 6N HCl for 24 hours and then diluted to 18.2 mL in Teflon[®] vials. A small aliquot $(100 \,\mu L)$ of this digestate was analyzed for ionic mercury and another aliquot $(25 \,\mu L)$ for methyl mercury by the methods previously described. The remainder of the sample was then cold oxidized with BrCl (0.5 mL/18.2 mL). A 1-mL aliquot of this digestate was analyzed for total mercury by the method described above.

The iodated-charcoal sorbent media was recovered from the tubes and placed into 18.2-mL Teflon[®] vials with 5 mL of 7:3 HNO₃/H₂SO₄. The samples were digested for three hours at 70° C in sealed containers, cooled, and diluted to 18.2 mL. Aliquots of these digestates were analyzed for total mercury, as described above.

Mercury Results and Discussion

Table D-5 presents the stack gas mercury concentrations determined by the Brooks Rand sampling system. For comparison, the results from the multi-metals train samples

		INTER COL		- 19-4 (
Component	Run 6a 3/7/91	Run 6b 3/7/91	Run 7a 3/8/91	Run 7b 3/8/91	Mean	95% CI	Percent of Vapor Hg
Rrooks Rand Samnling Train							
Darticulate Phase	0.021	0.008	0.011	0.025	0.016	0.013	;
Vanar Phase	0.59	2.7	1.5	1.7	1.6	1.4	ł
v aput 1 tiase Ionie Inerganie Ho	ND(0.02)	0.24	0.90	1.1	0.56	0.83	35%
Romental Ha	0.57	2.5	0.56	0.55	1.0	1.5	65%
Liturum 116 Methul He	0.018	0.002	0.005	0.008	0.008	0.011	0.5%
Total	0.61	2.7	1.5	1.7	1.6	1.4	ţ
	Run 3	Run 4	Run 5		Mean	17 Deg	Percent of Vanor Ho
	3/5/91	3/6/91	3/0/91		INICALI	ID ACC	gra rodary
Multi-Metals Train							
Particulate Phase	0.13	0.54	0.37		0.35	0.51	ł
Vannr Phase	6.6	7.2	7.6		7.1	1.2	ł
Ionic Hof	5.3	6.6	6.6		6.2	1.9	87%
Riemental Ho ^b	1.3	0.60	0.96		0.95	0.87	13%
Total	6.7	<i>L.</i> L	7.9		7.5	1.6	ť

Table D-5 Speciation of Mercury in Stack Gas - OFA Test • Mercury captured in the HNO₃/H₂O₂ impingers.

^b Mercury captured in the H₂SO₄/KMnO₄ impingers.

Appendix D: Additional Overfire Air Test Results

D-11

collected during the preceding two days are also provided. The mean total mercury concentrations are $1.6 \pm 1.4 \,\mu g/Nm^3$ for the Brooks Rand train and $7.5 \pm 1.6 \,\mu g/Nm^3$ for the multi-metals train. The multi-metals results agree much better with the coal measurements (71 ± 27 percent material balance closure), suggesting that the Brooks Rand results are biased very low.

The mercury speciation technique used for the OFA test was declared invalid by Nicholas Bloom in a communication to EPRI on July 7, 1993. The main problem with the system was the loss of mercury by condensation in the Teflon[®] tubing ahead of the KCl impingers. Also, some analytical interferences were presented by the KCl solution.

A much improved sampling system was used for the OFA/LNB test, the results of which were discussed in Section 3. The improvements included using heated quartz tubing ahead of the mercury traps (to prevent condensation) and a KCl/soda lime solid sorbent to replace the KCl impingers.

Size-Fractionated Particulate Matter Samples

Particulate matter samples were collected with particle sizing devices at both the ESP inlet and the stack during the OFA test. Specific size fractions were analyzed for metals by INAA.

Sampling and Analysis Methods

Stack particulate matter fractions were collected with a University of Washington Mark 5 cascade impactor to provide four particle size fractions: <0.8, 0.8-1.8, 1.8-12, and $>12 \,\mu$ m. The samples were collected isokinetically at a fixed point in the stack. Because of the low particulate loading at the stack, samples collected over three days (March 5-7, 1991) were combined to produce a single set of four fractions for analysis.

At the ESP inlet, size-fractionated particulate matter was collected using an Anderson High Capacity Stack Sampler in four approximate particle sizes: <2, 2-7, 7-13, and >13 μ m. The samples were collected isokinetically at a fixed point in the ESP inlet duct. Three sets of ESP inlet samples were submitted for analysis.

Following collection, the samples were desiccated and weighed until a constant weight was obtained. The samples were then analyzed as-collected using INAA. For those samples collected on a substrate, the entire substrate was placed in the analysis vial. Blank Kapton (polyamide) films, quartz filters, and glass fiber thimbles were analyzed so that the background concentrations of metals in these substrates could be subtracted from the sample values. All the results were corrected for the blank results for the substrates.

Size Fraction Results and Discussion

Table D-6 presents the results for the stack particulate matter, and Tables D-7, D-8, and D-9 show the results for the ESP inlet particulate matter. High background concentrations of aluminum, arsenic, barium, calcium, chromium, and manganese were observed in the blank glass fiber thimbles; therefore, those blank corrections were the most significant.

For the major elements such as aluminum, calcium, iron, and titanium and the nonvolatile trace elements such as barium, cobalt, manganese, nickel, and vanadium, relatively constant levels are expected in all of the size fractions. These elements should appear in the particulate matter at levels approximately equal to their concentrations in the ash fraction of the coal. However, many anomalous results can be seen in Tables D-6 through D-9.

Two ash standards were analyzed in the same batch as the samples. Ash masses of roughly 10 mg and 100 mg were chosen to approximate the masses of particulate matter found in the samples. As shown in Table D-10, the recoveries were, in general, poor. Of the 13 elements with a certified value, only one element (titanium) had a recovery in the acceptable range of 75-125% for both sample sizes. In addition, there is no consistent pattern in the recoveries. Many of the elements show high recoveries in one sample but low recoveries in the other.

The analysis of particulate matter samples by INAA was attempted for one other test site in the FCEM project, with similar findings. As a result, INAA is not considered a reliable technique for accurately determining the composition of small quantities of particulate matter. A different procedure, microwave digestion followed by ICP-MS, was used successfully to analyze the samples collected during the OFA/LNB test.

Table D-6Elemental Analysis of Stack Particulate Matter Fractions - OFA TestSamples Collected March 5-7, 1993

Sample No.	UWST4	UWST3	UWST2	UWST1
Sample ID	Stage 4	Stage 3	Stage 2	Stage 1
dP50 Size Range	<0.8 µ m	<u>0.8-1.8 μm</u>	1.8-12 μm	>12 µ m
Substrate	Quartz Filter	Kapton	Kapton	Quartz Filter
Mass Collected, g	0.012	0.008	0.046	0.088
% of Total Mass, g	8	5	30	57
Mass Analyzed, g	0.012	0.008	0.046	0.088
Gas Volume, Nm ³	0.9057	0.9057	0.9057	0.9057
Elemental Conc., mg	g/kg			
Aluminum	150,000	180,000	41,000	43,000
Calcium	8,300	20,000	7,600	7,500
Iron	130,000	200,000	50,000	67,000
Titanium	11,000	15,000	3,300	2,700
Arsenic	3,000	ND(12)	390	130
Barium	3,500	3,800	1,100	2,200
Cadmium	ND(130)	ND(110)	ND(28)	ND(18)
Chromium	660	1,100	280	270
Cobalt	130	210	59	44
Copper	2,500	1,600	330	410
Manganese	320	190	50	65
Mercury	ND(8)	ND(63)	11	ND(1.1)
Molybdenum	NC	3,000	ND(10)	NC
Nickel	320	640	120	210
Selenium	830	930	150	140
Vanadium	1,200	1,100	220	100

Table D-7Elemental Analysis of ESP Inlet Particulate Matter Fractions - OFA TestSamples Collected March 5, 1991

•

.

Sample No.	12	11	10	9
Sample ID	Stage 4	Stage 3	Stage 2	Stage 1
dP50 Size Range	<2.1 µ m	<u>2.1-7.5 μm</u>	7.5-14 μm	<u>>14 µm</u>
Substrate	Glass Thimble	None	None	None
Mass Collected, g	0.1288	0.32	0.461	3.492
% of Total Mass, g	3	7	11	79
Mass Analyzed, g	0.1288	0.1226	0.1906	0.4399
Gas Volume, Nm ³	0.6465	0.6465	0.6455	0.6465
				······
Elemental Conc., mg	/kg			
Aluminum	NC	160,000	160,000	130,000
Calcium	93,000	18,000	13,000	8,900
Iron	24,000	100,000	89,000	69,000
Titanium	ND(1,400)	11,000	9,800	7,200
Arsenic	1,600	660	280	63
Barium	540,000	1,700	1,300	750
Cadmium	ND(220)	ND(34)	ND(24)	ND(12)
Chromium	600	500	240	110
Cobalt	23	110	82	38
Copper	ND(6,400)	690	740	ND(370)
Manganese	34	240	190	130
Мегсигу	9.3	ND(3.4)	ND(2.4)	ND(1.2)
Molybdenum	110	180	56	ND(6.3)
Nickel	ND(640)	130	61	26
Selenium	480	28	19	23
Vanadium	570	630	430	220

Table D-8

Elemental Analysis of ESP Inlet Particulate Matter Fractions - OFA Test Sample Collected March 6, 1991

Sample No.	16	15	14	13
Sample ID	Stage 4	Stage 3	Stage 2	Stage 1
dP50 Size Range	<2.1 µ m	<u>2.1-7.3 μm</u>	7.3-13 μm	>13 µm
Substrate	Glass Thimble	None	None	None
Mass Collected, g	0.1288	0.269	0.263	2.155
% of Total Mass, g	1	10	10	79
Mass Analyzed, g	0.1288	0.1995	0.2232	0.4978
Gas Volume, Nm ³	0.4935	0.4935	0.4935	0.4935
Elemental Conc., mg	/kg			
Aluminum	NC	150,000	140,000	126,000
Calcium	ND(19,000)	15,000	17,000	8,400
Iron	1,000,000	93,000	99,000	84,000
Titanium	12,000	9,700	8,400	6,500
Arsenic	2,000	480	280	60
Barium	140,000	1,600	1,500	1,000
Cadmium	ND(370)	ND(25)	ND(22)	ND(18)
Chromium	4,000	370	260	120
Cobalt	2,500	89	84	49
Copper	ND(4,900)	600	760	310
Manganese	110	250	210	130
Mercury	7_	ND(2.4)	ND(2.2)	ND(1.8)
Molybdenum	1,200	97	65	ND(9.3)
Nickel	64,000	110	72	ND(39)
Selenium	5,500	26	29	11
Vanadium	760	560	380	200

Table D-9Elemental Analysis of ESP Inlet Particulate Matter Fractions - OFA TestSamples Collected on March 7, 1991

Sample No.	484	483	482	481
Sample ID	Stage 4	Stage 3	Stage 2	Stage 1
dP50 Size Range	< 2.0 µ m	2.0-7.0 μm	7.0-13 μm	>13 µ m
Substrate	Glass Thimble	None	None	None
Mass Collected, g	0.0646	0.499	0.553	3.871
% of Total Mass, g	1	10	11	78
Mass Analyzed, g	0.0646	0.1563	0.201	0.5022
Gas Volume, Nm ³	0.7764	0.7764	0.7764	0.7764
		<u></u>	<u>-</u>	
Elemental Conc., mg	/kg			
Aluminum	NC	150,000	150,000	150,000
Calcium	260,000	13,000	14,000	8,000
Iron	87,000	130,000	120,000	93,000
Titanium	ND(28,000)	10,000	8,900	6,500
Arsenic	5,300	760	420	88
Barium	1,200,000	2,600	1,800	960
Cadmium	ND(450)	ND(36)	ND(29)	ND(20)
Chromium	1,500	580	270	120
Cobalt	88	140	120	50
Copper	ND(13,000)	980	780	ND(590)
Manganese	34	ND(9.0)	160	120
Mercury	NC	ND(3.8)	ND(3.1)	ND(2.1)
Molybdenum	670	88	42	ND(7.5)
Nickel	ND(1,300)	640	310	ND(150)
Selenium	2,000	26	17	6.7
Vanadium	1,500	640	450	260

Table D-10 INAA Analysis of Standard Reference Material (SRM) Ash

Element	Reference Value (mg/kg)	0.096g Sample Recovery (%)	0.1013g Sample Recovery (%)
Aluminum	146,000	139	115
Calcium	18,600	149	38
Iron	55,700	149	48
Titanium	13,700	123	99
Arsenic	56.9	113	48
Barium	711	201	61
Cadmium	3.06	ND	ND
Chromium	183	158	47
Cobalt	49.8	153	53
Copper	157	ND	ND
Manganese	381	95	73
Nickel	117	ND	42
Vanadium	375	137	116

APPENDIX E: UNCERTAINTY FORMULAS

An error propagation analysis was performed on calculated results to determine the contribution of process, sampling, and analytical variability, and measurement bias, to the overall uncertainty in the result. This uncertainty was determined by propagating the bias and precision error of individual parameters through the calculation of the results. This uncertainty does not represent the total uncertainty in the result since many important bias errors are unknown and have been assigned a value of zero for this analysis. Also, this uncertainty is only for the period of time that the measurements were taken.

This method is based on ANSI/ASME PTC 19.1-1985, "Measurement Uncertainty."

Nomenclature

r =	Calculated result;
$S_{p_i} =$	Sample standard deviation of parameter i;
$\hat{\theta}_{i} =$	Sensitivity of the result to parameter i;
$\beta_{\rm pi} =$	Bias error estimate for parameter i;
$v_i =$	Degrees of freedom in parameter i;
v _r =	Degrees of freedom in result;
S, =	Precision component of result uncertainty;
$\beta_r =$	Bias component of result uncertainty;
t =	Student "t" factor (two-tailed distribution at 95% confidence);
$U_r =$	Uncertainty in r; and
$N_i =$	Number of measurements of parameter i.

For a result, r, the uncertainty in r is calculated as:

$$U_r = \sqrt{\beta_r^2 + (S_r * t)^2}$$
 (eq. 1)

Appendix E: Uncertainty Formulas

The components are calculated by combining the errors in the parameters used in the result calculation.

$$\beta_{r} = \sqrt{\sum_{i=1}^{j} (\theta_{i} * \beta_{\overline{pi}})^{2}} \qquad (eq. 2)$$

$$S_{r} = \sqrt{\sum_{i=1}^{j} (\theta_{i} * S_{\overline{pi}})^{2}} \qquad (eq. 3)$$

The sensitivity of the result to each parameter is found from a Taylor series estimation method:

$$\theta_i = \frac{\partial r}{\partial pi}$$
 (eq. 4)

Or using a perturbation method (useful in computer applications):

$$\theta_{i} = \frac{r(P_{i} + \Delta P_{i}) - r(P_{i})}{\Delta P_{i}}$$
(eq. 5)

Equation 5 was applied to the calculations in this report. The perturbation selected for each parameter was the larger of the normalized standard deviation, S_{pi} , or the bias, β_{pi} .

The standard deviation of the average for each parameter is calculated as:

$$S_{\overline{pi}} = \frac{S_{pi}}{\sqrt{N}}$$
 (eq. 6)

The degrees of freedom for each parameter is found from

$$\mathbf{v}_{i} = \mathbf{N}_{i} - 1 \tag{eq. 7}$$

and the degrees of freedom for the result is found by weighing the sensitivity and precision error in each parameter.

$$\mathbf{v}_{r} = \frac{\mathbf{S}_{r}^{4}}{\sum_{i=1}^{j} \left[\frac{(\mathbf{S}_{\overline{pi}} \times \boldsymbol{\theta}_{i})^{4}}{\mathbf{v}_{i}} \right]}$$
(eq. 8)

The student "t" in Equation 1 is associated with the degrees of freedom in the result.

The precision error terms are easily generated from the collected data. The bias error terms are more difficult to quantify. The following conventions were used for this report:

- 5% bias on coal and ash flow rates.
- No bias in gas flow rates.
- No bias in analytical results if the result is greater than the detection limit. One-half of the detection limit is used for both the parameter value and its bias in calculations if the result is below the detection limit.

Assignment of the flow rate bias values is based on engineering judgment. No bias is assigned to the analytical results (above the detection limit) or gas flow rate since a good estimate for magnitude of these terms is unknown. These bias terms may be very large (relative to the mean values of the parameters) and may represent a large amount of unaccounted uncertainty in each result. Analytical bias near the instrument detection limit may be especially large. The uncertainty values calculated for this report are, therefore, subject to these limitations.

The calculations assume that the population distribution of each measurement is normal and that the samples collected reflect the true population. Also, the uncertainty calculated is only for the average value over the sampling period. The uncertainty does not represent long-term process variations. In other words, the calculated uncertainty does not include a term to reflect the fact that the sampled system may not have been operating (and emitting) at conditions equivalent to the average conditions for that system over a longer period.

Improvements in bias estimates will be made as more data are collected and the QA/QC database is expanded. Spike and standard recoveries can be used to estimate analytical bias. Also, as the analytical methods improve, accuracy will improve, resulting in the true bias of the analytical results being closer to the zero bias now assigned. Accounting for long-term system variability will require repeated sampling trips to the same location.

APPENDIX F: QA/QC RESULTS

Table F.1-1

-

.

SUMMARY OF BLANK SAMPLE RESULTS FOR SITE 16

<u>Analyte Grouping</u>	Number of Blanks <u>Analyzed</u>	Number of <u>Detects</u>	Range of Compounds <u>Detected</u>	Method Detection <u>Limits</u>
Semivolatile Organics				
Lab Blanks (MM5)	1	0		
Field Blanks (MM5)	2		•	
Dibutylphthalate		2	0.83-106 ug	10 ug
Diethylphthalate		1	1.81 ug	10 ug
Bis(2-Ethylhexyl)				
phthalate		2	1.53-2.44 ug	10 ug
Naphthalene		1	2.05 ug	10 ug
Trip Blanks	2			
Dibutylphthalate		1	13.8 ug	10 ug
Diethylphthalate		1	3.32 ug	10 ug
Bis(2-Ethylhexyl)				10
phthalate		2	1.64-26.3 ug	10 ug
Naphthalene		1	1.82 ug	10 ug
Semivolatile Organics (Sol	ids)			
Lab Blanks	1			
Dibutylphthalate		1	0.104 ug/g	l ug
Semivolatile Organics (Wat	ters)			
Lab Blanks	1	0		
Field Blanks	1	0		
VOST				
Lab Blanks	4	0		
Field Blank - ESP Inlet	3			
Trichlorofluoromethane		1	25 ng	10 ng
Methylene Chloride		1	26 ng	10 ng
Field Blanks - Stack	3			
Trichlorofluoromethane		1	23 ng	10 ng
Methylene Chloride		1	410 ng	10 ng

Table F.1-1

(Continued)

	Number of Blanks	Number of	Range of Compounds	Method Detection
<u>Analyte Grouping</u>	<u>Analyzed</u>	<u>Detects</u>	<u>Detected</u>	<u>Limits</u>
Aldehydes				
Lab Blanks	1			
Formaldehyde		1	1.32 ug	0.48 ug
Field Blanks	1			
Formaldehyde		1	5.4 ug	2.4 ug
Trip Blanks	1		•	
Formaldehyde		1	3.06 ug	0.95 ug
Chloride				
Lab Blanks (Solids)	1	0		
Lab Blanks (Filters)	1	0		
Lab Blanks (Impingers)	1	0		
Lab Blanks (Waters)	1	0		
Field Blanks (Filters)	1	1	4.34 ug	
Trip Blanks (Filters)	1	1	3.75 ug	
Field Blanks (Impingers)	1	0		
Trip Blanks (Impingers)	1	1	0.0864 mg/L	0.036 mg/L
Field Blanks (Waters)	1	0		
Fluoride				
Lab Blanks (Solids)				
Lab Blanks (Filters)				
Lab Blanks (Impingers)	1	0		
Lab Blanks (Waters)	1	0		
Field Blanks (Filters)				
Trip Blanks (Filters)				
Field Blanks (Impingers)	1	0		
Trip Blanks (Impingers)	1	0		
Field Blanks (Waters)	1	0		

.

Table F.1-1

(Continued)

Analyta Cupyring	Number of Blanks	Number of	Range of Compounds	Method Detection
Analyte Grouping	Allalyzeu	Delecis	Detected	
Phosphate				
Lab Blanks (Solids)	1	0		
Lab Blanks (Filters)	1	0		
Lab Blanks (Impingers)	1	0		
Lab Blanks (Waters)	1	0		
Field Blanks (Filters)	1	0		
Trip Blanks (Filters)	2	0		
Field Blanks (Impingers)	1	1	0.228 mg/L	0.30 mg/L
Trip Blanks (Impingers)	1	0		
Field Blanks (Waters)	1	0		
Sulfate				
Lab Blanks (Solids)	1	0		
Lab Blanks (Filters)	1	0		
Lab Blanks (Impingers)	1	0		
Lab Blanks (Waters)	1	1		
Field Blanks (Filters)	1	1	0.0421 mg	0.0035 mg
Trip Blanks (Filters)	2	2	0.0398-0.052 mg	0.008 mg
Field Blanks (Impingers)	1	1	2.04 mg/L	2.4 mg/L
Trip Blanks (Impingers)	1	0		-
Field Blanks (Waters)	1	0		
Metals (ICP) - Coal				
Lab Blanks				
Aluminum	1	1	3.65 mg/kg	20 mg/kg
Barium	1	1	0.01 mg/kg	1.0 mg/kg
Calcium	1	1	2.32 mg/kg	100 mg/kg
Chromium	1	1	0.500 mg/kg	1.0 mg/kg
Cobalt	1	1	0.17 mg/kg	1.0 mg/kg
Copper	1	1	0.18 mg/kg	2.0 mg/kg
Iron	1	1	3.91 mg/kg	4.0 mg/kg
Magnesium	1	1	1.04 mg/kg	100 mg/kg
Manganese	1	1	0.07 mg/kg	1.0 ma/ka

Table F.1-1

(Continued)

Analyte Grouping	Number of Blanks Analvzed	Number of Detects	Range of Compounds Detected	Method Detection Limits
		<u></u>		<u> </u>
(Cont'd)				
Molybdenum	1	1	6.95 mg/kg	5.0 mg/kg
Nickel	1	1	0.92 mg/kg	2.0 mg/kg
Potassium	1	1	1.53 mg/kg	300 mg/kg
Sodium	1	1	4.90 mg/kg	100 mg/kg
Titanium	1	1	0.47 mg/kg	5.0 mg/kg
Thallium	1	1	0.27 mg/kg	10 mg/kg
Vanadium	1	1	0.99 mg/kg	2.0 mg/kg
Zinc	1	1	0.80 mg/kg	2.0 mg/kg
Metals (GFAAS, HGAAS, CVAA Coal	S) -			
Arsenic	1	0		
Cadmium	1	0		
Lead	1	0		
Mercury	1	0		
Selenium	1	0		
Metals (ICP) - Filters				
Field Blanks	1			
Aluminum		1	232 ug	20 ug
Barium		1	6.75 ug	1.0 ug
Calcium		1	101 ug	100 ug
Chromium		1	2.13 ug	1.0 ug
Copper		1	0.986 ug	2.0 ug
Iron		1	35 ug	4.0 ug
Magnesium		1	16.6 ug	100 ug
Manganese		1	1.28 ug	1.0 ug
Molybdenum		1	29.2 ug	5.0 ug
Nickel		1	2.10 ug	2.0 ug
Potassium		1	17.0 ug	300 ug
Sodium		1	110 ug	100 ug
Titanium		1	2.3 ug	0.30 ug

.

Table F.1-1

(Continued)

Analyte Grouping	Number of Blanks <u>Analyzed</u>	Number of <u>Detects</u>	Range of Compounds <u>Detected</u>	Method Detection <u>Limits</u>
Metals (ICP) - Filters (Cont'd)				
Vanadium		1	0.900 ug	2.0 ug
Zinc		1	3.74 ug	2.0 ug
Metals (GFAAS, HGAAS, CVAAS) Filters	-			
Field Blanks	1			
Arsenic		1	0.16 ug	0.40 ug
Cadmium		0		
Lead		1	0.56 ug	0.30 ug
Mercury		1	0.0082 ug	0.018 ug
Selenium		0		
Metals (ICP) - Filters				
Trip Blanks	1			
Aluminum		1	230 ug	20 ug
Barium		1	6.27 ug	1.0 ug
Calcium		1	114 ug	100 ug
Chromium		1	2.06 ug	1.0 ug
Copper		1	1.3 ug	2.0 ug
Iron		1	32.4 ug	4.0 ug
Magnesium		1	14.9 ug	100 ug
Manganese		1	1.17 ug	1.0 uq
Molybdenum		1	29.6 ug	5.0 ug
Nickel		1	1.95 ug	2.0 ug
Silicon		1	330000 ug	1000 ug
Sodium		1	105 ug	300 ug
Strontium		1	2.27 ug	0.3 ua
Vanadium		1	0.931 ug	2.0 ug
Zinc		1	3.03 ug	2.0 ug
Table F.1-1

(Continued)

Metals (GFAAS, HGAAS, CVAAS) - Filters Trip Blanks 1 Arsenic Cadmium Lead	1 0	0.13 ug	
Trip Blanks 1 Arsenic Cadmium Lead	1 0	0.13 ug	
Arsenic Cadmium Lead Momeumu	1 0	0.13 ug	
Cadmium Lead Momoumy	0		0.4 ug
Lead	1		
Маланиян	1	0.57 ug	0.030 ug
mercury	1	0.0424 ug	0.018 ug
Selenium	0		
Metals (ICP) - Probe and Nozzla Pinco			
Field Blanks			
	1	6 51 ug	20 ug
Rarium	1	1 46 ug	10 ug
Calcium	1	17 7 ug	1.0 ug
Chromium	1	0.53.00	100 ug
Cohalt	l	0.129 ug	1.0 ug
Conner	1		2.0 ug
Iron	1	30.8 ug	4.0 ug
Magnesium	1	4.47 ug	100 ug
Manganese	1	0.139 ug	1.0 ug
Molvbdenum	1	0.894 ug	5.0 ug
Nickel	1	0.700 ug	2.0 ug
Silicon	1	18500 ug	1000 ug
Sodium	1	71 ug	300 ug
Strontium	1	0 692 ug	0 3 ug
Vanadium	1	0.128 ug	2.0 ug
Zinc	1	1.22 ug	2.0 ug
Metals (GFAAS, HGAAS, CVAAS) - Probe and Nozzle Rinse			
Field Blanks 1			
Arsenic	1	0.63 ua	0.4 ua
Cadmium	n	0 100 10	0.1 ua

Table F.1-1

(Continued)

Analyte Grouping	Number of Number Ran Blanks of Com <u>yte Grouping Analyzed Detects Det</u>		Range of Compounds <u>Detected</u>	Method Detection Limits	
Metals (GFAAS, HGAAS, CVAA Probe and Nozzle Rinse (Co	NS) - ont'd)				
Lead		1	0.50 ug	0.3 ug	
Mercury		0			
Selenium		0			
Metals (ICP) - Probe and Nozzle Rinse					
Trip Blanks	1				
Barium		1	0.0652 ug	1.0 ug	
Beryllium		1	0.149 ug	0.2 ug	
Calcium		1	25.1 ug	100 ug	
Chromium		1	0.179 ug	1.0 ug	
Copper		1	0.34 ug	2.0 ug	
Iron		1	2.58 ug	4.0 ug	
Magnesium		1	7.16 ug	100 ug	
Manganese		1	0.142 ug	l ug	
Silicon		1	19000 ug	1000 ug	
Sodium		1	48.2 ug	100 ug	
Strontium		1	0.235 ug	0.3 ug	
Thallium		1	0.903 ug	10 ug	
Zinc		1	0.905 ug	2.0 ug	
Metals (GFAAS, HGAAS, CVA Probe and Nozzle Rinse	AS) -				
Trip Blanks	1				
Arsenic		1	0.14 ug	0.4 ug	
Cadmium		1	0.17 ug	0.1 ug	
Lead		0			
Mercury		1	0.0302 ug	0.009 ug	
Selenium		0			

-

Table F.1-1

•

(Continued)

	Number of Blanks	Number of	Range of Compounds	Method Detection
<u>Analyte Grouping</u>	<u>Analyzed</u>	<u>Detects</u>	Detected	<u>Limits</u>
Metals (ICP) - Impinger So	lutions			
Field Blanks	2			
Aluminum		2	0.213-0.228 mg/L	0.2 mg/L
Antimony		2	0.001-0.042 mg/L	0.1 mg/L
Barium		2	0.003-0.006 mg/L	0.01 mg/L
Calcium		2	0.145-0.154 mg/L	1.0 mg/L
Chromium		2	0.003-0.005 mg/L	0.01 mg/L
Cobalt		1	0.006 mg/L	0.01 mg/L
Copper		2	0.004-0.007 mg/L	0.02 mg/L
Iron		2	0.058-0.063 mg/L	0.04 mg/L
Magnesium		2	0.005-0.039 mg/L	1.0 mg/L
Manganese		2	0.017-0.025 mg/L	0.01 mg/L
Molybdenum		2	0.054 mg/L	0.05 mg/L
Nickel		2	0.016-0.017 mg/L	0.02 mg/L
Potassium		2	0.124-0.30 mg/L	3.0 mg/L
Silicon		2	0.578-0.975 mg/L	1.0 mg/L
Silver		2	0.001 mg/L	0.01 mg/L
Sodium		2	0.35-0.564 mg/L	1.0 mg/L
Strontium		2	0.009-0.0012 mg/L	0.003 mg/L
Thallium		2	0.049-0.064 mg/L	0.1 mg/L
Vanadium		2	0.006-0.008 mg/L	0.02 mg/L
Zinc		2	46-47.6 mg/L	0.02 mg/L
Metals (GFAAS, HGAAS, CVAA Impinger Solutions	lS) -			
Field Blanks	2			
Arsenic		2	0.0010-0.0011 mg/L	0.004 mg/L
Cadmium		2	0.0007-0.0009 mg/L	0.001 mg/L
Lead		2	0.0068-0.0122 mg/L	0.003 mg/L
Mercury		0		
Selenium		0		

•

•

Table F.1-1

•

(Continued)

	Number	Numbon	Damas of	Mathad
	UT Blanks	number	Kange of Compounds	Detection
Analyte Grouning	Didiiks Analyzed	Detects	Detected	limite
<u>And ty ce at oup the</u>	<u>Anulyzed</u>	Deteets	Detected	<u>L 1111 (5</u>
Metals (ICP) - Impinger So	lutions			
Trip Blanks	1			
Aluminum		1	0.206 mg/L	0.2 mg/L
Barium		1	0.005 mg/L	0.01 mg/L
Calcium		1	0.039 mg/L	1.0 mg/L
Chromium		1	0.005 mg/L	0.01 mg/L
Copper		1	0.015 mg/L	0.02 mg/L
Iron		1	0.027 mg/L	0.04 mg/L
Magnesium		1	0.016 mg/L	1.0 mg/L
Manganese		1	0.02 mg/L	0.01 mg/L
Molybdenum		1	0.059 mg/L	0.05 mg/L
Nickel		1	0.022 mg/L	0.02 mg/L
Potassium		1	0.244 mg/L	3.0 mg/L
Silicon		1	0.152 mg/L	1.0 mg/L
Sodium		1	0.197 mg/L	1.0 mg/L
Strontium		1	0.0009 mg/L	0.003 mg/L
Thallium		1	0.081 mg/L	0.1 mg/L
Vanadium		1	0.007 mg/L	0.02 mg/L
Zinc		1	53.7 mg/L	0.02 mg/L
Metals (GFAAS, HGAAS, CVA	AS) -			
Impinger Solutions				
Trip Blanks	1			
Arsenic		1	0.0001 mg/L	0.004 mg/l
Cadmium		1	0.0014 mg/L	0.001 mg/L
Lead		1	0.015 mg/L	0.003 mg/L
Mercury		0		
Selenium		0		
Metals (ICP) - Impinger S	olutions			
Lab Blanks	1			
Antimony		1	0.005 mg/L	0.1 mg/L
Calcium		1	0.036 mg/L	1.0 mg/L
Cobalt		1	0.002 mg/L	0.01 mg/l

Table F.1-1

Analyte Grouping	Number of Blanks <u>Analyzed</u>	Number of <u>Detects</u>	Range of Compounds <u>Detected</u>	Method Detection <u>Limits</u>
Metals (ICP) - Impinger So (Cont'd)	lutions			
Copper		1	0.002 mg/L	0.02 mg/L
Magnesium		1	0.001 mg/L	1.0 mg/L
Silicon		1	0.055 mg/L	1.0 mg/L
Zinc		1	0.015 mg/L	0.02 mg/L
Metals (GFAAS, HGAAS, CVAA	S) -			
Impinger Solutions				
Lad Blanks	1		0.005 (1	0.001 ()
Arsenic		1	0.005 mg/L	0.004 mg/L
		1	0.006 mg/L	0.001 mg/L
Lead		1	0.0049 mg/L	0.003 mg/L
Mercury Solonium		0		
Selenium		U		
Metals (ICP) - Solid Samples				
Lab Blanks	1			
Beryllium		1	2.4 mg/kg	2.0 mg/kg
Chromium		1	20.7 mg/kg	10 mg/kg
Molybdenum		1	58 mg/kg	50 mg/ kg
Silicon		1	159000 mg/kg	10000 mg/kg
Silver			56.4 mg/kg	40 mg/kg
Metals (ICP) - Water Sampl	es			
Lab Blanks	1			
Antimony		1	0.035 mg/L	0.10 mg/L
Cobalt		1	0.002 mg/L	0.01 mg/L
Copper		1	0.001 mg/L	0.02 mg/L
Magnesium		1	0.001 mg/L	0.01 mg/L
Manganese		1	0.037 mg/L	0.05 mg/L
Sodium		1	0.198 mg/L	1 mg/L

Analyte Grouping	Number of Blanks <u>Analyzed</u>	Number of <u>Detects</u>	Range of Compounds <u>Detected</u>	Method Detection <u>Limits</u>
Metals (ICP) - Water Samples	(Cont'd)			
Vanadium		1	0.004 mg/L	0.02 mg/L
Zinc		1	0.001 mg/L	0.001 mg/L
Metals (GFAAS, HGAAS, CVAAS) Water Samples	-			
Lab Blanks	1			
Arsenic		0		
Cadmium		0		
Metals (GFAAS, HGAAS, CVAAS) Water Samples (Cont'd)	-			
Lead	Þ	0		
Mercury		0		
Selenium		0		

SUMMARY OF QUALITY CONTROL CHECK SAMPLE RESULTS FOR SITE 16

		_			No.
Parameter	No. OCCS	Avg. % Rec	Std. Dev	No. Below	Above
<u>raname cer</u>	4000	<u>// nee.</u>	<u>0011</u>	<u>E 1111 (5</u>	<u>L INI 63</u>
Semi-Volatile Organics:					
Acenaphthene	2	66	2.8	0	0
4-Chloro-3-methylphenol	2	78	0.71	0	0
2-Chlorophenol	2	84 、	0.71	0	0
1,4-Dichlorobenzene	2	86	0.71	0	0
2,4-Dinitrotoluene	2	68	3.5	0	0
N-Nitrosodipropylamine	2	80	0.71	0	0
4-Nitrophenol	2	74	2.8	0	0
Pentachlorophenol	2	114	0	0	0
Phenol	2	78	2.1	0	0
Pyrene	2	82	0.71	0	0
1,2,4-Trichlorobenzene	2	94	0	0	0
Volatile Organics (VOST):					
Chloromethane	2	52	0.7	0	0
Vinyl Chloride	2	71	5.6	0	0
Bromomethane	2	76	3.5	0	0
Chloromethane	2	81	1.4	0	0
Trichlorofluoromethane	2	80	4.9	0	0
1,1-Dichloroethene	2	80	3.5	0	0
Carbon Disulfide	2	93	4.2	0	0
Acetone	2	94	21.9	0	0
Methylene Chloride	2	83	2.8	0	0
trans-1,2-Dichloroethene	2	82	2.1	0	0
1,1-Dichloroethane	2	90	6.4	0	0
Vinyl Acetate	2	69	53.7	0	0
2-Butanone	2	103	15.6	0	0
Chloroform	2	93	7.1	0	0
1,1,l-Trichloroethane	2	78	0.7	0	0
Carbon Tetrachloride	2	78	6.4	0	0
Benzene	2	90	14.1	0	0
1,2-Dichloroethane	2	116	14.1	0	0
Trichloroethene	2	105	19.8	0	0

۴

Table F.1-2

(Continued)

No.	Avg.	Std.	No. Below	No. Above
<u>QUUS</u>	<u>% ReC.</u>	<u>Dev.</u>	LIMIUS	Limits
2	106	13.4	0	0
2	100	2.1	0	0
2	105	9.9	0	0
2	74	5.6	0	0
2	96	2.1	0	0
2	82	11.3	0	0
2	121	14.1	0	0
2	80	13.4	0	0
2	102	1.41	0	0
2	102	0.7	0	0
2	94	1.4	0	0
2	86	2.1	0	0
2	87	2.8	0	0
2	104	9.2	0	0
2	120	4.2	0	0
2	120	2.1	0	0
2	94	36.1	0	0
2	96	7.1	0	0
2	96	9.2	0	0
2	93	2.8	0	0
1	98	-	0	0
3	111	11.3		
1	105	-		
1	101	-		
into Reage	nt			
5	92	1.6	0	0
5	84	15.5	1	0
5	95	3.7	0	0
5	89	4.3	0	0
	No. QCCS 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	No.Avg. % Rec.21062100210527429628221212802102210221022102294296294296294296296296296296296296296296296293198311111051101intoReagent592584595589	No. Avg. % Rec. Std. Dev. 2 106 13.4 2 100 2.1 2 105 9.9 2 74 5.6 2 96 2.1 2 105 9.9 2 74 5.6 2 96 2.1 2 82 11.3 2 121 14.1 2 80 13.4 2 102 0.7 2 94 1.4 2 102 0.7 2 94 1.4 2 102 0.7 2 94 1.4 2 104 9.2 2 120 4.2 2 96 7.1 2 96 7.1 2 96 9.2 2 93 2.8 1 101 - 3 <td< td=""><td>No. Avg. % Rec. Std. Dev. No. Below Limits 2 106 13.4 0 2 100 2.1 0 2 105 9.9 0 2 74 5.6 0 2 96 2.1 0 2 96 2.1 0 2 82 11.3 0 2 102 1.41 0 2 102 1.41 0 2 102 0.7 0 2 102 0.7 0 2 94 1.4 0 2 86 2.1 0 2 104 9.2 0 2 104 9.2 0 2 96 7.1 0 2 96 7.1 0 2 96 9.2 0 2 93 2.8 0 1 <td< td=""></td<></td></td<>	No. Avg. % Rec. Std. Dev. No. Below Limits 2 106 13.4 0 2 100 2.1 0 2 105 9.9 0 2 74 5.6 0 2 96 2.1 0 2 96 2.1 0 2 82 11.3 0 2 102 1.41 0 2 102 1.41 0 2 102 0.7 0 2 102 0.7 0 2 94 1.4 0 2 86 2.1 0 2 104 9.2 0 2 104 9.2 0 2 96 7.1 0 2 96 7.1 0 2 96 9.2 0 2 93 2.8 0 1 <td< td=""></td<>

Table F.1-2

<u>Parameter</u>	No. QCCS	Avg. <u>% Rec.</u>	Std. <u>Dev.</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals (Predigestion Spike i	nto Reag	ent			
Water) (Cont'd):					
Calcium	5	94	3.3	0	0
Chromium	5	94	4.4	0	0
Cobalt	5	94	4.2	0	0
Copper	5	95	5.2	0	0
Iron	5	93	3.4	0	0
Magnesium	5	90	2.2	0	0
Manganese	5	92	4.1	0	0
Molybdenum	5	107	26.3	0	1
Nickel	5	93	6.1	0	0
Potassium	5	92	6.8	0	0
Silicon	5	708	1038	0	1
Silver	5	31	33.6	3	0
Sodium	5	94	3.2	0	0
Strontium	5	95	3.0	0	0
Thallium	5	79	44.6	0	0
Titanium	4	89	4.1	0	0
Vanadium	5	93	4.1	0	0
Zinc	5	89	6.0	0	0
Metals by GFAAS and HGAAS (F Spike into Reagent Water):	Predigest	ion			
Arsenic	4	98	16.4	1	0
Cadmium	5	100	26.0	1	1
Lead	5	111	13.0	0	1
Selenium	2	79	17.0	1	0
Metals-ICAPES (NBS Fly Ash 1	1633A):				
Aluminum	2	91	1.1	0	0
Barium	2	82	3.7	0	0
Beryllium	2	91	4.4	0	0
Calcium	2	90	3.3	0	0
Chromium	2	93	6.4	0	0
Cobalt	2	108	8.3	0	0

Table F.1-2

Parameter	No. QCCS	Avg. <u>% Rec.</u>	Std. <u>Dev.</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals-ICAPES (NBS Fly Ash (Cont'd):	1633A)				
Copper	2	91	4.4	0	0
Iron	2	91	0	0	0
Magnesium	2	86	3.5	0	0
Manganese	2	88	0	0	0
Molybdenum	1	266	-	0	1
Nickel	2	94	0	0	0
Silicon	1	166	24.8	0	2
Sodium	2	92	5.4	0	0
Strontium	2	89	6.7	0	` 0
Titanium	2	95	1.0	0	0
Vanadium	2	92	1.1	0	0
Zinc	2	88	1.1	0	0
Metals -GFAAS, HGAAS, and (NBS Fly Ash 1633A)	CVAAS				
Arsenic	1	143	-	1	1
Cadmium	1	150	-	1	1
Mercury	1	69	-	1	0
Lead	1	131	-	0	1
Selenium	2	83	14.4	0	0
Metals-ICAPES (NBS Coal 1	632A):				
Aluminum	2	62	36.5	2	0
Barium	2	56	25.6	2	0
Beryllium	2	62	40.4	1	0
Calcium	2	107	29.5	0	1
Cobalt	1	114	-	0	0
Iron	2	78	14.2	1	0
Manganese	2	76	7.9	1	0
Nickel	2	260	116	0	1
Potassium	2	76	11.9	1	0
Sodium	2	92	16.4	0	0
Thallium	1	100	-	0	0

Table F.1-2

(Continued)

Parameter	No. QCCS	Avg. <u>%_Rec.</u>	Std. <u>Dev.</u>	No. Below Limits	No. Above <u>Limits</u>
Metals-ICAPES (NBS Coal 163 (Cont/d)	B2A):				
Titanium	1	88	-	0	0
Strontium	1	46	-	1	0
Zinc	1	87	-	0	0
Metals -GFAAS (NBS Coal 163	33A)				
Arsenic	2	288	144	0	2
Lead	2	58	17.2	2	1
Selenium	2	62	9.7	2	0

Table F.1-3

SUMMARY OF SPIKED SAMPLE RESULTS FOR SITE 16

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD <u>(Std. Dev.)</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Semivolatile Organics -					
Bottom Ash:	-				
Acenaphthene	2	64	7.8	0	0
4-Chloro-3-methylphenol	2	82	3.6	0	0
2-Chlorophenol	2	82	6.1	0	0
1,4-Dichlorobenzene	2	88	6.8	0	0
2,4-Dinitrotoluene	2	68	4.4	0	0
N-Nitrosodipropylamine	2	80	6.2	0	0
4-Nitrophenol	2	63	0	0	0
Pentachlorophenol	2	96	5.2	0	0
Phenol	2	84	3.6	0	0
Pyrene	2	125	11.2	0	0
1,2,4-Trichlorobenzene	2	94	3.2	0	0
Anions:					
Chloride – Solids	4	94	4.8	0	0
Chloride - Water	1	100	-	0	0
Fluoride - Gas	1	103	-	0	0
Fluoride – Solids	5	77	23.2	2	0
Fluoride - Waters	2	100	11.3	0	0
Phosphate - Waters	1	105	-	0	0
Phosphate - Solids	1	76	-	0	0
Metals by ICP in ESP Inlet (Impinger Solutions):	Gas				
Aluminum	2	92	0	0	0
Antimony	2	86	1.2	0	0
Barium	2	94	1.1	0	0
Beryllium	2	96	1.0	0	0
Boron	2	90	1.1	0	0
Calcium	2	96	0	0	. 0
Chromium	2	92	0	0	0
Cobalt	2	92	1.1	0	0
Copper	2	91	0	0	0

.

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD (Std. Dev.)	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by ICP in ESP Inlet	Gas				
(Impinger Solutions): (Con	t'd)				
Iron	2	96	0	0	0
Magnesium	2	90	1.1	0	0
Manganese	2	91	1.1	-0	0
Molybdenum	2	94	1.2	0	0
Nickel	2	94	1.1	0	0
Potassium	2	86	1.2	0	0
Silicon	2	98	1.0	0	0
Silver	2	86	1.2	0	0
Sodium	2	92	0	0	0
Strontium	2	95	2.1	0	0
Thallium	2	93	3.2	0	0
Titanium	2	94	1.1	0	0
Vanadium	2	92	0	0	0
Zinc	2	40	100	2	0
Metals by GFAAS, HGAAS, and CVAAS in Impinger Solutions	 \:				
Arsenic	2	74	1.3	1	0
Cadmium	2	106	7.5	0	0
Lead	2	74	1.3	2	0
Mercury	2	98	0	0	0
Selenium	2	70	5.1	2	0
Metals by ICP in Stack Gas (Probe and Nozzle Rinse):					
Antimony	2	84	2.4	0	0
Barium	2	94	1.0	0	0
Beryllium	2	86	0	0	0
Chromium	2	95	1.0	0	0
Cobalt	2	96	2.1	0	0
Copper	2	92	61.10	0	0
Manganese	2	91	0	0	0
Molybdenum	2	95	0	0	0

Table F.1-3

(Continued)

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD (Std. Dev.)	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by ICP in Stack Gas	(Probe				
and Nozzle Rinse) (Cont'd):	•			•	
Nickel	2	92	4.3	0	0
Silver	2	0	NC	2	0
Strontium	2	93	2.2	0	0
Thallium	2	91	2.2	0	0
Titanium	2	92	1.1	0	0
Vanadium	2	94	0	0	0
Zinc	2	86	0	0	0
Metals by GFAAS in Stack Ga (Probe and Nozzle Rinse):	\$				
Selenium	2	72	62.80	2	0
Metals by ICP in ESP Inlet Gas (Filter):					
Aluminum	2	63	15.9	2	0
Antimony	2	80	11.2	0	0
Barium	2	80	6.2	0	0
Beryllium	2	74	4.0	1	0
Calcium	2	88	4.5	0	0
Chromium	2	83	2.4	0	0
Cobalt	2	85	2.4	0	0
Copper	2	84	3.6	0	0
Iron	2	72	4.1	2	0
Magnesium	2	77	7.8	1	0
Manganese	2	84	6.0	0	0
Molybdenum	2	79	25	0	0
Nickel	2	82	1.2	0	0
Potassium	2	86	2.3	0	0
Silver	2	11	18.2	2	0
Sodium	2	90	5.6	0	0
Strontium	2	68	10.4	2	0
Thallium	2	82	11.0	0	0
Titanium	2	78	2.6	0	0

Table F.1-3

(Continued)

.

1

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD <u>(Std. Dev.)</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by ICP in ESP Inlet					
Gas (Filter): (Lont'd)	•			•	•
Vanadium	2	83	2.4	0	0
Zinc	2	80	2.5	0	0
Metals by GFAAS, HGAAS, and in ESP Inlet Gas (Filter):	CVAAS				
Arsenic	2	71	2.8	2	0
Cadmium	2	91	2.2	0	0
Lead	2	92	1.1	0	0
Mercury	2	310	2.9	0	2
Selenium	2	92	13.0	0	0
Metals by ICP in Coal:					
Aluminum	4	106	50	1	1
Antimony	4	92	5.0	0	0
Barium	4	102	7.5	0	0
Beryllium	4	84	5.4	0	0
Calcium	4	119	20.2	0	2
Chromium	4	86	14.6	1	0
Cobalt	4	94	3.9	0	0
Copper	4	98	33.1	2	1
Iron	4	52.5	49	2	0
Magnesium	4	83	13.4	1	0
Manganese	4	95	2.6	0	0
Molybdenum	4	95	2.9	0	0
Nickel	4	88	19.9	1	0
Potassium	4	99	· 9.0	0	0
Silver	4	12.6	7.4	4	0
Sodium	4	94	2.1	0	0
Strontium	4	109	9.1	0	0
Thallium	4	97	7.6	0	0

Table F.1-3

(Continued)

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD (Std. Dev.)	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by ICP in Coal: (Cont'd)					
Vanadium	4	97	4.2	0	0
Zinc	4	88	3.3	0	0
Metals by GFAAS, HGAAS, and CVAAS in Coal:					
Arsenic	4	108	8.3	0	0
Cadmium	4	99	12.3	0	0
Lead	4	116	32.3	0	2
Mercury	4	57	21.5	2	0
Selenium	4	110	15.9	0	1
Metals by ICP in Fly Ash:					
Aluminum	2	57	21.1	2	0
Antimony	2	68	10.4	2	0
Barium	2	80	7.5	0	0
Beryllium	2	80	0	0	0
Calcium	2	75	8.0	1	0
Chromium	2	86	0	0	0
Cobalt	2	90	1.1	0	0
Copper	2	90	0	0	0
Iron	2	82	4.9	0	0
Magnesium	2	61	9.8	2	0
Manganese	2	82	2.4	0	0
Molybdenum	2	84	3.6	0	0
Nickel	2	88	1.1	0	0
Potassium	2	92	1.1	0	0
Silver	2	25	20	2	0
Sodium	2	80	2.5	0	0
Strontium	2	68	14.7	2	0
Thallium	2	96	1.0	0	0
Vanadium	2	88	0	0	0
Zinc	2	85	5.9	0	0

Table F.1-3

Compound	No. of <u>Spikes</u>	% <u>Recovery</u>	Mean RPD <u>(Std. Dev.)</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by GFAAS, HGAAS, and CVAAS in Fly Ash:					
Arsenic	2	89	11.2	0	0
Cadmium	2	150	5.3	0	2
Lead	2	67	20.9	2	0
Mercury	2	97	2.1	0	0
Selenium	2	88	0	0	0
Metals by ICP in Bottom Ash Sluice Water:					
Aluminum	2	97	0	0	0
Antimony	2	92	0	0	0
Barium	2	97	1.0	0	0
Beryllium	2	96	0	0	0
Boron	2	96	1.0	0	0
Calcium	2	98	6.1	0	0
Chromium	2	94	1.1	0	0
Cobalt	2	94	0	0	0
Copper	2	94	0	0	0
Iron	2	96	1.1	0	0
Magnesium	2	94	1.1	0	0
Manganese	2	94	1.1	0	0
Molybdenum	2	92	1.1	0	0
Nickel	2	94	1.1	0	0
Potassium	2	94	1.1	0	0
Silicon	2	105	1.9	0	0
Silver	2	93	0	0	0
Sodium	2	97	21	0	0
Strontium	2	96	1.0	0	0
Thallium	2	96	0	0	0
Vanadium	2	93	0	0	0
Zinc	2	92	1.1	0	0

Table F.1-3

(Continued)

Compound	No. of <u>Spikes</u>	% <u>Reçovery</u>	Mean RPD <u>(Std. Dev.)</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>
Metals by ICP in Sluice Water Supply:					
Aluminum	2	99	1.0	0	0
Antimony	2	90	1.1	0	0
Barium	2	96	1.0	0	0
Beryllium	2	96	1.1	0	0
Boron	2	96	0	0	0
Calcium	2	98	5.1	0	0
Chromium	2	94	1.1	0	0
Cobalt	2	94	1.1	0	0
Copper	2	94	1.1	0	0
Iron	2	95	1.1	0	0
Magnesium	2	95	0	0	0
Manganese	2	94	1.1	0	0
Molybdenum	2	92	1.1	0	0
Nickel	2	96	1.0	0	0
Potassium	2	102	1.0	0	0
Silicon	2	108	1.0	0	• 0
Silver	2	92	1.1	0	D
Sodium	2	97	0	0	0
Strontium	2	96	0	0	0
Thallium	2	94	3.3	0	0
Vanadium	2	93	2.2	0	0
Zinc	2	92	1.1	0	0

SUMMARY OF SURROGATE RECOVERIES FOR VOLATILE ORGANIC ANALYSES

	No.	M 04		No.	No.	
Cita 16	OT Annikana	Mean %	Std.	Below	Above	QC Vinita (V
<u>Site 10</u>	Analyses	<u>ReC.</u>	<u>uev.</u>		<u>Limits</u>	Limits %
ESP Inlet Gas (VOST):						
1,4-Bromofluorobenzene	9	103	11.9	0	0	50-150
1,2-Dichloroethane-d14	9	92	13.4	0	0	5 0- 150
Toluene-d8	9	104	6.2	0	0	50-150
Stack Gas (VOST):						
1,4-Bromofluorobenzene	9	110	15.1	0	1	50-150
1,2-Dichloroethane-d14	9	94	6.6	0	0	50-150
Toluene-d8	9	109	10.6	0	0	50-150
VOST Field Blanks (ESP Inl	et):					
1,4-Bromofluorobenzene	3	100	16.2	0	0	50-150
1,2-Dichloroethane-d14	3	90	6.4	0	0	50-150
Toluene-d8	3	105	8.2	0	0	50-150
VOST Field Blanks (Stack):	:					
1,4-Bromofluorobenzene	3	118	11.2	0	0	50-150
1,2-Dichloroethane-d14	3	94	6.1	0	0	50-150
Tol uene-d8	3	109	8.5	0	0	50-150
VOST Lab Blanks:						
1,4-Bromofluorobenzene	4	110	13.6	0	0	50-150
1,2-Dichloroethane-d14	4	92	3.3	0	0	50-150
Toluene-d8	4	110	8.3	0	0	50-150

SUMMARY OF SURROGATE RECOVERIES FOR SEMIVOLATILE ORGANIC ANALYSES

<u>Site 16</u>	No. of <u>Analyses</u>	Mean % <u>Rec.</u>	Std. Dev.	No. Below <u>Limits</u>	No. Above <u>Limits</u>	QC Limits %
Gas:						
2-Fluorobiphenyl	12	88	23.2	1	0	30-115
2-Fluorophenol	12	70	23.2	1	0	25-121
Nitrobenzene-d5	12	80	22.5	1	D	23-120
Pheno1-d5	12	80	21.7	1	0	24-114
Terphenyl-d14	12	91	31.0	1	0	18-137
2,4,6-Tribromophenol	12	109	32.7	1	4	19-122
Solids:						
2-Fluorobiphenyl	8	102	2.6	0	0	30-115
2-Fluorophenol	8	86	3.4	0	0	25-121
Nitrobenzene-d5	8	92	2.2	0	0	23-120
Phenoì-d5	8	91	4.6	0	0	24-114
Terphenyl-d14	8	94	21.4	0	0	18-137
2,4,6-Tribromophenol	8	70	5.2	0	0	19-122
Waters:						
2-Fluorobiphenyl	12	67	7.3	0	0	43-116
2-Fluorophenol	12	65	4.1	0	0	21-100
Nitrobenzene-d5	12	86	5.0	0	0	35-114
Phenol-d5	12	85	2.7	0	0	10-94
Terpheny]-d14	12	130	14.2	0	2	33-141
2,4,6-Tribromophenol	12	81	13.8	0	0	10-123

<u>Site 16</u>	No. of <u>Analyses</u>	Mean % <u>Rec.</u>	Std. Dev.	No. Below <u>Limits</u>	No. Above <u>Limits</u>	QC Limits %
Field Blanks - Gas Samp	les:					
2-Fluorobiphenyl	2	92	1.4	0	0	30-115
2-Fluorophenol	2	51	46.7	1	0	25-121
Nitrobenzene-d5	2	98	3.5	0	0	23-120
Phenol-d5	2	58	54.4	1	0	24-114
Terphenyl-d14	2	89	1.4	0	0	18-137
2,4,6-Tribromophenol	2	64	2.8	0	0	19-122
Trip Blanks - Gas Sample	es:					
2-Fluorobiphenyl	2	99	8.5	0	0	30-115
2-Fluorophenol	2	72	2.1	0	0	25-121
Nitrobenzene-d5	2	80	6.4	0	0	23-120
Phenol-d5	2	86	6.4	0	0	24-114
Terphenyl-d14	2	96	3.5	0	0	18-137
2,4,6-Tribromophenol	2	85	41.0	0	0	19-122
Lab Blanks - Gas Samples	S					
2-Fluorobiphenyl	3	99	1.2	0	0	43-116
2-Fluorophenol	3	78	1.2	0	0	21-100
Nitrobenzene-d5	3	84	1.5	0	0	35-114
Phenol-d5	3	89	2.9	0	0	10-94
Terphenyl-d14	3	98	2.1	0	0	33-141
2,4,6-Tribromophenol	3	117	5.3	0	0	10-123

Table F.1-5

<u>Site 16</u>	No. of <u>Analyses</u>	Mean % <u>Rec.</u>	Std. <u>Dev.</u>	No. Below <u>Limits</u>	No. Above <u>Limits</u>	QC Limits _ <u>%</u>
Lab Blanks - Solid Samp	les					
2-Fluorobiphenyl	3	101	6.1	0	0	43-116
2-Fluorophenol	3	77	3.1	0	0	21-100
Nitrobenzene-d5	3	95	0.6	0	0	35-114
Phenol-d5	3	94	3.1	0	0	10-94
Terphenyl-d14	3	123	12.6	0	0	33-141
2,4,6-Tribromophenol	3	83	7.6	0	0	10-123
Field Blanks - Water Sa	mples					
2-Fluorobiphenyl	1	80	NC	0	0	43-116
2-Fluorophenol	1	68	NC	0	0	21-100
Nitrobenzene-d5	1	98	NC	0	0	35-114
Phenol-d5	1	88	NC	0	0	10-94
Terphenyl-d14	1	157	NC	0	1	33-141
2,4,6-Tribromophenol	1	66	NC	0	0	10-123
Lab Blanks - Water Samp	les:					
2-Fluorobiphenyl	1	67	NC	0	0	43-116
2-Fluorophenol	1	66	NC	0	0	21-100
Nitrobenzene-d5	1	70	NC	0	0	35-114
Phenol-d5	1	73	NC	0	0	10-94
Terphenyl-d14	1	90	NC	0	0	33-141
2,4,6-Tribromophenol	1	98	NC	0	0	10-123

Table F.1-6

SUMMARY OF DUPLICATE SAMPLE RESULTS

	No. of			
<u>Site 16</u>	<u>Pairs</u>	<u>Mean</u>	<u>RPD %</u>	
Chloride, Phosphate, and Total Phosphorus in Coal (mg/kg):				
Chloride	1	414.7	13.1	
Phosphate	1	<30	0	
Total Phosphorus	1	42.7	28.9	
Chloride, Fluoride, and Total Sulfur in Bottom Ash (mg/kg):				
Chloride	1	<134	>50	
Fluoride	1	14.8	23.6	
Total Sulfur	1	<0.005	NC	
Chloride, Fluoride, and Total Sulfur in Fly Ash (mg/kg):				
Chloride	1	<100	NC	
Fluoride	1	84.4	11.2	
Total Sulfur	1	0.211	12.8	
ICAPES Metals in ESP Inlet Gas Samples:				
Aluminum	1	69450	1.6	
Antimony	1	<50	0	
Barium	1	555	1.8	
Beryllium	1	10.25	2.9	
Calcium	1	19550	1.5	
Chromium	1	130	0.8	
Cobalt	1	31.7	3.8	
Copper	1	111.5	1.8	
Iron	1	39450	1.8	
Magnesium	1	8255	1.6	
Manganese	1	65.55	0.5	
Molybdenum	1	50.85	0.2	
Nicke]	1	60.65	1.5	
Potassium	1	13850	2.2	

	No. of	•	
<u>Site 16</u>	<u>Pairs</u>	Mean	<u>RPD %</u>
ICAPES Metals in ESP Inlet			
Gas Samples: (Cont'd)	_	_	-
Silver	1	<5	0
Sodium	1	48800	1.6
Strontium	1	487	2.0
Thallium	1	9.665	NC
Titanium	1	3605	1.9
Vanadium	1	151	2.6
Zinc	1	95.2	0.1
Arsenic and Lead by AAS in ESP			
Inlet Gas Samples:			
Arsenic	1	164	0
Cadmium	1	0.625	8.0
Lead	1	51.3	5.1
Mercury	1	0.55	18.2
Selenium	1	49.3	. 2.8
ICAPES Metals in Coal (mg/kg):			
Aluminum	1	9583	2.8
Antimony	1	<10	0
Barium	1	99.6	3.7
Beryllium	1	1.676	2.5
Calcium	1	814	1.3
Chromium	1	21.48	6.8
Cobalt	1	6.438	6.3
Copper	1	47.48	23.9
Iron	1	8252	0.2
Magnesium	1	355.6	2.9
Manganese	1	11.75	0.9
Molybdenum	1	2.305	34.9
Nickel	1	22.95	9.6
Potassium	1	1220	5.4
Silver	1	<1	0
Sodium	1	251.4	1.6

Table F.1-6

	No. of		
<u>Site 16</u>	Pairs	<u>Mean</u>	<u>RPD %</u>
ICAPES Metals in Coal (mg/k	g): (Cont'd)		
Strontium	1	1.552	NC
Titanium	1	600	2.8
Vanadium	1	24.52	0
Zinc	1	614.38	6.3
Metals by AAS and CVASS in			
Coal (mg/kg):			
Arsenic	1	31	1.9
Cadmium	1	<0.1	0
Mercury	1		
Lead	1	4.6	2.2
Selenium	1	1.4	6.9
ICAPES Metals in Bottom Ash	(mg/kg):		
Aluminum	1	128500	0.8
Antimony	1	<100	NC
Barium	1	882	2.5
Beryllium	1	11.2	0.9
Calcium	1	7295	1.0
Chromium	1	120	1.7
Cobalt	1	51.3	5.4
Copper	1	94.2	2.1
Iron	1	90550	0.1
Magnesium	1	4490	0.4
Manganese	1	141	1.4
Molybdenum	1	<200	NC
Nickel	1	103.5	1.0
Potassium	1	18315	1.0
Silicon	1	348500	0.3
Silver	1	<40	NC
Sodium	1	2475	8.5
Strontium	1	978	4.5
Thallium	1	<100	NC
Titanium	1	7215	0.1

.

Table F.1-6

	No. of		
<u>Site 16</u>	Pairs	Mean	<u>RPD %</u>
ICAPES Metals in Bottom Ash (mg/kg): (Cont'd)			
Vanadium	1	212	1.0
Zinc	1	31.25	8.6
Arsenic and Lead by AAS in Bottom Ash (mg/kg):			
Arsenic	1	84.15	6.8
Cadmium	1	<1	NC
Lead	1	21.4	0.5
Mercury	1	<0.045	NC
Selenium	1	<5	NC
ICAPES Metals in Fly Ash (mg/kg):			
Aluminum	1	104500	6.7
Antimony	1	<100	Nc
Barium	1	801	6.5
Beryllium	1	12.6	0
Calcium	1	7585	1.7
Chromium	1	491	157
Cobalt	1	58.95	5.6
Copper	1	153.5	9.8
Iron	1	78600	2.5
Magnesium	1	3305	22.1
Manganese	1	180	55.5
Molybdenum	1	<200	NC
Nickel	1	322	129
Potassium	1	15850	4.4
Silicon	1	310100	1.2
Silver	1	<40	NC
Sodium	1	2336	1.4
Strontium	1	886	6.4
Titanium	1	6595	1.4
Vanadium	1	242	0.4
Zinc	1	88.3	5.0

	No. of		
<u>Site 16</u>	<u>Pairs</u>	Mean	<u>RPD %</u>
Metals by AAS and CVASS in Fly Ash (mg/kg):			
Arsenic	1	224	0.4
Cadmium	1	<1	27.0
Mercury	1	0.263	6.8
Lead	1	59.4	64.8
Selenium	1	18.9	64.6
ICAPES Metals in Bottom Ash Sluice Water (mg/L):			
Aluminum	1	0.266	13.2
Antimony	1	<0.1	NC
Barium	1	0.094	1.1
Beryllium	1	0.0007	0
Boron	1	0.1125	6.2
Calcium	1	16.0	0.6
Chromium	1	0.002	NC
Cobalt	1	0.003	NC
Copper	1	0.006	0
Iron	1	0.335	25.1
Magnesium	1	2.84	1.7
Manganese	1	0.05	2.0
Molybdenum	1	<0.05	NC
Nicke]	1	0.008	NC
Potassium	1	2.24	2.2
Silicon	1	3.61	2.2
Silver	1	0.002	NC
Sodium	1	5.535	0.54
Strontium	1	0.169	1.1
Titanium	1	<0.1	NC
Vanadium	1	0.003	0
Zinc	1	0.014	NC

<u>Site 16</u>	No. of <u>Pairs</u>	<u>Mean</u>	<u>RPD %</u>
ICAPES Metals in Sluice Water Supply (mg/L):			
Aluminum	1	0.688	5.2
Antimony	1	<0.1	NC
Barium	1	0.150	0.7
Beryllium	1	0.0011	NC
Boron	1	0.160	NC
Calcium	1	21.25	1.4
Chromium	1	0.004	0
Cobalt	1	0.008	NC
Copper	1	0.034	NC
Iron	1	0.828	5.4
Magnesium	1	3.58	1.4
Manganese	1	0.092	1.1
Molybdenum	1	<0.05	0
Nickel	1	0.013	NC
Potassium	1	3.46	4.3
Silicon	1	3.96	4.3
Silver	1	0.006	NC
Sodium	1	9.04	0.1
Strontium	1	0.267	0.8
Titanium	1	<0.1	0
Vanadium	1	0.007	NC
Zinc	1	0.02	NC

RESULTS OF SITE 16 PERFORMANCE EVALUATION SAMPLE -FLY ASH 1633a ANALYZED BY ICP/AAS AND NEUTRON ACTIVATION

		ICP/A	AS	NAA		XR	F
<u>Parameter</u>	Certified Value <u>ug/G</u>	Result <u>uq/q</u>	Rec	Result <u>uq/q</u>	Rec	Result%	Rec
Aluminum	14.3%	12.0%	84	1.33%	90	14.3	102
Antimony	6.8	<100	-	6.199	90		
Barium	0.15%	0.115%	77	0.1839%	122	0.161	107
Beryllium	12	11.0	92	NA	-		
Calcium	1.11%	0.913%	82	1.49%	134	1.06	95
Chromium	196	450	229	182	90		
Cobalt	46	54.8	119	53.1	115		
Copper	118	115	97	<418	-		
Iron	9.4%	8.28%	88	9.58%	102	9.81	104
Magnesium	0.455%	0.336%	76	0.6572%	144	0.452	100
Manganese	1 79	186	104	160	89	0.079	440
Nickel	127	256	202	162	140		
Potassium	1.88%	1.66%	88	2.53%	134	1.88	100
Silicon	22.85%	34.8%	153	NA	-	23.2	101
Sodium	0.17%	0.156%	92	0.184%	108	0.11	65
Strontium	830	664	80	1021	123	0.27	320
Titanium	0.8%	0.772%	96	0.7259%	90	0.82	103
Vanadium	297	277	93	288	90		
Zinc	220	192	87	202	92		
Arsenic	145	214	148	157	108		
Cadmium	1.00	<1	-	<24.9	-		
Mercury	0.16	0.104	65	<3.6	-		
Lead	72.4	88.2	122	NA	-		
Selenium	10.3	<5	-	10.9	106		

SITE 16 PERFORMANCE EVALUATION SAMPLE -FLY ASH 1633a SAMPLE ASPIRATED INTO IMPINGER TRAIN

		<u>ICP/AA</u>	<u>S</u>	<u>ICP/AA</u>	<u>s</u>
		<u>Blank Corr</u>	<u>ected</u>	<u>Not Blank C</u>	<u>orrected</u>
	Certified	Result	Rec.	Result	Rec.
<u>Parameter</u>	<u>Value ug/G</u>	<u>ug/g</u>	%	<u>uq/q</u>	%
Aluminum	14.3%	9.70%	68	9.93%	69
Antimony	6.8				
Barium	0.15%	940	63	1021	68
Beryllium	12	9.61	80	9.6	80
Calcium	1.11%	8552	77	9723	88
Chromium	196	169	86	195	100
Cobalt	46	38.7	84	40.0	87
Copper	118	107.5	91	120.1	102
Iron	9.4%	6.2%	66	6.26%	67
Magnesium	0.455%	0.310%	68	0.3303%	73
Manganese	179	136	76	150	84
Molybdenum	29	27	93	324	1117
Nickel	127	103.6	82	131	103
Potassium	1.88%	1.36%	72	1.37%	73
Silicon	22.85				
Silver					
Sodium	0.17%	0.46	271	0.64%	376
Strontium	830	683	82	713	86
Titanium	0.8%	0.548%	69	0.548%	69
Vanadium	297	234	79	244	82
Zinc	220	228	104	277	126
Arsenic	145	209	144	217	150
Cadmium	1.00	1.98	1 9 8	2.97	297

·

ì

RESULTS OF SITE 16 PERFORMANCE EVALUATION SAMPLE - COAL 1632A NEUTRON ACTIVATION ANALYSIS

		PE Sa	ample			QC
	Certified			Certified		_
<u>Parameter</u>	Value	<u>Result</u>	<u>Rec. %</u>	Value	<u>Result</u>	<u>Rec. %</u>
Aluminum	8550	8572	100	29500	29121	99
Antimony	0.24	0.277	115	0.600	0.612	102
Barium	67.5	97.8	145	. 120	117	98
Beryllium						
Calcium	2040	1962	96	2410	2355	98
Chromium	11	12.8	117	34.3	33.5	98
Cobalt	2.29	2.57	112	6.7	6.55	9 8
Copper	6.28	55.1	878			
Iron	7590	8423	111	11100	10847	98
Magnesium	383	381	99	1150	1096	95
Manganese	12.4	12.35	100	28.0	26.7	95
Molybdenum	0.9	<1.15		3.85	4.02	104
Nickel	6.1	<15.8		19.4	<25	NC
Potassium	748	1017	136	4110	3918	95
Silicon						
Silver						
Sodium	515	526	102	828	844	102
Strontium	102	95.1	93.2	85	83.1	98
Titanium	454	519	114	1630	1554	95
Vanadium	14	15.5	111	44.0	41.9	95
Zinc	11.89	9.25	78	28.0	27.4	98
Arsenic	3.72	3.9	105	9.3	9.48	102
Cadmium	0.057	<3.2		0.17	<0.25	NC
Mercury				0.13	<0.25	
Lead						
Selenium	1.29	1.64	127	2.6	2.54	98

•

Table F.1-10

SITE 16 PERFORMANCE EVALUATION SAMPLE -EPA TRACE ELEMENTS IN WATER

<u>Parameter</u>	<u>Certified Value mg/L</u>	<u>Result_mg/L</u>	<u>Rec. %</u>
Aluminum	0.500	0.511	102
Beryllium	0.100	0.0973	97
Chromium	0.100	0.098	98
Cobalt	0.100	0.097	97
Copper	0.100	0.093	93
Iron	0.100	0.113	113
Manganese	0.100	0.094	94
Nickel	0.100	0.100	100
Vanadium	0.250	0.233	93
Zinc	0.100	0.098	98
Arsenic	0.100	0.092	92
Cadmium	0.025	0.028	112
Mercury	0.005	0.0055	110
Lead	0.102	0.102	102
Selenium	0.025	0.022	88

Sample ID	Analyte	Method	Result Unit	C S	et Lim	Sample
H-315	Acetaldehyde	HPLC	3.9 ug/	/mL 0.1 ug/s	ample	Stack gas, impingers
H-315	Formaldehyde	HPLC	17.0 ug/	/mL 0.1 ug/s	ample	Stack gas, impingers
H-111/H-109/H-125	5-methyl chrysene	HRGC/HRMS	7.6 n	Ď	7.6	ESP inlet gas
H-111/H-109/H-125	7H - dibenzo[c,g]carbazole	HRGC/HRMS	41.7 n	D	41.7	ESP inlet gas
H-111/H-109/H-125	Acenaphthene	HRGC/HRMS	469.0 n	Ð	I I	ESP inlet gas
H-111/H-109/H-125	Acenaphthylene	HRGC/HRMS	32.2 n	Ď	1	ESP inlet gas
H-111/H-109/H-125	Anthracene	HRGC/HRMS	19.7 n	D	19.7	ESP inlet gas
H-111/H-109/H-125	Benzofalpyrene	HRGC/HRMS	22.4 n	<u>ה</u>	22.4	ESP inlet gas
H-111/H-109/H-125	Benzolb.j&k)fluoranthenes	HRGC/HRMS	13.8 n	- - -	ī	ESP inlet gas
H-111/H-109/H-125	Benzolahilpervien e	HRGC/HRMS	25.1 n	5		ESP inlet gas
H-111/H-109/H-125	Benzlalanthracene	HRGC/HRMS	13.2 n	5	13.2	ESP intet gas
H-111/H~109/H-125	Chrysene	HRGC/HRMS	11.7 n	5		ESP inlet gas
H-111/H-109/H-125	Dibenzofa.elpvrene	HRGC/HRMS	11.6 n	Ď	11.6	ESP inlet gas
H-111/H-109/H-125	Dibenzola, hipvrene	HRGC/HRMS	30.2 n	0	30.2	ESP inlet gas
H-111/H-109/H-125	Dibenzola, il pyrene	HRGC/HRMS	8.5	ß	8.5	ESP inlet gas
H = 111/H ~ 109/H = 125	Dibenzia hlacridine	HRGC/HRMS	8.4 n	0	8.4	ESP inlet gas
H-111/H-109/H-125	Dibenzia, hianthracene	HRGC/HRMS	8.2 n	5	8.2	ESP inlet gas
H-111/H-109/H-125	Dibenzia ilacridine	HRGC/HRMS	15.3 n	D	15.3	ESP inlet gas
H-111/H-109/H-125	Fluoranthene	HRGC/HRMS	27.4 n	5		ESP inlet gas
H-111/H-109/H-125	Fluorene	HRGC/HRMS	35.4 n	5		ESP inlet gas
H = 111/H = 109/H = 125	Indeno[1,2,3-cd]pvrene	HRGC/HRMS	7.4 n	- - -		ESP inlet gas
H-111/H-109/H-125	Phenanthrene	HRGC/HRMS	119.9 n	5		ESP inlet gas
H-111/H-109/H-125	Pyrene	HRGC/HRMS	46.1 n	- - -		ESP intet gas
1944	Alterioten	ICP analysis by SW6010	2060.0 u	Ð	8.3	ESP inlet gas, th/PCR
1101	Antimonu	ICP analysis hy SW6010	ND		1.5	ESP inlet das, th/PCR
101	Arsenic	ICP analysis by SW6010	QN	, D	3.4	ESP inlet gas, th/PCR
1311	Barium	ICP analysis by SW6010	15.0 u	0	0.1	ESP inlet gas, th/PCR
H311/H303 COMP	Bervilium	BIF ICP for Metals Trains	0.4 u	0	0.2	ESP inlet gas, th/PCR
H311	Cadmium	BIF ICP for Metals Trains	0.1 u		0.3	ESP intet gas, th/PCR
H311	Calcium	ICP analysis by SW6010	159.0 u	0	26.0	ESP inlet gas, th/PCR
H311	Chromium	ICP analysis by SW6010	6.0 U	0	0.5	ESP inlet gas, th/PCR
H311/H303 COMP	Cobait	BIF ICP for Metals Trains	2.7 u	0	<u>8</u> .1	ESP inlet gas, th/PCR
Hatt/Hana COMP	Copper	BIF ICP for Metals Trains	4.1 V	0	1.0	ESP inlet gas, th/PCR
H311	Iron	ICP analysis by SW6010	1130.0 u	5	34.0	ESP inlet gas, th/PCR
H311	Lead	BIF ICP for Metals Trains	n QN	5	2.5	ESP inlet gas, th/PCR
H311/H303 COMP	Magnesium	BIF ICP for Metals Trains	92.6 U	5	10.8	ESP inlet gas, th/PCR
H311	Manganese	ICP analysis by SW6010	6.6 U	0	0.1	ESP intet gas, th/PCR
Hat I/H303 COMP	Mercury	Mercury, cold vapor SW7471	0.0	5	0.0	ESP infet gas, th/PCR
H311/H303 COMP	Molybdenum	BIF ICP for Metals Trains	0.0	6	1.0	ESP inlet gas, th/PCR
H311	Nickel	BIF ICP for Metals Trains	4.4	0	1.1	ESP inlet gas, th/PCR

Table F2-1. Summary of Field Blank Results

Appendix F: QA/QC Results

Samila ID	Analyte	Method	Result Units	Det Lim	Sample
Dampic to Hatt	Phosphorus	BIF ICP for Metals Trains	50.4 ug	7.3	ESP inlet gas, th/PCR
Hatthana COMP	Potassitum	BIF ICP for Metals Trains	240.0 ug	140.0	ESP inlet gas, th/PCR
	Selenium	BIF ICP for Metals Trains	0n QN	17.6	ESP inlet gas, th/PCR
	Selenium	BIF Se for Metals Trains	0.3 ug	0.1	ESP inlet gas, th/PCR
	Silicon	BIF ICP for Metals Trains	4360.0 ug	44.0	ESP intet gas, th/PCR
	Silver	BIF ICP for Metals Trains	DN ND	0.2	ESP inlet gas, th/PCR
H311 H311/H303 COMP	Sodium	BIF ICP for Metals Trains	53.7 ug	10.4	ESP inlet gas, th/PCR
	Strontium	ICP analysis by SW6010	14.4 ug	0.1	ESP inlet gas, th/PCR
	Thallinm	ICP analysis by SW6010	2.3 ug	2.0	ESP inlet gas, th/PCR
Hatt	Titanium	ICP analysis by SW6010	91.2 ug	0.5	ESP inlet gas, th/PCR
	Vanadium	BIF ICP for Metals Trains	3.4 ug	1.7	ESP inlet gas, th/PCR
	Zinc	ICP analysis by SW6010	3.6 ug	0.7	ESP inlet gas, th/PCR
	Chloride	Chloride by IC EPA300	278.0 ug/sam	I 3.3	Stack gas, filt+PNR
		Fliorida by FPA 340.2	15,1 ug/sam	3.3	Stack gas, filt+PNR
		Suitate on filters	438.0 ud/sam	9.9	Stack gas, filt + PNR
		BIE As for Metals Trains	41 ug	0.1	ESP inlet das, th/PCR
	Arsenic	DIE Colfor Matale Trains	0.2 ug	0.1	ESP inlet gas, th/PCR
		DIE Dh for Motale Trains	50	0.1	ESP inlet das. th/PCR
H311/H303 COMP	Lead				ESD inlet mae th/PCR
H311/H303 COMP	Nicket	BIF NI for Metals Trains	4.3 u g	0.1	Eor line gas, un con
		SMUS	, 10 ng	10	Stack Gas. 20L VOST
H-243				÷	Stack Gas 20L VOST
H243	1,1 Dichloroethene	GCMS		2 \$	Stack Gas 201 VOST
H-243	1,1,1 – Trichloroethane	GCMS	Su ul>	2 9	DIACH CAS, EUL VOOL
H-243	1,1,2 - Trichloroethane	GCMS	<10 ng	01	
H-243	1.1.2.2 – Tetrachloroethane	GCMS	<10 ng	10	Stack Gas, 20L VUSI
11 - 243	1 2 - Dichlorohenzene	GCMS	<10 ng	1	Stack Gas, 20L VOST
	t o Dichlorochane	GCMS	<10 ng	₽ ₽	Stack Gas, 20L VOST
H-243		COMS	<10 ng	10	Stack Gas, 20L VOST
H-243	1,2 - Dicritoroproparte		< 10 ng	10	Stack Gas, 20L VOST
H-243		SUMS SUMS	<10 ng	40	Stack Gas, 20L VOST
H - 243			<50 ng	50	Stack Gas, 20L VOST
H - 243		STOR STOR	<50 ng	50	Stack Gas, 20L VOST
H-243	2-Hexanone			50	Stack Gas. 20L VOST
H-243	4-Methyl-2-Pentanone				Stack Gas. 201 VOST
H-243	Acetone			ŝ÷	Charle Can POL VOST
H-243	Benzene	GCMS	siu ng	2 9	
H243	Bromodichloromethane	GCMS	<10 ng	01	Stack Gas, ZUL VUS I
H_243	Bromoform	GCMS	<10 ng	10	Stack Gas, 20L VOST
H - 243	Bromomethane	GCMS	<10 ng	10	Stack Gas, 20L VOST
	Carbon Disulfide	GCMS	<10 ng	10	Stack Gas, 20L VOST
640 H	Carbon Tatrachlorida	GCMS	<10 ng	10	Stack Gas, 20L VOST
H - 243		GOMS	<10 ng	10	Stack Gas, 20L VOST
H-243)		

.

Table F2-1. Summary of Field Blank Results

ESP inlet gas, imps 1&2 ESP inlet gas, imps 1&2 inlet gas, HCI rinse imps 1&2 gas, imps 1&2 gas, imps 1&2 gas, imps 1&2 ESP inlet gas, imps 1&2 ESP inlet gas, imps 1&2 imps 1&2 gas, imps 1&2 gas, imps 1&2 imps 1&2 imps 1&2 ESP inlet gas, imps 1&2 imps 1&2 imps 1&2 ESP inlet gas, imps 1&2 Stack Gas, 20L VOST Stack Gas, 20L VOST Stack Gas, 20L VOST Stack Gas, 20L VOST 20L VOST 20L VOST Stack Gas, 20L VOST Stack Gas, 20L VOST Stack Gas, 20L VOST 20L VOST 20L VOST 20L VOST Sample gas, Ì gas, gas, gas, ESP inlet gas, ESP inlet gas, Stack Gas. Stack Gas. Stack Gas, Stack Gas, Stack Gas, ESP inlet ESP inlet **ESP** inlet ESP i 0.02800 0.02400 0.02300 0.00065 0.00053 0.15000 0.00380 0.02400 22222 2 22222 0.00055 0.01500 0.00170 0.00250 0.00340 0.02700 0.00110 0.02300 0.00039 0.00024 <u>o</u> <u>o</u> <u>8</u>0 0.00024 0.00032 0.00072 2 9 Det Lim 0.00460 mg/L mg/f mg/L **Result Units** <10 <10 10 10 2 9 9 v ¢ ₹0 ₽¥ V 2700.0 v 10 9 v €~ 11.0 € 20 ¢ 10 4 10 55.0 < 50 <10 <10 g 9 g 0.00156 g g 2 2 v 2 22 0.07890 0.00021 0.02840 0.00078 0.03420 0.00640 0.00403 0.00000 0.01250 0.00916 0.00252 0.00670 Mercury, HNO3/H2O2 Impinger Mercury, HNO3/H2O2 Impinger Chromium by GF - SW7191 ICP analysis by SW6010 CP analysis by SW6010 CP analysis by SW6010 ICP analysis by SW6010 ICP analysis by SW6010 CP analysis by SW6010 CP analysis by SW6010 ICP analysis by SW6010 CP analysis by SW6010 Cadmium by SW7131 Arsenic by SW7060 Lead by SW7421 Method GCMS rans-1,3-Dichloropropene rans-1,2-Dichloroethene cis-1,3-Dichloropropene Dibromochloromethane **Irichlorofluoromethane Methylene Chloride Tetrachioroethene Frichloroethene** Chloromethane Ethy! Benzene Chloroethane **/inyl** Chloride /inyl Acetate Molybdenum m,p-Xylene Magnesium Manganese Chloroform o-Xylene Chromium Chromium Aluminum Cadmium Cadmium Beryllium Antimony Calcium **[oluene** Mercury Barium Mercury Copper Styrene Arsenic Arsenic Analyte Boron Cobait Lead ead lon Sample ID H-243 H-243 H -- 243 H -- 243 H -- 243 H--243 H--243 H--243 H-243 H-243 H-308 H-308 H-308 H-308 H-243 H-243 H-243 H-243 H-243 H-308 H-308 H-308 H-308 H-308 H-308 H-308 H-243 H-243 H - 243H-308 H - 308H--308 H-308 H - 308H - 308H-310 H-308 H-308 H-308 H-308

Table F2-1. Summary of Field Blank Results

1

. 1

1

Appendix F: QA/QC Results

Results
3lank F
Field E
narv of
Sumn
F2 - 1,
Table

Samula ID	Analyte	Method	Result Units	De	t Lim Sample
	Nickel	ICP analysis by SW6010	Sm UN	J/L 0.0	0990 ESP inlet gas, imps 1&2
H-308	Nickel	Nickel by GF, EPA 249.2	0.02690 mg	/۲ 0.0	0180 ESP inlet gas, imps 1&2
H_308	Phosphorus	ICP analysis by SW6010	0.10000 mg	J/L 0.B	4000 ESP Inlet gas, imps 1&2
H-308	Potassium	ICP analysis by SW6010	0.08330 mg	J/L 0.3	7000 ESP Inlet gas, imps 1&2
H_308	Selenium	Selenium by SW7740	Gm GN	¢۲ 0.0	0140 ESP Inlet gas, imps 1&2
H-308	Selenium	ICP analysis by SW6010	Sm QN	y'L 0.0	4200 ESP intet gas, imps 182
H - 308	Silican	ICP analysis by SW6010	1.60000 mg)/L 0.2	7000 ESP inlet gas, imps 1&2
H-308	Silver	ICP analysis by SW6010	Sữ QN	¢/ل 0.0	0490 ESP inlet gas, imps 1&2
H_308	Sodium	ICP analysis by SW6010	0.06200 mg	0.0 الر	4000 ESP inlet gas, imps 1&2
H-30B	Strontium	ICP analysis by SW6010	0.00035 mg	¢/۲ 0.0	0017 ESP inlet gas, imps 1&2
H_308	Thallium	ICP analysis by SW6010	0.00668 mg	0.0 T/¢	1700 ESP inlet gas, imps 1&2
H - 308	Titanium	ICP analysis by SW6010	QN	J/L 0.0	0100 ESP inlet gas, imps 1&2
H-308	Vanadium	ICP analysis by SW6010	Sữ QN	3/F 0.0	0240 ESP inlet gas, imps 1&2
H-308	Zinc	ICP analysis by SW6010	0.00458 mg	¢/L 0.0	0150 ESP intet gas, imps 16/2
H-309	Mercury	Mercury, cold vapor SW7470	GN QN)/r 0.0	0005 ESP inlet gas, imps 3,4,5
		ICP anatveis by SW6010	0.02210 mc	0.0 0.0	2800 ESP inlet gas, TL rinse
		ICP analysis by SW6010	UN UN	0.0	2400 ESP inlet gas. TL rinse
H-304	Ammony	Areanic hv SW7060		0'0 7/F	0065 ESP inlet gas, TL rinse
H-304	Aronio	ICP analysis by SW6010	0.02000 mg	1/F 0.0	2300 ESP inlet gas, TL rinse
	Rarium	ICP analysis by SW6010	0.00078 mg	0.0 1/L 0.0	0053 ESP inlet gas, TL rinse
	Bervilium	ICP analysis by SW6010	0.00038 mg	j/L 0.0	0055 ESP inlet gas, TL rinse
	Boron	ICP analysis by SW6010	0.04980 mg	¢ر 0.0	1500 ESP inlet gas, TL rinse
	Cadmium	Cadmium by SW7131	0.00331 mg	¢ر 0.0	0024 ESP inlet gas, TL rinse
	Cadmium	ICP analysis by SW6010	0.00061 mg	3/L 0.0	0170 ESP inlet gas, TL rinse
	Calcium	ICP analysis by SW6010	0.08740 mç	g/L 0.1	5000 ESP inlet gas, TL rinse
	Chromium	ICP analysis by SW6010	0.00435 mg	¢/۲ 0.0	0250 ESP inlet gas, TL rinse
H-304	Chromium	Chromium by GF - SW7191	0.00640 mg	¢/ل 0.0	0072 ESP inlet gas, TL rinse
H-304	Cobalt	ICP analysis by SW6010	Sử QN	0 ^{.0}	0340 ESP inlet gas, TL rinse
H - 304	Copper	ICP analysis by SW6010	0.00514 mg	J/F 0.0	0380 ESP inlet gas, IL rinse
H-304	Iron	ICP analysis by SW6010	0.08360 mg)را 10.00	0600 ESP intet gas, iL rinse
H-304	Lead	ICP analysis by SW6010	Sm ON	¢۲ 0.0	2700 ESP Iniet gas, iL finse
H-304	Lead	Lead by SW7421	0.04860 mg	¢/۲ 0.0	0110 ESP inlet gas, IL rinse
H-304	Magnesium	ICP analysis by SW6010	0.02720 mg	J/L 0.00	2300 ESP inlet gas, IL rinse
H-304	Mandanese	ICP analysis by SW6010	0.00174 mg	¢/۲ 0.0	0039 ESP inlet gas, TL rinse
H-304	Mercury	Mercury, cold vapor SW7470	0.00000 mg	J/L 0.0	2005 ESP inlet gas, IL rinse
H-304	Molvbdenum	ICP analysis by SW6010	0.00343 mg	J/L 0.0	0460 ESP inlet gas, TL rinse
H-304	Nickel	Nickel by GF, EPA 249.2	0.00190 mg	<i>ا</i> لا 0.0	0182 ESP inlet gas, TL rinse
H-304	Nickel	ICP analysis by SW6010	Sm ON)/F 0.0	0990 ESP inlet gas, iL rinse
H-304	Phosphorus	ICP analysis by SW6010	0.26200 mg	رالا مراجع 1.20 مراجع	1000 ESP inlet gas, iL rinse
H-304	Potassium	ICP analysis by SW6010	ND mg	/۲ ۵.3	7000 ESP inter gas, 1L mise
Table F2-1. Summary of Field Blank Results

ts Det Lim Sample	ig/L 0.00144 ESP intet gas, TL rinse	g/L 0.04200 ESP inlet gas, TL rinse	ig/L 0.02700 ESP inlet gas, TL rinse	ig/L 0.00490 ESP inlet gas, TL rinse	ig/L 0.04000 ESP inlet gas, TL rinse	g/L 0.00017 ESP inlet gas, TL rinse	ig/L 0.01700 ESP inlet gas, TL rinse	ig/L 0.00100 ESP inlet gas, TL rinse	g/L 0.00240 ESP inlet gas, TL rinse	g/L 0.00150 ESP inlet gas, TL rinse	g/L 0.02 Stack gas, anion imps	g/L 0.02 Stack gas, anion imps	g/L 0.06 Stack gas, anion imps	frap 2.1 Stack Gas	ftrap 7.6 Stack Gas	frap 0.9 Stack Gas	trap 8.0 Stack Gas
Result Un	m 06900.0	0.03500 m	0.33100 m	0.00423 m	0.41800 m	0.00132 m	E ON	0.00052 m	E QN	0.04 m	1.28 m	0.09 m	1.73 m	1.9 ng	12.4 ng	1.9 ng	16.2 ng
Method	Selenium by SW7740	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	ICP analysis by SW6010	Chloride, by IC EPA300	Fluoride by EPA 340.2	Sulfate, EPA 300.0	Bloom	Bloom	Bloom	Bloom
Analyte	Selenium	Selenium	Silicon	Silver	Sodium	Strontium	Thallium	Titanium	Vanadium	Zinc	Chloride	Fluoride	Sulfate	Mercury (0)	Mercury (II)	Methyl Mercury	Total Mercury
Sample ID	H304	H-304	H-304	H-304	H-304	H-304	H-304	H304	H304	H304	H-122	H-122	H-122	H383	H383	H383	H383

Table F2-2. Summary of Laboratory Method Blank Results

Sample ID	Analyte	Method	Result	Units	Det Lim	Sample
BLK93-682	Aluminum	ICP analysis by SW6010		mg/kg	73	Laboratory QC
BLK93682	Antimony	ICP analysis by SW6010		mg/kg	19	Laboratory QC
BLK93-682	Arsenic	ICP analysis by SW6010		mg/kg	160	Laboratory QC
BLK93+682	Rerium	ICP analysis by SW6010	0.30.1	mg/kg mg/kg	0.933	Laboratory QC
BLK93-682	Berium	ICP analysis by SW6010	0.03 0	ma/ka	0.8	Laboratory QC
BLK93~682	Beryllium	ICP analysis by SW6010	0.12 J	ma/ka	0.6	Laboratory QC
BLK93638	Beryllium	ICP analysis by SW6010	0.00 J	mg/kg	0.06	Laboratory QC
BLK93638	Beryllium	ICP analysis by SW6010	0.01 J	mg/kg	0.06	Laboratory QC
BLK93-682	Beryllium	ICP analysis by SW6010		mg/kg	0.6	Laboratory QC
BLK93682	Cadmium	ICP analysis by SW6010	3.39	mg/kg	2.9	Laboratory QC
BLK93638	Cadmium	ICP analysis by SW6010	0.00 J	mg/kg	0.29	Laboratory QC
DLN93030	Cadmium	Cadmium by SW2131	0.02 J	mg/xg	0.29	Laboratory QC
BL K93-682	Calcium	ICP analysis by SW6010		mg/kg mg/kg	250	Laboratory QC
BLK93-682	Calcium	ICP analysis by SW6010	18.3.3	ma/ka	250	Laboratory QC
BLK93629	Chloride	Chloride, potentiometric		ma/ka	100	Laboratory QC
BLK93525	Chloride	Chloride by SM 4500 Cl B	ND	mg/kg	100	Laboratory QC
BLK93-682	Chromium	ICP analysis by SW6010		mg/kg	5.5	Laboratory QC
BLK93-682	Chromium	ICP analysis by SW6010	3.12 J	mg/kg	5.5	Laboratory QC
BLK93638	Cobalt	ICP analysis by SW6010	0.39 J	mg/kg	0.52	Laboratory QC
BLK93638	Cobait	ICP analysis by SW6010	0.86	mg/kg	0.52	Laboratory QC
BLK93682	Cobart	ICP analysis by SW6010	6.33	mg/kg	5.2	Laboratory QC
BLN93636	Copper	ICP analysis by SW6010	0.88	mg/kg	0.25	Laboratory QC
BLK93682	Copper	ICP analysis by SW6010	2.52	mg/kg mg/kg	0.25	Laboratory QC
BI K93840	Fluoride	Fluoride by EPA 340 2	183	mg/kg	2.5	Laboratory QC
BLK93539	Fluoride	Fluoride by EPA 340.2	0.02	ma/ka	0.02	Laboratory QC
BLK93840	Fluoride	Fluoride by EPA 340.2	19.4	ma/ka	10	Laboratory QC
BL K93 - 682	Iron	ICP analysis by SW6010	93 J	mg/kg	330	Laboratory QC
BLK93-682	Iron	ICP analysis by SW6010		mg/kg	330	Laboratory QC
BLK93682	Lead	Lead by SW7421	0.1 J	mg/kg	1.1	Laboratory QC
BLK93-682	Lead	ICP enalysis by SW6010		mg/kg	3.4	Laboratory QC
BLK93638	Lead	ICP analysis by SW6010		mg/kg	2.5	Laboratory QC
BLK93-662	Lead	ICP analysis by SW6010	10.2	mg/kg	3.4	Laboratory QC
BLK93-802	Magnesium	ICP analysis by SW6010	2 47 1	mg/kg mg/ka	20	Laboratory QC
BLK93-682	Magnesium	ICP analysis by SW6010	2.4 J	ma/ka	26	Laboratory QC
BLK93638	Magnesium	ICP analysis by SW6010	0.59 J	ma/ka	2.7	Laboratory QC
BLK93-682	Manganese	ICP analysis by SW6010	0.14 J	mg/kg	1.4	Laboratory QC
BLK93-682	Manganese	ICP analysis by SW6010	1.88	mg/kg	1.4	Laboratory QC
BLK93649	Mercury	Mercury, cold vapor SW7471		mg/kg	0.012	Laboratory QC
BLK93682	Molybdenum	ICP analysis by SW6010		mg/kg	2.6	Laboratory QC
BLK93636	Molyodenum	ICP analysis by SW6010		mg/kg	0.26	Laboratory QC
DLN99002	Nickel	ICP analysis by SW6010	10 J	រាg/kg ភាព/kg	11	Laboratory QC
RI K93638	Nickel	ICP enalysis by SW6010	0.04 J	mg/kg ma/ka	1.1	Laboratory QC
BLK93682	Nickel	Nickel by GF. EPA 249.2		ma/ka	1 17	Laboratory QC
BLK93-682	Phosphorus	ICP analysis by SW6010		ma/ka	140	Laboratory QC
BLK93682	Potassium	ICP analysis by SW6010	308 J	mg/kg	350	Laboratory QC
BLK93638	Potassium	ICP analysis by SW6010	8.68 J	mg/kg	35	Laboratory QC
BLK93638	Potassium	ICP analysis by SW6010		mg/kg	35	Laboratory QC
BLK93638	Selenium	ICP analysis by SW6010		mg/kg	4.4	Laboratory QC
BLK93682	Selenium Selenium	CP analysis by SW6010 Selection by SM6740	3.81 J	mg/kg	44	Laboratory QC
BLK93632	Silicon	ICP analysis by SW6010	1210	mg/kg Omg/kg	1.10	Laboratory QC
BLK93682	Silicon	ICP analysis by SW6010	2630	Qma/ka	110	Laboratory QC
BLK93638	Silicon	ICP analysis by SW6010	3130	Qima/ka	11	Laboratory QC
BLK93638	Silver	ICP analysis by SW6010		mg/kg	0.18	Laboratory QC
BLK93682	Silver	ICP analysis by SW6010		mg/kg	1.8	Laboratory QC
BLK93638	Sodium	ICP analysis by SW6010	7.89	mg/kg	2.6	Laboratory QC
BLK93638	Sodium	ICP analysis by SW6010	8.86	mg/kg	2.6	Laboratory QC
DLVA2- 6655	Sodium	ICP analysis by SW6010	4.14 J	mg/kg	21	Laboratory QC
BLK93-682	Strontium	CP analysis by SW6010		mg/Kg mg/ka	21	Laboratory QC
BLK93-682	Strontium	ICP analysis by SW6010	0.23	mg/kg	0.0	Laboratory QC
BLK93-682	Thallium	ICP analysis by SW6010	41.9	l ma/ka	120	Laboratory QC
BLK93-682	Thallium	ICP analysis by SW6010	6.62	mg/kg	120	Laboratory QC
BLK93682	Thallium	ICP analysis by SW6010	33.8	l mg/kg	69	Laboratory QC
BLK93-682	Titanium	ICP analysis by SW6010		mg/kg	3.5	Laboratory QC

Table F2-2. Summary of Laboratory Method Blank Results

- {

Sample ID	Analyte	Method	Result	Units	Det Lim	Sample
BLK93682	Titanium	ICP analysis by SW6010	0.92 J	mg/kg	1.8	Laboratory QC
BLK93-682	Titanium	ICP analysis by SW6010	6.07	mg/kg	3.5	Laboratory QC
BLK93638	Vanadium	ICP analysis by SW6010	0.24 J	mg/kg	0.43	Laboratory QC
BLK93682	Vanadium	ICP analysis by SW6010	2.2 J	mg/kg	4.3	Laboratory QC
BL.K93638	Vanadium	ICP analysis by SW6010	0.12 J	mg/kg	0.43	Laboratory QC
BLK93-682	Zinc	ICP analysis by SW6010		mg/kg	19	Laboratory QC
BLK93-682	Zinc	ICP analysis by SW6010	9.77 J	mg/kg	19	Laboratory QC
BLK93-376	Aluminum	ICP analysis by SW6010	0. 00 J	mg/L	0.028	Laboratory QC
BLK93716	Aluminum	ICP analysis by SW6010	0.00 J	mg/L	0.028	Laboratory QC
BLK93716	Antimony	ICP analysis by SW6010	0.00 J	mg/L	0.024	Laboratory QC
BLK93-376	Antimony	ICP analysis by SW6010		mg/L	0.024	Laboratory QC
BLK93~376	Arsenic	ICP analysis by SW6010		mg/L	0.023	Laboratory QC
BLK93716	Arsenic	ICP analysis by SW6010	0.00 J	mg/L	0.023	Laboratory QC
BLN33042	Arsenic	Argenia by SM7060		mg/L	0.00065	Laboratory QC
BLK93-375	Register	ICP analysis by SWE010	0.00.1	mg/L mg/l	0.00063	Laboratory QC
BL K93-376	Berium	ICP analysis by SW6010	0.00 0	mg/L	0.00053	Laboratory QC
BL K93-376	Bendlium	ICP analysis by SW6010	0.00.1	ma/L	0.00055	Laboratory QC
BLK93716	Beryllium	ICP analysis by SW6010	0.00 J	ma/l	0.00055	Laboratory QC
BLK93716	Boron	ICP analysis by SW6010	0.02	ma/L	0.015	Laboratory QC
BLK93-376	Boron	ICP analysis by SW6010	ND	ma/L	0.015	Laboratory QC
BLK93-376	Cadmium	ICP analysis by SW6010	0.00	ma/L	0.0017	Laboratory QC
BLK93642	Cadmium	Cadmium by SW7131		ma/L	0.00024	Laboratory QC
BLK93716	Cadmium	ICP analysis by SW6010	0.00 J	mg/L	0.0017	Laboratory QC
BLK93379	Cadmium	Cadmium by SW7131	ND	mg/L	0.00032	Laboratory QC
BLK93716	Calcium	ICP analysis by SW6010	0.05 J	mg/L	0.15	Laboratory QC
BLK93-376	Calcium	ICP analysis by SW6010		mg/L	0.15	Laboratory QC
BLK93624	Chloride	Chloride, by IC EPA300		mg/L	0.02	Laboratory QC
BLK93744	Chloride	Chloride, by IC EPA300		mg/L	0.02	Laboratory QC
BLK93746	Chloride	Chloride, by IC EPA300		mg/L	0.02	Laboratory QC
BLK93566	Chloride	Chloride, by IC EPA300	ND	mg/L	0.02	Laboratory QC
BLK93-376	Chromium	ICP analysis by SW6010		mg/L	0.0025	Laboratory QC
BLK93-379	Chromium	Chromium by GF - SW7191	0.00	mg/L	0.00072	Laboratory QC
BLK93716	Chromium	Chanalysis by SW6010	0.00.1	mg/L	0.0025	Laboratory QC
DLN93042	Cobolt	Chromium by GF - SW/191	0.00 J	mg/L	0.00072	Laboratory CC
BLN93/10	Cobelt	ICP analysis by SW6010		mg/L mg/l	0.0034	Laboratory QC
BLK93-376	Copper	ICP analysis by SW6010		ma/l	0.0034	Laboratory QC
BLK93716	Copper	ICP analysis by SW6010	0.00	ma/L	0.0038	Laboratory QC
BLK93725	Fluoride	Fluoride by EPA 340.2	0.02	Bma/L	0.02	Laboratory QC
BLK93634	Fluoride	Fluoride by EPA 340.2	0.01 J	mg/L	0.02	Laboratory QC
BLK93635	Fluoride	Fluoride by EPA 340.2	0.01 J	mg/L	0.02	Laboratory QC
Lab Blank	Fluoride	Fluoride by EPA 340.2	0.01 J	mg/L	0.02	Laboratory QC
Lab Blank	Fluoride	Fluoride by EPA 340.2	0.02	mg/L	0.02	Laboratory QC
BLK93539	Fluoride	Fluoride by EPA 340.2	0.03	mg/L	0.02	Laboratory QC
BLK93716	lron	ICP analysis by SW6010	0.01	mg/L	0.006	Laboratory QC
BLK93-431	Iron	ICP analysis by SW6010	0.01 J	mg/L	0.024	Laboratory QC
BLK93482	Lead	Lead by SW7421	0.00	mg/L	0.0011	Laboratory QC
BLK93482	Lead	Lead by SW7421	0.00	mg/L	0.0011	Laboratory QC
BLK93-376	Lead	ICP analysis by SW6010		mg/L	0.027	Laboratory QC
BLK93-642	Lead	Lead by SW7421	0.00	I mg/L	0.0011	Laboratory QC
BLK93607	Lead	Leed by SW7421		mg/L	0.0011	Laboratory QC
DLK93/10	Leas	ICP analysis by SW6010		mg/∟	0.027	Laboratory QC
DLN93-3/6	Magnesium	ICP analysis by SW6010		mg/L	0.023	
BLK93718	Magnessan	ICP analysis by SW6010		mg/L	0.023	Laboratory QC
BI K93-376	Manganese	ICP analysis by SW6010	0.00	mo/l	0.00003	Laboratory QC
BI K93722	Mercury	Mercury cold vapor SW7470	0.00	ma/l	0.00004	Laboratory QC
BLK93596	Mercury	Mercury, cold vanor SW7470		ma/l	0.00004	Laboratory OC
BLK93587	Mercury	Mercury.HNO3/H2O2 Impinge	r	ma/L	0.00004	Laboratory QC
BLK93647	Mercury	Mercury, cold vapor SW7470		ma/L	0.00004	Laboratory QC
BL K93587	Mercury	Mercury, cold vapor SW7470		mg/L	0.00004	Laboratory QC
BLK93716	Molybdenum	ICP analysis by SW6010	0.00	J mg/L	0.0046	Laboratory QC
BLK93-376	Molybdenum	ICP analysis by SW6010		mg/L	0.0046	Laboratory QC
BLK93-376	Nickel	ICP analysis by SW6010		mg/L	0.0099	Laboratory QC
BLK93379	Nickel	Nickel by GF, EPA 249.2	0.00	J mg/L	0.0018	3 Laboratory QC
BLK93642	Nickel	Nickel by GF, EPA 249.2	0.00	J mg/L	0.00182	2 Laboratory QC
BLK93716	Nickel	ICP analysis by SW6010		mg/L	0.0099	Laboratory QC
BLK93-431	Phosphorus	ICP analysis by SW6010		mg/L	0.84	Laboratory QC

Table F2-2. Summary of Laboratory Method Blank Results

Sample ID	Analvte	Method	Result	Units	DetLim	Sample
BLK93782	Phosphorus	ICP analysis by SW6010	0.25 J	mg/L	0.84	Laboratory QC
BLK93716	Potassium	ICP analysis by SW6010		mg/L	0.37	Laboratory QC
BLK93-376	Potassium	ICP analysis by SW6010		mg/L	0.37	Laboratory QC
BLK93 - 376	Selenium	ICP analysis by SW6010		mg/L	0.042	Laboratory QC
BLK93716	Selenium	ICP analysis by SW6010	0.01 J	mg/L	0.042	Laboratory QC
BLK93642	Selenium	Selenium by SW7740		mg/L	0.00144	Laboratory QC
BLK93482	Selenium	Selenium by SW7740		mg/L	0.0014	Laboratory QC
BLK93–376	Silicon	ICP analysis by SW6010	0.02 J	mg/L	0.27	Laboratory QC
BLK93716	Silicon	ICP analysis by SW6010	0.05	mg/L	0.027	Laboratory QC
BLK93–376	Silver	ICP analysis by SW6010		mg/L	0.0049	Laboratory QC
BLK93716	Silver	ICP analysis by SW6010	0. 00 J	mg/L	0.0049	Laboratory QC
BLK93716	Sodium	ICP analysis by SW6010	0.01 J	mg/L	0.04	Laboratory QC
BLK93 – 376	Sodium	ICP analysis by SW6010	0.01 J	mg/L	0.04	Laboratory QC
BLK93716	Strontium	ICP analysis by SW6010	0.00	mg/L	0.00017	Laboratory QC
BLK93–376	Strontium	ICP analysis by SW6010	ND	mg/L	0.00017	Laboratory QC
BLK93624	Sulfate	Sulfate, EPA 300.0		mg/L	0.06	Laboratory QC
BLK93744	Sulfate	Sulfate on filters	0	mg/L	0.06	Laboratory QC
BLK93566	Sulfate	Sulfate, EPA 300.0	ND	mg/L	0.06	Laboratory QC
BLK93625	Sulfate	Sulfate, EPA 300.0		mg/L	0.06	Laboratory QC
BLK93716	Thaliium	ICP analysis by SW6010		mg/L	0.017	Laboratory QC
BLK93-376	Thallium	ICP analysis by SW6010	0.03	mg/L	0.017	Laboratory QC
BLK93716	Titanium	ICP analysis by SW6010		mg/L	0.001	Laboratory QC
BLK93 - 376	Titanium	ICP analysis by SW6010		mg/L	0.001	Laboratory QC
BLK93716	Vanadium	ICP analysis by SW6010		mg/L	0.0024	Laboratory QC
BLK93-376	Vanadium	ICP analysis by SW6010		mg/L	0.0024	Laboratory QC
BLK93-376	Zine	ICP analysis by SW6010	0.00 J	mg/L	0.0015	Laboratory QC
BLK93716	Zinc	ICP analysis by SW6010	0. 00 J	mg/L	0.0015	Laboratory QC
BLK93-638	Aluminum	ICP analysis by SW6010		ug	8.3	Laboratory QC
8LK93-638	Antimony	ICP analysis by SW6010		ug	1.5	Laboratory QC
BLK93638	Arsenic	BIF As for Metals Trains	0.65	ug	0.093	Laboratory QC
BLK93-638	Arsenic	ICP analysis by SW6010		ug	3.4	Laboratory QC
BLK93-638	Barium	ICP analysis by SW6010	0.09	ug	0.092	Laboratory QC
BLK93-638	Beryllium	ICP analysis by SW6010	0.03 J	ug	0.068	Laboratory QC
BLK93638	Cadmium	BIF Cd for Metals Trains		ug	0.1	Laboratory QC
BLK93-638	Calcium	ICP analysis by SW6010	4.53 J	ug	26	Laboratory QC
BLK93-638	Chromium	ICP analysis by SW6010	0.42 J	ug	0.47	Laboratory QC
BLK93-638	iron	ICP analysis by SW6010	3.06 J	ug	34	Laboratory QC
BLK93638	Lead	BIF Pb for Metals Irains		uġ	0.11	Laboratory QC
BLK93-638	Lead	ICP analysis by SW6010	0.25 J	ug	3.1	Laboratory QC
BLK93-638	Magnesium	ICP analysis by SW6010	1.67 J	ug	2.3	Laboratory QC
BLK93-638	Manganese	ICP analysis by SW6010	0.15	ug	0.13	Laboratory QC
BLN93040	Mercury	Mercury, cold vapor Sw/4/1		ug	0.0096	Laboratory QC
BLK93-038	NICKH Nickel	BIF NI TOP MOTUS TRAINS	A AA 1	ug	0.117	Laboratory QC
BLK93-636	Discontraction	DIF 111 TOF MINUS ITALIS	0.02 3	ug	0.117	Laboratory QC
BLK93-030	Phosphorus	DIF IOD to Motole Trains		ug	7.3	Laboratory QC
BLN93030	Priosphorus	DIF ICP for Metals trains		ug	7.29	Laboratory QC
BLK93030	Sedium	DIT SETOT MERIIS HAINS	19.1	ug	0.12	Laboratory QC
DL 133 - 030	Stantium	ICP analysis by SW6010	13.1	ug	4.0 0.0E	Laboratory QC
BLK93-638	Thellium	ICP analysis by SW6010	0.00	ug	0.05	Laboratory GC
BLK93-638	Titerium	CP analysis by SW6010	0.71	ug	0 47	Laboratory QC
BL K03 - 638	Titesium	ICP analysis by SW6010	0.71	ug	0.47	Laboratory QC
BLK93-638	Zinc	ICP analysis by SW6010	0.71	ug	0.97	Laboratory QC
02130-000	2000	icr analysis by orrooto	0.04 0	ug	0.07	Laboratory QC
Method Blank	Hevevelent chromium	Cr(VI) by BIE METHOD	ND	us/I	0.024	Leboratory OC
RI K93-381	Hexavalent chromium	Cr(VI) by BIE METHOD	ND	ug/L	0.024	Laboratory OC
02.000 000		0.(1,) 0, 0		09/ L	0.04	
Method Blank	Chlorida	Chlorida, by IC EPA300	ND	ua/semr	0.02	Laboratory QC
Leh Blank	Sulfate	Sulfate on filters	ND	ug/same	0.05	Laboratory OC
BI K93794	Sulfur	Total sultur		% Sulfur	0.00	Laboratory OC
		· artest warries			0.000	
Method Blani	Antimony	ICP-MS	0.05	ua	0.0004	Method Blank
Method Blani	Arsenic	ICP-MS	1.3	- 0 ua	0.0054	Method Blank
Method Blank	Barium	ICP-MS	0.14	-e	0.0021	Method Blank
Method Blani	k Beryllium	ICP-MS	0.4	- 0	0.0028	Method Blank
Method Blani	Cadmium	ICP-MS	0.03 6	Bug	0.0012	Method Blank
Method Blani	k Chromium	ICP-MS	4.59		0.0014	Method Blank
Method Blani	k Cobalt	ICP-MS	0.13	ug	0.0005	Method Blank
Method Blan	k Copper	ICP-MS	0.7	uğ	0.0016	Method Blank

.

Table F2-2. Summary of Laboratory Method Blank Results

Completion	An -1 - 4 -	Adapte and	Deput Laite	Det im Semale
Sample ID	Analyte	Metrico	Hesuit Onits	Derum Sample
Method Blank	Lead	ICP-MS	0.01 @ Ug	0.0009 Method Blank
Method Blarik	Manganese	ICP-MS	0.09 ug	0.0004 Method Blank
Method Blank	Mercury	ICP-MS	0.14 @ ug	0.0049 Method Blank
Method Blank	Molybdenum	ICP-MS	0.18 ug	0.0024 Method Blank
Method Blank	Nickel	ICP-MS	0.52 ug	0.0049 Method Blank
Method Blank	Selenium	ICP-MS	1.6 ug	0.009 Method Blank
Method Blank	Vanadium	ICP-MS	2.84 ug	0.002 Method Blank
Method Biank	Antimony	ICP-MS	0.02 ug	0.0003 Method Blank Dup
Method Blank	Arsenic	ICP-MS	0.97 ug	0.0037 Method Blank Dup
Method Biank	Barium	ICP-MS	0.15 ug	0.0017 Method Blank Dup
Method Blank	Beryllium	ICP-MS	0.34 ug	0.0024 Method Blank Dup
Method Blank	Cadmium	ICP-MS	0.00 ua	0.0005 Method Blank Dup
Method Blank	Chromium	ICP-MS	3.71 10	0.0108 Method Blank Dup
Method Blank	Cohett	ICP-MS	0.11 UD	0 0006 Method Blank Dup
Method Blank	Copper	ICP_MS	0.62 μg	0.0016 Method Blank Dup
Method Black	Lond		0.02 Ug	0.0000 Method Blank Dup
Method Diarik				0.0009 Method Blank Dup
Method Biank	Manganese		0.01 @ Ug	0.0004 Method Slank Dup
Method Blank	Mercury	ICP-MS	0.02 ug	0.0029 Method Blank Dup
Method Blank	Molybdenum	ICP-MS	0.05 @ ug	0.0018 Method Blank Dup
Method Blank	Nickel	ICP-MS	0.22 @ ug	0.0052 Method Blank Dup
Method Blank	ISelenium	ICP-MS	1.09 ug	0.02 Method Blank Dup
Method Blank	Nanadium	ICP-MS	2.07 ug	0.0026 Method Blank Dup
Lab Diamia	E as attack above and		1 00	10101
Lao Diank			1.29 Ng	
Lab Blank	/n-dibenzo(c,g)carbazole	HHGC/HHMS	21.1 ng	21.1/14 MM3 180 DINK
Lab Blank	Acenaphthene	HRGC/HRMS	1.55 ng	MM5 lab bink
Lab Blank	Acenaphthylene	HRGC/HRMS	2.06 ng	MM5 lab bink
Lab Blank	Anthracene	HRGC/HRMS	0.96 ng	MM5 lab bink
Lab Blank	Benzo(a)pyrene	HRGC/HRMS	5.59 ng	5.59659 MM5 lab bink
Lab Blank	Benzo(b,j&k)fluoranthenes	HRGC/HRMS	3.47 ng	– - MM5 lab bink
Lab Blank	Benzo[ghi]perylene	HRGC/HRMS	2.47 ng	MM5 lab blnk
Lab Blank	Benz[a]anthracene	HRGC/HRMS	1.55 ng	MM5 lab bink
Lab Blank	Chrysene	HRGC/HRMS	2.15 na	MM5 lab bink
Leb Biank	Dibenzola elovrene	HRGC/HRMS	1.04 ng	MM5 jeb bink
Leb Blank	Dibenzola blovrene	HRGC/HRMS	12.5 pg	12.5022 MM5 lab bink
Lab Blank	Dibenzola il pyrene	HPGC/HPMS	2.6 5	2 96588 MM5 leb blok
Lab Clark	Diberzela, ijpyrene Diberzia blassidina	HPGC/HPMS	2.30 Hg	MMS lab blok
Lab Blank	Diberz[a,h]achume		3.42 510	MM5 leb blok
Leb Blenk			0.10	
			9.12 Ng	
Lad Blank	riuoranmene	HHGC/HHMS	3.37 ng	MMO IAD DINK
Lab Blank	Fluorene	HRGC/HRMS	2.43 ng	MM5 lab bink
Lab Blank	Indeno[1,2,3-cd]pyrene	HRGC/HRMS	2.70 ng	MM5 lab bink
Lab Blank	Phenanthrene	HRGC/HRMS	7.40 ng	MM5 lab bink
Lab Blank	Pyrene	HRGC/HRMS	2.64 ng	MM5 lab bink
	A		6 4 4 ()	
			UII Ug/mL	v. i ug/samotack gas, impingers
Lab Blank	Formaldehyde	HPLC	0.13 @ ug/mL	0.1 ug/sanStack gas, impingers
	r1 1 - Dichlorosthese	GCMS	10	to VOST Lab Blackt
VOSTIAL		GCMS	10 19	10 VOST ab Blank1
VOCTILED	ar, - Diano outene art 1 t. Triables-stance	GONS	10 ng 10	10 VOST Lab Blask1
		GOME		
VUSILAD BI		GOMO	iu ng	
VOST LAD BI	ari, 1, 2, 2 - Ietrachioroethane	GCMS	10 ng	10 VUST LAD BIANK1
VOST Lab Bl	ar1,2-Dichtorobenzene	GCMS	10 ng	10 VOST Lab Blank1
VOST LAD B	ar1,2-Dichloroethane	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab Bl	ar1,2 - Dichloropropana	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab Bi	ar1,3-Dichlorobenzene	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab B	ar1,4 – Dichlorobenzene	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab BI	ar2 – Butanone	GCMS	50 ng	50 VOST Lab Blank1
VOST Lab B	ar2 – Hexanone	GCMS	50 ng	50 VOST Lab Blank1
VOST Lab B	ar4 – Methyl – 2 – Pentanone	GCMS	50 ng	50 VOST Lab Blank1
VOST LAD B	arAcetone	GCMS	50 ng	50 VOST Lab Blank1
VOST Lab B	arBenzene	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab BI	arBromodichioromethane	GCMS	10 na	10 VOST Lab Blank1
VOST Lab B	arBromoform	GCMS	10 ng	10 VOST Lab Blank1
VOST Lab B	arBromomethane	GCMS	10 00	10 VOST Lab Blank1
VOST Lab R	arCarbon Disulfide	GCMS	10 65	10 VOST Lab Blank 1
VOSTILA	arCarbon Tetrachloride	GCMS	10	10 VOST Lab Blank
VOG1 LAS D	archlorobeszese	GCMS	10 ng	10 VOST Lab Dianki
	arChloroethene	GCMS		
VUOILAD D		GONO	iu ng	IV YUSI LAD DIARKI

Table F2~2. Summary of Laboratory Method Blank Results

Sample ID Analyte	Method
VOST Lab BlarChloroform	GCMS
VOST Lab BlarChloromethane	GÇMS
VOST Lab Blarcis-1,3-Dichloropropene	GCMS
VOST Lab BlarDibromochioromethane	GCMS
VOST Lab Blanthyl Benzene	GCMS
VOST Lab Blarmenytene Chickles	GCMS
VOST Lab Blam + Xviene	GCMS
VOST Lab BlarStvrene	GCMS
VOST Lab BlarTetrachloroethene	GCMS
VOST Lab BiarToluene	GCMS
VOST Lab Blartrans-1,2-Dichloroethene	GCMS
VOST Lab Blartrans-1,3-Dichloropropene	GCMS
VOST Lab BlarTrichloroethene	GCMS
VOST Lab Blar Trichlorofluoromethane	GCMS
VOST Lab Blatvinyi Acetate	GCMS
VOST Lab Blart 1 - Dichloroothane	GCNS
VOST Lab Biart, 1 - Dichloroethene	GCMS
VOST Lab Blar1,1,1 - Trichloroethane	GCMS
VOST Lab Blar1,1,2-Trichloroethane	GCMS
VOST Lab Blar1, 1, 2, 2 - Tetrachioroethane	GCMS
VOST Lab Blar1,2-Dichlorobenzene	GCMS
VOST Lab Blar1,2-Dichloroethane	GCMS
VOST Lab Blar1,2-Dichloropropane	GCMS
VOST Lab Blart, 3~Dichlorobenzene	GCMS
VOST Lab Blar2 - Butenone	GCMS
VOST Lab Blar2-Hexanone	GCMS
VOST Lab Blar4-Methyl-2-Pentanone	GCMS
VOST Lab BlarAcetone	GCMS
VOST Lab BlarBenzene	GCMS
VOST Leb BlarBromodichioromethane	GCMS
	GCMS
VOST Lab Blackston Disulfide	GCMS
VOST Leb BlarCarbon Tetrachloride	GCMS
VOST Lab BlarChlorobenzene	GCMS
VOST Lab BlarChloroethane	GCMS
VOST Lab BlarChloroform	GCMS
VOST Lab BlarChloromethane	GCMS
VOST Lab Blarcis-1,3-Dichloropropene	GCMS
VOST Lab BiarDibromochioromethane	GCMS
VOST Lab BlarMethylene Chloride	GCMS
VOST Lab Blarm, p - Xviene	GCMS
VOST Lab Blaro-Xylene	GCMS
VOST Lab BlarStyrene	GCMS
VOST Lab BlarTetrachloroethene	GCMS
VOST Lab Blartoluene	GCMS
VOST Lab Blantrans~ 1,2-Dichloroethene	GCMS
VOST Lab BlarTrichloroethene	GCMS
VOST Lab BlarTrichlorofluoromethane	GCMS
VOST Lab BlarVinyl Acetate	GCMS
VOST Lab BlarVinyl Chloride	GCMS
VOST Lab Blar1,1-Dichloroethane	GCMS
VOST Lab Blart 1 1 Trichloroethene	GCMS
VOST Lab Blar1.1.2-Trichloroethane	GCMS
VOST Lab Biar1,1.2,2-Tetrachioroethane	GCMS
VOST Lab Blar1,2-Dichlorobenzene	GCMS
VOST Lab Blar1,2-Dichloroethane	GCMS
VOST Lab Blar1,2~Dichloropropane	GCMS
VUSI LED BIATI,3-Dichlorobenzene	GCMS
VOST Lab Blar2 - Butanone	GCMS
VOST Lab Blar2-Hexanone	GCMS
VOST Lab Blar4-Methyl-2-Pentanone	GCMS
VOST Lab BlarAcetone	GCMS

Result	Units	Det Lim Sample
10	nġ	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
11	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blank1
10	ng	10 VOST Lab Blankt
10	ng	10 VOST Leb Blankt
10	ng	10 VOSI Lab Blank1
50	ng	50 VOST Lab Blank1
10	ng	
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	
10	ng	
10	ng	
10	ng 80	10 VOST Lab Blank2
10	រាម្ន	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
50	ng	TO VOST LED Blank2
50	119	50 VOST Lab Blank2
50	119	50 VOST Lab Blank2
50	119	50 VOST Lab Blank2
10	11 <u>9</u>	10 VOST Lab Blank2
10	200	10 VOST Lab Blank2
10	70 70	10 VOST Lab Black2
10	ng	10 VOST Leb Blank2
10	00	10 VOST Lab Blank2
10	na	10 VOST Lab Blank2
10	na	10 VOST Lab Blank2
10	na	10 VOST Lab Blank2
10	na	10 VOST Leb Blank2
10	na	10 VOST Lab Blank2
10	na	10 VOST Lab Blank2
10	na	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
13	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Leb Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
13	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
50	ng	50 VOST Lab Blank2
10	ng	10 VOST Lab Blank2
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST LAD Blank3
10	ng	TO VUST LAD BIANKS
10	ng	10 VOST LAD Blank3
10	ng	
50	ng	
50	119	SO VOST LAD BIANKS
50		50 VOST Lab Blanks
50		U TOUL LAD DIENKS

Table F2-2. Summary of Laboratory Method Blank Results

-

Sample iD	Analyte	Method
VOST Lab	BlarBenzene	GCMS
VOST Lab	BlarBromodichloromethane	GCMS
VOST Lab	BiarBromoform	GCMS
VOST Lab	BiarBromomethane	GCMS
VOST Lab	BlarCarbon Disulfide	GCMS
VOST Lab	BlarCarbon Tetrachioride	GCMS
VOST Lab	BlarChlorobenzene	GCMS
VOST Lab	BlarChloroethane	GCMS
VOST Lab	BlarChloroform	GCMS
VOST Lab	BlarChloromethane	GCMS
VOST Lab	Blarcis-1,3-Dichloropropene	GCMS
VOST Lab	SlarDibromochloromethane	GCMS
VOST Lab	BlarEthyl Benzene	GCMS
VOST Lab	BlarMethylene Chloride	GCMS
VOST Lab	Blarm,p-Xylene	GCMS
VOST Lab	Biaro-Xylene	GCMS
VOST Lab	BlarStyrene	GCMS
VOST Lab	BlarTetrachloroethene	GCMS
VOST Lab	BiarToluene	GCMS
VOST Lab	Biartrans-1,2-Dichloroethene	GCMS
VOST Lab	Slartrans-1,3-Dichloropropene	GCMS
VOST Lab	BlarTrichloroethene	GCMS
VOST Lab	BierTrichlorofluoromethane	GCMS
VOST Lab	BlarVinyl Acetate	GCMS
VOST Lab	BlarVinyl Chloride	GCMS

Result	Units	Det Lim Sample
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
12	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Biank3
11	ng	10 VOST Lab Blank3
1Ò	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Blank3
10	ng	10 VOST Lab Biank3
10	ng	10 VOST Lab Blank3
50	ng	50 VOST Lab Blank3
10	ng	10 VOST Lab Blank3

Table F2--3. Summary of Reagent Blank Results

Sample ID	Analyte	Method		Result	Units	Det Lin	Sample
M-FF Blank	Antimony	ICP-MS		0.29	ug	0.00	Final Filter Blank 1
M-FF Blank	Amenic	ICP-MS		1.68	υg	0.00	Final Filter Blank 1
M-FF Blank	Barium	ICP-MS		27.25	uğ	0.02	Final Filter Blank 1
M-FF Blank	Beryllium	ICP-MS		0.63	υĝ	0.00	Final Filter Blank 1
M-FF Blank	Cadmium	ICP-MS		0.17	ug	0.00	Final Filter Blank 1
M-FF Blank	Chromium	ICP-MS		9.9	ug	0.00	Final Fitter Blank 1
M-FF Blank	Cobalt	ICP-MS		0.49	ug	0.00	Finel Filter Blank 1
M-FF Blank	Copper	ICP-MS		1.73	ug	0.00	Final Filter Blank 1
M-FF Blank	Lead	ICP-MS		1.24	ug	0.00	Final Filter Blank 1
M-FF Blank	Manganese	ICP-MS		2.69	ug	0.00	Finel Filter Blank 1
M-FF Blank	Mercury	ICP-MS		0.15 @	2 ug	0.00	Final Filter Blank 1
M-FF Blank	Molybdenum	ICP-MS		36.3	ug	0,03	Final Filter Slank 1
M-FF Blank	Nickel	ICP-MS		2.37	ug	0.00	Final Filter Blank 1
M-FF Blank	Selenium	ICP-MS		12.01	ug	0.01	Final Filter Blank 1
M-FF Blenk	Vanadium	icp-ms		4.51	ug	0.00	Final Filter Blank 1
N-FF Blank	Antimony	ICP-MS		0.18	ug	0.00	Final Filter Blank 2
N-FF Blank	Arsenic	ICP-MS		0.9	ug	0.00	Final Filter Blank 2
N-FF Blank	Barium	ICP-MS		25.85	ug	0.02	Final Filter Blank 2
N-FF Blank	Beryllium	ICP-MS		0.32	ug	0.00	Final Filter Blank 2
N-FF Blank	Cadmium	ICP-MS		0.03 (ន្ទ បក្ខ	0.00	Final Filter Blank 2
N-FF Blank	Chromium	ICP-MS		3.59	ug	0.01	Final Filter Blank 2
N-FF Blank	Cobalt	KP-MS		0.4	ug	0.00	Final Filter Blank 2
N-FF Blank	Copper	ICP-MS		0.69	ug	0.00	Final Filter Blank 2
N-FF Blank	Lead	ICP-MS		1.34	ug	0.00	Final Filter Blank 2
N-FF Blank	Manganese	ICP-MS		1.6	ug	0,00	Final Filter Blank 2
N-FF Blank	Mercury	ICP-MS		0.12 (🕑 ug	0.00	Final Filter Blank 2
N-FF Blank	Molybdenum	ICP-MS		29.93	ug	0.02	Final Filter Blank 2
N-FF Blank	Nickel	ICP-MS		0.5	្មពថ	0.00	Final Filter Slank 2
N-FF Blank	Selenium	ICP-MS		0.65 (g ug	0.03	Final Filter Blank 2
N-FF Blank	Vanadium	ICP-MS		1.66	ug	Q. D O	Final Filter Blank 2
H_198	Hevevelent chromium			40.2	unđ	24	Been blok KOH impe
H_197	Hereweient Chromium			42	990- UA/	24	Read bink, KOH impa
H-139	Hexavalent chromium			37 9	100/1	2.4	Beag bink, KOH imps
11-100				01.0	- .	_	tion & name in the
H-370	Acetaldehyde	HPLC		0.1	ид/п	nL0.1 ug/	Stack gas, impingers
H-339	Acetaidehyde	HPLC		0.0484	uğ/n	nL0.1 ug/	Stack gas, impingers
H-339	Formaldehyde	HPLC		0.0184	uğ/n	nL0.1 ug/	Stack gas, impingers
H-370	Formaldehyde	HPLC		0.02	ug/n	nL0.1 ug/	Stack gas, impingers
					_		
H-QB1	Aluminum	ICP analysis by SW6010		286	Bug	7.33	Stack gas, qtz filt bink 1
H-QB1	Antimony	BIF ICP for Metals Trains		5.91	ug	1.54	Stack gas, qtz filt bink 1
H-QB1	Arsenic	BIF As for Metals Trains		0.34	ug	0.09	Stack gas, qtz tilt pink 1
H-QB1	Arsenic	ICP analysis by SW6010	NU	*	ug	1.50	Stack gas, quz nit bink 1
H-QB1	Banum	ICP analysis by SW6010		5.36	ug	0,05	Stack gas, dtz nit pink 1
H-QB1	Beryillum	ICP analysis by SW6010	NU		Bug	0.05	Stack gas, qtz filt bink 1
H-QBT		ICP analysis by Sweetin	ND		вug	0.28	Stack gas, qtz tilt bink 1
		BIF CO for Metals 1 rains	NU	100	ug	0.04	Stack gas, qtz mt bink 1
	Calcium	ICP analysis by SW6010		168	Bug	23.8	Stack gas, qui fin bink 1
	Chromium	DIE Cates Mateia Trains		2.40	b ug	0.27	Stack gas, qtz fitt block 1
	Cobolt	DIF OF IOF METALS STAINS	ND	2.77	oug	0.00	Stack gas, qtz filt block 1
	Copper	CP analysis UY 3 WOULU		1 43	Bus	0.52	Stack Gas, die filt bink 1
	imo	ICP analysis by SW6010		95 0	Bua	0.24	Stack das att filt bink 1
8-081	Land	RIE Ph for Metale Traine		03.5	U ug	01.1	Stack das off filt blok 1
H-OB1	Lead	ICP analysis by SW6010		17 4	ug Do	0.11 9.47	Stack das, oft filt bink 1
H-Q81	Magnesium	ICP analysis by SW6010		10.4	un un	2.77	Stack das, otz filt blok 1
H-QB1	Manganese	ICP analysis by SW6010		1.36	Buo	0.01	Stack gas, atz filt bink 1
H-QB1	Mercury	Mercury, cold vapor SW7471		0.19	Buo	0.04	Stack gas, gtz fitt blnk 1
H-QB1	Molybdenum	BIF ICP for Metals Trains		28.7	Bua	0.29	Stack gas, qtz filt bink 1

.

Table F2-3. Summary of Reagent Blank Resu	lits	
---	------	--

Sample ID	Anaivte	Method		Result	Units	Det Lin	Sample
	Niciral	ICP analysis by SW6010		1 47	Bua	1.09	Stack das, dtz filt bink 1
	Nickel	BIE NI for Metals Trains		1 16	ug	0.11	Stack gas, qu int bink 1
	Bhosphonus	BIE ICP for Metals Trains	ND	1.10	110	7 29 9	Stack and att fit bink 1
	Priospriorus		ND		uy Dug	94.6	Stack ges, qui lit bink t
H-Q81	Potestium	ICP analysis by SW6010			Bug	34.0	Stack gas, qtz int Dirk 1
H-QB1	Selenium	RCP analysis by SW6010			o ug	4.42	
H-QB1	Selenium	BIF Se for Metals I fains	ND		_ug	0.11	Stack gas, qtz nit bink 1
H-QB1	Silver	ICP analysis by SW6010	ND		Bug	0.18	Stack gas, qtz filt bink 1
H-QB1	Sodium	ICP analysis by SW6010		86.4	Bug	2.59	Stack gas, qtz filt blnk 1 👘
H-QB1	Strontium	ICP analysis by SW6010		2.73	B ug	0.05	Stack gas, qtz filt bink 1
H-QB1	Thallium	ICP analysis by SW6010	ND		8 ug	6.94	Stack gas, qtz filt bink 1 👘
H-Q81	Titanium	BIF ICP for Metals Trains		2.79	Bug	0.47	Stack gas, qtz filt bink 1 👘
H-QB1	Vanadium	ICP analysis by SW6010		0.68	Bug	0.43	Stack gas, qtz filt bink 1
H-OB1	Zinc	ICP analysis by SW6010		7	Bua	0.29	Stack das, atz filt bink 1
					•		
H-OB2	Aluminum	ICP analysis by SW6010		286	Виа	7.33	Stack gas, gtz filt bink 2
H-082	Antimony	BIE ICP for Metals Trains		4.92		1.54	Steck ges, gtz filt bink 2
	Amenic	ICP englysis by SW6010	ND			1.58	Stack des ofz fit blok 2
		BIE As for Metale Traine	110	0 17		0.00	Stack ges at fit bink 2
H-Q82	Ansenic			4.00	ug	0.00	Stack gas, que fit black 2
H-Q82	Banum	ICP Enalysis by SW6010		4.39	_ ug	0.05	Stack gas, diz fit bink 2
H-QB2	Beryllium	ICP Englysis by Sweetu	NU		Bug	0.05	Stack gas, qtz nit bink 2
H-QB2	Cadmium	ICP analysis by SW6010	ND		Bug	0.25	Stack gas, qtz filt bink 2
H-Q82	Cadmium	BIF Cd for Metals Trains	ND		ug	0.04	Stack gas, qtz filt bink 2
H-QB2	Calcium	ICP analysis by SW6010		146	B ug	23.8	Stack gas, otz filt bink 2
H-Q82	Chromium	BIF Cr for Metals Trains		2.96	Bug	0.05	Stack gas, otz filt bink 2
H-QB2	Chromium	ICP analysis by SW6010		3.06	Bug	0.27	Stack gas, gtz filt bink 2
H-QB2	Cobalt	ICP analysis by SW6010	ND		Bug	0.52	Stack gas, qtz filt bink 2
H-OB2	Copper	ICP analysis by SW6010		1.66	Bua	0.24	Stack gas, gtz filt bink 2
H-QB2	Inon	ICP analysis by SW6010		38.8	Bug	31.1	Stack gas, gtz filt blnk 2
	Lead	BIF Pb for Metals Trains		0.25		0.11	Stack das, gtz filt bink 2
	Lead	ICP enalysis by SW6010		15.2		2 47	Stack das dtz filt blok 2
	Megnesium	ICP analysis by SW6010		21.5	-9	2 73	Stack ges gtz filt blok 2
	Magnosan	ICP analysis by SW6010		1.38	Bug	0.01	Stack das dtz filt bink 2
	Manyanovo	Mercury cold yapor SW7471		0 122	Rug	0.00	Stack one att filt blok 2
	Mercury	BIE ICB for Metale Traine		08.5	Dug	0.00	Stack gas, que in bink 2
H-QB2	Molypdenum			20.0	Dug	0.29	Steck gas, qui fit black 2
H-Q82	NICKEI	CP analysis by Swould		1.30	e ug	1.09	Stack gas, diz tit bink 2
H-Q82	NICKO	BIF NITOR METELS I FEIRIS		1.4	ug	0.11	Stack gas, qtz nit bink 2
H-QB2	Phosphorus	BIF ICP for Metals I rains	NU		_ug	7.29	Stack gas, gtz titt bink 2
H 082	Potassium	ICP analysis by SW6010		74.3	Bug	34.6	Stack gas, atz fin bink 2
H-QB2	Selenium	ICP analysis by SW6010	ND		Bug	4.42	Stack gas, qtz filt bink 2
H-Q82	Selenium	BIF Se for Metals Trains		0.17	Rug	0.11	Stack gas, qtz filt bink 2
H-QB2	Silver	ICP analysis by SW6010	ND		Bug	0.18	Stack gas, qtz filt bink 2
H-QB2	Sodium	ICP analysis by SW6010		99.4	Bug	2.59	Stack gas, qtz filt blnk 2
H-QB2	Strontium	ICP analysis by SW6010		2.73	Bug	0.05	Stack gas, qtz filt bink 2
H-Q82	Thallium	ICP analysis by SW6010	ND		Bug	6.94	Stack gas, qtz filt bink 2
H-Q82	Titanium	BIF ICP for Metals Trains		2.81	Bug	0.47	Stack gas, qtz filt bink 2
H-082	Vanadium	ICP analysis by SW6010		0.961	Bua	0.43	Stack das, dtz filt bink 2
H-082	Zinc	ICP analysis by SW6010		3.21	Bua	0.29	Stack gas, gtz filt bink 2
					,	÷	· · · · · · · · · · · · · · · · · · ·
H 083	Aluminum	ICP analysis by SW6010		286	Bug	7 35	Stack das lotz filt bink 3
H-QB3	Antimony	BIE ICP for Metals Trains		5 45	ing.	1.54	Stack gas, gtz filt blok 3
H-080	Amonio	CP ensivers by SW6010	ND	0.40	, vg	4 54	Stock one ott fit blok 3
	Americ	BIE As for Metain Trains	110	0.44	- ug 1	0.00	Stack ges, qtz filt bink 9
H-Q83	ArseniiC			U.444	• ug	0.05	F GUECK GES, QUE III. DINK 3
H-QB3	senum Senullium	ICP analysis by SW6010		4.47	r ug	0.0	Source gas, giz na bink 3
H-QB3	Beryillum	ICP analysis by SW6010	UN TH		ьnd	0.0	o otack gas, qtz nit bink 3
H-Q83	Caomium	ICP analysis by SW6010			вug	0.2	Stack gas, qtz nit bink 3
H-Q83	Cadmium	BIF CO TOT Metals Trains	ND	,	ug	0.0	STACK gas, qt2 hit bink 3
H-QB3	Calcium	ICP analysis by SW6010		14	Bug	23.	B Stack gas, gtz filt bink 3
H-Q83	Chromium	BIF Cr for Metals Trains		3.5	4 Bug	0.0	5 Stack gas, qtz filt bink 3
H-QB3	Chromium	ICP analysis by SW6010		3.4	6 Bug	0.2	7 Stack gas, qtz filt bink 3
H-QB3	Cobalt	ICP analysis by SW6010	ND		Bug	0.5	2 Stack gas, qtz filt bink 3

Table F2-3. Summary of Reagent Blank Results

Sample ID	Analyte	Method		Result	Units	Det Lin Sample	
H-QB3	Copper	ICP analysis by SW6010		1.51	8 ug	0.24 Stack gas, gtz filt bin	ik 3
H-QB3	tron	ICP analysis by SW6010		37.6	Bug	31.1 Stack gas, gtz filt bin	ik 3
H-QB3	Leed	BIF Pb for Metals Trains		0.29	uğ	0.11 Stack gas, gtz filt bin	ık 3
H-QB3	Lead	ICP analysis by SW6010		16.4	ug	2.47 Stack gas, gtz filt bin	ık 3
H-QB3	Magnesium	ICP analysis by SW6010		22.4	ug	2.73 Stack gas, gtz filt bin	ık 3
H-Q83	Manganese	ICP analysis by SW6010		1.42	Bug	0.01 Stack gas, gtz filt bir	ık 3
H-QB3	Mercury	Mercury, cold vapor SW7471		0.122	Bug	0.00 Stack gas, gtz filt bir	ık 3
H-QB3	Molybdenum	BIF ICP for Metals Trains		28.3	Bug	0.29 Stack gas, gtz filt bir	1k 3
H-QB3	Nickel	BIF NI for Metals Trains		1.48	ug	0.11 Stack gas, gtz filt bir	1K 3
H-Q83	Nickel	ICP analysis by SW6010	ND		Bug	1.09 Stack gas, gtz filt bir	143
H-QB3	Phosphorus	BIF ICP for Metals Trains	ND		uğ	7.29 Stack gas, gtz filt bir	nk 3
H-QB3	Potassium	ICP analysis by SW6010	ND	•	Bug	34.6 Stack gas, gtz filt bir	ık 3
H-QB3	Selenium	ICP analysis by SW6010	ND		Bug	4.42 Stack gas, dtz filt bir	ık 3
H-Q83	Selenium	BIF Se for Metals Trains	ND		uğ	0.11 Stack gas, ctz filt bir	nk S
H-QB3	Silver	ICP analysis by SW6010	ND		Bug	0.18 Stack gas, atz filt bli	nk S
H-QB9	Sodium	ICP analysis by SW6010		115	Bug	2.59 Stack gas, gtz filt bir	nk S
H~QB3	Strontium	ICP analysis by SW6010		2.76	Bug	0.05 Stack gas, qtz fift bli	nk 3
H~QB3	Thallium	ICP analysis by SW6010	ND		Bug	6.94 Stack gas, gtz filt bli	nk 3
H~QB3	Titanium	BIF ICP for Metals Trains		3.35	Bug	0.47 Stack gas, gtz filt bis	nk S
H~QB3	Vanadium	ICP analysis by SW6010		0.939	Bug	0.43 Stack ges, gtz filt bli	nk S
H-QB3	Zinc	ICP analysis by SW6010		3.33	Bug	0.29 Stack gas, qtz filt bli	nk 9

...

Sample ID	Analyte	Method	Result Units	Det Lim	Sample
H330-I-11	Antimony	ICP-M\$	0.04 @ ug	0.0013	Kapton Trip Blank 2
H394-L-11	Antimony	ICP-MS	0.01 @ ug	0.0013	Kapton Trip Blank 1
H330-1-11	Arsenic	ICP-MS	1.53 ug	0.0058	Kapton Trip Blank 2
H394-L-11	Arsenic	ICP-MS	1.32 ug	0.0058	Kapton Trip Blank 1
H330-I-11	Barium	ICP-MS	0.21 @ ug	0.0212	Kapton Trip Blank 2
H394-L-11	Barium	ICP-MS	0.18 @ ug	0.0212	Kapton Trip Blank 1
H330-I-11	Beryliium	ICP-MS	0.39 ug	0.0042	Kapton Trip Blank 2
H394-L-11	Beryllium	ICP-MS	0.6 ug	0.0042	Kapton Trip Blank 1
H330-I-11	Cadmium	ICP-MS	0.06 ug	0.029	Kapton Trip Blank 2
H394-L-11	Cadmium	ICP-MS	0.04 @ ug	0.0029	Kapton Trip Blank 1
H330-I-11	Chromium	ICP-MS	4.05 ug	0.0118	Kapton Trip Blank 2
H394-L-11	Chromium	ICP-MS	4.28 ug	0.0118	Kapton Trip Blank 1
H330-l-11	Cobalt	ICP-MS	0.17 @ ug	0.0073	Kapton Trip Blank 2
H394-L-11	Cobalt	ICP-MS	0.25 @ ug	0.0073	Kapton Trip Blank 1
H394-L-11	Copper	ICP-MS	1.04 ug	0.0018	Kapton Trip Blank 1
H330-I-11	Copper	ICP-MS	0.94 ug	0.0018	Kapton Trip Blank 2
H330-l-11	Lead	ICP-MS	0.04 ug	0.03	Kapton Trip Blank 2
H394-L-11	Lead	ICP-MS	0.08 @ ug	0.003	Kapton Trip Blank 1
H330- -11	Manganese	ICP-MS	0.04 ug	0.012	Kapton Trip Blank 2
H394-L-11	Manganese	ICP-MS	0.03 @ ug	0.0012	Kapton Trip Blank 1
H330-i-11	Mercury	ICP-MS	0.17 @ ug	0.0051	Kapton Trip Blank 2
H394-L-11	Mercury	ICP-MS	0.2 @ ug	0.0051	Kapton Trip Blank 1
H394-L-11	Molybdenum	ICP - MS	0.14 @ ug	0.0372	Kapton Trip Blank 1
H330-I-11	Molybdenum	ICP-MS	0.1 @ ug	0.0372	Kapton Trip Blank 2
H330-1-11	Nickel	ICP-MS	0.12 @ ug	0.0065	Kapton Trip Blank 2
H394-L-11	Nickel	ICP-MS	0.54 ug	0.0065	Kapton Trip Blank 1
H330-l-11	Selenium	ICP-MS	1.6 @ ug	0.0387	Kapton Trip Blank 2
H394-L-11	Selenium	ICP-MS	2.44 ug	0.0387	Kapton Trip Blank 1
H330-I-11	Vanadium	ICP – MS	2.1 ug	0.0026	Kapton Trip Blank 2
H394-L-11	Vanadium	ICP-MS	2.35 ug	0.0026	Kapton Trip Blank 1
H-368	Acetaldehyde	HPLC	< 0.01 ug/r	n0.1 ug/sa	Stack gas, impingers
H-368	Formaldehyde	HPLC	0.015 @ ug/r	n0.1 ug/sa	Stack gas, impingers

Table F2-4. Summary of Trip Blank Results

		Recovery
Analyte	Method code	(%)
Aluminum	ICPES	94
Aluminum	ICPES	97
Aluminum	ICPES	87
Aluminum	ICPES	83
Aluminum	ICPES	98
Aluminum	ICPES	95
Aluminum	ICPES	98
Aluminum	ICPES	97
Antimony	ICPES	96
Antimony	ICPES	98
Antimony	ICPES	94
Antimony	ICPES	92
Апйтопу	ICPES	91
Antimony	ICPES	91
Antimony	ICPES	88
Antimony	ICPES	94
Antimony	ICPES	91
Antimony	ICPES	95
Arsenic	ICPES	21
Arsenic	ICPES	96
Arsenic	ICPES	94
Arsenic	ICPES	96
Arsenic	ICPES	32
Artenic	ICPES	97
Artenic	ICPES	43
Arsenic	ICPES	33
Barium	ICPES	95
Barium	ICPES	98
Barium	ICPES	93
Berium	ICPES	97
Banum	ICPES	96
Barium	ICPES	97
Barium	ICPES	99
Barium	ICPES	97
Beryllium	ICPES	99
Beryllium	ICPES	87
Beryllium	ICPES	89
Beryilium	ICPES	96
Beryllium	ICPES	95
Beryllium	ICPES	89
Beryilium	CPES	80
Beryilium	ICPES	88
Beryllium	ICPES	81
Beryllium	ICPES	99
Boron	ICPES	97
Boron	ICPES	99
Boron	ICPES	96
Boron	ICPES	97
Cadmium	ICPES	76
Cadmium	ICPES	95
Cadmium	ICPES	85

Table F.2-5. Laboratory Control Sample Recovery Data for Metals and Anions

7

		Recovery
Anaiyte	Method code	(%)
Cadmium	ICPES	79
Cadmium	ICPES	68
Cadmium	ICPES	94
Cadmium	ICPES	95
Cadmium	ICPES	92
Calcium	ICPES	103
Calcium	ICPES	94
Calcium	ICPES	100
Calcium	ICPES	93
Calcium	ICPES	85
Calcium	ICPES	102
Calcium	ICPES	99
Calcium	ICPES	89
Chromium	ICPES	95
Chromium	ICPES	96
Chromium	ICPES	91
Chromium	ICPES	91
Chromium	ICPES	93
Chromium	ICPES	94
Chromium	ICPES	96
Chromium	ICPES	91
Cobait	ICPES	94
Cobalt	ICPES	95
Cobait	ICPES	81
Cobalt	ICPES	88
Cobalt	ICPES	80
Cobalt	ICPES	93
Cobalt	ICPES	88
Cobait	ICPES	95
Copper	ICPES	93
Copper	ICPES	91
Copper	ICPES	95
Copper	ICPES	96
Copper	ICPES	91
Copper	ICPES	94
Copper	ICPES	92
Copper	ICPES	96
Iron	ICPES	96
Iron	ICPES	94
iron	ICPES	97
Iron	ICPES	95
Iron	ICPES	92
Iron	ICPES	96
iron	ICPES	100
Iron	ICPES	99
Lead	ICPES	80
Lead	ICPES	89
Lead	ICPES	95
Lead	ICPES	96
Lead	ICPES	97
Lead	ICPES	98

Table F.2-5. Laboratory Control Sample Recovery Data for Metals and Anions

		Recovery
Analyte	Method code	(%)
Lead	ICPES	89
Lead	ICPES	75
Lead	ICPES	87
Lead	ICPES	86
Magnesium	ICPES	95
Magnesium	ICPES	98
Magnesium	ICPES	94
Magnesium	ICPES	95
Magnesium	ICPES	87
Magnesium	ICPES	82
Magnesium	ICPES	86
Magnesium	ICPES	98
Magnesium	ICPES	95
Magnesium	ICPES	89
Manganese	ICPES	94
Manganese	ICPES	95
Manganese	ICPES	96
Manganese	ICPES	94
Manganese	ICPES	92
Manganese	ICPES	93
Manganese	ICPES	94
Manganese	ICPES	96
Molvbdenum	ICPES	96
Molybdenum	ICPES	91
Molybdenum	ICPES	91
Molybdenum	ICPES	96
Molybdenum	ICPES	82
Molybdenum	ICPES	92
Molybdenum	ICPES	92
Molybdenum	ICPES	81
Nickel	ICPES	91
Nickel	ICPES	83
Nickel	ICPES	96
Nickel	ICPES	90
Nickel	ICPES	80
Nickel	ICPES	96
Nickel	ICPES	95
Nickel	ICPES	94
Phosphorus	ICPES	91
Phosphorus	ICPES	112
Phosphorus	ICPES	85
Phosphorus		110
Phosphorus	ICPES	96
Phosphorus	ICPES	83
Phosphorus	ICPES	109
Phosphorus	ICPES	93
Phosphorus	ICPES	87
Phosphorus	ICPES	86
Potassium	ICPES	91
Potassium	ICPES	94
Potassium	ICPES	95

.

-

•

		Recovery
Analyte	Method code	(%)
Potassium	ICPES	94
Potassium	ICPES	92
Potassium	ICPES	91
Potassium	ICPES	93
Potassium	ICPES	94
Selenium	ICPES	5
Selenium	ICPES	174
Selenium	ICPES	94
Selenium	ICPES	99
Selenium	ICPES	94
Selenium	ICPES	95
Selenium	ICPES	43
Selenium	ICPES	209
Silicon	ICPES	98
Silicon	ICPES	99
Silicon	ICPES	90
Silicon	ICPES	102
Silicon	ICPES	95
Silicon	ICPES	84
Silicon	ICPES	98
Silicon	ICPES	98
Silver	ICPES	64
Silver	ICPES	43
Silver	ICPES	93
Silver	ICPES	94
Silver	ICPES	62
Silver	ICPES	94
Silver	ICPES	26
Silver	ICPES	94
Sodium	ICPES	94
Sodium	ICPES	95
Sodium	ICPES	97
Sodium	ICPES	95
Sodium	ICPES	99
Sodium	ICPES	97
Sodium	ICPES	97
Sodium	ICPES	99
Sodium	ICPES	97
Sodium	ICPES	94
Strontium	ICPES	97
Strontium	ICPES	98
Strontium	ICPES	97
Strontium	ICPES	92
Strontium	ICPES	94
Strontium	ICPES	97
Strontium	ICPES	95
Strontium	ICPES	93
Thallium	ICPES	96
Thallium	ICPES	94
Thailium	ICPES	91
Thallium	ICPES	97

		Recovery
Analyte	Method code	(%)
Thallium	ICPES	91
Thallium	ICPES	101
Thailium	ICPES	88
Thallium	ICPES	92
Thailium	ICPES	96
Thallium	ICPES	85
Titanium	ICPES	96.4
Titanium	ICPES	92
Titanium	ICPES	90
Titanium	ICPES	96
Vanadium	ICPES	92
Vanadium	ICPES	94
Vanadium	ICPES	96
Vanadium	ICPES	86
Vanadium	ICPES	92
Vanadium	ICPES	88
Vanadium	ICPES	93
Vanadium	ICPES	96
Zinc	ICPES	92
Zinc	ICPES	95
Zinc	ICPES	89
Zinc	ICPES	9 6
Zinc	ICPES	93
Zinc	ICPES	89
Zinc	ICPES	91
Zinc	ICPES	94
Arsenic	GFAAS	97
Arsenic	GFAAS	106
Arsenic	GFAAS	99
Arsenic	GFAAS	109
Arsenic	GFAAS	102
Arsenic	GFAAS	95
Arsenic	GFAAS	99
Arsenic	GFAAS	98
Cadmium	GFAAS	112
Cadmium	GFAAS	121
Cadmium	GFAAS	99
Cadmium	GFAAS	102
Cadmium	GFAAS	122
Cadmium	GFAAS	103
Cadmium	GFAAS	114
Cadmium	GFAAS	100
Chromium	GFAAS	98
Chromium	GFAAS	102
Chromium	GFAAS	98
Chromium	GEAAS	98
	GEAAS	96
Cnromium	GFAAS	96
Lead	GEAAS	93
Lead	GRAAS	105

.

		Recovery
Analyte	Method code	(%)
Lead	GFAAS	91
Lead	GFAAS	129
Lead	GFAAS	104
Lead	GFAAS	98
Lead	GFAAS	92
Lead	GFAAS	93
Lead	GFAAS	107
Lead	GFAAS	104
Nickel	GFAAS	99
Nickel	GFAAS	101
Nickei	GFAAS	96
Nickel	GFAAS	101
Nickel	GFAAS	101
Nickel	GFAAS	108
Nickel	GFAAS	91
Nickel	GFAAS	92
Nickel	GFAAS	103
Selenium	GFAAS	95
Selenium	GFAAS	89
Selenium	GFAAS	98
Selenium	GFAAS	90
Selenium	GFAAS	90
Selenium	GFAAS	94
Selenium	GFAAS	94
Mercury	CVAAS	93
Mercury	CVAAS	94
Mercury	CVAAS	111
Mercury	CVAAS	112
Mercury	CVAAS OVAAS	111
Mercury	CVAAS	90
Mercury	CVAAS	132
Mercury	CVAAS	104
Mercury	CVARS	100
Mercury		107
Mercury		107
Mercury	01003	104
Fluoride	EPA 340.2	91
Fluoride	EPA 340.2	97
Fluoride	EPA 340.2	88
Fluoride	EPA 340.2	95
Fluoride	EPA 340.2	90
Fluoride	EPA 340.2	91
Fluoride	EPA 340.2	90
Fluoride	EPA 340.2	93
Fluoride	EPA 340.2	86
Fluoride	EPA 340.2	91
Fluoride	EPA 340.2	98

Table F.2-5. Laboratory Control Sample Recovery Data for Metals and Anions

•

		Recovery
Analyte	Method code	(%)
Fluoride	EPA 340.2	95
Fluoride	EPA 340.2	97
Fluoride	EPA 340.2	71
Fluoride	EPA 340.2	92
Fluoride	EPA 340.2	94
Fluoride	EPA 340.2	93
Fluoride	EPA 340.2	73
Hexavalent chromium	Cr(VI) by BIF METHOD	98
Hexavalent chromium	Cr(VI) by BIF METHOD	92
Chloride	Potentiometric	100
Chloride	Potentiometric	100
Chloride	ion Chromatography	103
Chloride	ion Chromatography	100
Chloride	Ion Chromatography	98
Chloride	Ion Chromatography	102
Chloride	Ion Chromatography	102
Chloride	ion Chromatography	90
Chloride	lon Chromatography	99
Chloride	lon Chromatography	101
Chloride	SM 4500 CI B	93
Chloride	SM 4500 CI B	99
Sulfur	Total sulfur	95
Sulfur	Total sulfur	92
Sulfate	Sulfate, EPA 300.0	97
Sulfate	Sulfate, EPA 300.0	97
Sulfate	Sulfate, EPA 300.0	96
Suifate	Suifate, EPA 300.0	96
Sulfate	Sulfate, EPA 300.0	96
Sulfate	Sulfate, EPA 300.0	96
Sulfate	Sulfate, EPA 300.0	95
Sulfate	Sulfate, EPA 300.0	96

Method Results (% Rec.) (% Rec.) <t< th=""><th>Summary</th><th></th><th>No. of</th><th>Mean</th><th>Min</th><th>Max</th></t<>	Summary		No. of	Mean	Min	Max
Aluminum ICPES 8 94 83 96 Antimony ICPES 10 93 88 96 Antimony ICPES 8 64 21 97 Barium ICPES 8 97 93 99 Barium ICPES 8 97 93 99 Cadmium ICPES 8 97 93 99 Cadmium ICPES 8 96 85 103 Calcium ICPES 8 96 85 103 Chromium ICPES 8 94 91 96 Cobait ICPES 8 94 91 96 Copper ICPES 8 94 92 96 Marganese ICPES 10 92 82 96 Marganese ICPES 8 90 81 96 Nickel ICPES 8 91 80 96 Nickel ICPES 8 92 96 Nickel		Method	Results	(% Rec)	(% Rec)	(% Rec)
Antimony ICPES 10 93 86 98 Arsenic ICPES 8 64 21 97 Barum ICPES 8 97 93 99 Baryum ICPES 8 97 96 99 Cadmium ICPES 4 97 96 99 Cadmium ICPES 8 86 68 95 Calcium ICPES 8 93 91 96 Chromium ICPES 8 94 91 96 Cobait ICPES 8 96 92 100 Lead ICPES 8 96 92 100 Lead ICPES 8 90 81 96 Magnesium ICPES 8 91 95 3112 Potassium ICPES 8 91 95 3112 Potassium ICPES 8 92 96 3112	Aluminum	ICPES	8	94	83	98
Arsenic ICPES 8 64 21 97 Barium ICPES 10 90 80 99 Baryllium ICPES 10 90 80 99 Boron ICPES 4 97 96 99 Boron ICPES 4 97 96 99 Cadmium ICPES 8 86 68 103 Calcium ICPES 8 96 85 103 Cobait ICPES 8 94 91 96 Icon ICPES 8 94 91 96 Icon ICPES 8 94 91 96 Icon ICPES 8 96 92 100 Lead ICPES 8 96 92 96 Marganese ICPES 8 90 81 96 Plosphorus ICPES 8 91 80 96 <t< td=""><td>Antimony</td><td>ICPES</td><td>10</td><td>93</td><td>88</td><td>98</td></t<>	Antimony	ICPES	10	93	88	98
Banum ICPES 8 97 93 99 Baryllium ICPES 10 90 80 99 Boron ICPES 4 97 96 99 Cadicium ICPES 8 96 65 103 Calcium ICPES 8 93 91 96 Cabit ICPES 8 94 91 96 Cobalt ICPES 8 94 91 96 Iron ICPES 8 94 91 96 Iron ICPES 10 92 82 98 Magnesium ICPES 10 92 82 98 Matgnesium ICPES 8 91 80 96 Nickel ICPES 8 91 80 96 Phosphorus ICPES 8 91 95 83 112 Sodium ICPES 8 92 89 <	Arsenic	ICPES	8	64	21	97
Beryllium ICPES 10 90 80 99 Boron ICPES 4 97 96 99 Cadmium ICPES 8 86 68 95 Calcium ICPES 8 96 85 103 Chromium ICPES 8 93 91 96 Cobalt ICPES 8 93 91 96 Cobalt ICPES 8 94 91 96 Copper ICPES 8 94 91 96 Iron ICPES 8 94 92 96 Maganese ICPES 8 94 92 96 Molybdenum ICPES 8 90 81 96 Plosphorus ICPES 8 91 95 96 Nickel ICPES 8 93 91 95 Silicon ICPES 8 96 84 102 </td <td>Barium</td> <td>ICPES</td> <td>8</td> <td>97</td> <td>93</td> <td>99</td>	Barium	ICPES	8	97	93	99
Boron ICPES 4 97 96 99 Cadmium ICPES 8 86 68 95 Calcium ICPES 8 96 85 103 Chomium ICPES 8 94 91 96 Cobait ICPES 8 94 91 96 Cobait ICPES 8 94 91 96 Copper ICPES 8 94 91 96 Magnesium ICPES 10 92 82 98 Magnesium ICPES 10 92 82 98 Magnesium ICPES 8 90 81 96 Nickel ICPES 8 91 80 96 Nickel ICPES 8 93 91 95 Selenium ICPES 8 93 91 95 Silver ICPES 8 91 92 92 <td>Beryllium</td> <td>ICPES</td> <td>10</td> <td>90</td> <td>80</td> <td>99</td>	Beryllium	ICPES	10	90	80	99
Cadimum ICPES 8 86 68 95 Calcium ICPES 8 96 85 103 Chromium ICPES 8 93 91 96 Cobait ICPES 8 93 91 96 Cobait ICPES 8 94 91 96 Cobait ICPES 8 94 91 96 Iron ICPES 8 94 92 96 Magnesium ICPES 8 94 92 96 Marganese ICPES 8 90 81 96 Nickel ICPES 8 91 80 96 Nickel ICPES 8 93 91 95 Selenium ICPES 8 96 84 92 Solum ICPES 8 96 84 92 Silicon ICPES 8 95 92 96	Boron	ICPES	4	97	96	99
Calcium ICPES 8 96 85 103 Chromium ICPES 8 93 91 96 Cobait ICPES 8 89 80 95 Copper ICPES 8 94 91 96 Iron ICPES 8 94 91 96 Iron ICPES 8 94 91 96 Iron ICPES 8 94 92 96 Magnesium ICPES 8 94 92 96 Malgnesium ICPES 8 90 81 96 Nickel ICPES 8 91 80 96 Nickel ICPES 8 91 95 53 112 Potassium ICPES 8 92 50 94 99 Silicon ICPES 8 96 84 102 5 209 Siliver ICPES <td< td=""><td>Cadmium</td><td>ICPES</td><td>8</td><td>86</td><td>68</td><td>95</td></td<>	Cadmium	ICPES	8	86	6 8	95
Chromium ICPES 8 93 91 96 Cobait ICPES 8 89 80 95 Copper ICPES 8 94 91 96 Iron ICPES 8 94 91 96 Iron ICPES 8 94 91 96 Magnesium ICPES 10 92 82 98 Magnesium ICPES 8 94 92 96 Molybdenum ICPES 8 91 80 96 Nickel ICPES 8 91 80 96 Potassium ICPES 8 91 95 83 112 Potassium ICPES 8 92 52 209 5 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 95 92 98 5 10 93 <td< td=""><td>Calcium</td><td>ICPES</td><td>8</td><td>96</td><td>85</td><td>103</td></td<>	Calcium	ICPES	8	96	85	103
Cobalt ICPES 8 89 80 95 Copper ICPES 8 94 91 96 Iron ICPES 8 96 92 100 Lead ICPES 10 89 75 98 Magnarese ICPES 10 92 82 98 Magnarese ICPES 10 92 82 98 Magnarese ICPES 8 94 92 96 Molybdenum ICPES 8 91 80 96 Nickel ICPES 8 91 95 83 112 Potassium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 Silicon ICPES 8 96 84 102 Silicon ICPES 8 95 92 98 Stortium ICPES 8 92 96	Chromium	ICPES	8	93	91	96
Copper ICPES 8 94 91 96 Iron ICPES 8 96 92 100 Lead ICPES 10 89 75 98 Magnesium ICPES 10 92 82 98 Magnesium ICPES 8 94 92 96 Molybdenum ICPES 8 91 96 Nickei ICPES 8 91 96 Nickei ICPES 8 91 96 Potassium ICPES 8 91 95 Selenium ICPES 8 92 5 Selenium ICPES 8 92 5 Silicon ICPES 8 95 92 98 Silicon ICPES 8 95 92 98 Sodium ICPES 8 92 86 96 Sodium ICPES 8 92 86	Cobait	ICPES	8	89	80	95
Iron ICPES 8 96 92 100 Lead ICPES 10 89 75 98 Magnesium ICPES 10 92 82 98 Magnesium ICPES 8 90 81 96 Molybdenum ICPES 8 90 81 96 Phosphorus ICPES 8 91 80 96 Phosphorus ICPES 8 93 91 95 Selenium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 Silicon ICPES 8 96 84 102 Silicon ICPES 8 92 98 71 26 94 Sodium ICPES 8 95 92 98 76 10 Silicon ICPES 8 92 86 96 101 101 102	Copper	ICPES	8	94	91	96
Lead ICPES 10 89 75 98 Magnesium ICPES 10 92 82 98 Manganese ICPES 8 94 92 96 Molybdenum ICPES 8 90 81 96 Nickei ICPES 8 91 80 96 Phosphorus ICPES 8 91 80 96 Potassium ICPES 8 91 95 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 95 92 98 96 101 90 96 Vanadium ICPES 8 92 86 96 2102 2106 93 102	iron	ICPES	8	96	92	100
Magnessium ICPES 10 92 82 98 Manganese ICPES 8 94 92 96 Molybdenum ICPES 8 90 81 96 Nickei ICPES 8 90 81 96 Phosphorus ICPES 8 91 80 96 Phosphorus ICPES 8 91 95 83 112 Potassium ICPES 8 93 91 95 209 Selenium ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Sodium ICPES 8 95 92 96 94 99 95 5 101 93 85 101 101 93 85 101 101 93 85 101 101 93 96 2 102 101	Lead	ICPES	10	89	75	98
Manganese ICPES 8 94 92 96 Molybdenum ICPES 8 90 81 96 Nickel ICPES 8 91 80 96 Phosphorus ICPES 10 95 83 112 Potassium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 Silicon ICPES 8 96 84 102 Silicon ICPES 8 96 84 102 Sodium ICPES 8 91 99 95 Strontium ICPES 8 95 92 98 Thallium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 86 96 Zinc ICPES 8 101 95 109 </td <td>Magnesium</td> <td>ICPES</td> <td>10</td> <td>92</td> <td>82</td> <td>98</td>	Magnesium	ICPES	10	92	82	9 8
Molybdenum ICPES 8 90 81 96 Nickei ICPES 8 91 80 96 Phosphorus ICPES 10 95 83 112 Potassium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silver ICPES 8 95 92 98 71 26 94 Sodium ICPES 8 95 92 98 71 76 94 99 Stronturn ICPES 8 95 92 98 70 96 Vanadium ICPES 8 92 86 96 102 Zinc ICPES 8 101 <	Manganese	ICPES	8	94	92	96
Nickei ICPES 8 91 80 96 Prosphorus ICPES 10 95 83 112 Potassium ICPES 8 93 91 95 Selenium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 294 Sodium ICPES 8 91 96 94 99 Strontium ICPES 10 93 85 101 11 Titanium ICPES 4 94 90 96 2 96 Arsenic AAS 8 101 95 109 2 2 Chromium AAS 10 102 91 129 96 102	Molybdenum	ICPES	8	90	81	96
Phosphorus ICPES 10 95 83 112 Potassium ICPES 8 93 91 95 Selenium ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 96 84 102 5 209 Silicon ICPES 8 971 26 94 99 Strontium ICPES 10 96 94 99 Strontium ICPES 10 93 85 101 Thallium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 86 96 Zinc ICPES 8 109 99 122 Chromium AAS 10 102 91 129 <td>Nickel</td> <td>ICPES</td> <td>8</td> <td>91</td> <td>80</td> <td>96</td>	Nickel	ICPES	8	91	80	96
Potassium ICPES 8 93 91 95 Selenium ICPES 8 102 5 209 Silicon ICPES 8 96 84 102 Silver ICPES 8 96 84 102 Sodium ICPES 8 71 26 94 Sodium ICPES 8 95 92 98 Thallium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 4 94 90 96 Zinc ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 6 98 96 102 Lead AAS 10 102 91 129	Phosphorus	ICPES	10	95	83	112
Selenium ICPES 8 102 5 209 Silicon ICPES 8 96 84 102 Silver ICPES 8 96 84 102 Sodium ICPES 8 71 26 94 Sodium ICPES 8 71 26 94 Sodium ICPES 8 95 92 98 Thallium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 100 99 122 Chromium AAS 10 102 91 129 Mercury AAS 10 92 99 91 <td>Potassium</td> <td>ICPES</td> <td>8</td> <td>93</td> <td>91</td> <td>95</td>	Potassium	ICPES	8	93	91	95
Silicon ICPES 8 96 84 102 Silver ICPES 8 71 26 94 Sodium ICPES 8 71 26 94 Sodium ICPES 8 95 92 98 Strontium ICPES 8 95 92 98 Thallium ICPES 8 92 85 101 Ttanium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Zinc ICPES 8 92 89 96 Zinc ICPES 8 91 99 91 122 Chromium AAS 6 98 96 102 129 Mercury AAS 10 102 91 129 Mercury AAS 9 99 91<	Selenium	ICPES	8	102	5	209
Silver ICPES 8 71 26 94 Sodium ICPES 10 96 94 99 Strontium ICPES 8 95 92 98 Thallium ICPES 8 95 92 98 Titanium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Selenium AAS 10 92 89 98 <	Silicon	ICPES	8	96	84	102
Sodium ICPES 10 96 94 99 Strontium ICPES 8 95 92 98 Thallium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 10 102 91 129 Mercury AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98	Silver	ICPES	8	71	26	94
Strontium ICPES 8 95 92 98 Thallium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 <td>Sodium</td> <td>ICPES</td> <td>10</td> <td>96</td> <td>94</td> <td>99</td>	Sodium	ICPES	10	96	94	99
Thallium ICPES 10 93 85 101 Titanium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluonide 340.2 18 90 71 98 Sulfate IC 8 96 95 97 Sulfate IC 2 94 92 95	Strontium	ICPES	8	95	92	98
Titanium ICPES 4 94 90 96 Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Thallium	ICPES	10	93	85	101
Vanadium ICPES 8 92 86 96 Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 10 102 91 129 Mercury AAS 10 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Sulfate IC 8 96 95 97 Sulfate IC 8 96 95 97 Sul	Titanium	ICPES	4	94	90	96
Zinc ICPES 8 92 89 96 Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 10 102 91 129 Mercury AAS 10 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Sulfate IC 8 96 95 97 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Vanadium	ICPES	8	92	86	96
Arsenic AAS 8 101 95 109 Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fiuoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Zinc	ICPES	8	92	89	96
Cadmium AAS 8 109 99 122 Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Arsenic	AAS	8	101	95	109
Chromium AAS 6 98 96 102 Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Cadmium	AAS	8	109	9 9	122
Lead AAS 10 102 91 129 Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Chromium	AAS	6	98	96	102
Mercury AAS 12 108 93 132 Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Lead	AAS	10	102	91	129
Nickel AAS 9 99 91 108 Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Mercury	AAS	12	108	93	132
Selenium AAS 10 92 89 98 Chloride IC/Pot. 12 99 90 103 Fluoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Nickel	AAS	9	99	91	108
Chloride IC/Pot. 12 99 90 103 Fiuoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Selenium	AAS	. 10	92	89	98
Fiuoride 340.2 18 90 71 98 Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Chloride	IC/Pot.	12	99	90	103
Hexavalent chromium Cr (IV) 2 95 92 98 Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Fiuoride	340.2	18	90	71	98
Sulfate IC 8 96 95 97 Sulfur Elem. Analyzer 2 94 92 95	Hexavalent chromium	Cr (IV)	2	95	92	98
Sulfur Elem. Analyzer 2 94 92 95	Sulfate	IC	- R	96	95	97
	Sulfur	Elem. Analyzer	2	94	92	95

Table F.2~5. Laboratory Control Sample Recovery Data for Metals and Anions

	LCS	LCSD	Mean	RPD
Analyte	(% Rec)	(% Rec)	(% Rec)	(% Rec)
Antimony	110	111	110	0.5
Arsenic	9 5	93	94	1.5
Barium	101	103	102	1.8
Beryllium	96	93	95	4.0
Cadmium	98	97	97	1.4
Chromium	97	101	99	3.8
Cobait	103	99	101	. 3.9
Copper	91	92	92	0.8
Lead	103	97	100	5.5
Manganese	100	102	101	2.2
Mercury	NA	NA	NA	NA
Molybdenum	104	95	99	8.2
Nickei	90	95	92	4.8
Selenium	94	94	94	0.7
Vanadium	108	106	107	1.2

Table F2-6. ICPMS Laboratory Control Sample Recovery Data for Metals

					WS	MSD	Mean	RPD
	Andres	Method	Samole	Run	(% Rec)	(% Rec)	(% Rec)	(%)
sample IU	Alldiyle		Dottom sch	20	80	64	72	22.2
H-294 MSD	Aluminum	ICP analysis by SW6U1U		ŝ	BA	102	18	19.4
H-294 MSD	Antimony	ICP analysis by SW6U10		9 6	106		tot Rot	- a
H-294 MSD	Arsenic	Arsenic by SW7060	Bottom asn	5 5	8	53		
H-294 MSD	Barium	ICP analysis by SW6010	Bottom ash	20	7.9		8	2
H_204 MSD	Bervílium	ICP analysis by SW6010	Bottom ash	20	87	84	86	ດ ເ
	Cadmittm	Cadmium by SW7131	Bottom ash	20	105	94	50	1.1
	Cadmium	Cadmium by SW7131	Bottom ash	2D	106	104	105	1.9
H-294 Mou	Caloinen	CD analysis hv SW6010	Bottom ash	20	102	82	92	21.7
H-294 MSU	Calcium	Chloride notentiometric	Bottom ash	20	66	97.7	96	1.3
H-294 MSU	Chromium	Undrate potentiamento ICD analysis hy SW6010	Bottom ash	2D	<u>9</u>	60	9	
H-294 MSU	Circumum	ICD analysis by SW6010	Bottom ash	20	68	91	6	2.2
H-294 MSU	Condit	ICD analysis by SW6010	Bottom ash	2D	92	91	92	. :
H294 MSU		Elitorido hy EPA 340 9	Bottom ash	2D	7	29	50	84.0
H-294 MSU	Fluoride	Fluoride by EPA 340 9	Rottom ash	2D	88	81	85	8.3
H-294 MSU		ICD analysis by SW6010	Bottom ash	2D	103	91	97	12.4
H-294 MSU			Rottom ash	2D	3 8	96	97	2
H-294 MSD	Lead	LEBU DY 3777421	Bottom ash	2	102	69	86	38.6
H-294 MSD	magnesum	ICF allarysis by Swoots ICF analysis by SW6010	Bottom ach	2D	91	06	91	1
H~294 MSD	Manganese				26	67	97	0.0
H-294 MSD	Mercury	Mercury, cold vapor SW/4/1		9 6	ŝ	104	103	•
H-294 MSD	Molybdenum	ICP analysis by SW6010	Bottom ash	5		3 5	3 8	- 0
H-294 MSD	Nickel	Nickel by GF, EPA 249.2	Bottom ash	2 (5 U 0 (5		4 C
H-294 MSD	Phosphorus	ICP analysis by SW6010	Bottom ash	20	95	ŝ	6 I 6	2.0
	Potassium	ICP analysis by SW6010	Bottom ash	20	68	S 2	2	4
	Salanium	Selenium by SW7740	Bottom ash	2	84	88	86	4.7
	Silicon	CP analysis by SW6010	Bottom ash	g	83	82	83	<u>н</u>
H-294 MSU	Cilitar	ICD analysis by SW6010	Bottom ash	2D	43	61	52	34.6
H-294 MSU	Suver	ICP analysis by SW6010	Bottom ash	2D	8 6	88	93	10.8
H-294 MSU	Soundru Presentium	ICD analysis by SW6010	Bottom ash	2D	80	17	79	3.8
H-294 MSU	The Nation	ICD analysis by SW6010	Bottom ash	20	91	88	66	3.4
	Variation	ICD analysis by SW6010	Bottom ash	2D	91	92	92	
H-294 MSU		ICD analysis by Credits	Bottom ash	20	9 6	85	98	– N
H~294 MSD		ICF allalysis by arresto						
	Chlorida	Chloride by SM 4500 CI B	Coal	0	95	97	96	<u>1</u>
	Elucido	Elitorida hv EPA 340 2	Coal	~	6	82.2	87	10.2
	Chorido	Fluctide hv EPA 340 2	Coal	2	49	60	55	20.2
HZU1//ZUZ MOU	Anionia			c	99	69	68	4.4
H201/202 MSD	Fluoride	Fluoride by EPA 340.2	0081	J	3	8	}	
	Aliminum	ICD analysis hy SW6010	Stack das. filt+PNR	ب	84	06	87	6.9
H302/H217/H210		ICF analysis vy vrvvrv ICB analysis hv SW6010	Stack das, imps 1&2	2	06	90.4	06	0.4
H-213 MSU	Aluminum	ICF allarysis by creeks						

Table F2-7. Matrix Spike Recovery Data for Metals and Anions

					MS	MSD	Меал	DO
Samule ID	Analyte	Method	Sample	Run	(% Rec)	(% Rec)	(% Rec)	(%)
H_944 MSD	Aluminum	ICP analysis by SW6010	ESP inlet gas, imps 1&2	~	9 3	92.1	93	0. F
	Aluminum	ICP analysis by SW6010	Stack das, imps 1&2	~	0 6	88.1	68	2.1
	Aluminum	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	92	91.2	92	0.9
	Aluminum	ICP analysis by SW6010	ESP inlet gas, th/PCR	2	101	10	101	1.0
	Antimony	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	87	87.4	87	0.5
	Antimony	ICP analysis by SW6010	ESP inlet das, imps 1&2	8	95	88	3 5	7.7
	Antimony	ICP analysis by SW6010	ESP inlet gas, th/PCR	2	96	95	96	1.0
	Antimony	ICP analysis by SW6010	Stack gas, imps 1&2	0	69	87.5	88	1.7
USW 612 - 1	Antimony	ICP analysis by SW6010	Stack gas, imps 1&2	0	06	89.1	6	1.0
	Antimony	ICP analysis by SW6010	Stack gas, filt+PNR	e 9	80	82	81	2.5
	Arsenic	Arsenic by SW7060	Stack gas, imps 1&2	2	93	94.2	94	1.3
H113/H949 D MSD	Arsenic	BIF As for Metals Trains	ESP inlet gas, th/PCR	2	100	95	86	5.†
H300/H017/H018 C	Arsenic	BIF As for Metals Trains	Stack gas, filt+PNR	e	94	96	95	5. 1. 1.
H-PAA MSD	Arsenic	Arsenic by SW7060	ESP inlet gas, imps 1&2	0	95	95	95	0.0
	Barium	ICP analysis by SW6010	Stack gas, filt+PNR	e	73	06	82	20.9
	Barium	ICP analysis by SW6010	ESP Inlet gas, th/PCR	0	109	107	108	1.9
H - 913 MSD	Barium	ICP analysis by SW6010	Stack gas, imps 1&2	2	85	87.2	86	2.6
H_013 MSD	Barium	ICP analysis by SW6010	Stack gas, imps 1&2	0	83	84,4	84	1.7
	Barium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	61	92	91.2	60	0.0
	Barium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	91	90.5	91	0.6
	Bervilium	BIF ICP for Metals Trains	Stack gas, tilt+PNR	e	6	6	6	0.0
H-244 MSD	Bervllium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	~	96	95.4	96	0.6
	Bervllium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	94	94.5	94	0.5
	Bervllium	ICP analysis by SW6010	Stack gas, imps 1&2	2	95	95	95	0.0
	Bervlium	BIF ICP for Metals Trains	ESP Inlet gas, th/PCR	2	69	68	68	0.0
H_013 MSD	Bervlium	ICP analysis by SW6010	Stack gas, imps 1&2	2	96	93.3	8	, 8
	Boron	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	84	80.2	62	4.6
H-244 MSD	Boron	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	82	80.9	81	*
H-213 MSD	Boron	ICP analysis by SW6010	Stack gas, imps 1&2	2	83	73	8/	12.6
H-213 MSD	Boron	ICP analysis by SW6010	Stack gas, imps 1&2	ຸ	80	23	2	30.2
H302/H917/H918 C	Cadmium	BIF Cd for Metals Trains	Stack gas, filt+PNR	ლ	86	97	86	0.
H - 360 MS	Cadmium	Cadmium by SW7131	ESP inlet gas, imps 1&2	4	108	109	109	0.9
H113/H249 D MSD	Cadmium	BIF Cd for Metals Trains	ESP Inlet gas, th/PCR	2	109	116	113	6.2
H - 213 MSD	Cadmium	Cadmium by SW7131	Stack gas, imps 1&2	2	109	112	Ŧ	2.7
HIII/HOAD MSD	Calcium	ICP analysis by SW6010	ESP inlet gas, th/PCR	2	93	92	8	-
	Calcium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	~	86	6 .96	67	
	Calcium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	0	66	8 6	66	,
H-213 MSD	Calcium	ICP analysis by SW6010	Stack gas, imps 1&2	8	96	94	96	0 -
H-213 MSD	Calcium	ICP analysis by SW6010	Stack gas, imps 1&2	8	96	96.4	<u> 36</u>	0 .4

.

Table F2-7. Matrix Spike Recovery Data for Metals and Anions

					WS	MSD	Mean	PD
	Analyte	Method	Sample	Bun (% Rec)	(% Rec)	(% Rec)	(%)
	Calchim	ICP analysis by SW6010	Stack gas, filt+PNR	0	85	98	86	1.2
	Chlorida	Chloride, by IC EPA300	ESP injet gas, th/PCR/TLR	e 0	100	100	100	0.0
	Chloride	Chloride, by IC EPA300	ESP inlet gas, th/PCR/TLR	2	114	103	109	0
01-11-203-L1-10	Chloride	Chloride, by IC EPA300	Stack gas, fift+PNR	2	108	106	107	6 .1
	Chloride	Chloride, by IC EPA300	Stack gas, anion imps	2	94	95.3	95	1.4
	Chloride	Chloride, by IC EPA300	ESP inlet gas, anion imps	2	100	6	95	10.5
	Chromium	Chromium by GF - SW7191	ESP inlet gas, imps 1&2	8	110	106	108	3.7
	Chrombum	Chromium by GF – SW7191	Stack gas, imps 1&2	8	104	96.8	100	7.2
	Chromium (VI)	Cr(VI) by BIF METHOD	Reag bink, KOH imps	RB BB	111	105	108	5.6
	Chromium (VI)	Cr(VI) by BIF METHOD	Stack gas, KOH imps	9	6	<u>6</u>	95	10.5
	Cobalt	ICP analysis by SW6010	Stack gas, imps 1&2	2	68	89.3	68	0.0
	Cobalt	ICP analysis by SW6010	Stack gas, imps 1&2	0	88	87.5	88	0.6
	Cobalt	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	91	89.9	6	- -
	Cobalt	BIF ICP for Metals Trains	ESP inlet gas, th/PCR	2	6	92	63	2.2
	Cobalt	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	68	88.4	68	0.7
MD17/H218 C	Cohalt	BIFICP for Metals Trains	Stack gas, filt+PNR	ი	88	88	88	0.0
	Conner	ICP analysis by SW6010	Stack gas, imps 1&2	2	68	87.8	88	-
	Conner	BIFICP for Metals Trains	ESP Inlet gas, th/PCR	2	94	7 6	8	0.0
	Conner	ICP analysis by SW6010	ESP inlet gas, imps 1&2	0	6	89.5	6	9.0 0
	Conner	ICP analysis by SW6010	Stack gas, imps 1&2	2	6	89.8	6	0.2
/H917/H918 C	Conner	BIF ICP for Metals Trains	Stack gas, filt+PNR	e	<u>10</u>	66	9	0. 2
	Copper	ICP analysis by SW6010	ESP Inlet gas, imps 1&2	0	91	6	91	-
	Fluoride	Fluoride by EPA 340.2	ESP intet gas, anion imps	2	6 3	98.1	96	ຕ ທີ່
	Fluoride	Fluoride by EPA 340.2	ESP intet gas, anion imps	2	102	94.1	86	8
0 H - 193 COMP	Fluoride	Fluoride by EPA 340.2	Stack gas, filt+PNR	N	82	88.9	87	4.5
0 H=209.H-16	Fluoride	Fluoride by EPA 340.2	ESP inlet gas, th/PCR/TLR	2	6	63	65	0 0
9 MSD	Fluoride	Fluoride by EPA 340.2	Stack gas, anion imps	2	96 1	93.5 	95 	
H217/H218 C	Iron	ICP analysis by SW6010	Stack gas, filt+PNR	ē,	74	62	12	0
H249 D MSD	Iron	ICP analysis by SW6010	ESP inlet gas, th/PCR	~	92	88	6	4
a MSD	Iron	ICP analysis by SW6010	Stack gas, imps 1&2	2	61	96.4	97	9.0
	Iron	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	66	98.8	66	0.0
	l ead	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	68	90.6	8	7
	Load Load	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	6	87.6	68	2.7
	Lead	Lead by SW7421	ESP inlet gas, imps 1&2	8	97	95,6	96	- 2
		RIF Pb for Metals Trains	Stack gas, filt+PNR	e	108	105	107	89. (V)
	Lead	Lead by SW7421	ESP inlet gas, imps 1&2	4	65	109	87	50.6
	Lead	Lead by SW7421	Stack das, imps 1&2	2	102	100 1	101	8 0 0
	Leau Load	RIF Ph for Metals Trains	ESP Inlet gas, th/PCR	5	8 6	9 2	97	3.1
A249 L Mou	Magnesium	ICP analysis by SW6010	Stack gas, imps 1&2	ଷ	68	87.8	88	4.4
0 1000	in a subscription of the s		•					

.

Table F2-7. Matrix Spike Recovery Data for Metals and Anions

.

.

Appendix F: QA/QC Results

					MS	MSD	Mean	RPD
			Sample	Run	(% Rec)	(% Rec)	(% Rec)	(%)
Sample ID	Analyte	merrou 	ESP inter ras imps 162	2	60	91.6	92 92	1.5
H-244 MSD	Magnesium	ICP analysis by avvouro	COLUMN STATE	0	96	5	5	0.0
H302/H217/H218 C	Magnesium	BIF ICP for metals intains	Clack as imported	0	06	6	0 6	0.0
H-213 MSD	Magnesium	ICP analysis by SWOUTU	ciach gas, mpo ton Eod intot gas th/PCR	•	6	94	94	0.0
H113/H249 D MSD	Magnesium	BIF ICP for Metals Trams	EST mer yas, we con ESD into the time 182		92	90.7	91	1,4
H-244 MSD	Magnesium	ICP analysis by SWOULD	EST milet gas, important		9	89.8	6	1,3
H-244 MSD	Manganese	ICP analysis by Swould	ESP inter gas,po test ESP inter rae, imps 1&2	I (N	6	88.5	69	1.7
H-244 MSD	Manganese	ICP analysis by avour o	Stark ras filt+PNR	, eo	82	8	63	2.4
H302/H217/H218 C	Manganese	ICP analysis by 200010	COD INTO and The PCB		92	8	91	2.2
H113/H249 D MSD	Manganese	ICP analysis by Sw6010	Cor lillet gas, up of		68	89.6	89	0.7
H-213 MSD	Manganese	ICP analysis by Swould	otach gas, imperior		88	87.2	88	0.9
H-213 MSD	Manganese	ICP analysis by Swould		1 00	112	126	119	11.8
H302/H217/H218 C	Mercury	Mercury, cold vapor SW/4/1	EDD into and impe 3 4 5	<u>م</u> ا	20	50	65	0.0
H-245 MSD	Mercury	Mercury, cold vapor SW/4/0		10	62	69	99	10.7
H-214 MSD	Mercury	Mercury, cold vapor SW/4/0	ctack gas, mige divide ECP intel as imps 182	1	104	106	105	6 .1
H-244 MSD	Mercury		ESP intel rate th/PCR	0	130	132	131	1.5
H113/H249 D MSD	Mercury	Mercury, cold Vapor SW/4/1	ctark nee imms 3.4.5	2	54	51	53	5.7
H-214 MSD	Mercury	Mercury, cold Vapor SW/4/0	ECD into rac tmps 3.4.5	2	83	71	77	15.6
H-245 MSD	Mercury	Mercury, cold vapor SW/4/0	Clark as impacted	2	78	85	82	8.6
H-213 MSD	Mercury	Mercury, cold Vapor SW/4/0	otach yes, migo	. 0	80	84	82	4.9
H-215 MSD	Mercury	Mercury, HNU3/H2U2 Impinger	Contrations The rule	। et	110	111	111	0.0
H-250 MSD	Mercury	Mercury, cold vapor SW/4/0	EST Intel gas, 15 mise	•	82	72	11	13.0
H-247 MSD	Mercury	Mercury, HN03/H2U2 Impinger	ESP Intel gas, not mise	10	9 8	91.5	91	0.5
H-213 MSD	Molybdenum	ICP analysis by SW6010	Stack gas, impartac	10	6	919	92	0.1
H-244 MSD	Molybdenum	ICP analysis by SWOUTU	ESF litter gas, mips for	ı م	8	90.4	6	4.0
H-213 MSD	Molybdenum	ICP analysis by Swould	Diack gas, mips for FOD inter and imme 18.0	10	65	91.3	92	0.8
H-244 MSD	Molybdenum	ICP analysis by SW6010	Contraction for the PNR	100	68	86	88	3.4
H302/H217/H218 C	Molybdenum	BIF ICP for Metals Trains	ESP inter mas th/PCR	0	93	63	66	0.0
H113/H249 D MSD	Molybdenum		ESP inter nas imps 162	N	96	94.6	95	1.5
H-244 MSD	Nickel	NICKEI DY GF, FF 7 249.2	Stack rae imps 1&2	2	68	86.8	88	2.5
H-213 MSD	Nickel	Nickel by Gr, FrA 249.2	Stack cas filter blank A	88	82	85	84	3.6
H392 A MSD	Nickel	Bit Ni for metals i tains	Ect inlat ras th/PCR	~	110	103	107	6.6
H113/H249 D MSD	Nickel	BIF NI TOT METALS THAT'S	ECP inter ree th/PCR	0	91	6	91	
H113/H249 D MSD	Nickel	BIF NI TOT METRIS I TRATIS		(6 7	96	60	88	18.2
H302/H217/H218 C	Nickel	BIF NI for Metals Frains	CCD injet ase Ti rinse	• ব	109	100	105	8 .6
H-313 MSD	Phosphorus		col mist gas, th/PCR	2	85	84	85	1 12
H113/H249 D MSD	Phosphorus		ECP inter ras imps 1&2	2	114	113	114	0.0
H-244 MSD	Phosphorus	ICP analysis by Swool o	Stark das imps 162	2	112	112	112	0.0
H-213 MSD	Phosphorus	ICP analysis by crevers	ESP inlat day, th/PCR	2	06	60	6	0.0
H113/H249 D MSD	Phosphorus	BIFICF JOT METARS TIAILS						

					MS	MSD	Меал	RPD
Samnle ID	Analyte	Method	Sample	Bun	(% Rec)	(% Rec)	(% Rec)	%
	Phosphorus	ICP analysis by SW6010	Stack gas, filt+PNR	e 1	79	62	79	0.0
	Polassium	tCP analysis by SW6010	Stack gas, imps 1&2	2	68	88.7	68	0.3
	Potecium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	9 3	90.7	92	2.5
	Potaceium	BIFICP for Metals Trains	Stack das, filt+PNR	e	103	100	102	3.0
	Potecium	ICP analysis by SW6010	Stack das, imps 1&2	2	68	86.9	88	2.4
	Potaceium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	93	90.9	92	2.3
H113/H940 D MSD	Potassium	BIF ICP for Metals Trains	ESP inlet gas, th/PCR	ŝ	102	102	102	0.0
H213 MSD	Selenium	ICP analysis by SW6010	Stack gas, imps 1&2	0	119	119	119	0.0
H=244 MSD	Selenium	Selenium by SW7740	ESP inlet gas, imps 1&2	0	76	79	78	3.9
H302/H217/H218 C	Selenium	BIF Se for Metals Trains	Stack gas, filt+PNR	ల	142	67	105	71.8
H113/H249 D MSD	Selenium	BIF Se for Metals Trains	ESP inlet gas, th/PCR	2	79	80	80	1.3
H309/H917/H918 C	Selenium	BIF Se for Metals Trains	Stack gas, filt+PNR	e)	137	135	136	1.5
H113/H249 D MSD	Silicon	BIF ICP for Metals Trains	ESP inlet gas, th/PCR	2	54	50	52	7.7
H=213 MSD	Silicon	ICP analysis by SW6010	Stack gas, imps 1&2	~	91	83	87	9.2
	Silicon	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	68	84.9	87	4.7
	Silicon	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	91	89.4	60	1.8
	Silicon	ICP analysis by SW6010	Stack gas, imps 1&2	2	91	89.7	66	4,1
	Silver	RIF ICP for Metals Trains	ESP inlet gas, th/PCR	2	51	52	52	1 .9
	Silver	ICP analysis by SW6010	Stack gas, imps 1&2	2	86	84.7	85	1.5
	Silver	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	88	87.9	88	0.1
	Silver	ICP analysis by SW6010	Stack gas, imps 1&2	2	87	87.2	87	0.2
	Silver	RIF ICP for Metals Trains	Stack gas, filt+PNR	6)	59	62	61	5.0
	Silver	ICP analysis by SW6010	ESP inlet das, imps 1&2	2	87	87	87	0.0
	Sodium	ICP analysis by SW6010	Stack das, imps 1&2	2	68	87.8	88	1.4
	Sodium	RIF ICP for Metals Trains	ESP inlet gas, th/PCR	2	8 6	66	66	1.0
	Sodium	RIFICP for Metals Trains	Stack gas, filt+PNR	9	101	100	101	1.0
	Sodium	ICP analysis by SW6010	Stack gas, imps 1&2	2	6	90.3	0 6	0.3
	Sodium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	9 5	94	95	
USM 442-11	Sodium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	95	93.8	94	с. С
	Strontium	ICP analysis by SW6010	ESP inlet gas, th/PCR	0	105	103	104	6 . -
H_244 MSD	Strontium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	~	92	91.5	92	0.5
	Strontium	ICP analysis by SW6010	Stack gas, imps 1&2	2	92	92.3	92	0.3
	Stronhum	ICP analysis by SW6010	Stack gas, imps 1&2	2	91	90.2	91	0.0
H302/H217/H218 C	Strontium	ICP analysis by SW6010	Stack gas, filt+PNR	ന	76	91	84	18.0
H_244 MSD	Strontium	ICP analysis by SW6010	ESP intet gas, imps 1&2	2	66	92.3	63	0.8
	Sultate	Sulfate, EPA 300.0	ESP inlet gas, anion imps	2	66	101	100	2.0
H-200 M-200 H-16	Sulfate	Sulfate on filters	ESP inlet gas, th/PCR/TLR	2	11	86	105	12.4
H-230 H-193 COMP	Sulfate	Sulfate on filters	Stack gas, filt+PNR	0	119	123	121	3.3
H-199 MSD	Sulfate	Sulfate, EPA 300.0	Stack gas, anion imps	2	107	96.2	102	10.6

					SW	MSD	Mean	RPD
Sample ID	Analyte	Method	Sample	Run	(% Rec)	(% Rec)	(% Rec)	(%)
	Suifate	Sulfate EPA 300.0	ESP inlet gas, anion imps	2	75	84.8	80	12.3
	Sulfate	Sulfate EPA 300.0	Stack das, anion imps	2	118	74	96	45.8
	Thellium	ICP analysis by SW6010	Stack das, filt+PNR	0	85	6	88	5.7
H113/H249 D MSD	Thalium	ICP analysis by SW6010	ESP iniet gas, th/PCR	N	66	68	6	÷
H-213 MSD	Thatitum	ICP analysis by SW6010	Stack gas, imps 1&2	2	91	85.2	88	6.6
H-213 MSD	Thallium	ICP analysis by SW6010	Stack gas, imps 1&2	2	91	87.7	68	3.7
H-244 MSD	Thallium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	88	91.9	66	4.3
H-244 MSD	Thallium	ICP analysis by SW6010	ESP iniet gas, imps 1&2	2	91	92.2	92	1.0
H-213 MSD	Titanium	BIF ICP for Metals Trains	Stack gas, imps 1&2	~	90.3	90.8	9	0.6
H-213 MSD	Titanium	BIF ICP for Metals Trains	Stack gas, imps 1&2	2	6.69	88.7	88	е. Г
H-244 MSD	Titanium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	91.7	90.9	91	0.0
H-244 MSD	Titanium	ICP analysis by SW6010	ESP Intet gas, imps 1&2	~	90.6	0 6	6	0.7
	Vanadium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	~	92	91.4	92	0.7
H_213 MSD	Vanadium	ICP analysis by SW6010	Stack gas, Imps 1&2	2	1 6	91.1	91	0.1
H113/H249 D MSD	Vanadium	BIFICP for Metals Trains	ESP inlet gas, th/PCR	2	96	3 2	96	- 10
	Vanadium	ICP analysis by SW6010	Stack gas, imps 1&2	2	9	89.1	6	2.1
H-244 MSD	Vanadium	ICP analysis by SW6010	ESP inlet gas, imps 1&2	ŝ	91	90.3	91	0.8
H300/H017/H018 C	Vanadium	BIFICP for Metals Trains	Stack gas, filt+PNR	0	8 6	96	97	2.1
	Zinc	ICP analysis by SW6010	Stack gas, imps 1&2	2	86	84.8	85	4.1
	Zinc	ICP analysis by SW6010	Stack gas, imps 1&2	0	87	87.2	87	0.2
	Zinc	ICP analysis by SW6010	Stack gas, filt+PNR	9	74	74	74	0.0
	Zinc	ICP analysis by SW6010	ESP inlet gas, th/PCR	2	87	86	87	1.2
	Zinc	ICP analysis by SW6010	ESP inlet gas, imps 1&2	2	88	86.7	87	1.5
H-244 MSD	Zinc	ICP analysis by SW6010	ESP inlet gas, imps 1&2	0	68	87.8	88	1.4
		•						

Table F2-7. Matrix Spike Recovery Data for Metals and Anions

Table F2–7. Matrix Spike Recovery Data for Metals and Anions Summary of Matrix Spike Recovery Data for Bottom Ash Samples

ļ

	No. of	Mean	Min	Max	RPD
America	MCD Paire	(% Rec)	(% Rec)	(% Rec)	(%)
		10	64	80	22.2
	- •	2.0	84	102	19.4
Ammony	- •	105	104	106	1.9
Alsello Barium		88	82	94	13.6
Bervlinm		86	84	87	3.5
Cadmium	· 01	102	94	106	6.5
Calcium		92	82	102	21.7
Chromium		91	06	91	1:1
Cohalt		06	68	91	2.2
Conner		92	91	92	Ξ
lron	. +-	97	91	103	12.4
lead	• •-	97	96	9 6	2.1
Macnesium	· .	86	69	102	38.6
Manganese	• •-	91	06	91	
Mercury	· .	97	97	97	0.0
Molvhdenim	· •-	103	102	103	1.0
Nickel	- -	06	69	91	2.2
Phosphorus	-	95	95	95	0.0
Potassium	• •	87	85	89	4.6
Selenium	• •	B 6	84	88	4.7
Silicon		83	82	83	1.2
Silver		52	43	61	34.6
Sodium		93	88	8 6	10.8
Strontium		56	11	60	3.8
Thallium	-	06	88	91	3.4
Vanadium	-	92	91	92	
Zinc	-	86	85	96	1.2
Chloride	-	9 8	86	66	1.3
Fluoride	2	67	29	88	46.1
Summary of Matrix Spike R	ecovery Data for Coal Samples				
	No. of	Mean	Min	Max	ЯРD
Analyta	MSD Pairs	(% Rec)	(% Rec)	(% Rec)	(%)
Chloride	-	96	95	97	2.1
Fluoride	3	70	49	91	11.6

Appendix F: QA/QC Results

Metals and Anions	as Samples
a for	for 0
y Dat	Data
COVEL	overy
e He	e Rec
k Spik	Spike
Matri	Aatrix
	y of N
le F2	nmar
Tab	Sun

GPD	(%)	21		t t v c		4.1	0.5	12.3	2.7	1.2	5.4	0.8	0.9	3.0	9.2	0.7	1.5	6.9	0.9	5.6	1.8	1.7	15.7	5.0	1.5	1.0	3.7	3.8	0.9	, 1,1	0.9	4.8	4.8	14.4	8.0
Мах	(% Rec)	101	30	0 0 0	200	601	96	84	116	66	110	94	101	66	109	94	92	132	66	110	114	103	142	91	88	101	105	92	92	8 6	68	114	102	123	111
Min	(% Hec)	84			7 (7 1	/3	68	69	97	85	97	88	88	74	65	88	82	50	86	80	19	87	67	50	51	88	76	85	68	68	74	06	85	74	06
Mean	(% Rec)	60	20	59	SB	06	93	78	107	94	104	06	92	91	95	92	88	86	91	92	97	94	103	81	11	94	92	68	06	60	85	101	93	101	102
No. of	MCC Daire		0	9	4	9	9	4	4	ÿ) () (C	? <u>-</u>					2) 10			. 6	ÿ	1				אנ) (2) (N
	A a b b b a c	Anaryte	Aluminum	Antimony	Arsenic	Barium	Bervllium	Boron	Cedmint	Calcium	Chromium	Cohalt	Conan	Lopper Lon		Meanonium	Magnesium	Manganese	Mercury	Maybaenum	Dhoonhoring	Prinspinus	Polassium Colonium	Cilicon	Shicon	Sodium	Strontium	Thallium	Titonium	Venedium	Valiauluin Zine	Chlorida	Unioride	Fiuoride Suifate	Chromium (VI)

			Meas'd	Ref Value	Recov	Dup Recov	RPD
Sample ID	Analyte	Method	(mg/kg)	(mg/kg)	(% Rec)	(% Rec)	(% Rec)
NBS 1632A	Aluminum	NAA	28159	29500	95		. ,
NBS 1632A	Antimony	NAA	0.6	0.6	101		
NBS 1632A	Arsenic	NAA	9,3	9.3	100		
NBS 1632A	Barium	NAA	117	120	97		
NBS 1632A	Bromine	NAA	42	41	103		
NBS 1632A	Calcium	NAA	2353	2410	98		
NBS 1632A	Cerium	NAA	30	29	102		
NBS 1632A	Cesium	NAA	2.3	2.3	99		
NBS 1632A	Chlorine	NAA	722	756	95		
NBS 1632A	Chromium	NAA	33.8	34.3	99		
NBS 1632A	Cobalt	NAA	7.1	6.7	106		
NBS 1632A	Copper	NAA	18.6	16.5	113		
NBS 1632A	Europium	NAA	0.6	0.52	110		
NBS 1632A	Hafnium	NAA	2	1.62	104		
NBS 1632A	lodine	NAA	1.7	1.8	95		
NBS 1632A	Iron	NAA	11390	-11100	103		
NBS 1632A	Lanthanum	NAA	16	15	104		
NBS 1632A	Lutetium	NAA	0.18	0.17	104		
NBS 1632A	Magnesium	NAA	1098	1150	95		
NBS 1632A	Manganese	NAA	27	' 28	95		
NBS 1632A	Molybdenum	NAA	4.02	3.85	104		
NBS 1632A	Neodymium	NAA	14	12	115		
NBS 1632A	Nickel	NAA	21.0) 19.4	108		
NBS 1632A	Potassium	NAA	4175	5 4110	102		
NBS 1632A	Rubidium	NAA	31	30	104	,	
NBS 1632A	Samarium	NAA	2.4	2.4	101		
NBS 1632A	Scandium	NAA	6.5	5 6.3	103	,	
NBS 1632A	Selenium	NAA	2.6	5 2.6	101		
NBS 1632A	Sodium	NAA	790) 828	95	5	
NBS 1632A	Strontium	NAA	79	9 85	93	3	
NBS 1632A	Tantalum	NAA	0.41	0.42	97	,	
NBS 1632A	Terbium	NAA	0.306	3 0.311	99	•	
NBS 1632A	Thorium	NAA	4.7	7 4.5	105	5	
NBS 1632A	Titanium	NAA	1556	6 1630	95	5	
NBS 1632A	Uranium	NAA	1.2	1.28	101		
NBS 1632A	Vanadium	NAA	42	2 44	95	5	
NBS 1632A	Ytterbium	NAA	1.1	5 1.08	107	7	
NBS 1632A	Zinc	NAA	20	8 28	98	3	
NBS 1632A	Zirconium	NAA	5	3 53	99	9	
SARM 20	Aluminum	NAA	5856	7 59600) 91	B 96	5 2.3
SARM 20	Antimony	NAA	0.0	6 0.4	16	2 148	3 9.3
SARM 20	Arsenic	NAA	5.	ษ 4.7	12	3 110	5 5.7
SARM 20	Barium	NAA	31	8 372	2 8	5	
SARM 20	Beryllium	ICP-A	LE 1.3	ษ <u>2.</u> รู	5 7	2	
SAHM 20	Bromine	NAA			2 23	0 20	10.7
SARM 20	Calcium	NAA	2113	2 13400	15	8	
SARM 20	Cerium	NAA	8	9 87	7 10	2	

Table F2-8. Coal Standard Reference Material Analysis Results for Metals

			Meas'd	Ref Value	Recov	Dup Recov	RPD
Sample ID	Analyte	Method	(mg/kg)	(mg/kg)	(% Rec)	(% Rec)	(% Rec)
SARM 20	Cesium	NAA	2	2	96		•
SARM 20	Chromium	NAA	71	67	106		
SARM 20	Cobalt	NAA	8.4	8.3	102		
SARM 20	Copper	NAA	44	18	244	256	4.9
SARM 20	Europium	NAA	1	1	122		
SARM 20	Hafnium	NAA	5.5	4.8	114		
SARM 20	iron	NAA	8253	8180	101		
SARM 20	Lanthanum	NAA	46	43	107	110	2.9
SARM 20	Lead	ICP-AE	27	26	104		
SARM 20	Magnesium	NAA	3265	2600	126	119	5.4
SARM 20	Manganese	NAA	64	80	80	83	3.7
SARM 20	Mercury	NAA	0.44	0.25	176	149	16.7
SARM 20	Mercury	DGA/CV	0.25	0.25	100		
SARM 20	Nickel	NAA	25	25	99		
SARM 20	Phosphorus	ICP-AE	630	611	103		
SARM 20	Potassium	NAA	2128	1160	183	,	
SARM 20	Rubidium	NAA	13	10	127		
SARM 20	Samarium	NAA	5.5	6.3	87	88	0.8
SARM 20	Scandium	NAA	12	10	124		
SARM 20	Selenium	NAA	2.0	0.8	244	,	
SARM 20	Sodium	NAA	2005	2000	100	108	7.5
SARM 20	Strontium	NAA	282	330	85	5	
SARM 20	Tantalum	NAA	1.2	1.2	99		
SARM 20	Terbium	NAA	1.0	0.9	116	5	
SARM 20	Thorium	NAA	19	18	103	3	
SARM 20	Tin	NAA	20	4	500)	
SARM 20	Titanium	NAA	3389	3780	90) 87	3.0
SARM 20	Uranium	NAA	5	4	121	127	4.7
SARM 20	Vanadium	NAA	50) 47	106	5 103	3.2
SARM 20	Ytterbium	NAA	0.3) 2	15	5	
SARM 20	Zinc	NAA	52	? 17	306	5	
SARM 20	Zirconium	NAA	266	6 18C) 148	3	

Table F2-8. Coal Standard Reference Material Analysis Results for Metals

		Result	Ref Value	Recoverv
Analyte	Method code	(ma/ka)	(ma/ka)	(%)
Aluminum	ICPES	142000	143000	99
Aluminum	ICPES	141000	143000	99
Aluminum	ICPES	140000	143000	98
Aluminum	ICPES	139000	143000	97
Aluminum	ICPES	143000	143000	100
Aluminum	ICPES	139000	143000	97
Aluminum	ICPES	131000	143000	92
Aluminum	ICPES	141000	143000	99
Aluminum	ICPES	139000	143000	97
Aluminum	ICPES	141000	143000	99
Aluminum	ICPES	139000	143000	97
Aluminum	ICPES	140000	143000	98
Arsenic	GFAAS	181	145	125
Arsenic	GFAAS	188	145	130
Arsenic	GFAAS	183	145	126
Arsenic	GFAAS	184	145	127
Arsenic	ICPES	99.6	145	69
Arsenic	ICPES	121	145	83
Arsenic	ICPES	121	145	83
Arsenic	ICPES	110	145	76
Arsenic	ICPES	94.3	145	65
Arsenic	ICPES	99.6	145	69
Arsenic	ICPES	94.3	145	65
Arsenic	ICPES	110	145	76
Barium	ICPES	1270	1500	85
Barium	ICPES	1310	1500	87
Barium	ICPES	1310	1500	87
Barium	ICPES	1320	1500	88
Barium	ICPES	1320	1500	88
Barium	ICPES	1270	1500	85
Barium	ICPES	1280	1500	85
Barium	ICPES	1280	1500	85
Barium	ICPES	1330	1500	89
Barium	ICPES	1340	1500	89
Barium	ICPES	1340	1500	89
Barium	ICPES	1320	1500	88
Beryllium	ICPES	16.4	12	137
Beryllium	ICPES	16.3	12	136
Beryllium	ICPES	15.9	12	133
Beryllium	ICPES	16.2	12	135
Beryllium	ICPES	16.3	12	136
Beryllium	ICPES	16.4	12	137
Beryilium	ICPES	16.4	12	137
Beryllium	ICPES	16.4	12	137
Beryllium	ICPES	15.6	12	130
Beryllium	ICPES	15.8	12	132
Beryllium	ICPES	16.4	12	137
Beryllium	ICPES	15.9	12	133
Calcium	ICPES	10900	11100	98
Calcium	ICPES	10700	11100	96
Calcium	ICPES	10700	11100	96

Table F2-9. Ash Recovery Data for Metals by ICPES and AAS

.

ı.

		Result	Ref Value	Recovery
Analyte	Method code	e (mg/kg)	(mg/kg)	(%)
Calcium	ICPES	10600	11100	95
Calcium	ICPES	9850	11100	89
Calcium	ICPES	10600	11100	95
Calcium	ICPES	11300	11100	102
Calcium	ICPES	10900	11100	98
Calcium	ICPES	10800	11100	97
Calcium	ICPES	11400	11100	103
Calcium	ICPES	10600	11100	95
Calcium	ICPES	10600	11100	95
Calcium	ICPES	10800	11100	97
Calcium	ICPES	10600	11100	95
Chromium	ICPES	179	196	91
Chromium	ICPES	183	196	93
Chromium	ICPES	185	196	94
Chromium	ICPES	185	196	94
Chromium	ICPES	179	196	91
Chromium	ICPES	181	196	92
Chromium	ICPES	177	196	90
Chromium	ICPES	181	196	92
Chromium	ICPES	183	196	93
Chromium	ICPES	180	196	92
Chromium	ICPES	181	196	92
Chromium	ICPES	178	196	91
iron	ICPES	98400	94000	105
iron	ICPES	91100	94000	97
Iron	ICPES	89500	94000	95
Iron	ICPES	92700	94000	99
iron	ICPES	94600	94000	101
Iron	ICPES	96200	94000	102
Iron	ICPES	93100	94000	99
Iron	ICPES	89500	94000	95
iron	ICPES	92000	94000	98
lron	ICPES	94600	94000	101
Iron	ICPES	96200	94000	102
iron	ICPES	94000	94000	100
Iron	ICPES	92000	94000	98
iron	ICPES	99900	94000	106
Lead	GFAAS	59.6	72	83
Lead	GFAAS	58.7	72	82
Lead	ICPES	64.3	72	89
lead	ICPES	44.7	72	62
Lead	ICPES	AA 7	72	62
Lead	ICPES	64.3	1 72	20
Lead	ICPES	59.3	- 72 - 79	74
Lead	ICPES	43 :	3 72	60
Lead	ICPES	53 :	- 12 3 70	74
lead	ICPES	43 1	- 12	03
lead	GFAAS		- /2 5 70	79
Lead	GFAAS	51. 51.	- /2 6 70	70
Lead	GFAAS	49	5 79	
lasri	GFAAS		- 12	74
2000		. v.	- 12	

Table F2-9. Ash Recovery Data for Metals by ICPES and AAS

		Result	Ref Value	Recovery
Analyte	Method code	(mg/kg)	(mg/kg)	(%)
Magnesium	ICPES	4160	4550	91
Magnesium	ICPES	4140	4550	91
Magnesium	ICPES	4250	4550	93
Magnesium	ICPES	4500	4550	99
Magnesium	ICPES	4240	4550	93
Magnesium	ICPES	4280	4550	94
Magnesium	ICPES	4280	4550	94
Magnesium	ICPES	4250	4550	93
Magnesium	ICPES	4240	4550	93
Magnesium	ICPES	4140	4550	91
Manganese	ICPES	174	179	97 °
Manganese	ICPES	180	· 179	101
Manganese	ICPES	174	179	97
Manganese	ICPES	172	179	96
Manganese	ICPES	175	179	98
Manganese	ICPES	176	179	98
Manganese	ICPES	174	179	97
Manganese	ICPES	180	179	101
Manganese	ICPES	175	179	98
Manganese	ICPES	173	179	97
Manganese	ICPES	172	179	96
Manganese	ICPES	167	179	93
Nickel	GFAAS	129	127	102
Nickel	GFAAS	131	127	103
Nickel	GFAAS	123	127	97
Nickel	GFAAS	124	127	98
Nickel	GFAAS	124	127	98
Nickel	GFAAS	132	127	104
Sodium	ICPES	2380	1700	140
Sodium	ICPES	2410	1700	142
Sodium	ICPES	2450	1700	144
Sodium	ICPES	2360	1700	139
Sodium	ICPES	2440	1700	144
Sodium	ICPES	2440	1700	144
Sodium	ICPES	2390	1700	141
Sodium	ICPES	2450	1700	144
Sodium	ICPES	2410	1700	1/42
Sodium	ICPES	2380	1700	140
Strontium	ICPES	793	830	96
Strontium	ICPES	804	830	97
Strontium	ICPES	793	830	96
Strontium	ICPES	799	830	96
Strontium	ICPES	799	830	96
Strontium	ICPES	799	830	96
Strontium	ICPES	804	830	97
Strontium	ICPES	809	830	97
Strontium	ICPES	801	830	97
Strontium	ICPES	808	830	97
Strontium	ICPES	808	830	97
Strontium	ICPES	760	830	92
Titanium	ICPES	7950	8000	99

Table F2-9. Ash Recovery Data for Metals by ICPES and AAS

7

		Result	Ref Value	Recovery	
Anaiyte	Method code	(mg/kg)	(mg/kg)	(%)	
Titanium	ICPES	8080	8000	101	
Titanium	CPES	7810	8000	98	
Titanium	ICPES	8076	8000	101	
Titanium	ICPES	7860	8000	98	
Titanium		8080	8000	101	
Titanium	ICPES	7950	8000	99	
Titanium	ICPES	7810	8000	98	
Titanium	ICPES	7860	8000	98	
Titanium	ICPES	7865	8000	98	
Titanium	ICPES	7952	8000	99	
Titanium	ICPES	7880	8000	99	
Titanium	ICPES	7880	8000	99	
Titanium	ICPES	7878	8000	98	
Titanium	ICPES	7809	8000	98	
Titanium	ICPES	7860	8000	98	
Titanium	ICPES	7951	8000	99	
Titanium	ICPES	7857	8000	98	
Zinc	ICPES	197	220	90	
Zinc	ICPES	189	220	86	
Zine	ICPES	193	220	88	
Zinc	ICPES	193	220	88	
Zinc	ICPES	197	220	90	
Zinc	ICPES	189	220	86	
Zinc	ICPES	196	220	89	
Zinc	ICPES	193	220	88	
Zinc	ICPES	193	220	88	
Zinc	ICPES	186	220	85	
Zinc	ICPES	196	220	89	
Zinc	ICPES	187	220	85	
Summary					
		No. of	Mean	_Min	N
	Method	Results	(% Rec)	(% Rec)	(% R
Aluminum	ICPES	12	98	92	1
Arsenic	ICPES	8	73	65	
Banum	ICPES	12	87	85	
Beryllium	ICPES	12	135	130	1
Calcium	ICPES	14	97	89	1
Chromium		12	92	90	
Iron	ICPES	14	100	95	•
Lead	ICPES	8	71	60	

. .

Table F2-9. Ash Recovery Data for Metals by ICPES and AAS

•		No. ot	Mean	Min	Max	Std Dev
	Method	Results	(% Rec)	(% Rec)	(% Rec)	(% Rec)
Aluminum	ICPES	12	98	92	100	2.1
Arsenic	ICPES	8	73	65	83	7.5
Barium	ICPES	12	87	85	89	1.7
Beryllium	ICPES	12	135	130	137	2.4
Calcium	ICPES	14	97	89	103	3.3
Chromium	ICPES	12	92	90	94	1.3
Iron	ICPES	14	100	95	106	3.3
Lead	ICPES	8	71	60	89	12.4
Magnesium	ICPES	10	93	91	9 9	2.3
Manganese	ICPES	12	97	93	101	2.0
Sodium	ICPES	10	142	139	144	1.9
Strontium	ICPES	12	96	92	97	1.6
Titanium	ICPES	18	99	98	101	1.1
Zinc	ICPES	12	87	85	90	1.7
Arsenic	GFAAS	4	127	125	130	2.0
Lead	GFAAS	6	75	69	83	5.7
Nickel	GFAAS	6	100	97	104	3.1

Table F2-10. PSD Ash Recovery Data for Metals

			Meas'd	Ref Value	Rec	Dup Rec	RPD
Sample ID	Analyte	Method	(mg/kg)	(mg/kg)	(%)	(%)	(%)
NIST 1633a	Antimony	ICPMS	5.2	6.8	76	76	1.0
NIST 1633a	Arsenic	ICPMS	128.6	145	89	88	0.7
NIST 1633a	Beryllium	ICPMS	10.1	12	84	86	2.7
NIST 1633a	Cadmium	ICPMS	0.5	1	51	60	16.7
NIST 1633a	Chromium	ICPMS	193.4	196	99	89	9.8
NIST 1633a	Cobalt	ICPMS	33.5	46	73	68	7.3
NIST 1533a	Copper	ICPMS	120.6	118	102	94	8.3
NIST 1633a	Lead	ICPMS	64.5	72.4	89	88	0.7
NIST 1633a	Manganese	ICPMS	195.1	179	109	104	4.5
NIST 1633a	Mercury	ICPMS	0.3	0.16	179	133	29.7
NIST 1633a	Molybdenum	ICPMS	24.9	29	86	97	12.0
NIST 1633a	Nickel	ICPMS	94.3	127	74	69	6.8
NIST 1633a	Selenium	ICPMS	10.0	10.3	98	108	9.8
NIST 1633a	Vanadium	ICPMS	253.5	297	85	82	3.7

•

able F2-1	1. Quartz Filter SI	pike Recove	rry Data for Metals		Car	Maan	Laa
				20	UCIN		
sample ID	Analyte	Method	Sample	(% Rec)	(% Rec)	(% Rec)	(%)
			Stack age at fit hink 1	83	82	83	1.2
1-QB1	Aluminum	ICFES	Older yas, yiz mu pink i	9	1	81	
1-0B1	Aluminum	ICPES	Stack gas, qtz nit pink i	5			
1-0B1	Arsenic	ICPES	Stack gas, qtz filt blnk 1	22	ł	2 9	9
1-0R1	Arsenic	ICPES	Stack gas, dtz filt blnk 1	79	11	R (D Q N Q
	Barium	ICPES	Stack gas, qtz filt blnk 1	06	88	68	N.N.
	Barium	ICPES	Stack das, dtz filt blnk 1	87		87	
			Stack das otz filt bink 1	8 6	85	96	1.2
	Deryllium Beryllium		Stack das, dtz filt blnk 1	83		83	
	Del ymui II		Stack nas otz filt blink 1	6 8		6 8	
			Stack das att fitt bink 1	72	91	82	23.3
			Stack das off filt bluk 1	87		87	
1-081			Stack dae of fit bink 1	89	88	6 8	
1-0B1	Calcium		stack das die in and in a	86		98	
1-0B1	Chromium		Diach gas, que in binn p Diach ann an bhairt		AA	88	0.0
1-QB1	Chromium	ICPES	Stack gas, qtz nn pink i				-
4-0B1	Cobalt	ICPES	Stack gas, dtz titt bink 1	5 0	0	38	i
4-0B1	Cobalt	ICPES	Stack gas, qtz filt blnk 1	83		50	
	Conner	ICPES	Stack gas, qtz filt blnk 1	83	1	EB	•
-081	Conder	ICPES	Stack gas, qtz filt blnk 1	86	85	99	N.
1-081	Iron	ICPES	Stack gas, qtz filt blnk 1	84		84	0
	Iron	ICPES	Stack gas, qtz filt blnk 1	96	. 86	86 1	0.0
	l ead	ICPES	Stack gas, gtz filt blnk 1	84	81	83	3.0
	Lead	ICPES	Stack gas, qtz filt bink 1	62		64	
	Mannesium	ICPES	Stack gas, dtz filt blnk 1	83	82	83	1.2
1-0B1	Magnesium	ICPES	Stack gas, qtz filt bink 1	81		19 19	
1-0B1	Manganese	ICPES	Stack gas, qtz filt blnk 1	85	1	8 8	0
	Manoanese	ICPES	Stack gas, qtz filt bink 1	87	87	JA I	
	Molvbdenum	ICPES	Stack gas, dtz filt blnk 3	92	40	99	78.8
	Molyhdanim	ICPES	Stack das, dtz filt blink 3	4		40	
	Nickel	ICPES	Stack das, dtz filt blnk 1	84		84	
	Nickel	ICPES	Stack gas, gtz filt blnk 1	87	98	87	<u>,</u>
	Dhoenhori is	10 PES	Stack das atz filt blnk 3	66		66	
	Phoenhorus	ICPES	Stack gas, gtz filt bink 3	101	66	100	2.0
	Driachtm	ICPES	Stack das. dtz filt bink 1	85		85	
	Potacelium	ICPES	Stack gas, gtz filt bink 1	87	87	87	0.0
	こうろうせんし	2					
	Salanium	NDPES	Stack das, dtz filt blnk 1	166		166	
-------	------------------------	--------------	----------------------------	------------	-----	------------	--------
	Selenium	ICPES	Stack das. dtz fitt blnk 1	102	170	136	50.0
	Silver	CPES	Stack one of filt bink 1	55		52	
	Silver		Stack das. dtz filt bink 1	55	56	56	1.8
	Sodi IT		Stack das, dtz filt bink 1	85	85	8 5	0.0
	Sodium	ICPES	Stack das. dtz filt blnk 1	63		8 3	
	Strontium	ICPES	Stack das. dtz filt bink 1	91	8	91	1.1
	Strontium	ICPES	Stack das, dtz filt blnk 1	88		88	
	Thallinm	SHOUL	Stack das, otz filt blnk 1	76		76	
	Thallium		Stack das, otz filt bink 1	74	11	76	4.0
	Vanadium		Stack das, dtz filt bink 1	68	68	68	0.0
	Variadium		Stack das, dtz filt blnk 1	87		87	
	Valiaurum Ziao		Stack nas oth fift bink 1	78	78	78	0.0
				76		76	
H-0B1	Zinc	ICPES	Stack gas, qtz nn pink 1	2		2	
	Arenio	GFAAS	Stack das, dtz filt blnk 3	114	111	113	2.7
		CEAAS	Stack das, dtz filt bink 1	107	105	106	1.9
	Caurinum		Stack das of filt block 3	8 6	88	6 3	10.8
		SEAS:	Stack das otz filt bink 1	67	61	64	9.4
	Leau Load		Stack das, otz fift bink 1	62	62	62	0.0
	Nickel	CEAAS	Stack das, dtz filt blok 3	110	106	108	3.7
	selection Selection		Stack das of 7 fit bink 2	76	24	50	104.0
			Stack das at fit hink 9	33	28	31	16.4
H-GB2	Selenium		Stack yas, 44 m Shirk S	5	24	38	73.7
H-QB2	Selenium	CT ANU	orack gas, que mu unin e	31		37	
H081	Mercury	CVAAS	Stack gas, qtz filt blnk 1	45	40	40	N N
	•						

•

Appendix F: QA/QC Results

Summary		Moon	Min	Max N	lean RPD
ICPES	Deerthe		(% Rec)	(% Rec)	(%)
Analyte	SUBSAL				,
	c	68	81	83	1.2
	4 C	5 5	75	62	2.6
Arsenic	~ ~		5. B	Co	2.2
Barium	N	00	5	98	12
Beryllium	N		90 52	6	23.3
Cadmium	N	6	12	68	
Calcium	N	86	5	88	0.0
Chromium	NC	10	8 8	85	1.2
Cobait	NC	4 B	88	98	4 12
Copper	NC		84	86	0.0
Iron	4 0	8 2	62	84	3.6
Lead	1 6	5	81	83	1.2
Magnesium	<i>N</i> C	A6	82	87	0.0
Manganese	N 0	8 2	40	92	78.8
Molybdenum	4 C	5	8	87	1.2
	N C	8 Ç	60	101	2.0
Phosphorus	NC	S B	85	87	0.0
Potassium	N C	151	102	170	50.0
Selenium	4 6	5	55	56	1.8
SIIVE	40	3 8	83	85	0.0
Sogium	40	5 2	88	91	1.1
Strontium		76	74	11	4.0
Venedium	10	88	87	68	0.0
Vallauluili Zinc	1 (1)	11	9/	82	0.0
2 17	ł				
AAS	-		***	114	77
Arsenic	-	5			. 0 i +
Cadmium	-	106		5	ה כ - י
Chromium	-	8 3	88	96	9.01 •
l ead	2	63	61	67	4.7
Marcurv		46	45	46	2.2
Nickel	-	108	106	110	3.7
Selenium	e	40	24	76	64.7

Appendix F: QA/QC Results

1

Table F2-12. Mercury Speciation Spike Recovery Data

Sample ID	Compound	Method	% Recovery
Method Spike	Mercury (0)	Bloom	88
Method Spike	Mercury (II)	Bloom	97
Method Spike	Methyl Mercury	Bloom	106
Method Spike	Mercury (II)	Bloom	94

.

.

.

		Duplicate Sample Results				
		Sample ID:	H201/202	H225/226	RPD	
Analyte	Method	Units	Meas'd	Meas'd	(%)	
Aluminum	NAA	mg/kg, dry	11964.9	11228.6	6.3	
Antimony	NAA	mg/kg, dry	1.3	1.4	5.1	
Arsenic	NAA	mg/kg, dry	23.4	42.7	58.5	
Barium	NAA	mg/kg, dry	94.5	100.1	5.8	
Beryllium	ICP-AES	mg/kg, dry	1.9	2.1	10.0	
Bromine	NAA	mg/kg, dry	5.7	5.6	2.4	
Cadmium	NAA	mg/kg, dry <	< 2.6	< 2.6	0.4	
Calcium	NAA	mg/kg, dry	1811.9	1509.0	18.2	
Cerium	NAA	mg/kg, dry	17.1	15.1	12.2	
Cesium	NAA	mg/kg, dry	1.4	1.1	20.9	
Chlorine	NAA	mg/kg, dry	473.2	448.0	5 .5	
Chromium	NAA	mg/kg, dry	14.7	16.2	10.2	
Cobalt	NAA	mg/kg, dry	5.9	6.3	5.2	
Copper	NAA	mg/kg, dry	28.8	34.1	17.0	
Europium	NAA	mg/kg, dry	0.3	0.3	4.0	
Hafnium	NAA	mg/kg, dry	0.7	0.7	2.6	
lodine	NAA	mg/kg, dry	0.2	0.2	8.8	
Iron	NAA	mg/kg, dry	14100.1	10368.8	30.5	
Lanthanum	NAA	mg/kg, dry	9.7	9.7	0.3	
Lead	ICP-AES	mg/kg, dry	8.0	7.0	13.3	
Lutetium	NAA	mg/kg, dry	0.1	0,1	1.1	
Magnesium	NAA	mg/kg, dry	538.3	483.7	10.7	
Manganese	NAA	mg/kg, dry	14.4	13.4	7.1	
Mercury	DGA/CVAAS	mg/kg, dry	0.2	0.1	37.0	
Mercury	NAA	mg/kg, dry	0.1	0.1	15.0	
Molybdenum	NAA	mg/kg, dry	3.2	3.3	4.6	
Neodymium	NAA	mg/kg, dry	11.1	11.6	4.3	
Nickel	NAA	mg/kg, dry	20.6	18.8	9.2	
Phosphorus	ICP-AES	mg/kg, dry	170.0	140.0	19.4	
Potassium	NAA	mg/kg, dry	822.7	860.0	4.4	
Rubidium	NAA	mg/kg, dry	21.8	16.8	26.1	
Samarium	NAA	mg/kg, dry	1.7	1.8	2.0	
Scandium	NAA	mg/kg, dry	3.6	3.5	3.7	
Selenium	NAA	mg/kg, dry	3.4	3.5	4.9	
Silver	NAA	mg/kg, dry	0.2	0.2	11.0	
Sodium	NAA	mg/kg, dry	310.0	303.6	2.1	
Strontium	NAA	mg/kg, dry	189.2	106.9	55.6	
Tantalum	NAA	mg/kg, dry	0.2	0.2	2.4	
Terbium	NAA ·	mg/kg, dry	0.2	0.2	0.7	
Thorium	NAA	mg/kg, dry	2.3	2.4	5.3	
Tin	NAA	mg/kg, dry	< 10.5	< 10.5	0.4	
Titanium	NAA	mg/kg, dry	739.1	723.5	2.1	
iungsten	NAA	mg/kg, dry	< 5.2	< 5.2	0.4	
Uranium	NAA	mg/kg, dry	1.9	1.6	17.1	
vanadium	NAA	mg/kg, ary	23.0	22.6	1.8	
YtterDium	NAA	mg/kg, dry	0.3	0.3	7.4	

Table F2-13. Coal Duplicate Sample Results for Metals

....

Table F2-13. Coal Duplicate Sample Results for Metals

.

	Duplicate Sample Results			
	Sample ID:	H201/202	H225/226	RPD
Method	Units	Meas'd	Meas'd	(%)
NAA	mg/kg, dry	17.5	16.5	6.0
NAA	mg/kg, dry	40.3	89.0	75.3
Ultimate	%	4.3	4.7	8.2
Ultimate	%, dry	77.3	77.3	0.1
Ultimate	%, dry	4.9	5.0	0.8
Ultimate	%, dry	1.5	1.5	0.7
Ultimate	%, dry	1.8	1.6	12.9
Ultimate	%, dry	9.5	9.2	2.9
Ultimate	%, dry	5.0	5.4	7.7
Proximate	%, dry	32.3	32.7	1.3
Proximate	%, dry	58.2	58.1	0.3
Proximate	Btu/lb, dry	13728.0	13779.0	0.4
	Method NAA NAA Ultimate Ultimate Ultimate Ultimate Ultimate Ultimate Proximate Proximate Proximate	MethodSample ID:MethodUnitsNAAmg/kg, dryNAAmg/kg, dryUltimate%Ultimate%, dryUltimate%, dryProximate%, dryProximate%, dryProximateBtu/lb, dry	Duplicate SamSample ID:H201/202MethodUnitsMeas'dNAAmg/kg, dry17.5NAAmg/kg, dry40.3Ultimate%4.3Ultimate%, dry77.3Ultimate%, dry1.5Ultimate%, dry1.5Ultimate%, dry1.8Ultimate%, dry9.5Ultimate%, dry5.0Proximate%, dry58.2Proximate%, dry13728.0	Duplicate Sample Results Sample ID: H201/202 H225/226 Method Units Meas'd Meas'd NAA mg/kg, dry 17.5 16.5 NAA mg/kg, dry 40.3 89.0 Ultimate % 4.3 4.7 Ultimate %, dry 77.3 77.3 Ultimate %, dry 1.5 1.5 Ultimate %, dry 1.8 1.6 Ultimate %, dry 9.5 9.2 Ultimate %, dry 32.3 32.7 Proximate %, dry 58.2 58.1 Proximate %, dry 13728.0 13779.0

		Run 2	Run 2D	
		Result	Result	RPD
Analyte	Method	(mg/kg)	(mg/kg)	(%)
Aluminum	ICPES	108000	131000	19.2
Antimony	ICPES	ND	ND	NC
Arsenic	GFAAS	5	4 13	191
Barium	ICPES	820	1020	20.7
Bervilium	ICPES	19.7	20.7	5.0
Cadmium	GEAAS	ND	0.478	
Calcium	ICPES	4000	5540	32.3
Chromium	ICPES	102	108	57
Cobalt	ICPES	61.5	57.1	74
Copper	ICPES	120	127	57
iron	ICPES	102000	113000	10.2
Lead	ICPES	61	19.9	106.2
Magnesium	ICPES	3780	5690	40.3
Manganese	ICPES	160	170	61
Mercury	CVAAS	ND	ND	NC
Molvodenum	ICPES	2 74	J 10.9.	
Nickel	ICPES	88.7	88	08
Phosphorus	ICPES	526	670	24.1
Potassium	ICPES	21200	25200	17.2
Selenium	ICPES	32	J 26.	J NC
Silicon	ICPES	211000	216000	B 2.3
Silver	ICPES	ND	ND	NC
Sodium	ICPES	3390	3460	2.0
Strontium	ICPES	679	875	25.2
Thallium	ICPES	4.34	J ND	NC
Titanium	ICPES	6657	6932	4.0
Vana dium	ICPES	200	228	13.1
Zinc	ICPES	51	51.3	0.6

Table F2-14. Botton Ash Duplicate Sample Results for Metals

Table F2-15. SVOC Internal Standard Recovery i	Data
--	------

1

Sample ID ESP inint Gee	Compound	% Rec.
H=317/981/938/919/977/978	Acenephibapo - d10	
H = 111/H = 100/H = 105		14
H = 037/039/494/469/035/036		2
H = 417/419/410/208/201/250/250	Acenaphinene-d10	17
H - 217/910/413/300/231/332/333	Acenaphinene-010	15
	Acenaphtnylene – ds	11
H = 237/233/101/103/235/236	Acenapmnylene-do	4
		6
H = 417/410/419/300/291/352/353	Acenapmnyiene – da	18
	Anthracene-d10	6
	Anthracene – d10	8
n = 417/418/419/300/291/352/353	Anthracene - d10	22
	Anminacene – d 10	26
	Benzojajpyrene – d12	4
H-111/H-109/H-125	Benzo[a]pyrene-d12	14
H - 237/233/181/163/235/236	Benzo[a]pyrene-d12	3
H-317/281/238/212/277/278	Benzo(a)pyrene-d12	11
H-317/281/238/212/277/278	Benzo[b&k]fluoranthenes-d12	23
H-111/H-109/H-125	Benzo(b&k)fluoranthenes-d12	22
H-417/418/419/306/291/352/353	Benzo(b&k)fluoranthenes-d12	1
H - 237/233/181/163/235/236	Benzo[b&k]fluoranthenes-d12	13
H-317/281/238/212/277/278	Benzo(ghi]peryiene—d12	12
H-237/233/181/163/235/236	Benzo[ghi]peryiene—d12	6
H-111/H-109/H-125	Benzo[ghi]perylene—d12	12
H-417/418/419/306/291/352/353	Benzo(ghi)perylene—d12	0
H-31 7/281/238/ 212/277/278	Benz[a]anthracene-d12	36
H-237/233/181/163/235/236	Benz[a]anthracene-d12	29
H-111/H-109/H-125	Benz(a)anthracene-d12	17
H-417/418/419/306/291/352/353	Benz[a]anthracene-d12	•5
H - 111/H - 109/H - 125	Chrysene-d12	17
H-317/281/238/212/277/278	Chrysene-d12	35
H-237/233/181/163/235/236	Chrysene-d12	37
H-417/418/419/306/291/352/353	Chrysene-d12	5
H-317/281/238/212/277/278	Dibenz(a,h)anthracene-d14	15
H-237/233/181/163/235/236	Dibenz(a,h)anthracene-d14	5
H-417/418/419/306/291/352/353	Dibenz(a,h)anthracened14	Ó
H-111/H-109/H-125	Dibenz[a,h]anthracene-d14	15
H-417/418/419/306/291/352/353	Fluoranthene — d10	17
H-317/281/238/212/277/278	Fluoranthene-d10	34
H-237/233/181/163/235/236	Fluoranthene-d10	49
H-111/H-109/H-125	Fluoranthene-d10	12
H-111/H-109/H-125	Fluorene-d10	3
H-417/418/419/306/291/352/353	Fluorene - d10	19
H-237/233/181/163/235/236	Fluorene-dt0	22
H-317/281/238/212/277/278	Fiuorene-d10	19
H-237/233/181/163/235/236	Indeno[1,2,3-cd]pyrene-d10	5
H-417/418/419/306/291/352/353	indeno[1,2,3-cd]pyrene-d10	1
H-111/H-109/H-125	Indeno[1,2,3-cd]pyrene-d10	16
H-317/281/238/212/277/278	Indeno[1,2,3-cd]pyrene-d10	14
H-237/233/181/163/235/236	Phenanthrene-d10	41
H-417/418/419/306/291/352/353	Phenanthrene - d 10	28
H-111/H-109/H-125	Phenanthrene - d10	7
H-317/281/238/212/277/278	Phenanthrene – d 10	40

Table F2~15. SVOC Internal Standard Recovery Data

Sample ID	Compound	% Rec.
E3F 11104 945	Bympo - d10	10
H - 417/981/998/940/281/992/993	Pyrene-di0	19
H = 517/201/200/212/217/270	Pyrene-d10	37
	Pyrene-dio	13
Stack Gas	Fylene-dio	53
H-227/194/119/223/224	Acenaphthene-d10	68
H-388/314/292/354/355	Acenaphthene-d10	60
H-316/228/211/279/280	Acenaphthene-d10	62
H-227/194/119/223/224	Acenaphthylene-d8	2
H-316/228/211/279/280	Acenaphthylene – d8	48
H-388/314/292/354/355	Acenaphthylene-d8	37
H-388/314/292/354/355	Anthracene – d10	39
H-316/228/211/279/280	Anthracene - d10	46
H-227/194/119/223/224	Anthracene - d10	1
H-388/314/292/354/355	Benzolalpyrene-d12	22
H-316/228/211/279/280	Benzo(a)pyrane - d12	47
H-227/194/119/223/224	Benzolejpyrene – d12	. 0
H-316/228/211/279/280	Benzolb&kifuorenthenes_d12	66
H-388/314/292/354/355	Banzo(b&k)fuoranthenes-d12	51
H-227/194/119/228/224	Renzo(b&k)fluorenthenes - d12	30
H-227/104/110/229/224		19
	Benzolghilperdene-d12	29
H = 318/228/211/279/280	Benzolabilperviene-d12	23
H - 388/314/202/954/355	Benzíalenthrasene-d12	74
H-227/104/110/209/204	Benzielenthreene -d12	
H = 318/229/241/270/280		94
H	Christens-d12	70
H 000/0 14/282/034/035	Chrysene-d12	73
H-318/228/244 (270/224		79
H-010/220/211/2/3/200	Citysene – 212 Diberzie blenthmeene – dt 4	73
11-227/134/113/223/224 U_318/999/911/970/980	Dibenzie blenthreene dit 4	. 13
1 - 388/914/200/954/955	Dibenzie bleetbreeze dit 4	37
H - 207/104/110/009/004	Diverzitione di C	29
H-388/314/909/854/355	Finorenthene-d10	
H-316/202/311/070/380	Fluoranthene - d10	53
11-010/220/211/2/3/200		33
H-907/194/110/999/994		01
H=318/228/911/270/280	Fluorene-div	97
H-227/104/110/229/200	Indepoid 2.2 - adjevance - dt0	92
H _ 388/914/909/954/955	indepold 2.3 - cdipyrene - d10	13
H-318/228/211/270/280	deno[1,2,3-cd]pyrefie=d10	20
H=227/194/119/223/234	Repeaters - dia	35
H+318/228/211/279/280		49
H-388/314/292/354/355	Phonosthese _ d10	
H-227/194/119/223/224	Pyrane-d10	
H-316/228/211/279/280	Pyrane-d10	51
H-388/314/292/354/355	Pyrane-d10	55
MM5 Trip Spike	- Jiene - 410	00
	Acenaphthene-d10	55 [^]
	Acenaphthylene-d8	57
	Anthracene – d10	60
	Benzo(a)pyrene-d12	87

•

Table F2-15. SVOC Internal Standard Recovery Data

Sample ID ESP Inlet Gas	Compound	% Rec.
	Benzo[b&k]fluoranthenes-d12	77
	Benzo[ghi]perylene-d12	82
	Benz(a)anthracene-d12	87
	Chrysene-d12	77
	Dibenz[a,h]anthracene-d14	89
	Fluoranthene-d10	72
	Fiuorene-d10	79
	Indeno[1,2,3-cd]pyrene-d10	85
	Phenanthrene – d10	60
	Pyrene-d10	78
MM5 Lab Spike		
	Acenaphthene-d10	61
	Acenaphthylene-d8	58
	Anthracene – d10	54
	Benzo[a]pyrene – d12	71
	Benzo[b&k]fluoranthenes-d12	78
	Benzo[ghi]perylene-d12	77
	Benz[a]anthracene-d12	68
	Chrysene-d12	69
	Dibenz(a,h)anthracene – d14	80
	Fiuoranthene-d10	84
	Fluorene-d10	67
	indeno[1,2,3-cd]pyrene-d10	77
	Phenanthrene – d10	54
	Pyrene – d10	86
MM5 Lab Blank		
	Acenaphthene-d10	101
	Acenaphthylene-d8	111
	Anthracene-d10	106
	Benzo[a]pyrene-d12	88
	Benzo[b&k]fluoranthenes-d12	99
	Benzo[ghi]perylene-d12	58
	Benz[a]anthracene-d12	95
	Chrysene-d12	87
	Dibenzia, hjanthracene – d14	58
	Fluoranthene – d10	83
	Fluorene-d10	79
	Indeno[1,2,3-cd]pyrene-d10	63
	rnenanthrene-d10	99
	Pyrene-d10	87

-

Summary for ESP and Stack Gas

-

·	No. of	Mean	Min	Max	Std Dev
	Results	(% Rec)	(% Rec)	(% Rec)	(% Rec)
Acenaphthene-d10	7	34	2	68	27.8
Acenaphthylene-d8	7	18	2	48	17.8
Anthracene – d10	7	21	1	46	17.1
Benzo (a) pyrene-d12	7	15	0	47	16.1
Benzo[b&k]fluoranthenes-d12	7	30	1	66	22.5
Benzo[ghi]perylene-d12	7	14	0	31	10.3
Benz(a)anthracene-d12	7	35	3	81	31.4
Chrysene-d12	7	46	5	79	29.8
Dibenz[a,h]anthracene-d14	7	17	0	37	12.7
Fluoranthene-d10	7	39	12	63	18.7
Fluorene – d10	7	46	3	97	38.6
Indeno[1,2,3-cd]pyrene-d10	7	16	1	35	11.6
Phenanthrene – d10	7	38	7	55	15.5
Pyrene – d10	. 7	39	13	60	19.1

.

.

Table F2-16. SVOC Trip Spike and Lab Spike Recovery Data

Sample ID Co	mpound	% Rec.
MM5 Trip Spik		
5-	-metnyi chrysene	93
71	1-dibenzo[c,g]carbazole	70
Ac	cenaphtnene	87
Ac	enaphthylene	90
Ar	nthracene	95
Be	enzo[a]pyrene	87
Be	enzo[b,j&k]fluoranthenes	95
Be	enzo[ghi]perylene	91
Be	enz[a]anthracene	89
CI	hrysene	103
Di	ibenzo[a,e]pyrene	62
Di	ibenzo[a,h]pyrene	23
Di	ibenzo[a,i]pyrene	49
Di	ibenz[a,h]acridine	91
Di	ibenz[a,h]anthracene	91
Di	ibenz[a,i]acridine	107
FI	uoranthene	92
FI	uorene	98
In	deno[1,2,3-cd]pyrene	85
PI	henanthrene	117
P	yrene	91
MM5 Lab Spil	ke	
5.	-methyl chrysene	83
71	H-dibenzo[c,g]carbazole	102
A	cenaphthene	100
A	cenaphthylene	94
A	nthracene	98
В	enzo[a]pyrene	89
В	enzolb.i&klfluoranthenes	84
В	enzo[ghi]pervlene	72
B	enzlalanthracene	81
Ē	hrvsene	76
		87
	ibenzola hipyrene	64
	ibenzola.ilpyrene	79
)ibenzía.hlacridine	88
Ē)ibenz[a,h]anthracene	82
	benz[a,i]acridine	97
F	luoranthene	80
F	luorene	96
Ir	deno[1.2.3-cd]ovrene	85
	henanthrene	89
P	vrene	84
	J	U 4

Table F2-17. SVOC Surrogate Spike Recovery Data

Sample ID ESP Inlet Gas	Compound	% Rec.
H-237/233/181/163/235/236	Biphenyl-d10	102
H-111/H-109/H-125	Biphenyl-d10	530
H-317/281/238/212/277/278	Biphenyl-d10	153
H-417/418/419/306/291/352/353	Biphenyl-d10	128
H-237/233/181/163/235/236	Hexachlorobenzene-13C6	66
H-317/281/238/212/277/278	Hexachlorobenzene – 13C6	108
H-111/H-109/H-125	Hexachlorobenzene – 13C6	160
H-417/418/419/306/291/352/353	Hexachlorobenzene-13C6	115
H-111/H-109/H-125	Perylene-d12	92
H-237/233/181/163/235/236	Perylene-d12	42
H-317/281/238/212/277/278	Perylene-d12	19
H-417/418/419/306/291/352/353	Perylene-d12	0
Stack Gas		
H-227/194/119/223/224	Biphenyl-d10	918
H-388/314/292/354/355	Biphenyl – d10	146
H-316/228/211/279/280	Biphenyl—d10	138
H-388/314/292/354/355	Hexachlorobenzene-13C6	106
H-316/228/211/279/280	Hexachlorobenzene-13C6	87
H-227/194/119/223/224	Hexachlorobenzene – 13C6	73
H-316/228/211/279/280	Perylene-d12	63
H-388/314/292/354/355	Perylene-d12	0
H-227/194/119/223/224	Perylene-d12	0
MM5 Trip Spike		•
	Biphenyl-d10	125
	Hexachiorobenzene – 13C6	95
	Perylene-d12	126
MM5 Lab Spike		
	Biphenyl-d10	137
•	Hexachlorobenzene-13C6	118
	Perylene-d12	136
MM5 Lab Blank		
	Biphenyl-d10	116
	Hexachlorobenzene – 13C6	102
	Perylene-d12	114
Ourseland the FOR and One of One		

Summary for ESP and Stack Gas

·	No. of Results	Mean (% Rec)	Min (% Rec)	Max (% Rec)	Std Dev (% Rec)
Biphenyl-d10	8	280	102	918	293.4
Hexachlorobenzene-13C6	8	101	66	160	29.5
Perylene-d12	8	43	0	126	47.4

Table F2-18. VOST Method Spike Recovery Data

•

	MS	MSD	Mean	RPD
Compound	(% Rec)	(% Rec)	(% Rec)	(%)
Methylene Chloride	130	130	130	ò.ó
1,2-Dichloroethane	130	130	130	0.0
1,3-Dichlorobenzene	128	130	129	1.6
1,1,2,2-Tetrachloroethane	128	128	128	0.0
trans-1,2-Dichloroethene	128	128	128	0.0
1,2-Dichlorobenzene	127	127	127	0.0
Tetrachloroethene	126	125	126	0.8
Trichlorofluoromethane	126	125	126	0.8
Chloroform	126	123	125	2.4
Chlorobenzene	125	123	124	1.6
1,1-Dichloroethane	123	123	123	0.0
1,4-Dichlorobenzene	123	122	123	0.8
Ethyl Benzene	122	122	122	0.0
o-Xylene	122	121	122	0.8
Acetone	120	121	121	0.8
1,1-Dichloroethene	119	120	120	0.8
Styrene	119	120	120	0.8
4-Methyl-2-Pentanone	114	120	117	5.1
2-Butanone	108	119	114	9.7
Carbon Tetrachloride	103	116	110	11.9
Bromodichloromethane	103.	116	110	11.9
Bromoform	102	114	108	11.1
1,1,2-Trichloroethane	100	111	106	10.4
1,1,1-Trichloroethane	99	107	103	7.8
Dibromochloromethane	99	105	102	5.9
Chloroethane	93	105	99	12.1
Trichloroethene	93	104	99	11.2
trans-1,3-Dichloropropene	92	103	98	11.3
cis-1,3-Dichloropropene	92	99	96	7.3
2-Hexanone	91	97	94	6.4
Benzene	89	95	92	6.5
1,2-Dichloropropane	89	95	92	6.5
Bromomethane	89	94	92	5.5
m,p-Xylene	86	92	89	6.7
Vinyl Chloride	70	91	81	26.1
Vinyl Acetate	50	88	69	55.1
Chloromethane	49	84	67	52.6
Carbon Disulfide	49	44	47	10.8
Toluene	142	137	140	3.6

Table F.2-19. VOST Surrogate Spike Recovery Data

0	O a man a sum d	01 D			
Sample		% necovery			
Statck Gas, 20L VOST Sur	1,2-Dichloroethane-04	92			
Stack Gas, 20L VOST Sur	1,2-Dichloroethane-04	105			
Stack Gas, 20L VOST Sui		106			
Stack Gas, 20L VOST Sur		115			
Stack Gas, 20L VOST Sur	1,2-Dichloroethane-d4	128			
Stack Gas, 20L VOST Sur	1,2-Dichloroethane-d4	105			
Stack Gas, 20L VOST Suf	1,2-Dichloroethane-d4	120			
Stack Gas, 20L VUST Sur	1,2-Dichloroethane-d4	123			
Stack Gas, 20L VOST Sur	1,2-Dichloroethane-d4	126			
Stack Gas, 20L VOST Sur	1,2-Dichloroethane-d4	128			
Stack Gas, 20L VOST Sur	4 - Bromofluorobenzene	100			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	83			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	82			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	96			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	96			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	85			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	113			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	85			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	117			
Stack Gas, 20L VOST Sur	4-Bromofluorobenzene	109			
Stack Gas, 20L VOST Sur	Toiuene-d8	81			
Stack Gas, 20L VOST Sur	Toluene – d8	87			
Stack Gas, 20L VOST Sur	Toluene – d8	84			
Stack Gas, 20L VOST Sur	Toluene – d8	92			
Stack Gas, 20L VOST Sur	Toluene – d8	92			
Stack Gas, 20L VOST Sur	Toluene – d8	86			
Stack Gas, 20L VOST Sur	Toluene – d8	93			
Stack Gas, 20L VOST Sur	Toluene-d8	82			
Stack Gas, 20L VOST Sur	Toluene-d8	97			
Stack Gas, 20L VOST Sur	Toluene – d8	80			
VOST Lab Blank1 Sur	1,2-Dichloroethane-d4	93			
VOST Lab Blank1 Sur	4-Bromofluorobenzene	78			
VOST Lab Blankt Sur	Toluene – d8	96			
VOST Lab Blank2 Sur	1,2-Dichloroethane-d4	95			
VOST Lab Blank2 Sur	4-Bromofluorobenzene	78			
VOST Lab Blank2 Sur	Toluene – d8	96			
VOST Lab Blank3 Sur	1,2-Dichloroethane-d4	128			
VOST Lab Blank3 Sur	4-Bromofluorobenzene	99			
VOST Lab Blank3 Sur	Toluene-d8	91			
VOST Meth Spk1 Sur	1,2-Dichloroethane-d4	118			
VOST Meth Spk1 Sur	4-Bromofluorobenzene	84			
VOST Meth Spk1 Sur	Toluene-d8	115			
VOST Meth Spk2 Sur	1,2-Dichloroethane-d4	96			
VOST Meth Spk2 Sur	4-Bromofluorobenzene	84			
VOST Meth Spk2 Sur	Toluene-d8	116			
		Number	Linon	Mi-	t fas
Summary		of Resulte	(% Rec)	(% Rec)	(% Bec)
			1.44	/	(· • · · • • • • • • • • • • • • • • •

mary		of Results	(% Rec)	(% Rec)	(% Rec)
-	1,2-Dichloroethane-d4	15	112	92	128
	4 – Bromofluorobenzene	15	93	78	117
	Toluene-d8	15	93	80	116

Table F.2-20. Spike Recovery Data for Aldehydes in Gas Samples

÷

١

Sample ID	Compound	Method	% Recovery
Stack gas, impingers LCS	Acetaldehyde	HPLC	100
Stack gas, impingers LCS	Formaldehyde	HPLC	89
Stack gas, impingers Trip Spike (H-369)	Formaldehyde	HPLC	83
Stack gas, impingers Trip Spike (H-369)	Acetaldehyde	HPLC	81
Stack gas, impingers Matrix Spike (H-307)	Acetaldehyde	HPLC	92
Stack gas, impingers Matrix Spike (H-307)	Formaldehyde	HPLC	90

. .

APPENDIX G: PROCESS STREAM FLOW RATES AND FLUE GAS SAMPLING DATA

Table G-1Process Stream Flows at Site 16 - OFA Test

Stream	Mean Flow Rate	Std. Dev.	Source
Coal (lb/hr, wet)	346,000	3,800	Measured*
Coal (lb/hr, dry)	330,000	3,900	Calculated ^b
Bottom Ash (lb/hr, dry)	6,770	79 0	Calculated ^e
ESP Hopper Ash (lb/hr, dry)	27,100	3,200	Calculated°
ESP Inlet Gas (dscfm)	1,290,000	28,000	Measured⁴
Stack Gas (dscfm)	1,110,000	61,000	Measured ^d

* Available from plant meters.

^b Calculated from wet coal flow rate and moisture content.

[°] Calculated assuming a 80:20 fly ash to bottom ash split of coal ash flow.

^d Average of gas flows measured during multi-metals and semivolatiles sampling, when duct was traversed.

Table G-2						
Process Stream	Flows	at	Site	16 -	OFA/LNB Tes	t

Stream	Mean Flow Rate	Std. Dev.	Source
Coal (lb/hr, wet)	328,000	1,200	Measured*
Coal (lb/hr, dry)	315,000	2,100	Calculated ^b
Bottom Ash (lb/hr, dry)	7,060	1,800	Calculated ^e
ESP Hopper Ash (lb/hr, dry)	23,700	1,100	Calculated ^d
ESP Inlet Gas (dscfm)	1,250,000	18,000	Measured
Stack Gas (dscfm)	1,080,000	7,600	Measured

* Available from plant meters.

^b Calculated from wet coal flow rate and moisture content.

^c Calculated as the difference between coal ash rate and ESP inlet ash rate.

^d Calculated as the difference between the ESP inlet ash rate and the stack ash rate.

^e Average of gas flows measured during multi-metals and semivolatiles sampling, when duct was traversed.

OFA Test Flue Gas Sampling Data

.

MULTI-METALS

PARAMETER	INLET		<u>STACK</u>	
Date	3/5/91		3/5/91	
Dry Standard Meter Volume	55.312	DSCF	65.224	DSCF
Percent Flue Gas Moisture	6.34	%	6.33	%
Flue Gas Molecular Weight (wet)	29.19	g/g-mole	29.28	g/g-mole
Average Gas Velocity	78.47	It/sec	64.17	ft/sec
Average Flue Gas Flow Rate	1,276,745	DSCFM	1,082,066	DSCFM
Adjusted Inlet Flue Gas Flow Rate	1,240,000	DSCFM		
Isokinetic Sampling Rate	101.7	%	99.6	%
Oxygen Concentration	7.5	%	9	96
Total Mass of Particulate Solids	5.6627	grams	0.2220	grams
Particulate Concentration	1.58	gr/DSCF	0.0525	gr/DSCF
Particulate Emissions	16.795	lb/hr	487	lb/hr
Impinger Volume (imp 1&2)	634.2	grams	630.3	grams
Impinger Volume (imp 3&4)	541.7	grams	553.4	grams

Date	3/6/91		3/6/91	
Dry Standard Meter Volume	55.602	DSCF	63.622	DSCF
Percent Flue Gas Moisture	6.17	%	6.86	%
Flue Gas Molecular Weight (wet)	29.00	g/g-mole	29.34	g/g-mole
Average Gas Velocity	82.62	ft/sec	83.95	ft/sec
Average Flue Gas Flow Rate	1,299,211	DSCFM	1,056,146	DSCFM
Adjusted Inlet Flue Gas Flow Rate	1,270.000	DSCFM		
Isokinetic Sampling Rate	100.4	%	99.5	%
Oxygen Concentration	7	%	8.5	%
Total Mass of Particulate Solids	4.4173	grams	0.5863	grams
Particulate Concentration	1.23	gr/DSCF	0.14	gr/DSCF
Particulate Emissions	13.348	Ib/hr	1,288	lb/hr
Impinger Volume (imp 1&2)	684.3	grams	644.4	grams
Impinger Volume (imp 3&4)	525.4	grams	568.7	grams

Date	3/6/91		3/6/91	
Dry Standard Meter Volume	51.725	DSCF	65.539	DSCF
Percent Flue Gas Moisture	6.45	%	6.27	%
Flue Gas Molecular Weight (wet)	29.06	g/g-mole	29.42	g/g-mole
Average Gas Velocity	83.14	ft/sec	84.1	ft/sec
Average Fiue Gas Flow Rate	1,307,402	DSCFM	1,058,476	DSCFM
Adjusted Inlet Flue Gas Flow Rate	1,270,000	DSCFM		
Isokinetic Sampling Rate	92.8	%	102.3	%
Oxygen Concentration	7.5	%	8.5	%
Total Mass of Particulate Solids	4.3543	grams	0.4498	grams
Particulate Concentration	1.30	gr/DSCF	0.11	gr/DSCF
Particulate Emissions	14,144	lb/hr	961	lb/hr
Impinger Volume (imp 1&2)	614.2	grams	589.3	grams
Impinger-Volume (imp 3&4)	607	grams	558	grams

ANIONS

INLET			STACK				
	Dry Stand	lard	Dry Stan	lard			
	Meter Vol	ume	Meter Vo	lume			
3/3/91	58.787	DSCF	56.882	DSCF			
3/5/91	59.168	DSCF	59.217	DSCF			
3/6/91-1	60.846	DSCF	55.876	DSCF			
3/6/91-2	58.317	DSCF	55.514	DSCF			
	Filter Wei	ght	Filter We	ight			
	Gain	-	Gain				
3/3/91	19.1000	gms	0.9849	gms			
3/5/91	14.644	gms	0.3099	gms			
3/6/91-1	18.41	gms	0.5078	gms			
3/6/91-2	14.5758	gms	0.4202	gms			
	Impinger		Impinger				
	volume		volume				
3/3/91	605.5	g	502.1	Q			
3/5/91	732.9	g	501.6	9			
3/6/91-1	730.8	g	545	9			
3/6/91-2	595.7	9	583.9	9			
				· · · · · · · · · · · · · · · · · · ·			

ALDEHYDES

	Dry Standard Meter Volume	Dry Standard Meter Volume
3/3/91	11.559 DSCF	57.216 DSCF
3/5/91	14.055 DSCF	21.563 DSCF
3/6/91	13.202 DSCF	19.489 DSCF

MODIFIED METHOD 5

PARAMETER	INLET		STACK	
Date	3/3/91		3/3/91	
Dry Standard Meter Volume	65.275	DSCF	71.552	DSCF
Percent Flue Gas Moisture	6.34	96	6.17	%
Flue Gas Molecular Weight (wet)	29.07	g/g-mole	29.17	g/g-mole
Average Gas Velocity	80.68	ft/sec	90.98	fl/sec
Average Flue Gas Flow Rate	1,279,777	DSCFM	1,174,695	DSCFM
Adjusted Inlet Flue Gas Flow Rate	1,220,000	DSCFM		
Isokinetic Sampling Rate	86.0	%	100.6	46
			and the second	

PARAMETER	INLET		STACK	
Date	3/4/91		3/4/91	
Dry Standard Meter Volume	77.691	DSCF	68.259	DSCF
Percent Flue Gas Moisture	6.17	%	6.86	%
Flue Gas Molecular Weight (wet)	29.09	g/g-mole	29.18	g/g-mole
Average Gas Velocity	82.18	ft/sec	87.21	f/sec
Average Flue Gas Flow Rate	1,300,588	DSCFM		
Adjusted Inlet Flue Gas Flow Rate	1,270,000	DSCFM	1,135,673	DSCFM
Isokinetic Sampling Rate	99.8	%	99.3	<u>44</u>
				. ·
PARAMETER	INLET		STACK	
Date	3/5/91		3/5/91	
Dry Standard Meter Volume	75.358	DSCF	70.908	DSCF
Percent Flue Gas Moisture	6.34	%	6.27	
Fiue Gas Molecular Weight (wet)	29.07	g/g-mole	29.38	g/g-mole

Average Gas Velocity	78.63	ft/sec	83.98	ft/sec
Average Fiue Gas Flow Rate	1,276,582	DSCFM	1,075,909	DSCFM
Adjusted Inlet Flue Gas Flow Rate	1,240,000	DSCFM		
Isokinetic Sampling Rate	98.6	%	107.3	46

VOST DATA

	Pair		Start	Stop	Volume		Meter	Probe	Bar.	as Volume
Run No.	No.	Date	Time	Time	at Meter (I)	DGMCF	Temp (deg F)	Temp (deg F)	Pressure (in. Hg)	Collected (Std I)
ESP INLET	A	3/3/91	1936	1938	1.15	1.037		278	28.9	1.21
RUN NO.1	в	3/3/91	1951	2001	5.06	1.037	44	281	28.9	5.31
	С	3/3/91	2012	2052	19.45	1.037	43	271	28,9	20.45
ESP INLET	A	3/4/91	1705	1745	18.96	1 037	53	277	29.22	19.76
BUN NO 2	R	3/4/91	1815	1825	4 86	1.007	52	292	29.22	5.70
	c	3/4/91	1830	1832	0.74	1.037	51	276	29.22	0.77
ESP INLET	A	3/5/91	1718	1758	18.91	1.0 37	71	284	29.44	19.19
RUN NO.3	в	3/5/91	1806	1816	4.67	1.037	71	282	29.44	4.74
	С	3/5/91	1820	1822	1.00	1.037	70	286	29.44	1.02
Clask		3/2/04	1006	1040						
	A D	3/3/01	1900	1940	20.02	0.962	00	304	28.65	18.51
HON NO. I	c	3/3/91	2005	2015	1.01	0.962	68	307 290	28.65 28.65	4.00 0.93
Stack	A	3/4/91	1653	1733	20.02	0.962	62	283	28.97	18.86
RUN NO.2	в	3/4/91	1822	1832	5.01	0.962	61	290	28.97	4.73
	С	3/4/91	1845	1847	1.00	0. 962	61	270	28.97	0.94
Stack	A	3/5/91	1 720	1800	19.98	1.038	78	267	29 .1	19. 80
RUN NO.3	8	3/5/91	1814	1824	5.03	1.038	84	268	29.1	4.93
	С	3/5/91	1834	1836	1.00	1.038	84	270	29 .1	0.98

Appendix G: Process Stream Flow Rates and Flue Gas Sampling Data

107.6 %

CHROME 6

!

1

i i i

Isokinetic Sampling Rate

	<u>Run 1-1</u>		<u>Run 1-2</u>		<u>Run 1-3</u>	
Date	3/07/91		3/07/91		3/07/91	
Sampling Port	South		West		North	
Dry Standard Meter Volume	59.491	DSCF	60.010	DSCF	58.629	DSCF
Isokinetic Sampling Rate	95.5	%	95. 9	%	94.7	%
	<u>Run 2-1</u>		<u>Run 2-2</u>		<u>Run 2-3</u>	
Date	3/07/91		3/07/91		3/07/91	
Sampling Port	South		North		West	·····
Dry Standard Meter Volume	57.569	DSCF	59.104	DSCF	59.351	DSCF
Isokinetic Sampling Rate	98.1	%	102.3	%	100.4	%
	<u>Run 3-1</u>		<u> Run 3-2</u>		<u>Run 3-3</u>	
Date	3/08/91		3/08/91		3/08/91	
Sampling Port	· South		North		West	
Dry Standard Meter Volume	58.921	DSCF	60.273	DSCF	59.182	DSCF
Isokinetic Sampling Rate	94.7	%	98.7	%	97.1	%
	<u>Run 4-1</u>		<u>Run 4-2</u>		<u>Run 4-3</u>	
Date	3/08/91		3/08/91		3/08/91	
Sampling Port	West		South		North	
Dry Standard Meter Volume	59.019	DSCF	60.782	DSCF	59.412	DSCF
Isokinetic Sampling Rate	95.0	%	96.7	%	95.8	%
	<u>Run 5-1</u>		<u>Run 5-2</u>		<u>Run 5-3</u>	
Date	3/08/91		3/08/91	···-	3/08/91	
Sampling Port	North		South		West	
Dry Standard Meter Volume	49.219	DSCF	57.884	DSCF	49.402	DSCF

98.5 %

91.9 %

ARSENIC

•

<u>3/07/91</u>	<u>Run 1-1</u>		<u>Run 1-2</u>		<u>Run 1-3</u>	
Sampling Port	North		West		South	
Dry Standard Meter Volume	60.784	DSCF	60.145	DSCF	57.676	DSCF
Isokinetic Sampling Rate	98.5	%	98.5	%	98.3	%
<u>3/07/91</u>	<u>Run 2-1</u>		<u>Run 2-2</u>		<u>Run 2-3</u>	
Sampling Port	West		North		South	
Dry Standard Meter Volume	56.952	DSCF	58.802	DSCF	24.966	DSCF
Isokinetic Sampling Rate	96.3	%	98.2	%	89.6	%
<u>3/08/91</u>	<u>Run 3-1</u>		<u>Run 3-2</u>		<u>Run 3-3</u>	
Sampling Port	North	_	West		South	
Dry Standard Meter Volume	59.844	DSCF	58.544	DSCF	59.314	DSCF
Isokinetic Sampling Rate	104.1	%	97.8	%	99.9	%
						1. 18 A. 19
3/08/91	<u>Run 4-1</u>		<u> Run 4-2</u>		<u>Run 4-3</u>	
Sampling Port	North		South		West	
Dry Standard Meter Volume	58.395	DSCF	58.892	DSCF	61.011	DSCF
Isokinetic Sampling Rate	101.3	%	99.8	%	99.5	%

Mercury Data

Run No.	Date	Start Time	Stop Time	Volume at Meter (I)	DGMCF	Meter Temp (deg F)	Probe Temp (deg F)	Bar. Pressure (in. Hg)	as Volume Collected (Std I)
Stack Run 1	3/07/91	1302	1622	100.00	1.038	87	289	29.09	97.42
Stack Run 2	3/07/91	1716	2036	100.00	1.038	90	293	29.09	96.94
Stack Run 3	3/08/91	7 37	1103	100.00	1.038	77	293	29 .17	9 9 .54
Stack Bun 4	3/08/91	1146	1506	100.00	1.038	79	297	29.17	99. 19

OFA/LNB Test Flue Gas Sampling Data

Appendix G: Process Stream Flow Rates and Flue Gas Sampling Data

Run No.	1	2	3	4	Average
Date	05-18-93	05-19-93	05-20-93	05-21-93	÷ –
Time Start	1258 CDT	0830 CDT	0725 CDT	0715 CDT	-
Time Finish	1639 CDT	1211 CDT	1106 CDT	1107 CDT	-
Operator	RLM	DHD	DHD	DHD	
Initial Leak Rate	0.008	0.004	0.006	0.009	-
Final Leak Rate	0.017	0.012	0.018	0.012	
Duct Dimensions (ft2)	425.7	425.7	425.7	425.7	-
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	1.008	1.008	1.008	1.008	1.008
Nozzle Diameter (inches)	0.1850	0.1850	0.1850	0.1810	-
Barometric Pressure ("Hg)	29.3	29.6	29.3	29.4	29.4
Static Pressure ("H2O)	-20.5	-19.5	-17.5	-19.5	-19.25
Meter Volume (acf)	127.573	122.975	121.984	119.711	123.061
Average square root of delta p	1.1868	1.1941	1.1819	1.1518	1.1787
Average delta H (" H2O)	1.27	1.23	1.17	1.13	1.20
Average Stack Temperature (F)	301	305	311	298	304
Average DGM Temp (F)	97.7	77.6	85.4	78.8	84.9
Test Duration (minutes)	216.0	216.0	216.0	216.0	216.0
Condensed Water (g)	208.8	203.8	176.9	177.2	191.7
Filter Weight Gain (g)		11.8018	12.8046	14.2535	12.9533
PNR Weight Gain (g)	Į	4.6916	4.7940	3.5414	4.3423
Impinger Residue (g)					ERR
% CO2	10.0	12.8	12.2	12.2	11.8
% O2	7.8	6.0	6.0	6.0	6.5
% N2	82.2	81.2	81.8	81.8	81.8
Meter Volume (dscf)	119.612	120.808	116.912	116.525	118.464
Flue Gas Moisture (%)	7.6	7.4	6.7	6.7	7.1
Gas Molecular Weight (Wet) (g/g-mole	29.01	29.38	29.38	29.38	29.29
Absolute Stack Pressure (" Hg)	27.79	28.17	28.01	27.97	27.98
Absolute Stack Temperature (R)	761	765	771	758	764
Average Gas Velocity (f/sec)	82.81	82.42	82.13	79.45	81.70
Avg Flow Rate (acfm)	2,115,168	2,105,117	2,097,695	2,029,434	2,086,853
Avg Flow Rate (dscfm)	1,258,769	1,267,221	1,255,498	1,232,217	1,253,426
Isokinetic Sampling Rate (%)	100.34	100.67	98.33	104.32	100.92
Particulate Concentration (gr/dscf)	0.00E+00	2.11E+00	2.32E+00	2.36E+00	1.70E+00
Particulate Concentration (lbs/dscf)	0.00E+00	3.01E-04	3.32E-04	3.37E-04	2.42E-04
Particulate Emission (grams/sec)	0.00	2883.94	3150.34	3136.78	2292.77
Particulate Emission (lbs/hour)	0.00	22888.93	25003.26	24895.62	18196.95

Location ESP Inlet - Particulate/Metals

Location	ESP	inlet -	Anions
----------	-----	---------	--------

Run No.	1	2	3	4	Average
Date	05-18-93	05-19-93	5-20-92	05-21-93	-
Time Start	1159 CDT	0750 CDT	0717 CDT	0705 CDT	-
Time Finish	1354 CDT	0950 CDT	0907 CDT	0900 CDT	-
Operator	LLR	LLR	LLR	LLR	-
Initial Leak Rate	0.000	0.004	0.003	0.002	-
Final Leak Rate	0.003	0.002	0.002	0.002	
Duct Dimensions (ft2)	425.7	425.7	425.7	425.7	-
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	1.007	1.007	1.007	1.007	1.007
Nozzle Diameter (inches)	0.181	0,181	0.181	0.181	0.181
Barometric Pressure ("Hg)	29.30	29.60	29.30	29.40	29.40
Static Pressure ("H2O)	-20.50	-19.50	-17.50	-19.00	-19.13
Meter Volume (acf)	65.631	66.300	63.697	67.710	65.835
Average square root of delta p	1.2751	1.2247	1,3038	1.3416	1.2863
Average delta H (" H2O)	1.21	1.05	1.20	1.25	1.18
Average Stack Temperature (F)	308	297	300	296	300
Average DGM Temp (F)	91.3	69.1	73.5	66.5	75.1
Test Duration (minutes)	110.0	120.0	110.0	115.0	113.8
Condensed Water (g)	120.4	120.6	112.0	117.3	117.6
% CO2	10.0	12.8	12.2	12.2	11.8
· % O2	7.8	6.0	6.0	6.0	6.5
% N2	82.2	81.2	81.8	81.8	81.8
Meter Volume (dscf)	62.174	66.085	62.357	67,400	64.504
Flue Gas Moisture (%)	8.4	7.9	7.8	7.6	7.9
Gas Molecular Weight (Wet) (g/g-mole	28.91	29.31	29.24	29.27	29.18
Absolute Stack Pressure (" Hg)	27.79	28.17	28.01	28.00	27.99
Absolute Stack Temperature (R)	768	757	760	756	760
Average Gas Velocity (f/sec)	89.50	84.23	90.17	92.53	89.11
Avg Flow Rate (acfm)	2,286,103	2,151,457	2,303,109	2,363,448	2276029.36
Avg Flow Rate (dscfm)	1,337,491	1,299,488	1,381,019	1,427,056	1361263.65
Isokinetic Sampling Rate (%)	100.70	100.98	97.81	97.86	99.34

Appendix G: Process Stream Flow Rates and Flue Gas Sampling Data

FORGION FOL MICL - MIMO	Location	ESP	Inlet	- MM5
-------------------------	----------	-----	-------	-------

1.1

1

Run No.	MM5-1	2	3	4	Average
Date	05-18-93	05-19-93	05-20-93	05-21-93	-
Time Start	1215 CDT	0750 CDT	0804 CDT	0755 CDT	-
Time Finish	1558 CDT	1133 CDT	1146 CDT	1159 CDT	-
Operator	DHD	DHD	DHD	DHD	-
Initial Leak Rate	0.008	0.009	0.010	0.003	-
Final Leak Rate	0.014	0.014	0.008	0.07	-
Duct Dimensions (ft2)	425.7	425.7	425.7	425.7	_
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.988	0.988	0.988	0.988	0.988
Nozzle Diameter (inches)	0.1970	0.1970	0.1970	0.1970	-
Barometric Pressure ("Hg)	29.3	29.6	29.3	29.4	29.4
Static Pressure ("H2O)	-20.5	-19.5	-17.5	-19.5	-19.25
Meter Volume (acf)	144.449	141.926	142.550	141.032	142.489
Average square root of delta p	1.2009	1.1913	1.1773	1.1726	1.1855
Average delta H (" H2O)	1.48	1.47	1.45	1.46	1.47
Average Stack Temperature (F)	310	308	305	298	305
Average DGM Temp (F)	99.9	78.0	94.2	83.5	88.9
Test Duration (minutes)	216.0	216.0	216.0	216.0	216.0
Condensed Water (g)	206.6	208.7	181.4	174.8	192.9
% CO2	10.0	12.8	12.2	12.2	11.8
% O2	7.8	6.0	6.0	6.0	6.5
% N2	82.2	81.2	81.8	81.8	81.8
Meter Volume (dscf)	132.291	136.640	131.890	133.497	133.579
Flue Gas Moisture (%)	6.9	6.7	6.1	5.8	6.4
Gas Molecular Weight (Wet) (g/g-mole	29.09	29.46	29.45	29.48	29.37
Absolute Stack Pressure (" Hg)	27.79	28.17	28.01	27.97	27.98
Absolute Stack Temperature (R)	770	768	765	758	765
Average Gas Velocity (f/sec)	84.14	82.32	81.43	80.75	82.16
Avg Flow Rate (acfm)	2,149,131	2,102,535	2,080,011	2,062,567	2,098,561
Avg Flow Rate (dscfm)	1,274,834	1,268,256	1,261,022	1,263,814	1,266,981
Isokinetic Sampling Rate (%)	96.64	100.33	97.40	98.37	98.19

		and the second se			
Run No.	1	2	3	4	Average
Date	05-18-93	05-19-93	05-20-93	05-21-93	-
Time Start	1325 CDT	0704 CDT	0703 CDT	0838 CDT	-
Time Finish	1852 CDT	1108 CDT	1041 CDT	1250 CDT	-
Operator	EBZ	TMP	TMP	EBZ	-
Initial Leak Rate	0.000	0.005	0.005	0.000	-
Final Leak Rate	0.002	0.005	0.000	0.005	-
Stack Diameter (ft)	21.0	21.0	21.0	21.0	-
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	1.017	1.017	0.994	0,994	1.006
Nozzle Diameter (inches)	0.1880	0.1880	0.1870	0.1870	-
Barometric Pressure ("Hg)	29.1	29.4	29.0	29.20	29.175
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	95.770	116.385	116.206	118.502	111.716
Average square root of delta p	1.2133	1.2032	1.2157	1.2209	1.2133
Average delta H (" H2O)	1.27	1.28	1.35	1.34	1.31
Average Stack Temperature (F)	314	302	308	308	308
Average DGM Temp (F)	92.2	83.9	84.0	79.2	84.8
Test Duration (minutes)	160.0	192.0	192.0	192.0	184.0
Condensed Water (g)	145.5	171.2	119.7	160.8	149.3
Filter Weight Gain (g)	0.2381	0.2395	0.3033	0.2416	0.2556
PNR Weight Gain (g)	Į	0.1028	0.1838	0.1551	0.1472
Impinger Residue (g)					ERR
% CO2	9.6	10.8	11.0	11.0	10.6
% O2	8.0	8.8	8.0	8.0	8.2
% N2	82.4	80.4	81.0	81.0	81.2
Meter Volume (dscf)	90.873	113.272	109.027	112.955	106.532
Flue Gas Moisture (%)	7.0	6.7	4.9	6.3	6.2
Gas Molecular Weight (Wet) (g/g-mole	29.02	29.28	29.48	29.32	29.28
Absolute Stack Pressure (" Hg)	28.90	29.20	28.80	29.00	28.98
Absolute Stack Temperature (R)	774	762	768	768	768
Average Gas Velocity (f/sec)	83.68	81.56	83.02	83.31	82.89
Avg Flow Rate (acfm)	1,739,050	1,695,029	1,725,340	1,731,317	1,722,684
Avg Flow Rate (dscfm)	1,065,344	1,069,826	1,085,138	1,081,017	1,075,331
Isokinetic Sampling Rate (%)	95.81	99.10	95.05	98.85	97.20
Particulate Concentration (gr/dscf)	4.04E-02	2 4.66E-02	6.90E-02	5.42E-02	5.26E-02
Particulate Concentration (lbs/dscf)	5.78E-06	6.66E-06	9.85E-06	7.74E-06	7.51E-06
Particulate Emission (grams/sec)) 46.53	3 53.89	80.81	63.29	61.13
Particulate Emission (lbs/hour)	369.30	427.72	2 641.40	502.28	485.17

Location Stack Outlet - Metals/Particulate

Appendix G: Process Stream Flow Rates and Flue Gas Sampling Data

Run No.	1	2 3		4	Average
Date	05-18-93	05-19-93	05-20-93	05-21-93	-
Time Start	1400 CDT	1400 CDT 0720 CDT 0		0847 CDT	-
Time Finish	1543 CDT	0906 CDT	0930 CDT	1022 CDT	-
Operator	EBZ	TMP	TMP	EBZ	-
Initial Leak Rate	0.002	0.003	0.005	0.000	- 1
Final Leak Rate	0.000	0.005	0.003	0.005	-
Stack Diameter (ft)	21.0	21.0	21.0	21.0	-
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.982	0.982	0.982	0.982	0.982
Nozzle Diameter (inches)	0.1840	0.1863	0.1850	0.1960	-
Barometric Pressure ("Hg)	29.1	29.4	29	29.2	29.18
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	68.220	64.531	67.991	63.537	66.070
Average square root of delta p	1.2247	1.1671	1.1571	1.2247	1.1934
Average delta H (" H2O)	1.23	1.20	1.17	1.37	1.24
Average Stack Temperature (F)	315	299	299	300	303
Average DGM Temp (F)	<u>97.5</u>	102.5	94.6	98.8	98.3
Test Duration (minutes)	103.0	106.0	114.0	95.0	104.5
Condensed Water (g)	102.6	104.2	93.0	78.2	94.5
% CO2	9.6	10.8	11.0	11.0	10.6
% O2	8.0	8.8	8.0	8.0	8.2
% N2	82.4	80.4	81.0	81.0	81.2
Meter Volume (dscf)	61.900	58.624	61.798	57.734	60.014
Flue Gas Moisture (%)	7.3	7.7	6.6	6.0	6.9
Gas Molecular Weight (Wet) (g/g-mole	29.00	29.14	29.28	29.35	· 29.19
Absolute Stack Pressure (" Hg)	28.90	29.20	28.80	29.00	28.98
Absolute Stack Temperature (R)	775	759	759	760	763
Average Gas Velocity (f/sec)	84.55	79.15	78.83	83.10	81.41
Avg Flow Rate (acfm)	1,757,064	1,644,963	1,638,137	1,726,897	1,691,765
Avg Flow Rate (dscfm)	1,072,683	1,029,808	1,023,959	1,092,591	1,054,760
Isokinetic Sampling Rate (%)	105.11	98.28	98.25	91.96	98.40

Location Stack Outlet - Anions

Location Stack Outlet - MM5

Run No.	1	2	3	4	Average
Date	05-18-93	05-19-93	05-20-93	05-21-93	-
Time Start	1229 CDT	0810 CDT	0807 CDT	0700 CDT	- 1
Time Finish	1720 CDT	1205 CDT	1223 CDT	1100 CDT	-
Operator	TMP	EBZ	EBZ	TMP	-
Initial Leak Rate	0.001	0.010	0.005	0.005	-
Final Leak Rate	0.005	0.000	0.000	0.000	-
Stack Diameter (ft)	21.0	21.0	21.0	21.0	-
Pitot Tube Correction Factor (Cp)	0.84	0. 84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.994	0.994	1.017	1.017	1.006
Nozzle Diameter (inches)	0.1890	0.1853	0.1870	0.1850	-
Barometric Pressure ("Hg)	29.1	29.4	29.0	29.2	29.175
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	109.012	116.550	120.203	109.284	113.762
Average square root of delta p	1.2137	1.2230	1.2481	1.1966	1.2204
Average delta H (" H2O)	1.28	1.30	1.34	1.18	1.28
Average Stack Temperature (F)	317	308	303	297	306
Average DGM Temp (F)	96.0	89.4	78.3	70.6	83.6
Test Duration (minutes)	183.0	192.0	192.0	192.0	189.8
Condensed Water (g)	161.4	136.2	150.4	137.8	146.5
% CO2	9.6	10.8	11.0	11.0	10.6
% O2	8.0	8.8	8.0	8.0	8.2
% N2	82.4	80.4	81.0	81.0	81.2
Meter Volume (dscf)	100.398	109.765	116.626	108.260	108.762
Flue Gas Moisture (%)	7.1	5.5	5.7	5.7	6.0
Gas Molecular Weight (Wet) (g/g-mole	29.02	29.41	29.39	29.40	29.30
Absolute Stack Pressure (" Hg)	28.90	29.20	28.80	29.00	28.98
Absolute Stack Temperature (R)	777	768	763	757	766
Average Gas Velocity (f/sec)	83.89	83.06	85.09	80.97	83.25
Avg Flow Rate (acfm)	1,743,277	1,726,196	1,768,255	1,682,758	1,730,121
Avg Flow Rate (dscfm)	1,063,274	1,093,383	1,110,160	1,072,769	1,084,897
Isokinetic Sampling Rate (%)	91.75	96.72	99.38	97.54	96.35

Run No.	1	2	3	4	Average
Date	05-19-93	05-20-93	05-21-93	05-21-93	-
Time Start	0933 CDT	1002 CDT	0650 CDT	1137 CDT	-
Time Finish	1121 CDT	1145 CDT	0829 CDT	1317 CDT	-
Operator	TMP	EBZ	EBZ	EBZ	-
Initial Leak Rate	0.005	0.002	0.000	0.005	-
Final Leak Rate	0.002	0.000	0.005	0.000	-
Stack Diameter (ft)	21.0	21.0	21.0	21.0	-
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.982	0.982	0.982	0.982	0.982
Nozzle Diameter (inches)	0.1863	0.1850	0.1960	0.1960	-
Barometric Pressure ("Hg)	29.4	29.0	29.2	29.2	29.2
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	65.175	62.510	64.580	70.283	65.637
Average square root of delta p	1.1771	1.1474	1.1704	1.2178	1.1782
Average delta H (" H2O)	1.22	1.16	1.43	1.54	1.34
Average Stack Temperature (F)	301	305	291	306	301
Average DGM Temp (F)	108.5	104.4	87.0	100.3	100.1
Test Duration (minutes)	108.0	103.0	99.0	100.0	102.5
Condensed Water (g)	96.2	74.6	86.9	83.1	85.2
% CO2	10.8	11.0	11.0	11.0	11.0
% O2	8.8	8.0	8.0	8.0	8.2
% N2	80,4	81.0	81.0	81.0	80.9
Meter Volume (dscf)	58.587	55.821	59.957	63.717	59.521
Flue Gas Moisture (%)	7.2	5.9	6.4	5.8	6.3
Gas Molecular Weight (Wet) (g/g-mole	29.21	29.36	29.31	29.38	29.32
Absolute Stack Pressure (" Hg)	29.20	28.80	29.00	29.00	29.00
Absolute Stack Temperature (R)	761	765	751	766	761
Average Gas Velocity (f/sec)	79.84	78.36	79.02	82.89	80.03
Avg Flow Rate (acfm)	1,659,219	1,628,373	1,642,080	1,722,625	1,663,074
Avg Flow Rate (dscfm)	1,042,327	1,017,546	1,046,693	1,084,477	1,047,761
Isokinetic Sampling Rate (%)	95.24	98.84	95.67	97.14	96.72

Location Stack Outlet - Aldehyde

. .

Run No.	1	2	3	Average
Date	05-19-93	05-20-93	05-21-93	-
Time Start	1456 CDT	1332 CDT	1412 CDT	-
Time Finish	1652 CDT	1532 CDT	1600 CDT	-
Operator	TMP	TMP	TMP	-
Initial Leak Rate	0.005	0.001	0.005	-
Final Leak Rate	0.005	0.001	0.000	-
Stack Diameter (ft)	21.0	21.0	21.0	-
Pitot Tube Correction Factor (Cp)	0.84	• 0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.994	0.994	0.994	0.994
Nozzle Diameter (inches)	0.1870	0.1840	0.1870	-
Barometric Pressure ("Hg)	29.4	29.0	29.2	29.2
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	57.618	56.622	54.780	56.340
Average square root of delta p	1.2508	1.2134	1.1434	1.2025
Average delta H (" H2O)	1.42	1.25	1.17	1.28
Average Stack Temperature (F)	313	317	313	315
Average DGM Temp (F)	88.8	86.5	80.1	85.2
Test Duration (minutes)	96.0	96.0	96 .0	96.0
Condensed Water (g)	92.0	98.0	91.0	93.7
% CO2	10.8	11.0	11.2	11.0
% O2	8.8	8.0	7.8	8.2
% N2	80.4	81.0	81.0	80.8
Meter Volume (dscf)	54.333	52.868	52.105	53.102
Flue Gas Moisture (%)	7.4	8.0	7.6	7.7
Gas Molecular Weight (Wet) (g/g-mole	29.19	29.11	29.18	29.16
Absolute Stack Pressure (" Hg)	29.20	28.80	29.00	29.00
Absolute Stack Temperature (R)	773	777	773	775
Average Gas Velocity (f/sec)	85.56	83.90	78.49	82.65
Avg Flow Rate (acfm)	1,778,061	1,743,646	1,631,193	1,717,633
Avg Flow Rate (dscfm)	1,096,755	1,047,988	996,825	1,047,189
Isokinetic Sampling Rate (%)	93.73	98.59	98.90	97.07

Location Stack Outlet - Chrome VI
																				Average	1	1	1	1	0.991	29.10	ł	1	1.19	303	96.1	10.6	8.2	81.2	19.145	0.0178	6.2	29.28	763
2-4	05-19-93	1030	1110	W	0.991	29.10	3.000	23.610	1.20	302	101.7	10.8	8.8	80.4	18.731	0.0174	6.7	29.27	762	4-4	05-21-93	1033	1117	RVW	0.991	29.20	88.000	110.330	1.20	300	93.0	11.0	8.0	81.0	20.682	0.0192	6.3	29.32	760
2-3	05-19-93	0830	1010	RW	0.991	29.10	80.000	100.030	1.20	300	100.0	10.8	8.8	80.4	18.258	0.0170	6.7	29.27	760	4-3	05-21-93	0935	1015	RVW	0.991	29.20	67.000	87.040	+ 8	300	91.3	11.0	8.0	81.0	18.619	0.0173	6.3	29.32	260
2-2	05-19-93	0823	0903	W M	0.991	29.10	66.000	86.020	1.20	299	98.3	10.8	8.8	80.4	18.303	0.0170	6.7	29.27	759	4-2	05-21-93	0841	0921	W	0.991	29.20	46.000	66.120	1.28	297	86.0	11.0	8.0	81.0	18.874	0.0175	6.3	29.32	757
2-1	05-19-93	0710	0750	WM	0.991	29.10	45.000	65.060	1.30	300	92.7	10.8	8.8	80.4	18.532	0.0172	6.7	29.27	760	4-1	05-21-93	0736	0816	W	0.991	29.20	25.500	45.890	1.20	293	76.3	11.0	8.0	81.0	19.474	0.0181	6.3	29.32	753
1-4	05-18-93	1645	1725	RVW	0.991	29.10	22.000	44.020	1.20	319	106.7	9.6	8.0	82.4	19.834	0.0184	7.0	29.03	779	3-4	05-20-93	1155	1239	WM	0.991	29.00	92.000	114.320	1.20	305	0.66	11.0	8.0	81.0	20.312	0.0189	4.9	29.49	765
1-3	05-18-93	1533	1613	RW	0.991	29.10	1.500	21.550	1.00	315	104.3	9.6	8.0	82.4	18.128	0.0168	7.0	29.03	775	3-3	05-20-93	1047	1127	Ŵ	0.991	29.00	71.000	91.130	1.20	305	98.0	11.0	8.0	81.0	18.352	0.0170	4.9	29.49	765
1-2	05-18-93	1345	1425	WW	0.991	29.10	0.000	19.965	1.20	313	103.3	9.6	8.0	82.4	18,092	0.0168	7.0	29.03	773	3-2	05-20-93	0935	1021	W	0.991	29.00	47.000	70,000	1.20	302	95.7	11.0	8.0	81.0	21.056	0.0195	4.9	29.49	762
1-1-1	05-18-93	1245	1325	WMH	0.991	29.10	0.000	20.250	1.20	305	102.3	9.6	8.0	82.4	18.383	0.0171	7.0	29.03	765	3-1	05-20-93	0830	0914	N N	0.991	29.00	24.000	46.350	1.20	300	89.7	11.0	8.0	81.0	20.684	0.0192	4.9	29.49	760
	Date	Time Start	Time Finish	Operator	Drv Gas Meter Calibration (Yd)	Barometric Pressure ("Ho)	Initial Meter Volume (Liters)	Final Meter Volume (Liters)	Average delta H (* H2O)	Average Stack Temperature (F)	Average DGM Temp (F)	% CO2	% 02	% N2	Mater Volume (STD iters 68F)	Meter Volume (Nm3. 0C)	Flue Gas Moisture (%)	Gas MW (Wet) (q/q-mole)	Absolute Stack Temperature (P	Run No.	Date	Time Start	Time Finishi	Operator	Drv Gas Meter Calibration (Yd)	Barometric Pressure ("Hq)	Initial Meter Volume (Liters)	Final Meter Volume (Liters)	Average delta H (" H2O)	Average Stack Temperature (F)	Average DGM Temp (F)	% CO2	% 02	% N2	Meter Volume (STD Liters, 68F)	Meter Volume (Nm3, 0C)	Flue Gas Moisture (%)	Gas MW (Wet) (a/a-mole)	Absolute Stack Temperature (F)

Plant Name: Site 16 - OFA/LNB Test Location: Stack Outlet - VOST

Run No.	1	2	3	4
Date	05-18-93	05-19-93	05-20-93	05-21-93
Time Start	1212	0815	1027	0915
Time Finish	1538	1334	1407	1259
Operator	RVW	RVW	RVW	RVW
Initial Flow (L/min)	0.500	0.451	0.500	0.500
Final Flow (L/min)	0.499	0.462	0.501	0.486
Sample Volume (Nm3)	0.0941	0.0964	0.0972	0.0983

Plant Name: Site 16 – OFA/LNB Test Location: Stack Outlet – Mercury Speciation

		2	3	4	5	9	Average
Date	05-18-93	05-19-93	05-19-93	05-20-93	05-20-93	05-21-93	ı
Time Start	1145 CDT	0807 CDT	1100 CDT	0705 CDT	0832 CDT	0730 CDT	I
Time Finish	1910 CDT	0950 CDT	1435 CDT	0756 CDT	1210 CDT	1116 CDT	•
Operator	RVW/EBZ	EBZ	EBZ	EBZ	EBZ	EBZ	1
Initial Leak Rate	0.000	0.005	0.005	0.005	0.008	0.005	
Final Leak Rate	AN	0.007	0.003				1
Stack Diameter (ft)	21.0	21.0	21.0	21.0	21.0	21.0	1
Pitot Tube Correction Factor (Cp)	0.84	0.84	0.84	0.84	0.84	0.84	0.84
Dry Gas Meter Calibration (Yd)	0.995	0.995	0.995	0.995	0.995	0.995	0.995
Nozzle Diameter (inches)	0.2430	0.1207	0.1910	0.1910	0.1910	0.1910	•
Barometric Pressure ("Hg)	29.1	29.4	29.4	29.0	29.0	29.2	29.2
Static Pressure ("H2O)	-2.7	-2.7	-2.7	-2.7	-2.7	-2.7	-2.7
Meter Volume (acf)	302.170	32.278	150.987	21.395	150.885	157.000	135.786
Average square root of delta p	1.2083	1.4491	1.4490	1.3962	1.4077	1.3941	1.3841
Average delta H (" H20)	1.50	0.29	1.87	1.74	1.76	1.74	1.48
Average Stack Temperature (F)	315	306	304	303	303	296	304
Average DGM Temp (F)	103.3	85.8	92.7	70.0	85.2	79.5	86.1
Test Duration (minutes)	445.0	103.0	200.0	51.0	209.0	226.0	205.7
Condensed Water (g)	326.9	40.0	258.2	30.0	140.5	188.3	164.0
% CO2	9.6	10.8	10.8	11.0	11.0	11.0	10.7
% 02	8.0	8.8	8.8	8.0	8.0	8.0	8.3
% N2	82.4	80.4	80.4	81.0	81.0	81.0	81.0
Meter Volume (dscf)	275.133	30.551	141.678	20.646	141.547	149.859	126.569
Flue Gas Moisture (%)	5.3	5.8	6.7	6.4	4.5	5.6	5.9
Gas Molecular Weight (Wet) (g/g-mole	29.23	29.38	29.12	29.30	29.54	29.40	29.33
Absolute Stack Pressure (" Hg)	28.90	29.20	29.20	28.80	28.80	29.00	28.98
Absolute Stack Temperature (R)	775	766	764	763	763	756	764
Average Gas Velocity (Vsec)	83.11	98.33	98.60	95.31	95.76	94.27	94.23
Avg Flow Rate (acfm)	1,727,238	2,043,532	2,049,150	1,980,720	1,990,063	1,959,057	1,958,293
Avg Flow Rate (dscfm)	1,075,900	1,294,238	1,272,723	1,234,735	1,265,070	1,251,308	1,232,329
Isokinetic Sampling Rate (%)	61.81	99.92	96.91	57.08	93.21	92.26	83.53

1

.

Location Stack Outlet - PSD

Appendix G: Process Stream Flow Rates and Flue Gas Sampling Data

. .

,

APPENDIX H: PROCESS DATA TREND PLOTS

.



Process Stability, 3 March 1991

March 3rd Process Data



March 4th Process Data



March 5th Process Data



March 6th Process Data



March 7th Process Data



Process Stability, 8 March 1991

March 8th Process Data







- 850	800	N 150	100	1 650	009		1 200	450	400	320 11111	300		<u>Ы</u> 200	20	
0010110	The first of the form to the form		ir Heater B Inlet Gas Temp. (^o F)	· · · ·					.1	ir Heater A Outlet Gas Temp. (^o F)				14 16 18 2	Time (Hours)
	Č		Sec. A							Sec. A		2ec. 4	-	12	
)) سلید		<u>سلیہ</u>		<u>سلس</u> 0	<u>ساب</u>	<u></u> 0	<u>l</u> 0		0	يىلىي 0		10	
19 0	80 8	75	70	65	60	55	50	45	40	35	30	25	0	2	

Unit 4 Process Data 5/18/93 Appendix H: Process Data Trend Plots





Unit 4 Process Data



850	800	750	700	650	600	550	500	450	400	350	300	250	200)) 	
						1								16	
	np. (^o F)		р. (^о F)							emp. (^o F)		erp. (^o F)	-	14	
	nlet Gas Ten		let Gas Tem							Dutlet Gas T		utlet Gas Te		12	(Hours)
	r Heater A Ir		r Heater B In							ir Heater A C		r Heater B O	_	10	Time (
	Sec <u>A</u> i		Sec. Ai							Sec. Ai		Sec. Ai	1	8	
			}										_	9	
0 E O	800	750	700	650	600	550	500	450	400	350	300	250		2002	

Unit 4 Process Data 5/19/93







. 850	800	750	700	650	600	550	500	450	400	350	300	250	200		
									1				-	16	
	[emp. (^o F)		[emp. (^o F)						,	s Temp. (^o F)		; Temp. (^o F)		14	
0010	A Inlet Gas 1		B Inlet Gas 1							A Outlet Gas		B Outlet Gas		12	Hours)
1	c. Air Heater		.c. Air Heater							.c. Air Heater		.c. Air Heater		10	Time (
	Se		Se							Se		Se	-	80	
]	ر بر این	.				_	9	
0 E O	800	750	700	650	600	550	500	450	400	350	300	250	000	2004	

Unit 4 Process Data 5/20/93 Appendix H: Process Data Trend Plots





Appendix H: Process Data Trend Plots



H-22

1 850	800	150 1	100			1 550	1 500	450	400	350	<u> </u> 300		J 200	16	
		Sec. Air Heater A Inlet Gas Temp. (^v F)	Sec. Air Heater B Inlet Gas Temp. (^o F)							Sec. Air Heater A Outlet Gas Temp. (^o F) -		Sec. Air Heater B Outlet Gas Temp. (^o F)		6 8 10 12 14 1	Time (Hours)
020		750 K	200 200	650 FT	600 F	550 F	500 F	450 1	400	350 [-	300	250 F	لب 200	>	

Unit 4 Process Data 5/21/93

APPENDIX I: BLANK CORRECTION DATA

For many of the substances of interest to this program, small traces are present in the reagents and filter media used for sampling and analysis. Therefore, some of the gas stream results in this report have been blank corrected. For the OFA test, blank corrections were routinely applied to the aldehydes, metals, and anions results. Tables I-1 through I-5 show the details of the blank corrections, including the ratio of the blank to the uncorrected result. If the uncorrected value was already below the detection limit, the result is not included in these tables. In many instances, the blank correction was a very small percentage of the result. Instances where the blank values exceed 50% of the measured values are denoted with a "B".

For the OFA/LNB test, only the chromium(VI) results were blank corrected. The details of these corrections are included in Table I-6.

Appendix I: Blank Correction Data

Substance	Stream	Run	Method	Uncorr. ug	Blank, ug	DL, ug	% Blank
Amenic	10 ab about and						
Arsenic	High dust gas	3	GFAAS	1,844	13	196	0.71
Amenic	High dust gas	-	GFAAS	1,117	14	180	1.3
Arsenic	Steck des	2	GFAAS	1,503	15	183	1.0
Arsenic	Stack gas	5	GEAAS	123	0.79	18	0.64
Barium	High dust gas	3	ICP_AES		0.79	20	0.30
Barium	High dust gas	4		0,000	2/6	50	4.1
Barium	High dust gas	5	ICP-AES	4,000	308	46	6.3
Barium	Stack cas	3	ICP-AES	290			6.3
Barium	Stack cas	4	ICP-AFS	550	0.0	3.0	2.8
Barium	Stack gas	5	ICP-AES	504		4.9	1.4
Beryllium	High dust gas	3	ICP-AES	96	22	10.0	1.0
Beryllium	High dust gas	4	ICP-AES	90	2.5	9.2	2.5
Beryllium	High dust gas	5	ICP-AES	94	2.6	93	2.0
Cadmium	High dust gas	3	GFAAS	6.5	0.85	50	13
Cadmium	High dust gas	5	GFAAS	5.7	0.98	4.7	17
Cadmium	Stack gas	3	GFAAS	0.69	0.10	0.30	15
Cadmium	Stack gas	5	GFAAS	1.5	0.10	0.51	6.5
Chromium	High dust gas	3	ICP-AES	1,301	393	50	
Chromium	HIGN dust gas	4	ICP-AES	1,153	438	46	38
Chromium	nign aust gas	5	ICP-AES	1,192	457	47	38
Chromium	Stack gas	3	ICP-AES	48	2.6	3.0	5.4
Chromium	Stack gas	4	ICP-AES	92	2.6	4.9	2.8
Cobalt	Stack gas	5	ICP-AES		2.7	5.1	2.9
Cobalt	Stack gas		ICP-AES	13	0.13	3.0	0.98
Cobalt	Stack gas	5	ICP AES	29	0.13	4.9	0.45
Copper	High dust gas		ICP-AES	2/	0.13	5.1	0.48
Copper	High dust gas	3	ICP-AES	1,117	6.3	100	0.56
Copper	High dust gas		ICP-AES	1 010	7.0	92	0.70
Copper	Stack das	3	ICP-AES	1,013	/.3	93	0.71
Copper	Stack gas	4	ICP-AFS	101	1.3	<u> </u>	2.6
Copper	Stack gas	5	ICP-AES	103	12	9.1	1.2
Lead	High dust gas	3	GFAAS	271			
Lead	High dust gas	4	GFAAS	497	49	54	
Lead	High dust gas	5	GFAAS	469	51	28	9.9
Lead	Stack gas	3	GFAAS	73	1.0	88	
Lead	Stack gas	5	GFAAS	63	1.1	5.5	17
Manganese	High dust gas	3	ICP-AES	884	51	50	5.7
Manganese	High dust gas	4	ICP-AES	570	56	46	9.9
Manganese	High dust gas	5	ICP-AES	599	59	47	9.8
Manganese	Stack gas	3	ICP-AES	35	1.4	3.0	4.0
Manganese	Stack gas	4	ICP-AES	62	1.4	4.9	2.2
Mercup	Stack gas	5	ICP-AES	53	1.4	5.1	2.6
Mercury	High dust gas	- 3	CVAAS	3.5	0.44	0.90	13
Mercury	Stack care		CVAAS	5.1	0.51	0.84	9.9
Mercurv	Stack cas		CVAAS	0.23	0.0080	0.05	3.5
Mercury	Stack cas		CVAAS	0.91	0.0079	0.09	0.87
Molyboenum	High dust cas	3	ICP-AFS	U.04	0.0082	0.09	1.3
Molybdenum	High dust gas	4	ICP-AFS	424	341	244	<u> </u>
Molybdenum	High dust gas	5	ICP-AES	465	300	230	<u>88 B</u>
Molybdenum	Stack gas	3	ICP-AES	42	200	233	
Molybdenum	Stack gas	4	ICP-AES	53	20		/V B
Molybdenum	Stack gas	5	ICP-AES	57		24	J3 B
Nickel	High dust gas	3	ICP-AES	735	44	100	0.0
Nickel	High dust gas	4	ICP-AES	514	49	92	9.5
Nickel	High dust gas	5	JCP-AES	555	51	93	9.2
Nickel	Stack gas	3	ICP-AES	33	2.7	6.0	8.2
Nickel	Stack gas		ICP-AES	59	2.7	9.7	4.6
Vonedium	Stack gas	5	ICP-AES	59	2.8	10	4.7
Vanadium	nigh dust gas	3	ICP-AES	1,723	7.8	100	0.45
Vanadium	High dust gas	. 4	ICP-AES	1,323	8.7	92	0.66
Vanadium	Steck con		ICP-AES	1,380	9.1	93	0.66
Vanadium	Stack car		ICP-AES	88	1.0	6.0	1.1
Vanadium	Stack ras		ICP-AES	174	0.99	9.7	0.57
	- mon yas	3	IVE-AES	173	1.0	10	0.59

Table 1~1: Metals Solid Phase Blank Corrections - OFA Test*

Includes filter and probe/nozzle rinses
 B indicates that blank correction exceeds 50% of uncorrected result.

-

Substance	Stream	Run	Method	Uncorr. ug	Blank ug	DL, ug	Blank %
Cadmium	High dust gas	5	GFAAS	0.69	0.39	0.61	57 B
Cadmium	Stack Gas	3	GFAAS	0.85	0.39	0.63	46
Cadmium	Stack Gas	5	GFAAS	0.76	0.39	0.59	52 B
Copper	High dust gas	5	ICP-AES	13	2.7	12	21
Lead	High dust gas	3	GFAAS	3.1	4.7	1.9	152 B
Lead	High dust gas	5	GFAAS	8.3	4.7	1.8	57 B
Lead	Stack Gas	3	GFAAS	6.2	4.7	1.9	76 B
Lead	Stack Gas	4	GFAAS	2.6	4.7	1.9	181 B
Lead	Stack Gas	5	GFAAS	2.1	4.7	1.8	229 B
Manganese	High dust gas	3	ICP-AES	9.8	10	6.3	105 B
Manganese	High dust gas	4	ICP-AES	13	10	6.8	82 B
Manganese	High dust gas	5	ICP-AES	13	10	6.1	78 B
Manganese	Stack Gas	3	ICP-AES	10	10	6.3	103 B
Manganese	Stack Gas	4	ICP-AES	9.6	10	6.4	107 B
Manganese	Stack Gas	5	ICP-AES	10	10	5.9	99 B
Nickel	High dust gas	5	ICP-AES	14	7.9	12	58 B

Metals Vapor Phase Blank Coorections - OFA Test* Table 1-2:

.

* Results of all impingers combined. B indicates that blank correction exceeds 50% of uncorrected result.

Appendix I: Blank Correction Data

.

Substance	Stream	Run	Method	Uncorr. mg	Blank mg	DL, mg	Blank %
Chloride	High dust gas	3	IC	27	1.9	0.0030	7.0
Chloride	High dust gas	4		25	1.6	0.0029	6.3
Chloride	High dust gas	5	IC	27	1.9	0.0028	7.0
Chloride	Stack gas	3	IC	0.031	0.0044	0.0012	14
Chloride	Stack gas	4	IC	0.053	0.0044	0.0012	8.4
Chloride	Stack gas	5	IC	0.088	0.0043	0.0013	4.9
Fiuoride	High dust gas	3	ISE	2.3	1.1	0.025	49
Fluoride	High dust gas	4	ISE	2.6	0.96	0.024	37
Fluoride	High dust gas	5	ISE	2.4	1.1	0.024	47
Fluoride	Stack gas	3	ISE	1.00	0.0031	0.010	0.32
Fluoride	Stack gas	4	ISE	0.76	0.0032	0.010	0.42
Fluoride	Stack gas	5	ISE	0.72	0.0031	0.011	0.43

.

Table I-3: Anions Solid Phase Blank Corrections - OFA Test

Substance	Stream	Run	Method	Uncorr. ug	Blank, ug	DL, ug	Blank %
Fluoride	High dust gas	3	ISE	2624	8.1	73	0.31
Fluoride	High dust gas	4	ISE	3604	8.1	73	0.22
Fluoride	High dust gas	5	ISE	2551	8.1	60	0.32
Fluoride	Stack gas	3	ISE	11164	8.1	50	0.072
Fluoride	Stack gas	4	ISE	9919	8.1	55	0.081
Fluoride	Stack gas	5	ISE	9957	8.1	58	0.081

.

Table I-4: Anions Vapor Phase Blank Corrections - OFA Test

Appendix I: Blank Correction Data

Table I-5: Aldehydes Blank Corrections – OFA Test

Substance	Stream	Run	Method	Uncorr. ug	Blank ug	DL, ug	Blank %
Formaldehvde	High dust gas	1	HPLC	9.5	5.4	2.4	57 B
Formaldehyde	High dust gas	3	HPLC	6.3	5.4	2.4	86 B
Formaldehyde	Stack gas	1	HPLC	9.5	5.4	2.4	57 B

.

B indicates that blank correction exceeds 50% of uncorrected result.

Table 1-6: Chromium(VI) Blank Corrections - OFA/LNB Test

I

1

Substance	Stream	Run	Method	Uncorr. ug	Blank ug	DL, ug	Blank %
Chromium(VI)	Stack gas	2	BIF Cr6	22	12	0.95	55 B
Chromium(VI)	Stack gas	3	BIF Cr6	22	18	0.84	83 B
Chromium(VI)	Stack gas	4	BIF Cr6	25	12	0.70	48

B indicates that blank correction exceeds 50% of uncorrected result.